CareFirst BlueChoice, Inc.

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

IN-NETWORK EVIDENCE OF COVERAGE FOR A QUALIFIED HEALTH PLAN

This Qualified Health Plan is being offered through the SHOP Exchange.

This is the In-Network Evidence of Coverage for the jointly offered Point-of-Service product with In-Network HMO benefits administered by CareFirst BlueChoice, Inc. and Out-of-Network indemnity benefits administered by Group Hospitalization and Medical Services, Inc., doing business as, CareFirst BlueCross BlueShield (CareFirst). Each time that services are sought the Member may choose to receive In-Network HMO benefits or to receive Out-of-Network indemnity benefits.

This In-Network Evidence of Coverage, including any notices, amendments and riders, is a part of the In-Network Group Contract issued to the Group through which Members are enrolled for covered health benefits. In addition, the In-Network Group Contract includes other provisions that explain the duties of CareFirst BlueChoice and the Group. The Group's payment to the SHOP Exchange and CareFirst BlueChoice's issuance of the In-Network Group Contract make the In-Network Group Contract's terms and provisions binding on CareFirst BlueChoice and the Group.

The Group reserves the right to change, modify, or terminate the plan, in whole or in part.

Members should not rely on any oral description of the plan because the written terms in the Group's plan documents always govern.

CareFirst BlueChoice recommends that the Member familiarizes himself or herself with the CareFirst BlueChoice complaint and appeal procedure, and make use of it before taking any other action.

NOTE: It is a crime to provide false or misleading information to an insurer for the purpose of defrauding the insurer or any other person. Penalties include imprisonment and/or fines. In addition, CareFirst BlueChoice may deny insurance benefits if false information materially related to a claim was provided by the applicant.

Group Name:	Sample
Group Number:	Sample
Product Name:	BlueChoice Advantage Gold 3000
Group Effective Date:	<u>January 1, 2019</u>
	CareFirst BlueChoice, Inc.
	[Signature]
	[Name] [Title]

SECTION	TABLE OF CONTENTS	PAGE
1	Definitions	3
2	Eligibility and Enrollment	14
3	Termination of Coverage	23
4	Continuation of Coverage	27
5	Coordination of Benefits (COB); Subrogation	29
6	General Provisions	36
7	Service Area	43
ATTACHMENTS		
A	Benefit Determinations and Appeals	A-1
В	Description of Covered Services	B-1
С	Schedules of Benefits	C-1
	Amendments/Notices/Riders	
D	Eligibility Schedule	D-1

SECTION 1 DEFINITIONS

The underlined terms, when capitalized, are defined as follows:

<u>Adoption</u> means the earlier of a judicial decree of adoption, or the assumption of custody, pending adoption, of a prospective adoptive child by a prospective adoptive parent.

<u>Affordable Care Act</u> means the Patient Protection and Affordable Care Act of 2010 (Pub. L. 111–148), as amended by the Health Care and Education Reconciliation Act of 2010 (Pub. L. 111–152).

Allowed Benefit means:

- A. For a Contracting Provider, the Allowed Benefit for a Covered Service is the amount agreed upon between CareFirst BlueChoice and the Contracting Provider which, in some cases, will be a rate set by a regulatory agency. The benefit is payable to the provider and is accepted as payment in full, except for any applicable Deductible, Copayment, or Coinsurance amounts, for which the Member is responsible.
- B. For a Non-Contracting Provider that is a health care practitioner, the Allowed Benefit for a Covered Service is based upon the lesser of the provider's actual charge or the established fee schedule. The benefit is payable to the Member or to the provider at the discretion of CareFirst BlueChoice. The Member is responsible for any applicable Deductible, Copayment, or Coinsurance amounts stated in the Schedule of Benefits and the difference between the Allowed Benefit and the practitioner's actual charge. The provider may bill the Member directly for such amounts. It is the Member's responsibility to apply any CareFirst BlueChoice payments to the claim from the Non-Contracting Provider charge.
- C. For a Non-Contracting Provider that is a health care facility, the Allowed Benefit for a Covered Service is based upon either the provider's actual charge or the established fee schedule. The benefit is payable to the Member or to the facility, at the discretion of CareFirst BlueChoice. Benefit payments to Department of Defense and Veteran Affairs providers will be made directly to the provider. The Member is responsible for any applicable Deductible, Copayment, or Coinsurance amounts stated in the Schedule of Benefits and, unless negotiated, for the difference between the Allowed Benefit and the provider's actual charge. It is the Member's responsibility to apply any CareFirst BlueChoice payments to the claim from the Non-Preferred Facility.
 - In some cases, and on an individual basis, CareFirst BlueChoice is able to negotiate a lower rate with an eligible provider. In that instance, the CareFirst BlueChoice payment will be based on the negotiated fee and the provider agrees to accept the amount as payment in full except for any applicable Deductible, Copayment, or Coinsurance amounts, for which the Member is responsible.
- D. For a Covered Service rendered by a Non-Contracting Provider of ambulance services, the Allowed Benefit for a Covered Service is based upon the lesser of the provider's actual charge or the established fee schedule. The benefit is payable to the Member or to the Non-Contracting Provider of ambulance services, at the discretion of CareFirst BlueChoice. When benefits are paid to the Member, it is the Member's responsibility to apply any CareFirst BlueChoice payments to the claim from the Non-Contracting Provider of ambulance services. The Member is responsible for any applicable Deductible, Copayment, or Coinsurance amounts stated in the Schedule of Benefits and the difference between the Allowed Benefit and the provider's actual charge. The provider may bill the Member directly for such amounts.

For Emergency Services provided by a Non-Contracting Provider, the Allowed Benefit for a Covered Service is based upon the lesser of the provider's actual charge, or the

amount that would be paid to a Contracting Provider for the Covered Service. The Member is responsible for any applicable Deductible, Copayment, or Coinsurance amounts stated in the Schedule of Benefits and the difference between the Allowed Benefit and the practitioner's actual charge. The provider may bill the Member directly for such amounts.

Pediatric Dental Allowed Benefit means:

A. For Preferred Dentists, the Allowed Benefit payable to a Preferred Dentist for a Covered Dental Service will be the amount agreed upon between CareFirst BlueChoice and the Preferred Dentist. The benefit payment is made directly to the Preferred Dentist and accepted as payment in full, except for any applicable Deductible and Coinsurance for which the Subscriber is responsible as stated in the Schedule of Benefits. The Subscriber is responsible for any applicable Deductible and Coinsurance, and both Preferred and Non-Preferred Dentists may bill the Subscriber directly for such amounts.

Pediatric Vision Allowed Benefit means:

For a Contracting Vision Provider, the Vision Allowed Benefit for a Covered Vision Service is the lesser of:

- A. The Contracting Vision Provider's actual charge; or
- B. The benefit amount, according to the Vision Care Designee's Contracting Vision Provider rate schedule for Covered Vision Service that applies on the date that the service is rendered.

The Vision Allowed Benefit is made directly to a Contracting Vision Provider and is accepted as payment in full.

Prescription Drug Allowed Benefit means the lesser of:

- A. The Pharmacy's actual charge; or
- B. The benefit amount, according to the CareFirst BlueChoice fee schedule, for covered Prescription Drugs that applies on the date the service is rendered.

If the Member purchases a covered Prescription Drug or diabetic supply from a Contracting Pharmacy Provider, the benefit payment is made directly to the Contracting Pharmacy Provider and is accepted as payment in full, except for any applicable Deductible, Copayment, or Coinsurance. The Member is responsible for any applicable Deductible, Copayment, or Coinsurance and the Contracting Pharmacy Provider may bill the Member directly for such amounts.

If the Member purchases a covered Prescription Drug from a Non-Contracting Pharmacy Provider, the Member is responsible for paying the total charge and submitting a claim to CareFirst BlueChoice or its designee for reimbursement. Members will be entitled to reimbursement from CareFirst BlueChoice or its designee in the amount of the Allowed Benefit, minus any applicable Deductible, Copayment, or Coinsurance. Members may be responsible for balances above the Allowed Benefit.

Annual Open Enrollment Period means the period of no less than thirty (30) days each year prior to the Group's Contract Renewal Date during which an individual may enroll or change coverage in this Qualified Health Plan through the SHOP Exchange.

<u>Benefit Period</u> means the Contract Year during which coverage is provided for Covered Services, Covered Dental Services and Covered Vision Services.

<u>Bereavement Counseling</u> means counseling provided to the Immediate Family or Family Caregiver of the Member after the Member's death to help the Immediate Family or Family Caregiver cope with the death of the Member.

<u>Brand Name Drug</u> means a Prescription Drug that has been given a name by a manufacturer or distributor to distinguish it as produced or sold by a specific manufacturer or distributor and may be used and protected by a trademark.

Calendar Year means January 1 through December 31 of each year.

<u>Cardiac Rehabilitation</u> means inpatient or outpatient services designed to limit the physiologic and psychological effects of cardiac illness, reduce the risk for sudden death or reinfarction, control cardiac symptoms, stabilize or reverse atherosclerotic process, and enhance the psychosocial and vocational status of eligible Members.

<u>CareFirst BlueCross BlueShield (CareFirst)</u> means the business name of Group Hospitalization and Medical Services, Inc. (GHMSI) the corporate entity which underwrites the Out-of-Network benefits under the Out-of-Network Evidence of Coverage.

<u>Civil Union</u> means a same-sex relationship similar to marriage that is recognized by law. The Subscriber's partner in a Civil Union is eligible for coverage to the same extent as an eligible Spouse.

<u>Coinsurance</u> means the percentage of the Allowed Benefit allocated between CareFirst BlueChoice and the Member, whereby CareFirst BlueChoice and the Member share in the payment for Covered Services, Covered Dental Services, or Covered Vision Services.

<u>Contract Renewal Date</u> means the date on which the In-Network Group Contract renews and each anniversary of such date.

Contract Year means the twelve (12) month period beginning on the Group Effective Date.

<u>Contracting Pharmacy Provider</u> means a separate independent Pharmacist or Pharmacy that has contracted with CareFirst BlueChoice or its designee to provide covered Prescription Drugs.

<u>Contracting Physician</u> means a licensed doctor who has entered into a contract with CareFirst BlueChoice to provide Covered Services to Members and has been designated by CareFirst BlueChoice as a Contracting Physician.

<u>Contracting Provider</u> means any physician, health care professional, health care facility, or Contracting Pharmacy Provider that has contracted with CareFirst BlueChoice, Inc. to render Covered Services to Members. A Preferred Dentist who provides Covered Dental Services or a Contracting Vision Provider who provides Covered Vision Services is not a Contracting Provider for the purposes of this definition.

<u>Contracting Vision Provider</u> means any optometrist or ophthalmologist licensed as such by the duly constituted authority in the jurisdiction in which Covered Vision Services are rendered when acting within the scope of such license and that has contracted with the Vision Care Designee to provide Covered Vision Services.

<u>Convenience Item</u> means any item that increases physical comfort or convenience without serving a Medically Necessary purpose (e.g., elevators, hoyer/stair lifts, ramps, shower/bath benches, and items available without a prescription).

<u>Copayment (Copay)</u> means the fixed dollar amount that a Member must pay for certain Covered Services, Covered Dental Services or Covered Vision Services.

<u>Cosmetic</u> means a service or supply which is provided with the primary intent of improving appearance, not restoring bodily function or correcting deformity resulting from disease, trauma, or previous therapeutic intervention, as determined by CareFirst BlueChoice.

<u>Covered Dental Services</u> means Medically Necessary services or supplies listed in Section 2 of the Description of Covered Services.

<u>Covered Service</u> means Medically Necessary services or supplies provided in accordance with the terms of this In-Network Evidence of Coverage, other than Covered Dental Services or Covered Vision Services.

<u>Covered Vision Services</u> means Medically Necessary services or supplies listed in Section 3of the Description of Covered Services.

<u>Custodial Care</u> means care provided primarily to meet the personal needs of the patient. Custodial Care does not require skilled medical or paramedical personnel. Such care includes help in walking, bathing, or dressing. Custodial Care also includes preparing food or special diets, feeding, administering medicine, or any other care not requiring continuing services of medically trained personnel.

<u>Decertification or Decertified</u> means the termination by the SHOP Exchange of the certification and offering of this Qualified Health Plan.

<u>Deductible</u> means the dollar amount of the Allowed Benefits payable during a Benefit Period for Covered Services or Covered Dental Services that must first be incurred by the Member before CareFirst BlueChoice will make payments for Covered Services or Covered Dental Services.

<u>Dental Director</u> is a Dentist appointed by the Medical Director of CareFirst BlueChoice to perform administrative duties with regard to the dental services listed in this In-Network Evidence of Coverage.

<u>Dental Plan</u> means the dental program under which the Covered Dental Services are made available to Members. The Dental Plan is offered in conjunction with Group Hospitalization and Medical Services, Inc. doing business as CareFirst BlueCross BlueShield (CareFirst) which contracts with Preferred Dentists and provides claims processing and administrative services under the Dental Plan.

<u>Dentist</u> means an individual who is licensed to practice dentistry as defined by the respective jurisdiction where the practitioner provides care.

<u>Dependent</u> means a Member who is covered as an eligible Spouse or Dependent Child as defined in Sections 2.2 and 2.3. The eligibility of Dependents to enroll is stated in the Eligibility Schedule.

<u>Dependent Child</u> or <u>Dependent Children</u> means one or more eligible individuals as defined in Section 2.3.

<u>Diabetic Supply or Diabetic Supplies</u> means all Medically Necessary and appropriate supplies prescribed by a health care provider for the treatment of diabetes, including but not limited to lancets, alcohol wipes, test strips (blood and urine), syringes and needles.

<u>Effective Date</u> means the date on which the Member's coverage becomes effective. Covered Services, Covered Dental Services, and Covered Vision Services rendered on or after the Member's Effective Date are eligible for coverage.

Emergency Medical Condition means:

- A. A medical condition manifesting itself by acute symptoms of sufficient severity (including severe pain) such that the absence of immediate medical attention could reasonably be expected to result in placing the health of the individual (or, with respect to a pregnant woman, the health of the woman or her unborn child) in serious jeopardy, serious impairment to bodily functions, or serious dysfunction of any bodily organ or part; or
- B. With respect to a pregnant woman who is having contractions: there is inadequate time to effect a safe transfer to another hospital before delivery, or transfer may pose a threat to the health or safety of the woman or the unborn child.

Emergency Services means, with respect to an Emergency Medical Condition:

- A. A medical screening examination (as required under section 1867 of the Social Security Act, 42 U.S.C. 1395dd) within the capability of the emergency department of a hospital, including ancillary services routinely available to the emergency department to evaluate such Emergency Medical Condition; and
- B. Such further medical examination and treatment, to the extent they are within the capability of the staff and facilities available at the hospital, as are required under section 1867 of the Social Security Act (42 U.S.C. 1395dd(e)(3)) to stabilize the Member. The term to "stabilize" with respect to an Emergency Medical Condition, has the meaning given in section 1867(e)(3) of the Social Security Act (42 U.S.C. 1395dd(e)(3))).

<u>Exclusive Specialty Pharmacy Network</u> means a pharmacy network that is limited to certain specialty Pharmacies that have been designated as "Exclusive" by CareFirst BlueChoice. Members may contact CareFirst BlueChoice for a list of Pharmacies in the Exclusive Specialty Pharmacy Network.

<u>Experimental/Investigational</u> means a service or supply in the developmental stage and in the process of human or animal testing excluding patient costs for clinical trials as stated in the Description of Covered Services. Services or supplies that do not meet all five of the criteria listed below are deemed to be Experimental/Investigational:

- A. The Technology* must have final approval from the appropriate government regulatory bodies;
- B. The scientific evidence must permit conclusions concerning the effect of the Technology on health outcomes;
- C. The Technology must improve the net health outcome;
- D. The Technology must be as beneficial as any established alternatives; and
- E. The improvement must be attainable outside the Investigational settings.

<u>FDA</u> means the United States Food and Drug Administration.

<u>Family Caregiver</u> means a relative by blood, marriage, or Adoption who lives with or is the primary Caregiver of the terminally ill Member.

<u>Family Counseling</u> means counseling given to the Immediate Family or Family Caregiver of the terminally ill Member for the purpose of learning to care for the Member and to adjust to the impending death of the Member.

<u>Generic Drug</u> means any Prescription Drug approved by the FDA that has the same bio-equivalency as a specific Brand Name Drug.

<u>Group</u> means the Qualified Employer to which CareFirst BlueChoice has issued the In-Network Group Contract and the In-Network Evidence of Coverage.

Group Contract Effective Date means the effective date of the In-Network Group Contract.

<u>Habilitative Services</u> mean health care services and devices that help a person keep, learn, or improve skills and functioning for daily living. Examples include therapy for a child who is not walking or talking at the expected age. These services may include physical and occupational therapy, speech-language

^{* &}quot;Technology" includes drugs, devices, processes, systems, or techniques.

pathology and other services for people with disabilities in a variety of inpatient and/or outpatient settings.

<u>Home Health Care or Home Health Care Services</u> means the continued care and treatment of a Member in the home by a licensed Home Health Agency if:

- A. The institutionalization of the Member in a hospital, related institution, or Skilled Nursing Facility would otherwise have been required if Home Health Care Services were not provided; and
- B. The Plan of Treatment covering the Home Health Care Service is established and approved in writing by the health care provider, and determined to be Medically Necessary by CareFirst BlueChoice.

<u>Immediate Family</u> means the Spouse, parents, siblings, grandparents, and children of the terminally ill Member.

<u>Infusion Services</u> means treatment provided by placing therapeutic agents into the vein, and parenteral administration of medication and nutrients. Infusion services also include enteral nutrition, which is the delivery of nutrients by tube into the gastrointestinal tract. Infusion Services include all medications administered intravenously and/or parenterally.

<u>In-Network</u> means Point-of-Service benefits provided to the Member under this In-Network Evidence of Coverage.

<u>In-Network Evidence of Coverage</u> means this agreement, including all duly authorized attachments, notices, amendments and riders, issued to the Group and the Subscriber by CareFirst BlueChoice under the In-Network Group Contract between the Group and CareFirst BlueChoice.

<u>In-Network Group Contract</u> means the contract, including all duly authorized attachments, notices, amendments and riders, between the Group and CareFirst BlueChoice.

<u>Limiting Age</u> means the maximum age up to which a Dependent Child may be covered as stated in the Eligibility Schedule.

<u>Low Vision</u> means a significant loss of vision but not total blindness. Ophthalmologists and optometrists specializing in Low Vision care can evaluate and prescribe optical devices, and provide training and instruction to maximize the remaining usable vision for Members with Low Vision.

<u>Maintenance Drug</u> means a Prescription Drug anticipated being required for six (6) months or more to treat a chronic condition.

Mastectomy means the surgical removal of all or part of the breast.

<u>Medical Child Support Order (MCSO)</u> means an order issued in the format prescribed by federal law and issued by an appropriate child support enforcement agency to enforce the health insurance coverage provisions of a child support order. An order means a judgment, decree, or a ruling (including approval of a settlement agreement) that:

- A. Is issued by a court or administrative child support enforcement agency of any state or the District of Columbia; and
- B. Creates or recognizes the right of a child to receive benefits under a parent's health insurance coverage or establishes a parent's obligation to pay child support and provide health insurance coverage for a child.

<u>Medical Director</u> means a board certified physician who is appointed by CareFirst BlueChoice. The duties of the Medical Director may be delegated to qualified persons.

<u>Medically Necessary or Medical Necessity</u> means health care services or supplies that a health care provider, exercising clinical judgment, renders to or recommends for a patient for the purpose of preventing, evaluating, diagnosing or treating an illness, injury, disease, or its symptoms. These health care services or supplies are:

- A. In accordance with generally accepted standards of medical practice;
- B. Clinically appropriate, in terms of type, frequency, extent, site and duration, and considered effective for a patient's illness, injury, or disease;
- C. Not primarily for the convenience of a patient or health care provider; and
- D. Not more costly than an alternative service or sequence of services that are at least as likely to produce equivalent therapeutic or diagnostic results in the diagnosis or treatment of the patient's illness, injury, or disease.

For these purposes, "generally accepted standards of medical practice" means standards based on credible scientific evidence published in peer-reviewed medical literature generally recognized by the relevant medical community, physician specialty society recommendations and views of health care providers practicing in relevant clinical areas, and any other relevant factors.

The fact that a health care provider may prescribe, authorize or direct a service does not of itself make it Medically Necessary or covered by the In-Network Group Contract.

<u>Medical Nutrition Therapy</u> provided by a licensed dietitian-nutritionist involves the assessment of the Member's overall nutritional status followed by the assignment of an individualized diet, counseling, and/or specialized nutrition therapies to treat a chronic illness or condition. The licensed dietitian-nutritionist, working in a coordinated, multidisciplinary team effort with the Primary Care Physician takes into account a Member's condition, food intake, physical activity, course of any medical therapy, including medications and other treatments, individual preferences, and other factors.

<u>Member</u> means an individual who meets all applicable eligibility requirements of Section 2, is enrolled for coverage and for whom the CareFirst BlueChoice receives the premiums and other required payments. A Member can be either a Subscriber or Dependent.

Minimum Essential Coverage has the meaning given in the Affordable Care Act, 26 U.S.C. §5000A(f).

Non-Contracting Physician means a licensed doctor who is not contracted with CareFirst BlueChoice to provide Covered Services to Members.

<u>Non-Contracting Provider</u> means any health care provider that has not contracted with CareFirst BlueChoice to provide Covered Services to Members.

Non-Preferred Dentist means any Dentist who is not a Preferred Dentist.

<u>Out-of-Network</u> means Point-of-Service benefits provided to the Member under the Out-of-Network Evidence of Coverage issued to the Subscriber by CareFirst BlueCross BlueShield under the Out-of-Network Group Contract between the Group and CareFirst BlueCross BlueShield.

<u>Out-of-Network Evidence of Coverage</u> means the Point-of-Service in-network agreement, including all duly authorized attachments, notices, amendments and riders, if any, issued to the Group and the Subscriber by CareFirst BlueCross BlueShield under the Out-of-Network Group Contract between the Group and CareFirst BlueCross BlueShield.

<u>Out-of-Network Group Contract</u> means the Point-of-Service out-of-network contract, including all duly authorized attachments, notices, amendments and riders, between the Group and CareFirst BlueCross BlueShield.

Out-of-Pocket Maximum means the maximum amount the Member will have to pay for his/her share of benefits in any Benefit Period. The Out-of-Pocket Maximum does not include premiums, the cost of services that are not Covered Services, or any amounts paid to providers in excess of the Allowed Benefit, the Pediatric Dental Allowed Benefit, the Vision Allowed Benefit or the Prescription Drug Allowed Benefit. Once the Member meets the Out-of-Pocket Maximum, the Member will no longer be required to pay Copayments, Coinsurance or Deductible for the remainder of the Benefit Period.

Over-the-Counter means any item or supply, as determined by CareFirst BlueChoice, available for purchase without a prescription, unless otherwise a Covered Service. This includes, but is not limited to, non-prescription eye wear, family planning and contraception items for men, cosmetics or health and beauty aids, food and nutritional items, support devices, non-medical items, foot care items, first aid and miscellaneous medical supplies (whether disposable or durable), personal hygiene supplies, incontinence supplies, and related over-the-counter medications, solutions, items or supplies.

<u>Pharmacist</u> means an individual licensed to practice pharmacy regardless of the location where the activities of practice are performed.

<u>Pharmacy</u> means an establishment in which prescription or nonprescription drugs or devices are compounded, dispensed, or distributed.

<u>Point-of-Service</u> means a benefit program that offers two benefit levels: In-Network that provides HMO-level benefits; and an Out-of-Network that provides indemnity level of benefits. Point-of-Service means that each time Members receive Covered Services, Covered Dental Services or Covered Vision Services they can choose to receive either In-Network benefits or Out-of-Network benefits.

<u>Preferred Brand Name Drug</u> means a Brand Name Drug that is included on CareFirst BlueChoice's Preferred Drug List.

<u>Preferred Dentist</u> means one of a network of Dentists who, at the time of rendering a Covered Dental Service to the Member, has a written agreement with CareFirst BlueChoice, or the Dental Plan, for the rendering of such service.

<u>Preferred Drug List</u> means the list of Brand Name Drugs and Generic Drugs issued by CareFirst BlueChoice and used by health care providers when writing, and Pharmacists, when filling, prescriptions. All Generic Drugs are included in the Preferred Drug List. Not all Brand Name Drugs are included in the Preferred Drug List. CareFirst BlueChoice may change this list periodically without notice to Members. A copy of the Preferred Drug List is available to the Member upon request.

Prescription Drug means:

- A. A drug, biological or compounded prescription intended for outpatient use that carries the FDA legend "may not be dispensed without a prescription";
- B. Drugs prescribed for treatments other than those stated in the labeling approved by the FDA, if the drug is recognized for such treatment in standard reference compendia or in the standard medical literature as determined by CareFirst BlueChoice;
- C. A covered Over-the-Counter medication or supply;
- D. Any Diabetic Supply;
- E. Prescription Drugs do not include:
 - 1 Compounded bulk powders that contain ingredients that:
 - a) Do not have FDA approval for the route of administration being compounded, or

- b) Have no clinical evidence demonstrating safety and efficacy, or
- c) Do not require a prescription to be dispensed.
- 2. Compounded drugs that are available as a similar commercially available Prescription Drug unless:
 - a) There is no commercially available bioequivalent Prescription Drug; or
 - b) The commercially available bioequivalent Prescription Drug has caused or is likely to cause the Member to have an adverse reaction.

<u>Prescription Guidelines</u> means the limited list of Prescription Drugs issued by CareFirst BlueChoice for which providers, when writing, and Pharmacists, when filling prescriptions, must obtain prior authorization from CareFirst BlueChoice and the quantity limits that CareFirst BlueChoice has placed on certain drugs. A copy of the Prescription Guidelines is available to the Member upon request.

<u>Preventive Drug</u> means a Prescription Drug or Over-the-Counter medication or supply dispensed under a written prescription by a health care provider that is included on the CareFirst BlueChoice Preventive Drug List.

<u>Preventive Drug List</u> means the list issued by CareFirst BlueChoice of Prescription Drugs or Over-the-Counter medications or supplies dispensed under a written prescription by a health care provider that have been identified in the current recommendations of the United States Preventive Services Task Force that have in effect a rating of "A" or "B" or as provided in the comprehensive guidelines for women's preventive health supported by the Health Resources and Services Administration. CareFirst BlueChoice may change this list periodically and without notice to Members. A copy of the Preventive Drug List is available to the Member upon request.

<u>Primary Care Dependent</u> means an unmarried grandchild, niece or nephew for whom the Subscriber provides primary care including food, shelter and clothing on a regular and continuous basis during the time the District of Columbia public schools are in regular session.

<u>Primary Care Physician (PCP)</u> means a Contracting Provider selected by a Member to provide and manage the Member's health care. PCP means health care practitioners in the following disciplines:

- A. General internal medicine;
- B. Family practice medicine;
- C. General pediatric medicine; or
- D. Geriatric medicine.

Services rendered by Specialists in the disciplines above will be treated as PCP visits for Member payment purposes.

<u>Professional Nutritional Counseling</u> means individualized advice and guidance given to a Member who is at nutritional risk due to nutritional history, current dietary intake, medication use, or chronic illness or condition, about options and methods for improving nutritional status. Professional Nutritional Counseling must be provided by a licensed dietitian-nutritionist, physician, physician assistant, or nurse practitioner.

<u>Qualified Employee</u> means an eligible individual who has been offered health insurance coverage by the Group through the SHOP Exchange. The Group's eligibility requirements for a Qualified Employee are stated in the Eligibility Schedule.

<u>Qualified Employer</u> means the employer that the SHOP Exchange has determined to be qualified to offer Qualified Health Plan(s).

<u>Qualified Health Plan</u> means a health plan certified by the SHOP Exchange as having met the standards established by the U.S. Department of Health and Human Services.

<u>Qualified Home Health Agency</u> means a licensed program approved for participation as a home health agency under Medicare, or certified as a home health agency by the Joint Commission on Accreditation of Healthcare Organizations, its successor, or the applicable state regulatory agency.

Qualified Hospice Care Program means a coordinated, interdisciplinary program provided by a hospital, Qualified Home Health Agency, or other health care facility licensed or certified by the state in which it operates as a hospice program and is designed to meet the special physical, psychological, spiritual, and social needs of terminally ill individuals and their families, by providing palliative and supportive medical, nursing, and other health services through home or inpatient care during the illness and bereavement period. Benefits are available to:

- A. Individuals who have no reasonable prospect of cure as estimated by a physician; and
- B. The immediate families or Family Caregivers of those individuals.

<u>Qualified Medical Support Order (QMSO)</u> means a Medical Child Support Order, issued under state law or the laws of the District of Columbia, that is issued to an employer sponsored health plan that complies with section 609(A) of the Employee Retirement Income Security Act of 1974 (ERISA), as amended.

<u>Rescind</u> or <u>Rescission</u> means a termination, cancellation or discontinuance of coverage that has retroactive effect. For example, a cancellation that treats coverage as void from the time of the individual's or group's enrollment is a Rescission. As another example, a cancellation that voids benefits paid up to a year before the cancellation is also a Rescission for this purpose. Coverage is not Rescinded and a cancellation or discontinuance of coverage is not a Rescission if:

- A. The termination, cancellation or discontinuance of coverage has only a prospective effect; or
- B. The termination, cancellation or discontinuance of coverage is effective retroactively to the extent it is attributable to a failure to timely pay charges when due, by the Group.

<u>Respite Care</u> means temporary care provided to the terminally ill Member to relieve the Caregiver/Family Caregiver from the daily care of the Member.

SHOP Exchange means the District of Columbia Health Benefit Exchange (DC HBX).

<u>Skilled Nursing Facility</u> means a licensed institution (or a distinct part of a hospital) accredited or approved under Medicare or The Joint Commission and provides continuous Skilled Nursing Care and related services for Members who require medical care, Skilled Nursing Care, or rehabilitation services. Inpatient skilled nursing is for patients who are medically fragile with limited endurance and require a licensed health care professional to provide skilled services in order to ensure the safety of the patient and to achieve the medically desired result. Inpatient skilled nursing services must be provided on a 24 hour basis, 7 days a week.

<u>Sound Natural Teeth</u> means teeth restored with intra- or extra-coronal restorations (fillings, inlays, onlays, veneers, and crowns) that are in good condition; absent decay, fracture, bone loss, periodontal disease, root canal pathology or root canal therapy and excludes any tooth replaced by artificial means (implants, fixed or removable bridges, dentures).

<u>Special Enrollment Period</u> means a period during which an eligible individual who experiences certain qualifying events may enroll in, or change enrollment in, a Qualified Health Plan through the SHOP Exchange outside of any Annual Open Enrollment Periods.

<u>Specialist</u> means a licensed health care provider who is certified or trained in a specified field of medicine.

<u>Specialty Drugs</u> means high-cost injectables, infused, oral, or inhaled Prescription Drugs for the ongoing treatment of a chronic condition, including but not limited to, the following: Hemophilia, Hepatitis C, Multiple Sclerosis, Rheumatoid Arthritis, Psoriasis, Crohn's Disease, Cancer (oral medications), and Growth Hormones. These Prescription Drugs usually require specialized handling (such as refrigeration).

<u>Spouse</u> means a person of the same or opposite sex who is legally married to the Subscriber under the laws of the state or jurisdiction in which the marriage took place. A marriage legally entered into in another jurisdiction will be recognized as a marriage in the District of Columbia. The Subscriber's partner in a Civil Union or domestic partnership is eligible for coverage to the same extent as an eligible Spouse. Whenever the term "Spouse" appears in this Evidence of Coverage, the provision includes a Civil Union partner or domestic partner.

<u>Subscriber</u> means a Member who is enrolled as a Qualified Employee or eligible former Qualified Employee rather than as a Dependent.

<u>Urgent Care</u> means treatment for a condition that is not a threat to life or limb but does require prompt medical attention. Also, the severity of an urgent condition does not necessitate a trip to the hospital emergency room. An Urgent Care facility is a freestanding facility that is not a physician's office and which provides Urgent Care.

<u>Vision Care Designee</u> means the entity with which CareFirst BlueChoice has contracted to administer Vision Care. CareFirst BlueChoice's Vision Care Designee is Davis Vision, Inc. Davis Vision, Inc. is an independent company and administers the Vision Care benefits on behalf of CareFirst BlueChoice.

<u>Waiting Period</u> means the period of time, stated in the Eligibility Schedule, that must pass before an eligible employee or any Dependent is eligible for coverage under the terms of the In-Network Group Contract.

SECTION 2 ELIGIBILITY AND ENROLLMENT

- 2.1 <u>Requirements for Coverage</u>. To be covered, <u>all</u> of the following conditions must be met:
 - A. A Subscriber must be an eligible Qualified Employee of the Group. To enroll as a Subscriber, the individual must, at the time of enrollment, meet the eligibility requirements established by the Group. The Group's eligibility requirements for a Qualified Employee are stated in the Eligibility Schedule.
 - B. The Subscriber must work or reside in the Service Area;
 - C. Any other Member must be an eligible individual who is a Dependent of a Subscriber. To enroll as a Dependent, the individual must, at the time of enrollment meet the eligibility requirements established by the Group. The eligibility of Dependents to enroll is stated in the Eligibility Schedule.
 - D. For each Subscriber and Member, the SHOP Exchange must receive premium payments as required by the In-Network Group Contract.
- 2.2 <u>Eligibility of Subscriber's Spouse</u>. If the Group has elected to include coverage for the Subscriber's Spouse, then a Subscriber may enroll a Spouse as a Dependent. A Subscriber cannot cover a former Spouse once divorced or if the marriage had been annulled. Premium changes resulting from the enrollment of a Spouse will be effective as of the Effective Date of the Spouse's enrollment.
- 2.3 <u>Eligibility of Dependent Children</u>. If the Group has elected to include coverage for Dependent Children of the Subscriber or a Subscriber's covered Spouse, then a Subscriber may enroll one or more Dependent Children. A <u>Dependent Child</u> means an individual who:
 - A. Is:
 - 1. The natural child, stepchild, or adopted child of the Subscriber;
 - 2. A child placed with the Subscriber or the Subscriber's covered Spouse for legal Adoption;
 - 3. A child under testamentary or court appointed guardianship, other than temporary guardianship for less than twelve (12) months' duration, of the Subscriber or the Subscriber's covered Spouse; or
 - 4. An unmarried grandchild, niece or nephew, who meets the requirements for coverage as the Subscriber's Primary Care Dependent as stated below:
 - a) The child must be the Subscriber's unmarried grandchild, niece, or nephew;
 - b) The child is under the Subscriber's Primary Care. <u>Primary Care</u> means the Subscriber provides food, clothing and shelter for the child on a regular and continuous basis during the time District of Columbia public schools are in regular session; and,
 - c) If the child's legal guardian is someone other than the Subscriber, the child's legal guardian is not covered under any other health insurance policy.

The Subscriber must provide CareFirst BlueChoice with proof upon application that the child meets the requirements for coverage as a Primary Care Dependent,

- including proof of the child's relationship and primary dependency on the Subscriber and certification that the child's legal guardian does not have other coverage. CareFirst BlueChoice reserves the right to verify whether the child is and continues to qualify as a Primary Care Dependent, and
- 5. A child who becomes a Dependent of the Subscriber through a child support order or other court order.
- B. Is under the Limiting Age as stated in the Eligibility Schedule; or
- C. Is a disabled Dependent Child who is older than the Limiting Age and the Subscriber provides proof that: (1) the Dependent Child is incapable of self-support or maintenance because of a mental or physical incapacity; (2) the Dependent Child is primarily dependent upon the Subscriber or the Subscriber's covered Spouse for support and maintenance; and (3) the Dependent Child had been covered under the Subscriber's or the Subscriber's Spouse's prior health insurance coverage since before the onset of the mental or physical incapacity.
- D. Is the subject of a Medical Child Support Order (MCSO) or Qualified Medical Support Order (QMSO) that creates or recognizes the right of the child to receive benefits under the health insurance coverage of the Subscriber or the Subscriber's covered Spouse.
- E. A child whose relationship to the Subscriber is not listed above, including, but not limited to, foster children or children whose only relationship is one of legal guardianship (except as provided above) is not eligible to enroll and is not covered under this In-Network Evidence of Coverage, even though the child may live with the Subscriber and be dependent upon him or her for support.
- 2.4 Limiting Age for Covered Dependent Children.
 - A. All covered Dependent Children are eligible for coverage up to the Limiting Age for Dependent Children as stated in the Eligibility Schedule.
 - B. A Dependent Child covered under this In-Network Evidence of Coverage will be eligible for coverage past the Limiting Age if, at the time coverage would otherwise terminate:
 - 1. The Dependent Child is unmarried and is incapable of self-support or maintenance because of a mental or physical incapacity;
 - 2. The Dependent Child is primarily dependent upon the Subscriber or the Subscriber's covered Spouse for support and maintenance;
 - 3. The mental or physical incapacity occurred before the covered Dependent Child reached the Limiting Age; and
 - 4. The Subscriber provides CareFirst BlueChoice with proof of the Dependent Child's mental or physical incapacity within thirty-one (31) days after the Dependent Child reaches the Limiting Age for Dependent Children. CareFirst BlueChoice has the right to verify whether the child is and continues to qualify as an incapacitated Dependent Child.
 - C. Dependents' coverage will automatically terminate if there is a change in their age, status, or relationship to the Subscriber, such that they no longer meet the eligibility requirements of this In-Network Evidence of Coverage or the Eligibility Schedule. Coverage of an ineligible Dependent will terminate as stated in the Eligibility Schedule.

- 2.5 <u>Open Enrollment Opportunities and Effective Dates</u>. A Qualified Employee may elect coverage for himself or herself or for an eligible Dependent only during the following times and under the following conditions.
 - A. <u>Annual Open Enrollment</u>. During an Annual Open Enrollment Period, a Qualified Employee or eligible individual may enroll as a Subscriber or Member.
 - B. <u>Newly Eligible Subscriber</u>. If a Qualified Employee is a new employee or a newly eligible employee of the Group, the new Qualified Employee may enroll him or herself and any eligible Dependent within thirty (30) days after a new Qualified Employee first becomes eligible. The eligibility requirements for a new Qualified Employee in the Group are stated in the Eligibility Schedule.
 - C. Special Enrollment. If a Qualified Employee or Dependent does not enroll during an Annual Open Enrollment Period or during the enrollment period for newly eligible Qualified Employees, he or she may only enroll during a Special Enrollment Period
 - 1. Thirty (30) or Thirty-One (31) Day Special Enrollment Period. Except as otherwise specified, an eligible individual may enroll as a Subscriber or Dependent upon the occurrence of one of the following qualifying events:
 - a) The Qualified Employee or a Dependent:
 - (1) Loses Minimum Essential Coverage. A loss of Minimum Essential Coverage includes those circumstances described in 26 CFR 54.9801-6(a)(3)(i) through (iii).

Loss of coverage described herein includes those circumstances described in 26 CFR 54.9801-6(a)(3)(i) through (iii) and in paragraphs (d)(1)(ii) through (iv) of 45 CFR §155.420. Loss of coverage does not include voluntary termination of coverage or other loss due to:

- (a) Failure to pay premiums on a timely basis, including COBRA premiums prior to the expiration of COBRA coverage; or
- (b) Situations allowing for a Rescission.
- (2) Is enrolled in any non-Calendar Year health insurance coverage even if the Qualified Employee or his or her Dependent has the option to renew such coverage. The date of the loss of coverage is the last day of the plan or policy year.
- (3) Loses pregnancy-related coverage described in 1902(a)(10)(A)(i)(IV) and (a)(10)(A)(ii)(IX) of the Social Security Act (42 U.S.C. 1396a(a)(10)(A)(i)(IV), (a)(10)(A)(ii)(IX)). The date of the loss of coverage is the last day the consumer would have pregnancy-related coverage; or
- (4) Loses medically needy coverage as described under section 1902(a)(10)(C) of the Social Security Act only once per calendar year. The date of the loss of coverage is the last day the consumer would have medically needy coverage.
- (5) At the option of the SHOP Exchange, the enrollee who is the Qualified Employee or the Spouse of the Qualified Employee if the enrollee loses a Dependent or is no longer considered to be a

Dependent due to divorce or legal separation; or if the employee or his or her Dependent dies.

- b) The Qualified Employee or a Dependent gains, or becomes, an eligible Dependent through marriage, birth, adoption, placement for adoption, or grant of court or testamentary guardianship or child support order (MCSO or QMSO) or other court order.
- c) The Qualified Employee or a Dependent has a foster child placed by an accredited foster child agency. (Note: The foster child is not eligible for coverage.)
- d) The Qualified Employee's or Dependent's enrollment in another Qualified Health Plan or non-enrollment, as evaluated and determined by the SHOP Exchange, is unintentional, inadvertent, or erroneous and is the result of an error, misrepresentation, misconduct, or inaction of an officer, employee, or agent of the SHOP Exchange or the United States Department of Health and Human Services or its instrumentalities, or a non-Exchange entity providing enrollment assistance or conducting enrollment activities.
- e) The Qualified Employee or Dependent, who is an enrollee in another Qualified Health Plan, demonstrates to the SHOP Exchange that the other Qualified Health Plan in which he or she has enrolled substantially violated a material provision of its contract in relation to the Qualified Employee or Dependent.
- f) The Qualified Employee or Dependent gains access to new Qualified Health Plans as a result of a permanent move.
- g) The Qualified Employee is an Indian, as defined in section 4 of the Indian Health Care Improvement Act, who may enroll in a Qualified Health Plan or change coverage from one Qualified Health Plan to another one time per month.
- h) The Qualified Employee or Dependent demonstrates to the SHOP Exchange, in accordance with guidelines issued by the United States Department of Health and Human Services that he or she meets other exceptional circumstances determined by the SHOP Exchange.
- i) The Qualified Employee or Dependent enrolled in the same Qualified Health Plan is determined to be newly eligible or newly ineligible for advance premium tax credit or has a change in eligibility for cost-sharing reductions

For purposes of this provision, advance premium tax credit means tax credits specified under section 1401 of the Affordable Care Act which are provided on an advance basis to an eligible individual enrolled in a Qualified Health Plan through the District of Columbia Health Benefit Exchange. For purposes of this provision, cost-sharing reduction means an affordability program under Section 1402 of the Affordable Care Act.

- 2. The Special Enrollment Period shall be as follows:
 - a) For the qualifying event listed in Section 2.5C.1.b) and c) (Qualified Employee or Dependent gains, or becomes, an eligible Dependent through marriage, birth, adoption, placement for adoption, placement for foster care, grant of court or testamentary guardianship), or child support

- order (MCSO or QMSO) or other court order: The thirty-one (31) day period from the date of the qualifying event.
- b) If an eligible employee or Dependent meets the requirements for the triggering event described in 2.5C.1.d), the exchange may take any action necessary to correct or eliminate the effects of the error, misrepresentation, or inaction.
- c) For the qualifying event listed in Section 2.5C.1.g) (Qualified Employee is an Indian, as defined in section 4 of the Indian Health Care Improvement Act): One time per month.
- d) For all other qualifying events listed in Section 2.5C.1: The thirty (30) day period from the date of the qualifying event, unless otherwise determined by the SHOP Exchange.
- 3. Sixty (60) Day Special Enrollment Period. In addition, an eligible individual may enroll as a Subscriber or Dependent upon the occurrence of one of the following qualifying events:
 - a) The Qualified Employee or Dependent loses eligibility for coverage under a Medicaid plan under title XIX of the Social Security Act or a state child health plan under title XXI of the Social Security Act;
 - b) The Qualified Employee or Dependent becomes eligible for assistance, with respect to coverage under a Qualified Health Plan offered through the SHOP Exchange, under such Medicaid plan or a state child health plan (including any waiver or demonstration project conducted under or in relation to such a plan); or
 - c) The Qualified Employee or Dependent is enrolled in an employersponsored health benefit plan that is not qualifying coverage in an eligible employer sponsored plan and is allowed to terminate existing coverage.
- 4. With the exception of Section 2.5.C.3.c), the Special Enrollment Period for the qualifying events listed in Section 2.5C.3 shall be the sixty (60) day period from the date of the qualifying event. In the event of a qualifying event under Section 2.5.C.3.c), the Special Enrollment Period shall be sixty (60) days prior to the end of coverage under the employer sponsored plan.

D. Effective Dates

- 1. The Effective Date for a Qualified Employee who enrolls during an Annual Open Enrollment Period shall be the first day of the new policy year, unless otherwise provided by the SHOP Exchange.
- 2. The Effective Date for a newly eligible Qualified Employee and any Dependent who timely enrolls as provided in Section 2.5B shall be as stated in the Eligibility Schedule.
- 3. The Effective Date for a Dependent Child who timely enrolls during a Special Enrollment Period is the Dependent Child's First Eligibility Date (unless otherwise provided by the SHOP Exchange).
 - a) First Eligibility Date means:

- (1) For a newborn Dependent Child, the child's date of birth;
- (2) For a non-newborn Dependent Child who is a stepchild, the date the stepchild became a Dependent of the Subscriber;
- (3) For a newly adopted Dependent Child, the earlier of;
 - (a) A judicial decree of Adoption; or
 - (b) Placement of the child in the Subscriber's home as the legally recognized proposed adoptive parent;
- (4) For a Dependent Child for whom guardianship of at least twelve (12) months' duration has been granted by court or testamentary appointment, the date of the appointment;
- (5) For a Dependent Child who has been placed with the Subscriber through a child support order (MCSO or QMSO) or other court order, the date that the court order is effective;
- (6) For all other Dependent Children, the first day of the month following the receipt of enrollment by the SHOP Exchange.
- b) The Dependent Child will be covered automatically, but only for the first thirty-one (31) days following the child's First Eligibility Date. The Subscriber must enroll such a Dependent Child within thirty-one (31) days of the child's First Eligibility Date when an additional premium is due for the enrollment of the Dependent Child. Otherwise, the Dependent Child will not be covered and cannot be enrolled until the next Annual Open Enrollment Period. (An additional premium will be due unless there are three (3) or more Dependent Children under the age of twenty-one (21) already enrolled by the Subscriber).
- 4. The Effective Date for a Qualified Employee or Dependent who gains, or becomes, an eligible Dependent through marriage and who timely enrolls during a Special Enrollment Period shall be the first day of the month following the receipt of enrollment by the SHOP Exchange.
- 5. The Effective Date for a Qualified Employee (and eligible Dependents) with whom a child was placed through a foster care program shall be the date that the child was placed with the Qualified Employee (and eligible Dependents) in foster care (unless otherwise provided by the SHOP Exchange). Note that foster children are not eligible for coverage.
- 6. The Effective Date for a Qualified Employee or Dependent who loses other coverage, or who gains access to new Qualified Heath plans due to a permanent move, who timely enrolls during a Special Enrollment Period shall be the first day of the month following the receipt of enrollment by the SHOP Exchange, or as stated in Section 2.5D.8. below, at the option of the SHOP Exchange.

- 7. The Effective Date for a Qualified Employee or Dependent who timely enrolls due to a qualifying event stated in Section 2.5C.1.c) (enrollment or non-enrollment was unintentional, inadvertent, or erroneous and is the result of an error by the SHOP Exchange or the United States Department of Health and Human Services), Section 2.5C.1.d) (a Qualified Health Plan substantially violated a material provision of its contract), Section 2.5C.1.g) (other exceptional circumstances as determined by the SHOP Exchange), , shall be the appropriate date based on the circumstances of the Special Enrollment Period as determined by the Exchange.
- 8. If an enrollee or his or her Dependent dies as stated in 2.5C.1.a)(5), the Effective Date of coverage is the first day of the month following plan selection or provided in Section 2.5D.6.b) of this provision as determined by the SHOP Exchange.
- 9. In all other cases, the Effective Date for a Qualified Employee or Dependent who timely enrolls during a Special Enrollment Period will be:
 - a) For enrollment received by the SHOP Exchange between the first (1st) and the fifteenth (15th) day of the month, the first day of the following month; and
 - b) For enrollment received by the SHOP Exchange between the sixteenth (16th) and the last day of the month, the first day of the second following month.
- 10. Premium changes resulting from the enrollment of a Subscriber or a Dependent during a Special Enrollment Period will be effective as of the Effective Date of the Subscriber's or the Dependent's enrollment.

2.6 Child Support Orders (MCSO or QMSO).

A. Eligibility and Termination.

- 1. Upon receipt of an MCSO or QMSO, CareFirst BlueChoice will accept enrollment of a Dependent Child that is the subject of an MCSO or QMSO and the Qualified Employee parent of such child, without regard to enrollment period restrictions, within the time period prescribed by law. Coverage will be effective as of the effective date of the order, and the premium will be adjusted as needed. If the Subscriber does not enroll the child then CareFirst BlueChoice will accept enrollment from the non-Subscriber custodial parent; or, the appropriate child support enforcement agency of any state or the District of Columbia. If the Subscriber is subject to a Waiting Period, the child will not be enrolled until the end of the waiting period.
- 2. Enrollment for such a child will not be denied because the child:
 - a) Was born out of wedlock;
 - b) Is not claimed as a dependent on the Subscriber's federal tax return;
 - c) Does not reside with the Subscriber; or
 - d) Is covered under any Medical Assistance or Medicaid program.
- 3. Coverage required by an MCSO or QMSO will be effective as of the date of the

order.

- 4. Termination. Unless coverage is terminated for non-payment of the premium, a covered child subject to an MCSO or QMSO may not be terminated unless written evidence is provided to CareFirst BlueChoice stating:
 - a) The MCSO or QMSO is no longer in effect;
 - b) The child has been or will be enrolled under other comparable health insurance coverage that will take effect not later than the effective date of the termination of coverage;
 - c) The Group has eliminated family members' coverage for all its employees; or
 - d) The employer no longer employs the insuring parent, except if the parent elects to exercise the provisions of the Consolidated Omnibus Budget Reconciliation Act of 1985 (COBRA), coverage shall be provided for the child consistent with the Group's plan for postemployment health insurance coverage for Dependents.
- B. <u>Administration</u>. When the child subject to an MCSO or QMSO does not reside with the Subscriber, CareFirst BlueChoice will:
 - 1. Send to the non-insuring custodial parent the identification cards, claim forms, the applicable In-Network Evidence of Coverage, and any information needed to obtain benefits;
 - 2. Allow the non-insuring custodial parent or a provider of a Covered Service to submit a claim without the approval of the Subscriber; and
 - 3. Provide benefits directly to:
 - a) The non-insuring parent;
 - b) The provider of the Covered Services, Covered Dental Services or Covered Vision Services; or
 - c) The appropriate child support enforcement agency of any state or the District of Columbia.
- 2.7 <u>Clerical or Administrative Error</u>. If an individual is ineligible for coverage, the individual cannot become eligible just because CareFirst BlueChoice, the Group or the SHOP Exchange made a clerical or administrative error in recording or reporting information. Likewise, if a Member is eligible for coverage, the Member will not lose coverage because CareFirst BlueChoice, the Group or the SHOP Exchange made an administrative or clerical error in recording or reporting information.
- 2.8 Cooperation and Submission of Information. CareFirst BlueChoice may require verification from the Group and/or Subscriber pertaining to the eligibility of a Subscriber or Dependent enrolled hereunder. The Group and/or Subscriber agree to cooperate with and assist CareFirst BlueChoice and the SHOP Exchange, including providing CareFirst BlueChoice and the SHOP Exchange with reasonable access to Group records upon request. At any time coverage under this In-Network Evidence of Coverage is in effect, CareFirst BlueChoice reserves the right to request documentation substantiating eligibility as described in this In-Network Evidence of Coverage and to provide any information it receives regarding a Member's eligibility to the Group or the SHOP Exchange.

2.9	<u>Proof of Eligibility</u> . CareFirst BlueChoice retains the right to require proof of relationships or facts to establish eligibility. CareFirst BlueChoice will pay the reasonable cost of providing such proof.

SECTION 3 TERMINATION OF COVERAGE

- A. In the manner permitted by the SHOP Exchange, the Subscriber may terminate his or her enrollment, or (except as provided in Section 3.5) the enrollment of a Dependent, during the Annual Open Enrollment Period by notifying the Group or the SHOP Exchange. Upon termination of the Subscriber's enrollment, the enrollment of all enrolled Dependents will also be terminated.
- B. Termination of Enrollment by the Subscriber Due to Qualifying Events.
 - 1. If certain life events occur, a Subscriber may be able to make a mid-year change to reduce and/or terminate the coverage of the Subscriber or Dependent. The following is a list of qualifying life events that allow the Subscriber to reduce or terminate coverage. The changes in coverage must satisfy the consistency requirements as described below.
 - a) Qualifying Life Events:
 - (1) Legal marital status. A change in a Member's legal marital status, including marriage, divorce, death of spouse, a legal separation or an annulment.
 - (2) Employment status. A change in a Subscriber's, Spouse's or Dependent's employment status due to termination or commencement of employment, a strike or lockout, an unpaid leave of absence, or a change in worksite.
 - (3) Dependent status. A change in status of a Dependent that results in the Dependent's eligibility or ineligibility for coverage because of age or similar circumstances.
 - (4) Any of the qualifying events listed in Section 2.5C of the In-Network Evidence of Coverage, not otherwise specified in this provision.
 - (5) Any reduction or termination that a Subscriber makes must be consistent with the life event. The life event must affect eligibility for coverage under the plan or under a plan of the spouse or Dependent, which covers the spouse or Dependent as a Subscriber. The change in coverage must correspond with the life event.
 - 2. Under certain circumstances, a Subscriber may make mid-year reduction or termination to coverage for reasons, such as coverage cost or Medicare eligibility as described below.
 - a) Coverage Events:
 - (1) If there is reduction or elimination of coverage during the Benefit Period.
 - (2) If the Spouse has a cafeteria plan which allows a Subscriber and Dependents to make an enrollment change during that plan's annual open enrollment period, the Subscriber may make a corresponding mid-year change.

- b) Cost Events: If the cost of coverage increases or decreases significantly during a Benefit Period (including a Subscriber's change from part-time to full-time work or vice versa) and the Group does not offer a similar, but less costly, coverage option.
- c) Entitlement to Medicaid, the federal child health insurance plan (CHIP) or a State-funded low-income basic health plan (known as a BHP). If a Subscriber or Dependent becomes eligible for Medicaid, the federal child health insurance plan (CHIP) or a State-funded low-income basic health plan (known as a BHP) mid-year, a Subscriber or Dependent *may* (but is not required) terminate coverage.
- C. The effective date of the termination will be as follows:
 - 1. On the date stated by the Subscriber, if the Subscriber has given reasonable notice. For purposes of this provision, reasonable notice is defined as fourteen (14) days from the requested termination date.
 - 2. Fourteen (14) days after the date the Subscriber requested termination if the Subscriber does not provide reasonable notice.
 - 3. If the Subscriber and Dependents give notice of termination of enrollment in order to enroll in another Qualified Health Plan, the day before the effective date of coverage under the new Qualified Health Plan.
 - 4. If the Subscriber and Dependents are newly eligible for Medicaid, the federal child health insurance plan (CHIP) or a State-funded low-income basic health plan (known as a BHP), the day before coverage under one of these programs begins.
 - 5. If the Subscriber terminates enrollment during an Annual Open Enrollment Period, on the last day of the Benefit Period.
- 3.2 <u>Termination of Individual Enrollment by the CareFirst BlueChoice or the SHOP Exchange.</u>
 CareFirst BlueChoice or the SHOP Exchange may terminate the coverage of a Subscriber and/or a Dependent under the following circumstances by providing the Subscriber at least thirty (30) days' notice prior to the last day of coverage:
 - A. Termination of Individual Enrollment for Ineligibility:
 - 1. The Subscriber is no longer eligible for coverage for any reason. In this circumstance, the enrollment of the Subscriber and any Dependents will be terminated.
 - 2. A Dependent is no longer eligible for coverage as a Dependent due to a change in the Dependent's age, status or relationship to the Subscriber, or no longer meets the eligibility requirements established by the Group.
 - 3. The date of termination is stated in the Eligibility Schedule.
 - 4. The Subscriber is responsible for notifying the SHOP Exchange of any changes in the status of a Dependent as an eligible individual; or his or her eligibility for coverage, except when the Dependent Child reaches the Limiting Age. These changes include a divorce and the marriage of a Dependent Child. If the Subscriber knows of a Dependent's ineligibility for coverage and intentionally fails to notify the SHOP Exchange of these types of changes, CareFirst BlueChoice has the right to seek Rescission of the coverage of the Dependent under Section 3.3 as of the initial date of the Dependent's ineligibility and

recover the full value of the services and benefits provided during the period of Dependent's ineligibility. CareFirst BlueChoice can recover these amounts from the Subscriber and/or from the Dependent, less the premium paid during the period of ineligibility, at the option of CareFirst BlueChoice.

- B. Termination of In-Network Group Contract due to the Decertification of the In-Network Evidence of Coverage as a Qualified Health Plan. If this In-Network Evidence of Coverage is Decertified as a Qualified Health Plan, the date of termination shall be the date established by the SHOP Exchange after written notice has been provided to the Subscriber and the Subscriber has been afforded an opportunity to enroll in other coverage.
- C. <u>Accommodation for Persons with Disabilities</u>. Notwithstanding the termination provisions above, CareFirst BlueChoice, when required by the SHOP Exchange, shall make reasonable accommodation of these provisions for all individuals with disabilities (as defined by the Americans with Disabilities Act) before terminating the coverage of such individuals.
- 3.3 <u>Rescission of Individual Enrollment for Fraud or Misrepresentation</u>. Coverage of a Member may be Rescinded if:
 - A. The Member has performed an act, practice, or omission that constitutes fraud;
 - B. The Member has made an intentional misrepresentation of material fact; or
 - C. An act, practice or omission that constitutes fraud includes, but is not limited to, fraudulent use of CareFirst BlueChoice's identification card by the Member, the alteration or sale of prescriptions by the Member, or an attempt by a Subscriber to enroll non-eligible persons.

CareFirst BlueChoice will provide thirty (30) days advance written notice of any Rescission. CareFirst BlueChoice shall have the burden of persuasion that its Rescission complies with applicable local law. The Rescission shall either (i) void the enrollment of the Member as of the Member's Effective Date (for fraudulent acts, practices, or omissions that occur at the time of enrollment); or (ii) in all other cases, void the enrollment of the Member as of the first date the Member performed an act, practice or omission which constituted fraud or made an intentional misrepresentation of material fact. The Subscriber will be responsible for payment of any voided benefits paid by CareFirst BlueChoice, net of applicable premiums paid.

- 3.4 <u>Death of Subscriber</u>. In the event of the Subscriber's death, coverage of any Dependents may continue under the Subscriber's enrollment as stated in this In-Network Evidence of Coverage. The date of termination of the Subscriber's enrollment and, if applicable, the enrollment of any Dependents will be as stated in the Eligibility Schedule.
- 3.5 <u>Medical Child Support Orders or Qualified Medical Support Orders.</u> Unless coverage is Rescinded or terminated for non-payment of the premium, a child subject to an MCSO/QMSO may not be terminated unless written evidence is provided to CareFirst BlueChoice or the SHOP Exchange that:
 - A. The MCSO/QMSO is no longer in effect;
 - B. The child has been or will be enrolled under other comparable health insurance coverage that will take effect not later than the date of the termination of coverage;
 - C. The Group has eliminated family member coverage for all Members; or

- D. The Group no longer employs the Subscriber, except if the Subscriber elects continuation coverage under applicable state or federal law the child will continue in this postemployment coverage.
- 3.6 <u>Termination of In-Network Evidence of Coverage upon Termination of In-Network Group Contract</u>. This In-Network Evidence of Coverage, and the enrollment of the Member(s), will terminate automatically upon the date of the termination of the In-Network Group Contract by the Group, the SHOP Exchange or CareFirst BlueChoice for any reason.
- 3.7 <u>Effect of Termination</u>. Except as provided in Section 4.3, no benefits will be provided for any services received on or after the date on which the Member's coverage terminates. This includes services received for an injury or illness that occurred before the date of termination.
- 3.8 <u>No Reinstatement</u>. Upon termination, enrollment will not reinstate automatically under any circumstances.

SECTION 4 CONTINUATION OF COVERAGE

- 4.1 <u>Continuation of Eligibility upon Loss of Group Coverage.</u>
 - A. Federal Continuation of Coverage under COBRA. If the Group health benefit plan provided under this In-Network Evidence of Coverage is subject to the Consolidated Omnibus Budget Reconciliation Act of 1985 (COBRA), as amended from time to time, and a Member's coverage terminates due to a "Qualifying Event" as described under COBRA, continuation of participation in this Group health benefit plan may be possible. The employer offering this Group health benefit plan is the plan administrator. It is the plan administrator's responsibility to notify a Member concerning terms, conditions and rights under COBRA. If a Member has any questions regarding COBRA, the Member should contact the plan administrator.
 - B. <u>Uniformed Services Employment and Reemployment Rights Act (USERRA)</u>. USERRA protects the job rights of individuals who voluntarily or involuntarily leave employment positions to undertake military service or certain types of service in the Natural Disaster Medical System. USERRA also prohibits employers, and insurers, from discriminating against past and present members of the uniformed services, and applicants to the uniformed services.

If an eligible employee leaves his or her job to perform military service, the eligible employee has the right to elect to continue his or her Group coverage including any Dependents for up to twenty-four (24) months while in the military. Even if continuation of coverage was not elected during the eligible employee's military service, the eligible employee has the right to be reinstated in their Group coverage when re-employed, without any Waiting Periods or pre-existing condition exclusion periods except for service connected illnesses or injuries. If an eligible employee has any questions regarding USERRA, the eligible employee should contact the plan administrator. The plan administrator determines eligible employees and provides the information to CareFirst BlueChoice.

- 4.2. <u>District of Columbia Continuation of Health Coverage (DCCHC)</u>. This provision applies to Subscribers enrolled in an employer-maintained health benefit plan for less than twenty (20) employees.
 - A. The Subscriber and any Dependents have the right to continue coverage under the In-Network Group Contract for a period of three (3) months, or for the period of time during which the Subscriber is eligible for premium assistance under the American Recovery and Reinvestment Act of 2009, as amended, unless:
 - 1. The Subscriber's employment was terminated for gross misconduct;
 - 2. The Member is eligible for any extension of coverage required under COBRA; or
 - 3. The Member fails to complete timely election and payment as provided below.
 - B. <u>Duties of the Group.</u>
 - 1. The Group shall furnish Subscribers whose coverage terminates with written notification of the Subscriber's eligibility to continue coverage under DCCHC. Such notice shall be furnished no later than fifteen (15) days of the date coverage under this In-Network Evidence of Coverage would otherwise terminate. Failure by the Group to furnish the required notification shall not extend the right to continue coverage beyond the three-month period, or for the period of time during which the Subscriber is eligible for premium assistance under the American Recovery and Reinvestment Act of 2009, as amended.

2. The Group shall forward to CareFirst BlueChoice the names of Members who apply for DCCHC Continuation of Coverage within fifteen (15) days from the date of application.

C. <u>Duties of the Subscriber</u>.

- 1. Individuals who elect coverage under this Section shall bear the cost of the continued coverage and such cost shall not exceed one hundred two percent (102%) of the Group's rate.
- 2. An individual who elects to continue coverage shall tender to the Group the amount described above within forty-five (45) days from the date coverage under this In-Network Evidence of Coverage would otherwise terminate.
- D. <u>Termination of Continued Coverage</u>. Coverage under this provision shall continue without interruption for the continued eligibility period and shall not terminate unless:
 - 1. The Member establishes residence outside CareFirst BlueChoice Service Area;
 - 2. The Member fails to make timely payment of the required cost of coverage;
 - 3. The Member violates a material condition of this In-Network Evidence of Coverage;
 - 4. The Member becomes covered under another group health benefits plan that does not contain any exclusion or limitation with respect to any preexisting condition that affects the Member;
 - 5. The Member becomes entitled to Medicare; or
 - 6. The Group no longer offers group coverage to any employee.
- 4.3 Extension of Benefits: Medical Services, Covered Dental Services, and Covered Vision Services. In the event of termination of this In-Network Evidence of Coverage, any Member who became totally disabled while enrolled under this In-Network Evidence of Coverage and who continues to be totally disabled at the date of termination will, upon payment of premium, be entitled to continued coverage under this In-Network Evidence of Coverage until the first of the following:
 - A. The date the Member is, in the judgment of CareFirst BlueChoice, no longer totally disabled;
 - B. The date that a succeeding carrier elects to provide replacement coverage to that Member without limitation as to the disabling condition; or
 - C. One-hundred and eighty (180) days following termination.
- 4.4 Right to Continue Coverage Under Only One Provision. If a Member is eligible to continue coverage under the In-Network Group Contract under more than one continuation provision as described in Sections 4.2 and 5.1, the Member will receive only one such continuation coverage. The In-Network Group will select the continuation option the member will receive.

SECTION 5 COORDINATION OF BENEFITS (COB); SUBROGATION

5.1 <u>Coordination of Benefits (COB)</u>.

A. Applicability.

- 1. This Coordination of Benefits (COB) provision applies to this CareFirst BlueChoice Plan when a Member has health care coverage under more than one Plan
- 2. If this COB provision applies, the Order of Benefit Determination Rules should be reviewed first. Those rules determine whether the benefits of this CareFirst BlueChoice Plan are determined before or after those of another Plan. The benefits of this CareFirst BlueChoice Plan:
 - Shall not be coordinated when, under the order of determination rules, this CareFirst BlueChoice Plan determines its benefits before another Plan;
 - b) May be coordinated when, under the order of determination rules, another Plan determines its benefits first. The coordination is explained in Section 6.1D.2.
- B. <u>Definitions</u>. For the purpose of this COB section, the following terms are defined. The definitions of other capitalized terms are found in the definitions section of this In-Network Evidence of Coverage.

Allowable Expenses means any health care expense, including deductibles, coinsurance or copayments that are covered in whole or in part by any of the Plans covering the Member. This means any expense or portion of an expense not covered by any of the Plans is not an Allowable Expense. If this CareFirst BlueChoice Plan is advised by a Member that all Plans covering the Member are high-deductible health plans and the Member intends to contribute to a health savings account, the primary Plan's deductible is not an Allowable Expense, except for any health care expense incurred that may not be subject to the deductible, as stated in Section 223(c)(2)(C) of the Internal Revenue Code of 1986.

<u>CareFirst BlueChoice Plan</u> means this In-Network Evidence of Coverage.

<u>Dental Plan</u> means any dental insurance policy, including those of nonprofit health service plans, and those of commercial group, blanket and individual policies, any subscriber contracts issued by Health Maintenance Organizations (HMOs), and any other established programs under which the insured may make a claim. The term Dental Plan includes coverage under a governmental plan, or coverage required to be provided by law. This does not include a State plan under Medicaid (Title XIX, Grants to States for Medical Assistance Programs, of the United States Social Security Act, as amended from time to time.)

<u>Intensive Care Policy</u> means a health insurance policy that provides benefits only when treatment is received in a specifically designated health care facility of a hospital that provides the highest level of care and which is restricted to those patients who are physically, critically ill or injured.

<u>Plan</u> means any health insurance policy issued on a group basis, including those of a nonprofit health service plan, those of a commercial, group, and blanket policy, any subscriber contracts issued by health maintenance organizations, and any other established programs under which the insured may make a claim. The term Plan includes

coverage required or provided by law and coverage under a governmental plan, except a governmental plan which, by law, provides benefits in excess of those of any private insurance plan or other non-governmental plan. This does not include a state plan under Medicaid (Title XIX, Grants to States for Medical Assistance Programs, of the United States Social Security Act, as amended from time to time).

The term Plan does not include:

- 1. An individually underwritten and issued, guaranteed renewable, specified disease policy;
- 2. An intensive care policy, which does not provide benefits on an expense incurred basis:
- 3. Coverage regulated by a motor vehicle reparation law;
- 4. The first one-hundred dollars (\$100) per day of a hospital indemnity contract; or
- 5. An elementary and/or secondary school insurance program sponsored by a school or school system.

<u>Primary Plan or Secondary Plan</u> means the order of benefit determination rules stating whether this CareFirst BlueChoice Plan is a Primary Plan or Secondary Plan as to another Plan covering the Member.

- 1. When this CareFirst BlueChoice Plan is a Primary Plan, its benefits are determined before those of the other Plan and without considering the other Plan's benefits.
- 2. When this CareFirst BlueChoice Plan is a Secondary Plan, its benefits are determined after those of the other Plan and may be coordinated because of the other Plan's benefits.
- 3. When there are more than two Plans covering the Member, this CareFirst BlueChoice Plan may be a Primary Plan as to one of the other Plans, and may be a Secondary Plan as to a different Plan or Plans.

<u>Specified Disease Policy</u> means a health insurance policy that provides (1) benefits only for a disease specified in the policy or for the treatment unique to a specific disease; or (2) additional benefits for a disease specified in the policy or for treatment unique to a specified disease.

C. Order of Benefit Determination Rules.

- 1. General. When there is a basis for a claim under this CareFirst BlueChoice Plan and another Plan, this CareFirst BlueChoice Plan is a Secondary Plan which has its benefits determined after those of the other Plan, unless;
 - a) The other Plan has rules coordinating benefits with those of this CareFirst BlueChoice Plan; and
 - b) Both those rules and this CareFirst BlueChoice Plan's rules require this CareFirst BlueChoice Plan's benefits be determined before those of the other Plan.
- 2. Rules. This CareFirst BlueChoice Plan determines its order of benefits using the first of the following rules which applies:

- a) Non-dependent/dependent. The benefits of the Plan which covers the person as an employee, member or subscriber (that is, other than as a dependent) are determined before those of the Plan which covers the person as a dependent; except if the person is also a Medicare beneficiary, and the result of the rule established by Title XVIII of the Social Security Act and implementing regulations, Medicare is:
 - (1) Secondary to the Plan covering the person as a dependent, and
 - (2) Primary to the Plan covering the person as other than a dependent (e.g., retired employee),

Then the benefits of the Plan covering the person as a dependent are determined before those of the Plan covering the person as other than a dependent.

- b) Dependent child covered by more than one Plan. Unless there is a court decree stating otherwise, when this CareFirst BlueChoice Plan and another Plan cover the same child as a dependent, the order of benefits shall be determined as follows:
 - (1) For a dependent child whose parents are married or are living together:
 - (a) The benefits of the Plan of the parent whose birthday falls earlier in a year are determined before those of the Plan of the parent whose birthday falls later in the year; but
 - (b) If both parents have the same birthday, the benefits of the Plan that covered one parent longer are determined before those of the Plan that covered the other parent for a shorter period of time.
 - (2) For a dependent child whose parents are separated, divorced, or are not living together:
 - (a) If the specific terms of a court decree state one of the parents is responsible for the health care expenses or health care coverage of the child, and the entity obligated to pay or provide the benefits of the Plan of that parent has actual knowledge of those terms, the benefits of that Plan are determined first. If the parent with responsibility has no health care coverage for the dependent child's health care expenses, but the parent's Spouse does, that parent's Spouse's plan is the primary plan. This paragraph does not apply with respect to any claim for services rendered before the entity has actual knowledge of the terms of the court decree.

The rule described in (1) above also shall apply if: i) a court decree states both parents are responsible for the dependent child's health care expenses or health care coverage, or ii) a court decree states the parents have joint custody without specifying one parent has responsibility for the health care expenses or coverage of the dependent child.

(b) If there is no court decree setting out the responsibility

for the child's health care expenses or health care coverage, the order of benefits for the dependent child are as follows:

- i) The Plan of the parent with custody of the child;
- ii) The Plan of the Spouse of the parent with the custody of the child;
- iii) The Plan of the parent not having custody of the child; and then
- iv) The Plan of the Spouse of the parent who does not have custody of the child.
- (3) For a dependent child covered under more than one plan of individuals who are not the parents of the child, the order of benefits shall be determined, as applicable, under the rules stated in (1) and (2) of this paragraph as if those individuals were parents of the child.
- c) Active/inactive employee. The benefit of a Plan which covers a person as an employee who is neither laid off nor retired is determined before those of a Plan that covers that person as a laid off or retired employee. The same would hold true if a person is a dependent of a person covered as an employee who is neither laid off nor retired or a person covered as a laid off or retired employee. If the other Plan does not have this rule, and if, as a result, the Plans do not agree on the order of benefits, this rule is ignored.
- d) Continuation coverage. If a person whose coverage is provided under the right of continuation pursuant to federal, state or local law also is covered under another Plan, the following shall be the order of benefits determination:
 - (1) First, the benefits of a Plan covering the person as an employee, retiree, member or subscriber (or as that person's dependent);
 - (2) Second, the benefits under the continuation coverage.

If the other Plan does not have the rule described above, and if, as a result, the Plans do not agree on the order of benefits, this rule is ignored.

- e) Longer/shorter length of coverage. If none of the above rules determines the order of benefits, the benefits of the Plan that covered the person longer are determined before those of the Plan that covered that person for the shorter term.
- f) Medical and Dental Plan. When one of the plans is a medical plan and the other is a Dental Plan, and a determination cannot be made in accordance with the above, the medical plan should be considered as the Primary Plan.
- D. Effect on the Benefits of this CareFirst BlueChoice Plan.
 - 1. When this Section Applies. This section applies when, in accordance with the prior section, order of benefits determination rules, this CareFirst BlueChoice Plan is a Secondary Plan as to one or more other Plans. In such an event, the

- benefits of this CareFirst BlueChoice Plan may be coordinated under this section. Any additional other Plan or Plans are referred to as "the other Plans" immediately below.
- 2. Coordination in this CareFirst BlueChoice Plan's Benefits. When this CareFirst BlueChoice Plan is the Secondary Plan, the benefits under this CareFirst BlueChoice Plan *may* be coordinated so that the total benefits would be payable or provided by all the other Plans do not exceed one hundred percent (100%) of the total Allowable Expenses. If the benefits of this CareFirst BlueChoice Plan are coordinated, each benefit is coordinated in proportion. It is then charged against any applicable benefit limit of this CareFirst BlueChoice Plan.
- E. <u>Right to Receive and Release Needed Information</u>. Certain facts are needed to apply these COB rules. CareFirst BlueChoice has the right to decide which facts it needs. It may get the needed facts from or give them to any other organization or person for purposes of treatment, payment, and health care operations. CareFirst BlueChoice need not tell, or get the consent of, any person to do this. Each person claiming benefits under this CareFirst BlueChoice Plan must give this CareFirst BlueChoice Plan any facts it needs to pay the claim.
- F. Facility of Payment. A payment made under another Plan may include an amount that should have been paid under this CareFirst BlueChoice Plan. If it does, this CareFirst BlueChoice Plan may pay the amount to the organization that made that payment. The amount will then be treated as though it were a benefit paid under this CareFirst BlueChoice Plan. This CareFirst BlueChoice Plan will not have to pay the amount again. The term "payment made" includes providing benefits in the form of services, in which case "payment made" means the reasonable cash value of the benefits provided in the form of services.
- G. <u>Right of Recovery</u>. If the amount of the payments made by this CareFirst BlueChoice is more than it should have paid under this COB provision, it may recover the excess from one or more of:
 - 1. The persons it has paid or for whom it has paid;
 - 2. Insurance companies; or
 - 3. Other organizations.

The "amount of the payments made" includes the reasonable cash value of any benefits provided in the form of services.

- 5.2 <u>Medicare Eligibility</u>. This provision applies to Members who are entitled to Part A and/or Part B of Medicare. A Member will not be terminated as a result of reaching the age of sixty-five (65) or becoming eligible for Medicare. Benefits not covered by Medicare will be provided as described in the In-Network Evidence of Coverage. Benefits covered by Medicare are subject to the provisions in this section.
 - A. <u>Coverage Secondary to Medicare</u>. Except where prohibited by law, the benefits under this CareFirst BlueChoice Plan are secondary to Medicare.
 - B. Medicare as Primary.
 - 1. When benefits for Covered Services, Covered Dental Services or Covered Vision Services are paid by Medicare as primary, this CareFirst BlueChoice Plan will not duplicate those payments. CareFirst BlueChoice will coordinate and pay benefits based on Medicare's payment (or the payment Medicare would have paid). When CareFirst BlueChoice coordinates the benefits with Medicare, CareFirst

BlueChoice's payments will be based on the Medicare allowance (if the provider is a participating provider in Medicare) or the Medicare maximum limiting charge (if the provider is not a participating provider in Medicare), less any claim reduction or denial due to a Member's failure to comply with Medicare's administrative requirements. CareFirst BlueChoice's right to coordinate is not contingent on any payment actually being made on the claim by Medicare. Members enrolled in Medicare agree to, and shall, complete and submit to Medicare, CareFirst BlueChoice, and/or any health care providers all claims, consents, releases, assignments and other documents required to obtain or assure such claim payment by Medicare.

- 2. If a Medicare-eligible Member has not enrolled in Medicare Part A and/or Part B, CareFirst BlueChoice will not "carve-out," coordinate, or reject a claim based on the amount Medicare would have paid had the Member actually applied for, claimed, or received Medicare benefits.
- 5.3 <u>Employer or Governmental Benefits</u>. Coverage does not include the cost of services or payment for services for any illness, injury, or condition for which, or as a result of which, a Benefit (as defined below) is provided or is required to be provided either:
 - A. Under any federal, state, county or municipal workers' compensation or employer's liability law or other similar program; or
 - B. From any federal, state, county or municipal or other government agency, including, in the case of service-connected disabilities, the United States Department of Veterans Affairs, to the extent that benefits are payable by the federal, state, county or municipal or other government agency, but excluding Medicare benefits and Medicaid benefits.

<u>Benefit</u> as used in this provision includes a payment or any other benefit, including amounts received in settlement of a claim for Benefits.

5.4 <u>Subrogation</u>. CareFirst BlueChoice has subrogation and reimbursement rights. Subrogation requires the Member to turn over to CareFirst BlueChoice any rights the Member may have against a third party. A third party is any person, corporation, insurer or other entity that may be liable to a Member for an injury or illness. This right applies to the amount of benefits paid by CareFirst BlueChoice for injuries or illnesses where a third party could be liable.

Recovery means to be successful in a lawsuit, to collect or obtain an amount; to obtain a favorable or final judgment; to obtain an amount in any legal manner; an amount finally collected or the amount of judgment as a result of an action brought against a third-party or involving uninsured or underinsured motorist claims. A Recovery does not include payments made to the Member under the Member's personal injury protection policy. CareFirst BlueChoice will not recover medical expenses from a Subscriber unless the Subscriber or Member recovers for medical expenses in a cause of action or settlement.

- A. The Member shall notify CareFirst BlueChoice as soon as reasonably possible that a third-party may be liable for the injuries or illnesses for which benefits are being provided or paid.
- B. To the extent actual payments made by CareFirst BlueChoice result from the occurrence that gave rise to the cause of action, CareFirst BlueChoice shall be subrogated and succeed to any right of recovery of the Member against any person or organization.
- C. The Member shall pay CareFirst BlueChoice the amount recovered by suit, settlement, or otherwise from any third-party's insurer, any uninsured or underinsured motorist coverage, or as permitted by law, to the extent any actual payments made by CareFirst BlueChoice result from the occurrence that gave rise to the cause of action.

- D. The Member shall furnish information and assistance, and execute papers that CareFirst BlueChoice may require to facilitate enforcement of these rights. The Member shall not commit any action prejudicing the rights and interests of CareFirst BlueChoice.
- E. In a subrogation claim arising out of a claim for personal injury, the amount recovered by CareFirst BlueChoice may be reduced by:
 - 1. Dividing the total amount of the personal injury recovery into the total amount of the attorney's fees incurred by the injured person for services rendered in connection with the injured person's claim; and
 - 2. Multiplying the result by the amount of CareFirst BlueChoice's subrogation claim. This percentage may not exceed one-third (1/3) of CareFirst BlueChoice's subrogation claim.
- F. On written request by CareFirst BlueChoice, a Member or Member's attorney who demands a reduction of the subrogation claim shall provide CareFirst BlueChoice with a certification by the Member that states the amount of the attorney's fees incurred.
- G. These provisions do not apply to residents of the Commonwealth of Virginia.

SECTION 6 GENERAL PROVISIONS

6.1 <u>Entire Certificate; Changes</u>. The entire In-Network Evidence of Coverage includes: (a) this In-Network Evidence of Coverage; (b) Benefit Determinations and Appeals; (c) the Description of Covered Services; (d) Schedule of Benefits; (e) Eligibility Schedule; and (f) any additional duly authorized notices, amendments and riders.

No amendment or modification of any term or provision of this In-Network Evidence of Coverage is effective unless authorized in writing by an executive officer of CareFirst BlueChoice. Any duly authorized notice, amendment or rider will be issued by CareFirst BlueChoice to be attached to the In-Network Evidence of Coverage. No agent has authority to change this Agreement or to waive any of its provisions. Any waiver of an In-Network Evidence of Coverage term or provision shall only be given effect for its stated purpose and shall not constitute or imply any subsequent waiver.

Oral statements cannot be relied upon to modify or otherwise affect the benefits, limitations and/or exclusions of this In-Network Evidence of Coverage, or increase or void any coverage or reduce any benefits. Such oral statements cannot be used in the prosecution or defense of a claim.

6.2 Claims and Payment of Claims.

A. <u>Claim Forms</u>. CareFirst BlueChoice does not require a written notice of claims. A claim form can be requested by calling the Member and Provider Service telephone number on the identification card during regular business hours. CareFirst BlueChoice shall provide claim forms for filing proof of loss to each claimant or to the Group for delivery to the claimant. If CareFirst BlueChoice does not provide the claim forms within fifteen (15) days after notice of claim is received, the claimant is deemed to have complied with the requirements of the policy as to proof of loss if the claimant submits, within the time fixed in the policy for filing proof of loss, written proof of the occurrence, character, and extent of the loss for which the claim is made.

When a child subject to a Medical Child Support Order or a Qualified Medical Support Order does not reside with the Subscriber, CareFirst BlueChoice will:

- 1. Send the non-insuring, custodial parent identification cards, claims forms, the applicable certificate of coverage or member contract, and any information needed to obtain benefits;
- 2. Allow the non-insuring, custodial parent or a provider of a Covered Service to submit a claim without the approval of the Subscriber; and
- 3. Provide benefits directly to:
 - a) The non-insuring, custodial parent;
 - b) The provider of the Covered Services, Covered Dental Services, or Covered Vision Services; or
 - c) The appropriate child support enforcement agency of any state or the District of Columbia.

B. Proof of Loss.

For Covered Services provided by Contracting Providers, Preferred Dentists, Contracting Vision Providers, and Contracting Pharmacies, Members are not required to submit claims in order to obtain benefits.

For Covered Services provided by Non-Contracting Providers, Non-Participating Dentists, Non-Contracting Vision Providers, and Non-Contracting Pharmacies, Members must furnish written proof of loss, or have the provider submit proof of loss, to CareFirst BlueChoice within one (1) year after the date of the loss. The Member is also responsible for providing information requested by CareFirst BlueChoice, including, but not limited to, medical records.

Failure to furnish proof within the time required shall not invalidate or reduce any claim if it was not reasonably possible to give proof within the required time, provided proof is furnished as soon as reasonably possible and in no event, except in the absence of legal capacity, later than one (1) year from the time proof is otherwise required.

CareFirst BlueChoice will honor claims submitted for Covered Services, Covered Dental Services, or Covered Vision Services by any agency of the federal, state, or local government that has the statutory authority to submit claims beyond the time limits established under this In-Network Evidence of Coverage. These claims must be submitted to CareFirst BlueChoice before the filing deadline established by the applicable statute on claims forms that provide all of the information CareFirst BlueChoice deems necessary to process the claims. CareFirst BlueChoice provides forms for this purpose.

- C. <u>Time of Payment of Claims</u>. Benefits payable will be paid immediately after receipt of written proof of loss.
- D. <u>Claim Payments Made in Error</u>. If CareFirst BlueChoice makes a claim payment to or on behalf of a Member in error, the Member is required to repay CareFirst BlueChoice the amount paid in error. If the Member has not repaid the full amount owed CareFirst BlueChoice and CareFirst BlueChoice makes a subsequent benefit payment, CareFirst BlueChoice may subtract the amount owed CareFirst BlueChoice from the subsequent payment.
- E. Payment of Claims. Payments for Covered Services will be made by CareFirst BlueChoice directly to Contracting Vision Providers, Preferred Dentists and Contracting Providers. Direct payments will also be made by CareFirst BlueChoice to providers from the United States Department of Defense and the United States Department of Veteran Affairs. If a Member receives Covered Services from Non-Contracting Vision or Non-Preferred or Non-Participating Providers, CareFirst BlueChoice reserves the right to pay either the Member or the provider. If the Member has paid the health care provider for services rendered, benefits will be payable to the Member. The payment will, in either case, be full and complete satisfaction of CareFirst BlueChoice's obligation, unless an appeal or grievance has been filed by, or on behalf of, the Member.
- 6.3 No Assignment. A Member cannot assign any benefits or payments due under this In-Network Evidence of Coverage to any person, corporation or other organization, except as specifically provided by this In-Network Evidence of Coverage or required by applicable law.
- 6.4 <u>Legal Actions</u>. A Member cannot bring any lawsuit against CareFirst BlueChoice to recover under this In-Network Evidence of Coverage before the expiration of sixty (60) days after written proof of loss has been furnished, and not after three (3) years from the date written proof of loss is required to be submitted to CareFirst BlueChoice.
- 6.5 Events Outside of CareFirst BlueChoice's Control. If CareFirst BlueChoice, for any reason beyond the control of CareFirst BlueChoice, is unable to provide the coverage promised, CareFirst BlueChoice is liable for reimbursement of the expenses necessarily incurred by any Member in procuring the services through other providers, to the extent prescribed by law.
- 6.6 Physical Examinations and Autopsy. CareFirst BlueChoice, at its own expense, has the right and

opportunity to examine the Member when and as often as it may reasonably require during the pendency of a claim hereunder and to make an autopsy in case of death where it is not forbidden by law.

- 6.7 <u>Identification Card.</u> Any cards issued to Members are for identification only.
 - A. Possession of an identification card confers no right to benefits.
 - B. To be entitled to such benefits, the holder of the card must, in fact, be a Member on whose behalf all applicable premiums have actually been paid.
 - C. Any person receiving benefits to which he or she is not then entitled will be liable for the actual cost of such benefits.
- 6.8 Member Medical Records. It may be necessary to obtain Member medical records and information from hospitals, Skilled Nursing Facilities, physicians or other practitioners who treat the Member. When a Member becomes covered, the Member (and, if the Member is legally incapable of giving such consent, the representative of such Member) automatically gives CareFirst BlueChoice permission to obtain and use such records and information, including medical records and information requested to assist CareFirst BlueChoice in determining benefits and eligibility of Members.
- Member Privacy. CareFirst BlueChoice shall comply with state, federal and local laws pertaining to the dissemination or distribution of non-public personally identifiable financial, medical or health related data. In that regard, CareFirst BlueChoice will not provide to unauthorized third parties any personally identifiable financial or medical information without the prior written authorization of the Member or parent/guardian of the Member or as otherwise permitted by law. Personal information, including email addresses and phone numbers, may be used and shared with other businesses who work with CareFirst BlueChoice to administer and/or provide benefits under this plan. Personal information may also be used to notify enrollees about treatment options, health-related services, and/or coverage options. Enrollees may contact CareFirst BlueChoice to change the information used to communicate with them.
- Relationship of CareFirst BlueChoice to Health Care Providers. Health care providers, including Contracting Providers, Preferred Dentists, Contracting Vision Providers, and Contracting Pharmacy Providers, are independent contractors or organizations and are related to CareFirst BlueChoice by contract only. Contracting Providers, Preferred Dentists, Contracting Vision Providers, and Contracting Pharmacy Providers are not employees or agents of CareFirst BlueChoice and are not authorized to act on behalf of or obligate CareFirst BlueChoice with regard to interpretation of the terms of the In-Network Evidence of Coverage, including eligibility of Members for coverage or entitlement to benefits. Contracting Providers, Preferred Dentists, Contracting Vision Providers, and Contracting Pharmacy Providers maintain a provider-patient relationship with the Member and are solely responsible for the professional services they provide. CareFirst BlueChoice is not responsible for any acts or omissions, including those involving malpractice or wrongful death of Contracting Providers, Preferred Dentists, Contracting Vision Providers, Contracting Pharmacy Providers, or any other individual, facility or institution which provides services to Members or any employee, agent or representative of such providers.
- 6.11 <u>Provider and Services Information</u>. Listings of current Contracting Providers, Preferred Dentists, Contracting Vision Providers, and Contracting Pharmacy Providers will be made available to Members at the time of enrollment. Updated listings are available to Members upon request. The listing of Contracting Providers, Preferred Dentists and Contracting Vision Providers is updated every fifteen (15) days on the CareFirst BlueChoice website (www.CareFirst BlueChoice.com).
- 6.12 <u>Administration of In-Network Evidence of Coverage</u>. CareFirst BlueChoice may adopt reasonable policies, procedures, rules and interpretations to promote the orderly and efficient administration of this In-Network Evidence of Coverage.

- 6.13 <u>CareFirst BlueChoice's Relationship to the Group</u>. The Group is not an agent or representative and is not liable for any acts or omissions by CareFirst BlueChoice or any health care provider. CareFirst BlueChoice is not an agent or representative of the Group and is not liable for any act or omission of the Group.
- 6.14 <u>Delivery of In-Network Evidence of Coverage</u>. Unless CareFirst BlueChoice makes delivery directly to the Member, CareFirst BlueChoice will provide to the Group, for delivery to each Subscriber, a statement that summarizes the essential features of the coverage and states to whom benefits under the In-Network Evidence of Coverage are payable. Only one (1) statement will be issued for each family unit, except in the instance of an eligible child who is covered due to an MCSO/QMSO. In this instance, an additional In-Network Evidence of Coverage will be delivered to the custodial parent upon request.
- 6.15 <u>In-Network Evidence of Coverage Binding on Members</u>. The In-Network Evidence of Coverage can be amended, modified or terminated in accordance with any provision of the In-Network Evidence of Coverage or by mutual agreement between CareFirst BlueChoice and the Group without the consent or concurrence of Members. By electing coverage under this In-Network Evidence of Coverage, or accepting benefits under this In-Network Evidence of Coverage, Members are subject to all terms, conditions, and provisions of the In-Network Group Contract and In-Network Evidence of Coverage.
- 6.16 Payment of Contributions. The In-Network Group Contract is issued to the Group on a contributory basis in accordance with the Group's policies. The Group has agreed to collect from Members any contributory portion of the premium and pay to the SHOP Exchange the premium as specified in the In-Network Group Contract for all Members.
- Retirement Income Security Act of 1974, as amended (ERISA), the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended (COBRA), the Health Insurance Portability and Accountability Act of 1996 (HIPAA), and/or the Affordable Care Act) that relates to the health benefits provided under this In-Network Group Contract. For the purposes of ERISA and/or COBRA, the Group is the "plan administrator." As the plan administrator, it is the Group's responsibility to provide Members with certain information, including access to and copies of plan documents describing Member's benefits and rights to coverage under the Group health plan. Such rights include the right to continue coverage upon the occurrence of certain "Qualifying Events."
 - In any event, the Member should check with the Group to determine the Member's rights under ERISA, COBRA, HIPAA and/or the Affordable Care Act, as applicable.
- 6.18 Representations and not Warranties. All statements made by the Subscriber shall be deemed to be representations and not warranties. No statement made for the purpose of obtaining coverage shall void such coverage or reduce benefits unless contained in a written instrument signed by the Subscriber, a copy of which has been furnished to CareFirst BlueChoice.
- 6.19 <u>Rules for Determining Dates and Times</u>. The following rules will be used when determining dates and times:
 - A. All dates and times of day will be based on the dates and times applicable to the Washington, DC area, i.e., Eastern Standard Time or Eastern Daylight Savings Time, as applicable.
 - B. When reference is made to coverage being effective on a particular date, this means 12:01 a.m. on that date.
 - C. When reference is made to termination being effective on a particular date, this means 12:00 midnight on that date.

- D. "Days" mean calendar days, including weekends, holidays, etc., unless otherwise noted.
- E. "Year" refers to Calendar Year, unless a different benefit year basis is specifically stated.

6.20 Notices.

- A. <u>To the Member</u>. Notice to Members will be sent via electronic mail, if the Member has consented to receive such notices via electronic mail, or by first class mail to the most recent address or electronic address for the Member in CareFirst BlueChoice's files. It is the Subscriber's responsibility to notify the Group, and the Group's responsibility to notify CareFirst BlueChoice, of an address change. The notice will be effective on the date mailed, whether or not the Member in fact receives the notice or there is a delay in receiving the notice. The notice will be effective on the date mailed, whether or not the Member in fact receives the notice.
- B. <u>To CareFirst BlueChoice</u>. When notice is sent to CareFirst BlueChoice, it must be sent by first class mail to:

CareFirst BlueChoice, Inc. 840 First Street, NE Washington, DC 20065

- 1. Notice will be effective on the date of receipt by CareFirst BlueChoice, unless the notice is sent by registered mail, in which case the notice is effective on the date of mailing, as certified by the Postal Service.
- 2. CareFirst BlueChoice may change the address at which notice is to be given by giving written notice thereof to the Subscriber.
- 6.21 <u>Amendment Procedure.</u> Amendments must be consistent with federal and state law. CareFirst BlueChoice may amend the In-Network Evidence of Coverage with respect to any matter by the means and within the time frame allowed by the procedures established by the SHOP Exchange.
 - A. CareFirst BlueChoice will give notice of any amendment at least sixty (60) days before the effective date that coverage will be renewed. If a material modification required to conform the In-Network Evidence of Coverage to changes in applicable state or federal law is made at a time other than renewal and it affects the content of the summary of benefits and coverage, CareFirst BlueChoice will provide at least sixty (60) days advanced notice of the modification.

Regardless of when the amendment is received, the In-Network Evidence of Coverage is considered to be automatically amended as of the date specified in the contract amendment or the notice (if not stated in the contract amendment), unless otherwise mandated to conform with any applicable changes to state or federal law.

- B. No agent or other person, except an officer of CareFirst BlueChoice, has authority to waive any conditions or restrictions of the Evidence of Coverage, or to bind CareFirst BlueChoice by making any promise or representation or by giving or receiving any information. No change in the In-Network Evidence of Coverage will be binding on CareFirst BlueChoice, unless evidenced by an amendment signed by an authorized representative of CareFirst BlueChoice.
- 6.22 Regulation of CareFirst BlueChoice. CareFirst BlueChoice is subject to regulation in the District of Columbia by the Department of Insurance, Securities and Banking pursuant to Title 31 of the District of Columbia Code and the District of Columbia Department of Health pursuant to Reorganization Plan No. 4 of 1996, as amended.

- 6.23 <u>Conformity to Law.</u> Any provision in this In-Network Evidence of Coverage that is in conflict with the requirements of any state or federal law that applies to this In-Network Evidence of Coverage is automatically changed to satisfy the minimum requirements of such law.
- 6.24 <u>Selection of a Primary Care Physician</u>.
 - A. A Member must select a Primary Care Physician and may select any Primary Care Physician from CareFirst BlueChoice's current list of Contracting Providers. If the Primary Care Physician is not available, CareFirst BlueChoice will assist the Member in making another selection.
 - B. A Member may change his or her Primary Care Physician at any time by notifying CareFirst BlueChoice. If the Member notifies CareFirst BlueChoice by the twentieth (20th) day of the month, CareFirst BlueChoice will make the change effective the first day of the next month. If the Member notifies CareFirst BlueChoice after the twentieth (20th) day of the month, CareFirst BlueChoice will make the change effective the first day of the second month following the notice.
 - C. CareFirst BlueChoice may require a Member to change to a different Primary Care Physician if:
 - 1. The Member's Primary Care Physician is no longer available as a Primary Care Physician; or
 - 2. CareFirst BlueChoice determines that the furnishing of adequate medical care is jeopardized by a seriously impaired physician-patient relationship between the Member and his or her Primary Care Physician due to any of the following:
 - a. The Member engages in threatening or abusive behavior toward the physician, the physician's staff or other patients in the office; or
 - b. The Member attempts to take unauthorized controlled substances from the physician's office or to obtain these substances through fraud, misrepresentation, and forgery or by altering the physician's prescription order
 - D. If a change in Primary Care Physician is required, advance written notice will be given to the Member. The change is effective upon written notice to the Member. However, the Member may request a review of the action under Benefit Determinations and Appeals.
 - E. If a Member is required to change to another Primary Care Physician due to any of these circumstances, and there is a recurrence of the same or a similar situation with another Primary Care Physician, CareFirst BlueChoice may terminate the Member's coverage upon 31 days written notice. The Member's refusal to follow a prescribed course of treatment will not, in and of itself, be cause for termination.
 - CareFirst BlueChoice will not furnish any further benefits or services for a particular condition if the Member refuses to follow a prescribed course of treatment for that condition. If the Member disagrees with a prescribed course of treatment, the Member shall be permitted to receive a second opinion from another Contracting Physician. If the second physician disagrees with the prescribed course of treatment, CareFirst BlueChoice may not refuse to provide services or benefits for that particular condition, subject to this In-Network Evidence of Coverage and CareFirst BlueChoice's utilization review protocols and policies.
- 6.26 <u>Credit Monitoring.</u> CareFirst BlueChoice is offering credit monitoring to the Subscriber and eligible Dependents at no additional charge through services administered by Experian. Credit monitoring is available on an opt-in basis for all eligible Members during the effective Benefit

Period of their CareFirst BlueChoice health insurance policy. Eligible Members may enroll by calling the number on the back of their ID card or visiting www.carefirst.com.					

SECTION 7 SERVICE AREA

CareFirst BlueChoice's Service Area is a clearly defined geographic area in which CareFirst BlueChoice has arranged for the provision of health care services to be generally available and readily accessible to Members.

The Service Area is as follows: the District of Columbia; the state of Maryland; in the Commonwealth of Virginia, the cities of Alexandria and Fairfax, Arlington County, the town of Vienna and the areas of Fairfax and Prince Williams Counties in Virginia lying east of Route 123.

If a Member temporarily lives out of the Service Area (for example, if a Dependent goes to college in another state), the Member may be able to take advantage of the CareFirst BlueChoice Away From Home Program. This Program may allow a Member who resides out of the Service Area for an extended period of time to utilize the benefits of an affiliated Blue Cross and Blue Shield HMO. This Program is not coordination of benefits. A Member who takes advantage of the Away From Home Program will be subject to the rules, regulations and plan benefits of the affiliated Blue Cross and Blue Shield HMO. If the Member makes a permanent move, he/she does not have to wait until the Annual Open Enrollment Period to change plans. Please call 888-452-6403 or visit www.bcbs.com for more information on the Away from Home Program.

CareFirst BlueChoice, Inc.

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

ATTACHMENT A

BENEFIT DETERMINATIONS AND APPEALS

This attachment contains certain terms that have a specific meaning as used herein. These terms are capitalized and defined in Section A below, and/or in the In-Network Evidence of Coverage to which this document is attached.

These procedures replace all prior procedures issued by CareFirst BlueChoice, which afford CareFirst BlueChoice Members recourse pertaining to denials and reductions of claims for benefits by CareFirst BlueChoice.

These procedures only apply to claims for benefits. Notification required by these procedures will only be sent when a Member requests a benefit or files a claim in accordance with CareFirst BlueChoice procedures.

An authorized representative may act on behalf of the Member in pursuing a benefit claim or appeal of an Adverse Benefit Determination. CareFirst BlueChoice may require reasonable proof to determine whether an individual has been properly authorized to act on behalf of a Member. In the case of a claim involving Urgent/Emergent Care, a Health Care Provider with knowledge of a Member's medical condition is permitted to act as the authorized representative.

TABLE OF CONTENTS

- A. **DEFINITIONS**
- B. BENEFIT DETERMINATIONS
- C. INTERNAL GRIEVANCE PROCEDURE
- D. FAIR AND FULL REVIEW
- E. DEEMED EXHAUSTION OF INTERNAL CLAIMS AND APPEAL PROCESS
- F. EXTERNAL APPEAL PROCEDURE

A. **DEFINITIONS**

Adverse Benefit Determination means, as used in this attachment, the following:

- 1. A denial, reduction, or termination of, or a failure to provide or make payment (in whole or in part) for, a benefit, including any such denial, reduction, termination, or failure to provide or make payment that is based on a determination of a Member's eligibility to participate in this plan. An Adverse Benefit Determination includes a Rescission.
- 2. A denial, reduction, or termination of, or a failure to provide or make payment (in whole or in part) for, a benefit resulting from the application of any utilization review, as well as a failure to cover an item or service for which benefits are otherwise provided because it is determined to be Cosmetic, Experimental or Investigational, or not Medically Necessary or appropriate.

<u>Health Care Provider</u> means a physician or other health care professional licensed, accredited, or certified to perform specified health services consistent with State law.

<u>Pre-Service Claim</u> means any claim for a benefit when the receipt of the benefit, in whole or in part, is conditioned on the prior approval of the service in advance by CareFirst BlueChoice. These are

services that must be "preauthorized" or "precertified" by CareFirst BlueChoice under the terms of the Member's contract.

<u>Post Service Claim</u> means any claim for a benefit that is not a Pre-Service Claim.

<u>Rescission</u> means, as used in this attachment, a cancellation or discontinuance of coverage that has retroactive effect, except to the extent it is attributable to a failure to pay required premiums or contributions towards the cost of coverage.

<u>Urgent/Emergent Care</u> means a Pre-Service or Concurrent Care claim for medical care or with respect to which the application of the time periods for making non-Urgent/Emergent Care determinations:

- 1. Could seriously jeopardize the life or health of the Member or the ability of the Member to regain maximum function; or,
- 2. In the opinion of a Health Care Provider with knowledge of the Member's medical condition, would subject the Member to severe pain that cannot be adequately managed without the care or treatment that is the subject of the claim.

Whether a claim involves Urgent/Emergent Care is to be determined by an individual acting on behalf of CareFirst BlueChoice applying the judgment of a prudent layperson who possesses an average knowledge of health and medicine. If a Health Care Provider with knowledge of the Member's medical condition determines that a claim involves Urgent/Emergent Care then CareFirst BlueChoice will treat the claim as one that involves Urgent/Emergent Care.

B. BENEFIT DETERMINATIONS

- 1. Request for Urgent/Emergent Care Coverage. When the Member or authorized representative requests a pre-service determination regarding Urgent/Emergent Care, then CareFirst BlueChoice will notify the Member or authorized representative of the benefit determination (whether adverse or not) as soon as possible, taking into account the medical exigencies, the earlier of:
 - a. 24 hours after CareFirst BlueChoice's receipt of the necessary information to make the benefit determination, or
 - b. 72 hours after receipt of the request for coverage.

If a Member fails to provide sufficient information for CareFirst BlueChoice to determine whether benefits are covered or payable, CareFirst BlueChoice will notify the Member as soon as possible, but not later than 24 hours after receipt of the claim, of the specific information necessary to complete the claims. The Member shall be afforded a reasonable amount of time, taking into account the circumstances, but not less than 48 hours, to provide the specified information. CareFirst BlueChoice will notify the Member of the benefit determination as soon as possible, but in no case later than 48 hours after the earlier of:

- a. CareFirst BlueChoice's receipt of the specified information, or
- b. The end of the period afforded the Member to provide the specified additional information.
- 2. <u>Pre-Service Claims</u>. In the case of a Pre-Service Claim, CareFirst BlueChoice shall notify the Member of the benefit determination (whether adverse or not) within a reasonable period of time appropriate to the medical circumstances, but no later than 15 days after receipt of the claim.

This period may be extended one time by CareFirst BlueChoice for up to 15 days, provided that such an extension is necessary due to matters beyond the control of CareFirst BlueChoice and CareFirst BlueChoice notifies the Member, prior to the expiration of the initial 15-day period, of the circumstances requiring the extension of time and the date by which CareFirst BlueChoice expects to render a decision. If such an extension is necessary due to a failure of the Member to submit the information necessary to decide the claim, the notice of extension shall specifically describe the required information, and the Member will have at least 45 days from receipt of the notice within which to provide the specified information.

In the case of a failure by a Member or authorized representative to follow CareFirst BlueChoice procedures for filing a Pre-Service Claim, the Member or authorized representative shall be notified of the failure and the proper procedures to be followed in filing a claim for benefits. This notification shall be provided to the Member or authorized representative, as appropriate, as soon as possible, but not later than 5 working days following the failure. Notice will be sent within 24 hours in the case of a failure to file a claim involving Urgent/Emergent Care. Notification may be oral, unless written notification is requested by the Member or authorized representative.

This paragraph shall apply only in the case of a communication:

- a. By a Member or authorized representative that is received by CareFirst BlueChoice or its authorized agent customarily responsible for handling benefit matters; and,
- b. That names a specific Member; a specific medical condition or symptom; and a specific treatment, service, or product for which approval is requested.
- 3. Post-Service Claims. In the case of a Post-Service Claim, CareFirst BlueChoice shall notify the Member of the CareFirst BlueChoice's Adverse Benefit Determination within a reasonable period of time, but not later than 30 days after receipt of the claim. This period may be extended one time by CareFirst BlueChoice for up to 15 days, provided that CareFirst BlueChoice both determines that such an extension is necessary due to matters beyond the control of CareFirst BlueChoice and notifies the Member, prior to the expiration of the initial 30-day period, of the circumstances requiring the extension of time and the date by which CareFirst BlueChoice expects to render a decision. If such an extension is necessary due to a failure of the Member to submit the information necessary to decide the claim, the notice of extension shall specifically describe the required information, and the Member shall be afforded at least 45 days from receipt of the notice within which to provide the specified information.
- 4. <u>Concurrent Care Decisions</u>. If CareFirst BlueChoice has approved an ongoing course of treatment to be provided over a period of time or number of treatments:
 - a. CareFirst BlueChoice will notify the Member of any reduction or termination of such course of treatment (other than by a change in the plan's coverage by amendment or termination of coverage) before the end of such period of time or number of treatments and at a time sufficiently in advance of the reduction or termination to allow the Member to appeal and obtain a determination on review before the benefit is reduced or terminated.
 - b. Any request by a Member to extend the course of treatment beyond the period of time or number of treatments that is a claim involving Urgent/Emergent Care will be decided as soon as possible, taking into account the medical exigencies. CareFirst BlueChoice will notify the Member of the benefit determination, whether adverse or not, within 24 hours after receipt of the claim, provided that any such claim is made to CareFirst BlueChoice at least

24 hours prior to the expiration of the prescribed period of time or number of treatments.

- 5. <u>Rescissions.</u> If CareFirst BlueChoice has made an Adverse Determination that is a Rescission, CareFirst BlueChoice shall provide 30 days advance written notice to any covered person who would be affected by the proposed Rescission.
- 6. <u>Calculating Time Periods</u>. For purposes of this Part B, the period of time within which an Adverse Benefit Determination is required to be made shall begin at the time a claim is filed in accordance with CareFirst BlueChoice procedures. The time is counted regardless to whether all the information necessary to make a benefit determination accompanies the filing. In the event that a period of time is extended due to a Member's failure to submit information necessary to decide a claim, the period for making the benefit determination shall be tolled from the date on which the notification of the extension is sent to the Member until the date on which the Member responds to the request for additional information.

C. INTERNAL GRIEVANCE PROCEDURE

- 1. A grievance must be filed within 180 days from the date of receipt of the written notice of any Adverse Benefit Determination.
- 2. A Member or authorized representative should first contact CareFirst BlueChoice about a denial of benefits. CareFirst BlueChoice can provide information and assistance on how to file a grievance. All grievances filed should be in writing, except grievances involving Urgent/Emergent Care which may be submitted orally or in writing.
- 3. The Member or authorized representative may submit written comments, documents, records, and other information relating to a claim for benefits.
- 4. The grievance decision for Urgent/Emergent Care claim shall be made as soon as possible but no later than the earlier of 24 hours after CareFirst BlueChoice's receipt of the necessary information to make the decision regarding request for coverage, or 72 hours after receipt of the request for coverage.
- 5. A Member shall be provided, upon request and free of charge, reasonable access to, and copies of, all documents, records, and other information relevant to the Member's claim for benefits. A document, record, or other information shall be considered relevant to a Member's claim if it:
 - a. Was relied upon in making the benefit determination;
 - b. Was submitted, considered, or generated in the course of making the benefit determination, without regard to whether such document, record, or other information was relied upon in making the benefit determination; or,
 - c. Demonstrates compliance with the administrative processes and safeguards designed to ensure and verify that benefit claim determinations are made in accordance with governing plan documents and that, where appropriate, the plan provisions have been applied consistently with respect to similarly situated members.
- 6. A grievance and any applicable documentation should be sent to the correspondence address stated on the reverse of the Member identification card.
- 7. Timing of CareFirst BlueChoice responses. The time limits for responding to a grievance will begin at the time an appeal is filed in accordance with these procedures,

without regard to whether all the information necessary to make a decision is initially included. CareFirst BlueChoice will make a grievance decision and written notification will be sent.

- a. Within 30 days after receipt of the grievance for a case involving a Pre-Service Claim;
- b. Within 60 days after receipt of the grievance for a case involving a Post-Service Claim; and

In the case of an expedited appeal regarding a claim relating to a prescription for the alleviation of cancer pain, the appeal decision shall be made as soon as possible but no later than 24 hours after receipt of the appeal.

- 8. When more information is needed for a decision. CareFirst BlueChoice will send notice within 5 working days of the receipt of the appeal that it cannot proceed with its review unless the additional information is provided. CareFirst BlueChoice will assist in gathering the necessary information. The response deadlines described above may be extended one time by CareFirst BlueChoice for up to 15 days, provided that CareFirst BlueChoice both:
 - a. determines that such an extension is necessary due to matters beyond the control of CareFirst BlueChoice; and,
 - b. notifies the Member, prior to the expiration of the initial 15-day period, of the circumstances requiring the extension of time and the date by which CareFirst BlueChoice expects to render a decision.

If such an extension is necessary due to a failure of the Member to submit the information necessary to decide the claim, the notice of extension shall specifically describe the required information, and the Member shall be afforded at least 45 days from receipt of the notice within which to provide the specified information.

In the event that a period of time is extended due to a Member's failure to submit necessary information, the period for responding to a grievance shall be tolled from the date on which the notification of the extension is sent to the Member until the date on which the Member responds to the request for additional information.

The Member must agree to this extension in writing. The Member will be asked to sign a consent form.

D. FAIR AND FULL REVIEW

CareFirst BlueChoice will provide a review that:

- 1. Takes into account all comments, documents, records, and other information submitted by the Member relating to the claim, without regard to whether such information was submitted or considered in the initial benefit determination;
- 2. Does not afford deference to the initial Adverse Benefit Determination and is conducted by an appropriate named fiduciary of CareFirst BlueChoice who is neither the individual who made the Adverse Benefit Determination that is subject to the appeal, nor the subordinate of such individual;
- 3. In deciding an appeal of an Adverse Benefit Determination that is based in whole or in part on a medical judgment, including determinations with regard to whether a particular treatment, drug, or other item is Cosmetic, Experimental, Investigational, or not Medically Necessary, the appropriate named fiduciary shall consult with a Health

Care Provider who has appropriate training and experience in the field of medicine involved in the medical judgment;

- 4. Provides for the identification of medical or vocational experts whose advice was obtained on behalf of CareFirst BlueChoice in connection with a Member's Adverse Benefit Determination, without regard to whether the advice was relied upon in making the benefit determination; and,
- 5. The Health Care Provider engaged for purposes of a consultation is an individual who is neither an individual who was consulted in connection with the Adverse Benefit Determination, nor the subordinate of any such individual.

E. DEEMED EXHAUSTION OF INTERNAL CLAIMS AND APPEAL PROCESS

In the case of a plan that fails to adhere to the minimum requirements for employee benefit plan procedures relating to Claims for Benefits, the Member is deemed to have exhausted the internal claims and appeals processes of paragraph C and D herein. Accordingly the Member may initiate an external review under paragraph F of this section, as applicable. The Member is also entitled to pursue any available remedies under section 502(a) of ERISA or under State law, as applicable, on the basis that the plan has failed to provide a reasonable internal claims and appeals process that would yield a decision on the merits of the Claim for Benefits. If a Member chooses to pursue remedies under section 502(a) of ERISA under such circumstances, the Claim for Benefits, Grievance, or Appeal is deemed denied on review without the exercise of discretion by an appropriate fiduciary.

F. EXTERNAL APPEAL PROCEDURE

A Member who is dissatisfied with a decision rendered in a final internal grievance process shall have the opportunity to pursue an appeal before an external independent review organization if filed within 4 months of the final grievance decision.

If a Member is dissatisfied with the resolution reached through CareFirst BlueChoice's internal grievance system regarding medical necessity, the Member may contact the Director, Office of Health Care Ombudsman and Bill of Rights, at the following:

District of Columbia Department of Health Care Finance Office of Health Care Ombudsman and Bill of Rights One Judiciary Square 441 4th St. NW, 900 South Washington, DC 20001 ((877) 685-6391; (202) 442-6724 or fax (202) 478-1397)

If a Member is dissatisfied with the resolution reached through CareFirst BlueChoice's internal grievance system regarding all other grievances, the Member may contact the Commissioner at the following:

Commissioner, Department of Insurance, Securities and Banking 1050 First St. N.E., Suite 801 Washington, D.C. 20002 (202) 727-8000 Fax: (202) 354-1085

A Member shall also have the option to contact the District of Columbia Department of Insurance, Securities and Banking to request an investigation or file a complaint with the Department at any time during the internal claims and appeal process.

CareFirst BlueChoice, Inc.

[Signature]
[Name] [Title]

CareFirst BlueChoice, Inc.

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

ATTACHMENT B IN-NETWORK DESCRIPTION OF COVERED SERVICES

The services described herein are eligible for coverage under the In-Network Evidence of Coverage. CareFirst BlueChoice will provide the benefits described in the In-Network Schedule of Benefits for Medically Necessary Covered Services incurred by a Member, including any extension of benefits for which the Member is eligible.

It is important to refer to the In-Network Schedule of Benefits to determine the percentage of the Allowed Benefit that CareFirst BlueChoice will pay and any specific limits on the number of services that will be covered. The In-Network Schedule of Benefits also lists important information about Deductibles, the Out-of- Pocket Maximum, and other features that affect Member coverage, including specific benefit limitations.

Refer to the In-Network Evidence of Coverage for additional definitions of capitalized terms included in this In-Network Description of Covered Services.

CareFirst BlueChoice, Inc.

curer not blue choice, me.
[Signature]
[Name]
[Title]

SECTION	TABLE OF CONTENTS	PAGE
1	Outpatient Facility, Office, and Professional Services	В-3
2	Pediatric Dental Services	B-15
3	Pediatric Vision Services	B-20
4	Inpatient Hospital Services	B-22
5	Skilled Nursing Facility Services	B-25
6	Home Health Care Services	B-26
7	Hospice Care Services	B-29
8	Mental Health and Substance Abuse Services	B-31
9	Emergency Services and Urgent Care	B-33
10	Medical Devices and Supplies	B-35
11	Prescription Drugs	B-39
12	Patient-Centered Medical Home	B-40
13	Complex Chronic or High Risk Acute Disease Management	B-41
14	General Provisions	B-45
15	Exclusions and Limitations	B-49

SECTION 1 OUTPATIENT FACILITY, OFFICE, AND PROFESSIONAL SERVICES

CareFirst BlueChoice provides coverage for the services listed below in Contracting Provider's offices or other Contracting Provider facilities. Unless otherwise provided in this Evidence of Coverage, benefits for In-Network Covered Services will only be provided when rendered by Contracting Providers.

See Section 14, General Provisions, for Covered Services that require prior authorization.

1.1 Office Visits

Benefits are available for office visits for the diagnosis and treatment of a medical condition, including care and consultation provided by Primary Care Physicians or other Contracting Providers.

- 1.2 <u>Laboratory Tests, X-Ray/Radiology Services, Specialty Imaging, and Diagnostic Procedures</u>
 Coverage is provided for laboratory and pathology services, x-ray/radiology services, specialty imaging, and diagnostic procedures. Covered services include mammograms, ultrasounds, nuclear medicine, CAT Scans, MRIs, EKGs, EEGs, MRAs, MRSs, CTAs, PET scans, SPECT scans, nuclear cardiology, and related professional services for lab interpretation, x-ray reading, and scan reading rendered by designated Contracting Providers, whether ordered by a Contracting Provider or a Non-Contracting Provider.
 - A. It is the Member's responsibility to locate and utilize a Contracting Provider.
 - B. For purposes of this provision, specialty imaging includes MRI's, MRA's and MRS's, PET scans, CAT scans and nuclear medicine studies.
 - C. Sleep Studies.
 - 1. Coverage is provided for electro-diagnostic tests used to diagnose sleep disorders, including obstructive sleep apnea. These tests may also be used to help adjust a treatment plan for a sleep disorder that has been previously diagnosed. These tests may be done at home, freestanding facilities, outpatient hospital facilities, or at a sleep disorder unit within a hospital.
 - 2. Prior authorization is required for facility-based sleep tests, independent sleep clinic services, and inpatient sleep tests. Prior authorization is not required for home sleep tests.

1.3 Preventive Services

In addition to the benefits listed in this provision, CareFirst BlueChoice will provide benefits for health exams and other services for the prevention and detection of disease, at intervals appropriate to the Member's age, sex, and health status, in accordance with the Patient Protection and Affordable Care Act, as amended, and the Health Care and Education Reconciliation Act of 2010, as amended, as well as CareFirst BlueChoice preventive guidelines. At a minimum, benefits for preventive services listed in this provision will be provided once per Benefit Period.

Benefits will be provided for evidence-based items or services that have in effect a rating of "A" or "B" in the current recommendations of the United States Preventive Services Task Force (USPSTF). This includes benefits for preventive maternity care. CareFirst BlueChoice will update new recommendations to the preventive benefits listed in this provision at the schedule established by the Secretary of Health and Human Services.

Benefits for preventive care include the following:

A. Cancer Screening Services Benefits include:

1. Prostate Cancer Screening

Benefits are available when rendered in accordance with the most current American Cancer Society's guidelines and include a medically recognized diagnostic examination, annual digital rectal examinations, and the prostate-specific antigen (PSA) tests.

2. Colorectal Cancer Screening

Colorectal cancer screening provided in accordance with the latest guidelines issued by the American Cancer Society.

3. Pap Smears

Benefits are available for pap smears, including tests performed using FDA approved gynecological cytology screening technologies, at intervals appropriate to the Member's age and health status, as determined by CareFirst BlueChoice.

4. Breast Cancer Screening

At a minimum, benefits will be provided for breast cancer screening in accordance with the latest screening guidelines issued by the American Cancer Society. The current recommendations of the United States Preventive Services Task Force regarding breast cancer screening, mammography, and prevention of breast cancer will be considered the most current other than those issued in or around November 2009.

B. Human Papillomavirus Screening Test

- 1. Coverage is provided for a Human Papillomavirus Screening Test at the screening intervals supported by the Health Resources and Services Administration of the U.S. Department of Health and Human Services.
- 2. Human Papillomavirus Screening Test means any laboratory test that specifically detects for infection by one or more agents of the human papillomavirus and is approved for this purpose by the FDA.

C. Immunizations

Coverage is provided for immunizations for routine use in children, adolescents, and adults that have in effect a recommendation from the Advisory Committee on Immunization Practices of the Centers for Disease Control and Prevention with respect to the individual involved. Immunizations required solely for travel or work are not covered.

A recommendation from the Advisory Committee on Immunization Practices of the Centers for Disease Control and Prevention is considered to be:

- 1. In effect after it has been adopted by the Director of the Centers for Disease Control and Prevention; and
- 2. For routine use if it is listed on the immunization schedules of the Centers for Disease Control and Prevention.

D. Well Child Care

With respect to infants, children, and adolescents, coverage is provided for evidence-informed preventive care and screenings in the Recommendations for Preventive Pediatric Health by the American Academy of Pediatrics and the Recommended Uniform Screening Panels by the Secretary's Advisory Committee on Heritable Disorders in Newborns and Children.

E. Adult Preventive Care

Benefits include health care services incidental to and rendered during an annual visit at intervals appropriate to the Member's age, sex ,and health status, including evidence-based items or services that have in effect a rating of "A" or "B" in the current recommendations of the United States Preventive Services Task Force (USPSTF).

F. Preventive Gynecological Care

Benefits include recommended preventive services that are age and developmentally appropriate as provided for in comprehensive guidelines supported by the Health Resources and Services Administration.

G. Prevention and Treatment of Obesity

Benefits will be provided for:

- 1. Well child care visits for obesity evaluation and management;
- 2. Evidence-based items or services for preventive care and screening for obesity that have in effect a rating of "A" or "B" in the current recommendations of the United States Preventive Services Task Force (USPSTF);
- 3. For infants, children, and adolescents, evidence-informed preventive care and screening for obesity provided for in the comprehensive guidelines supported by the Health Resources and Services Administration; and,
- 4. Office visits for the treatment of obesity.

H. Osteoporosis Prevention and Treatment Services

1. Definitions

<u>Bone Mass Measurement</u> means a radiologic or other scientifically proven technology for the purpose of identifying bone mass or detecting bone loss.

Qualified Individual means a Member:

- a. Who is estrogen deficient and at clinical risk for osteoporosis;
- b. With a specific sign suggestive of spinal osteoporosis, including roentgenographic osteopenia or roentgenographic evidence suggestive of collapse, wedging, or ballooning of one or more thoracic or lumbar vertebral bodies, who is a candidate for therapeutic intervention or for an extensive diagnostic evaluation for metabolic bone disease;
- c. Receiving long-term glucocorticoid (steroid) therapy;
- d. With primary hyperparathyroidism; or,
- e. Being monitored to assess the response to, or efficacy of, an approved osteoporosis drug therapy.
- 2. Covered Benefits

Benefits for Bone Mass Measurement for the prevention, diagnosis, and treatment of osteoporosis are covered when requested by a Health Care Provider for a Qualified Individual.

1.4 <u>Professional Nutritional Counseling and Medical Nutrition Therapy</u> Benefits are available for Medically Necessary Professional Nutritional Counseling and Medical Nutrition Therapy as determined by CareFirst BlueChoice.

1.5 <u>Family Planning Services</u>

Benefits will be provided for:

- A. Non-Preventive Gynecological Care
 Benefits are available for Medically Necessary gynecological care. Benefits for preventive gynecological care are described in Section 1.3.F.
- B. Contraceptive Methods and Counseling Covered Benefits
 - 1. Contraceptive patient education and counseling for all Members with reproductive capacity.
 - 2. Benefits will be provided for all FDA approved contraceptive drugs and devices for all Members, and sterilization procedures and other contraceptive methods for female Members that must be administered to the Member in the course of a covered outpatient or inpatient treatment.
 - 3. Coverage will be provided for the insertion or removal, and any Medically Necessary examination associated with the use of any contraceptive devices or drugs that are approved by the FDA.
 - 4. Voluntary sterilization for male Members and surgical reversal of voluntary sterilization for all members.
 - 5. Elective abortion.

See Section 11, Prescription Drugs, for coverage for self-administered FDA-approved contraceptive drugs and devices.

C. Maternity Services

The following maternity services are provided for all female members.

- 1. Preventive Services
 - Routine outpatient obstetrical care of an uncomplicated pregnancy, including prenatal evaluation and management office visits and one postpartum office visit;
 - b) Prenatal laboratory tests and diagnostic services related to the outpatient care of an uncomplicated pregnancy, including those identified in the current recommendations of the United States Preventive Services Task Force that have in effect a rating of "A" or "B" or provided in the comprehensive guidelines for women's preventive health supported by the Health Resources and Services Administration;
 - c) Preventive laboratory tests and services rendered to a newborn during a covered hospitalization for delivery, identified in the current recommendations of the United States Preventive Services Task Force that have in effect a rating of "A" or "B," the Periodicity Schedule of the Bright Futures Recommendations for Pediatric Preventive Health Care, and the Uniform Panel of the Secretary's Advisory Committee on Heritable Disorders in Newborns and Children, including the collection of adequate samples for hereditary and metabolic newborn screening and newborn hearing screening; and
 - d) Breastfeeding support, supplies, and consultation.

2. Non-Preventive Services

- a) Outpatient obstetrical care and professional services for all prenatal and post-partum complications. Services include prenatal and post-partum office visits and ancillary services provided during those visits, such as Medically Necessary laboratory tests and diagnostic services;
- b) Inpatient care for delivery;
- c) Non-preventive routine professional services rendered to the newborn during a covered hospitalization for delivery. Non-routine care of the newborn, either during or following the mother's covered hospitalization, requires that the newborn be covered as a Member in the newborn's own right. The In-Network Evidence of Coverage describes the steps, if any, necessary to enroll a newborn Dependent child.
- 3. Postpartum Home Visits. See Section 6.3.C., Home Health Care Services.
- D. Newborn Coverage Coverage includes:
 - 1. Medically Necessary routine newborn visits including admission and discharge exams and visits for the collection of adequate samples for hereditary and metabolic newborn screening;
 - 2. Medically Necessary care and treatment of medically diagnosed congenital defects and birth abnormalities; and
 - 3. Routine hearing screening consisting of one of the following:
 - a. Auditory brain stem response;
 - b. Otoacustic emissions; or
 - c. Other appropriate, nationally recognized, objective physiological screening test.

Additionally, benefits will be provided for infant hearing screenings and all necessary audiological examinations provided using any technology approved by the United States Food and Drug Administration, and as recommended by the most current standards addressing early hearing detection and intervention programs by the National Joint Committee on Infant Hearing. Such coverage includes follow-up audiological examinations as recommended by a physician or audiologist and performed by a licensed audiologist to confirm the existence or absence of hearing loss. Infant as used here is defined according to the most current recommendation of the American Academy of Pediatrics.

E. Infertility Services

Benefits are available for the diagnosis of infertility. Benefits are limited to the following:

- A. Infertility counseling; and
- B. Testing.

1.6 Allergy Services

Benefits are available for allergy testing and treatment, including allergy serum and the administration of injections.

1.7 Rehabilitation Services

A. Definitions

<u>Physical Therapy (PT)</u> includes the short-term treatment that can be expected to result in a significant improvement of a condition. Physical Therapy is the treatment of disease or injury through the use of therapeutic exercise and other interventions that focus on improving a person's ability to go through the functional activities of daily living, to develop and/or restore maximum potential function, and to reduce disability following an illness, injury, or loss of a body part. These may include improving posture, locomotion, strength, endurance, balance, coordination, joint mobility, flexibility, and alleviating pain.

Occupational Therapy (OT) means the use of purposeful activity or interventions designed to achieve functional outcomes that promote health, prevent injury or disability, and that develop, improve, sustain or restore the highest possible level of independence of an individual who has an injury, illness, cognitive impairment, psychosocial dysfunction, mental illness, developmental or learning disability, physical disability, loss of a body part, or other disorder or condition. Occupational Therapy services do not include the adjustment or manipulation of any of the osseous structures of the body or spine.

<u>Speech Therapy (ST)</u> means the treatment of communication impairment and swallowing disorders. Speech Therapy services facilitate the development and maintenance of human communication and swallowing through assessment, diagnosis, and rehabilitation, including cognitive rehabilitation.

B. Covered Benefits

Coverage includes benefits for rehabilitation services including Physical Therapy, Occupational Therapy, and Speech Therapy for the treatment of individuals who have sustained an illness or injury that CareFirst BlueChoice determines to be subject to improvement.

The goal of rehabilitation services is to return the individual to his/her prior skill and functional level.

1.8 <u>Spinal Manipulation</u> Coverage is provided for Medically Necessary spinal manipulation, evaluation, and treatment for the musculoskeletal conditions of the spine when provided by a licensed chiropractor, doctor of osteopathy (D.O.), or other eligible practitioner who is a Contracting Provider. Benefits will not be provided for spinal manipulation services other than for musculoskeletal conditions of the spine.

1.9 Habilitative Services

- A. For Members from birth to age 21.
 - 1. Coverage for Habilitative Services include health care services and devices that help a person keep, learn, or improve skills and functioning for daily living. Examples include therapy for a child who is not walking or talking at the expected age. These services may include physical and occupational therapy, speech-language pathology and other services for people with disabilities in a variety of inpatient and/or outpatient settings.

- 2. Benefits are not available for Habilitative Services delivered through early intervention and school services.
- B. For Members age 21 and over.
 - Coverage for Habilitative Services include health care services and devices that help a person keep, learn, or improve skills and functioning for daily living. These services may include physical and occupational therapy, speech-language pathology and other services for people with disabilities in a variety of inpatient and/or outpatient settings.
 - 2. Prior authorization is required.
- 1.10 Outpatient Therapeutic Treatment Services. Benefits are available for outpatient services rendered in a health care provider's office, in the outpatient department of a hospital, in an ambulatory surgical facility, or other facility in connection with a medical or surgical procedure covered under Section 1, Outpatient Facility, Office and Professional Services.

Benefits include:

- A. Hemodialysis and peritoneal dialysis;
- B. Radiation therapy, including radiation administration;
- C. Cardiac Rehabilitation benefits for Members who have been diagnosed with significant cardiac disease, or who have suffered a myocardial infarction, or have undergone invasive cardiac treatment immediately preceding referral for Cardiac Rehabilitation. Cardiac Rehabilitation is a comprehensive program involving medical evaluation, prescribed exercise, cardiac risk factor modification, education and counseling. Benefits include:
 - 1. Continuous EKG telemetric monitoring during exercise, EKG rhythm strip with interpretation, physician's revision of exercise prescription, and follow up examination for physician to adjust medication or change regimen; and
 - 2. Increased outpatient rehabilitation services (physical therapy, speech therapy and occupational therapy) for Cardiac Rehabilitation of ninety (90) visits per therapy per Benefit Period
 - 3. Services must be provided at a place of service equipped and approved to provide Cardiac Rehabilitation.
- D. Pulmonary rehabilitation benefits for Members who have been diagnosed with significant pulmonary disease.
 - 1. Limited to one (1) program per lifetime
 - 2. Services must be provided at a place of service equipped and approved to provide pulmonary rehabilitation services.
- E. Transfusion services and Infusion Services, including home infusions, infusion of therapeutic agents, medication and nutrients, enteral nutrition into the gastrointestinal tract, chemotherapy, and prescription medications;
- F. Radioisotope treatment.
- 1.11 Blood and Blood Products

Benefits are available for blood and blood products (including derivatives and components) that are not replaced by or on behalf of the Member.

1.12 Organ and Tissue Transplants

- A. Coverage is provided for all Medically Necessary, non-Experimental/Investigational bone marrow, solid organ transplant, and other non-solid organ transplant procedures.

 Medical Necessity is determined by CareFirst BlueChoice. Except for cornea transplants and kidney transplants, prior authorization must be obtained from CareFirst BlueChoice.
- B. Covered Services include the following:
 - 1. The expenses related to registration at transplant facilities. The place of registry is subject to review and determination by CareFirst BlueChoice.
 - 2. Organ procurement charges including harvesting, recovery, preservation, and transportation of the donated organ.
 - 3. Cost of hotel lodging and air transportation for the recipient Member and a companion (or the recipient Member and two companions if the recipient Member is under the age of eighteen (18) years) to and from the site of the transplant.
 - 4. There is no limit on the number of re-transplants that are covered.
 - 5. If the Member is the recipient of a covered organ/tissue transplant, CareFirst BlueChoice will cover the Donor Services (as defined below) to the extent that the services are not covered under any other health insurance plan or contract.
 - <u>Donor Services</u> means services which are related to the transplant surgery, including evaluating and preparing the actual donor, regardless of whether the transplant is attempted or completed, and recovery services after the donor procedure which are directly related to donating the organ or tissue.
 - 6. Immunosuppressant maintenance drugs are covered when prescribed for a covered transplant.

1.13 High Dose Chemotherapy/Bone Marrow or Stem Cell Transplant

Benefits will be provided for high dose chemotherapy bone marrow or stem cell transplant treatment that is not Experimental/ Investigational, when performed pursuant to protocols approved by the institutional review board of any United States medical teaching college including, but not limited to, National Cancer Institute protocols that have been favorably reviewed and utilized by hematologists or oncologists experienced in dose-intensive chemotherapy/autologous bone marrow transplants or stem cell transplants.

1.14 Clinical Trial Patient Cost Coverage

A. Definitions

<u>Cooperative Group</u> means a formal network of facilities that collaborate on research projects and have an established NIH-approved peer review program operating within the Group. Cooperative Group includes the National Cancer Institute Clinical Cooperative Group, National Cancer Institute Community Clinical Oncology Program, AIDS Clinical Trials Group, and Community Programs for Clinical Research in AIDS.

<u>Multiple Project Assurance Contract</u> means a contract between an institution and the federal Department of Health and Human Services that defines the relationship of the institution to the federal Department of Health and Human Services, and sets out the responsibilities of the institution and the procedures that will be used by the institution to protect human subjects.

NIH means the National Institutes of Health.

<u>Qualified Individual</u>, as used in this section, means a Member who is eligible to participate in an approved clinical trial according to the trial protocol, with respect to the treatment of cancer or other life-threatening disease or condition, and the provider who recommended the Member for the clinical trial has concluded that the Member's participation in such trial is appropriate to treat the disease or condition, or the Member's participation is based on medical and scientific information.

Routine Patient Costs means the costs of all Medically Necessary items and health care services consistent with the Covered Services that are typically provided for a Qualified Individual who is not enrolled in a clinical trial that are incurred as a result of the treatment being provided to the Qualified Individual for purposes of the clinical trial. Routine Patient Costs do not include the investigational item, device, or service itself; items or services that are provided solely to satisfy data collection and analysis needs and that are not used in the direct clinical management of the patient; or a service that is clearly inconsistent with widely accepted and established standards of care for a particular diagnosis.

B. Covered Services

- 1. Benefits for Routine Patient Costs to a Qualified Individual in a clinical trial will be provided if the Qualified Individual's participation in the clinical trial is the result of:
 - Treatment provided for a life-threatening disease or condition, or chronic disease; or
 - b) Prevention, early detection, treatment, and monitoring studies on cancer.
- 2. Coverage for Routine Patient Costs will be provided only if:
 - a) The treatment is being provided or the studies are being conducted in a Phase I, Phase II, Phase III, or Phase IV clinical trial for cancer; or,
 - b) The treatment is being provided in a Phase I, Phase II, Phase III, or Phase IV clinical trial for any other life-threatening disease or condition;
 - c) The treatment is being provided in a federally funded or approved clinical trial approved by one of the National Institutes of Health, the Centers for Disease Control and Prevention, the Agency for Health Care Research and quality, the Centers for Medicare and Medicaid Services, an NIH Cooperative Group, an NIH Center, the FDA in the form of an investigational new drug or device application, the federal Department of Defense, the federal Department of Veterans Affairs, the federal Department of Energy or a qualified nongovernmental research entity to which the National Cancer Institute has awarded a support grant,, or an institutional review board of an institution in a state that has a Multiple Project Assurance Contract approved by the Office of Protection from Research Risks of the NIH;
 - d) The treatment is being provided under a drug trial that is exempt from the requirement of an investigational new drug application.
 - e) The facility and personnel providing the treatment are capable of doing so by virtue of their experience, training, and volume of patients treated to maintain expertise;

3. Coverage is provided for the Routine Patient Costs incurred for drugs and devices that have been approved for sale by the FDA whether or not the FDA has approved the drug or device for use in treating the Qualified Individual's particular condition, to the extent that the drugs or devices are not paid for by the manufacturer, distributor, or provider of that drug or device.

1.15 <u>Diabetes Equipment and Supplies, and Self-Management Training</u>

- A. If deemed necessary, diabetes outpatient self-management training and educational services, including Medical Nutrition Therapy, will be provided through an in-person program supervised by an appropriately licensed, registered, or certified CareFirst BlueChoice-approved facility or health care provider whose scope of practice includes diabetes education or management.
- B. Coverage information for diabetic equipment and supplies is located in Section 10, Medical Devices and Supplies and Section 11, Prescription Drugs.

1.16 Dental Services

Pediatric dental benefits for Members up to age 19 are described in Section 2. Benefits will be provided to all Members for the following:

Accidental Injury

A. Covered Benefits

Dental benefits will be provided to repair or replace Sound Natural Teeth that have been damaged or lost due to injury if the injury did not arise while or as a result of biting or chewing, and treatment is commenced within six (6) months of the injury or, if due to the nature of the injury, treatment could not begin within six (6) months of the injury, treatment began within six (6) months of the earliest date that it would be medically appropriate to begin such treatment.

As used in this provision, accidental injury means an injury to Sound Natural Teeth as a result of an external force or trauma resulting in damage to a tooth or teeth, surrounding bone and/or jaw.

B. Conditions and Limitations

Benefits are limited to Medically Necessary dental services as a restoration of the tooth or teeth or the initial placement of a bridge or denture to replace the tooth or teeth injured or lost as a direct and sole result of the accidental bodily injury. Except as listed here, or in Section 1.18, describing benefits for the treatment of cleft lip or cleft palate or both, dental care is excluded from coverage. Benefits for oral surgery are described in Section 1.17.

C. Exclusions

Injuries to teeth that are not Sound Natural Teeth are not covered. Injuries as a result of biting or chewing are not covered.

1.17 Oral Surgery

Benefits include:

- A. Medically Necessary procedures, as determined by CareFirst BlueChoice, to attain functional capacity, correct a congenital anomaly (excluding odontogenic congenital anomalies or anomalies limited to the teeth), reduce a dislocation, repair a fracture, excise tumors, non-odontogenic cysts or exostoses, or drain abscesses involving cellulitis and are performed on the lips, tongue, roof, and floor of the mouth, sinuses, salivary glands or ducts, and jaws.
- B. Medically Necessary procedures, as determined by CareFirst BlueChoice, needed as a result of an accidental injury, when the Member requests oral surgical services or dental services for Sound Natural Teeth and supporting structures or the need for oral surgical services or dental services for Sound Natural Teeth and supporting structures is identified

in the patient's medical records within sixty (60) days of the accident. Benefits for such oral surgical services will be provided up to three (3) years from the date of injury.

C. Surgical treatment for temporomandibular joint syndrome (TMJ) if there is clearly demonstrable radiographic evidence of joint abnormality due to an illness.

All other procedures involving the teeth or areas surrounding the teeth including the shortening of the mandible or maxillae (orthognathic surgery) for Cosmetic or other purposes or for correction of the malocclusion unrelated to a functional impairment that cannot be corrected non-surgically are excluded.

1.18 Treatment for Cleft Lip or Cleft Palate or Both

Benefits will be provided for inpatient or outpatient expenses arising from orthodontics, oral surgery, otologic, audiological, and speech/language treatment for cleft lip or cleft palate or both.

1.19 Outpatient Surgical Procedures

- A. Benefits are available for surgical procedures performed by Contracting Providers on an outpatient basis.
- B. Benefits are available for services in a hospital outpatient department or in an ambulatory surgical facility, in connection with a covered surgical procedure, including:
 - 1. Use of operating room and recovery room.
 - 2. Use of special procedure rooms.
 - 3. Diagnostic procedures, laboratory tests, and radiology services.
 - 4. Drugs, medications, solutions, biological preparations, and services associated with the administration of the same.
 - 5. Medical and surgical supplies.
 - 6. Blood, blood plasma and blood products, and related donor processing fees that are not replaced by or on behalf of the Member. Administration of infusions is covered.
- 1.20 Anesthesia Services for Medical or Surgical Procedures. Benefits are available for the administration of general anesthesia in connection with a covered medical or surgical procedure. To be eligible for separate coverage, a health care provider other than the operating surgeon or assistant at surgery must administer the anesthesia. For example, a local anesthetic used while performing a medical or surgical procedure is not generally viewed as a separately covered charge.

1.21 <u>Reconstructive Surgery</u>

Benefits for reconstructive surgery are limited to surgical procedures that are Medically Necessary, as determined by CareFirst BlueChoice, and operative procedures performed on structures of the body to improve or restore bodily function or to correct a deformity resulting from disease, trauma, or previous therapeutic intervention.

1.22 Reconstructive Breast Surgery

Benefits will be provided for reconstructive breast surgery resulting from a Mastectomy.

- A. Reconstructive breast surgery means surgery performed as a result of a Mastectomy to reestablish symmetry between the two breasts. Reconstructive breast surgery includes:
 - 1. Augmentation mammoplasty;

- 2. Reduction mammoplasty; and
- 3. Mastopexy.
- B. Benefits are provided for all stages of reconstructive breast surgery performed on the non-diseased breast to establish symmetry with the diseased breast when reconstructive breast surgery on the diseased breast is performed.
- C. Benefits are provided regardless of whether the Mastectomy was performed while the Member was covered under the In-Network Evidence of Coverage.
- D. Coverage will be provided for treatment of physical complications at all stages of Mastectomy, including lymphedemas, in a manner determined in consultation with the Member and the Member's attending Contracting Physician.

1.23 Retail Health Clinics

Coverage is provided for treatment of common conditions or ailments, which require rapid and specific treatment that can be administered in a limited duration of time. Retail Health Clinics are mini-medical office chains typically staffed by nurse practitioners with an on-call physician. Retail Health Clinic services are non-emergency and non-urgent services for common ailments for which a reasonable, prudent layperson who possesses an average knowledge of health and medicine would seek in a Retail Health Clinic, including, but not limited to: ear, bladder, and sinus infections; pink eye; flu, and strep throat.

1.24 Telemedicine Services

- A. Coverage shall be provided for the use of interactive audio, video, or other electronic media for the purpose of consultation, diagnosis, or treatment of the patient.
- B. Benefits for telemedicine shall be provided by a health care provider to deliver health care services within the scope of the provider's practice at a site other than the site where the patient is located.
- C. Benefits for telemedicine are not subject to any annual dollar maximum or annual visit limitation.
- D. CareFirst BlueChoice shall not exclude a service from coverage solely because the service is provided through telemedicine and is not provided through face-to-face consultation or contact between a health care provider and a patient for services appropriately provided through telemedicine.

SECTION 2 PEDIATRIC DENTAL SERVICES

- 2.1 Subject to the terms and conditions of the In-Network Evidence of Coverage, benefits will be provided for the following Covered Dental Services when rendered and billed for by a Dentist as specified in the attached Schedules of Benefits.
- 2.2 Pediatric dental benefits for Members up to the end of the calendar year in which the Member turns age 19 will be provided in accordance with the High Option dental benefits of the Federal Employees Dental and Vision Insurance Program (FEDVIP) as specified in the In-Network Schedule of Benefits.
- 2.3 Class I Preventive and Diagnostic Services
 - A. Services limited to twice per Benefit Period.
 - 1. Oral examination including oral health risk assessment
 - 2. Routine cleaning of teeth (dental prophylaxis)
 - 3. Topical application of fluoride
 - 4. Bitewing x-ray (not taken on the same date as those in B. below)
 - 5. Pulp vitality tests; additional tests may be allowed for accidental injury and trauma, or other emergency
 - B. Services limited to one per 60 months
 - 1. Intraoral complete series x-ray (full mouth x-ray including bitewings)
 - 2. One panoramic x-ray and one additional set of bitewing x-rays
 - C. Services limited to once per tooth per 36 months: sealants on permanent molars
 - D. Space maintainers when Medically Necessary due to the premature loss of a posterior primary tooth
 - E. Services as required
 - 1. Palliative Treatments once per date of service
 - 2. Emergency Oral Exam once per date of service
 - 3. Periapical and occlusal x-rays limited to the site of injury or infection
 - 4. Professional consultation rendered by a Dentist, limited to one consultation per condition per Dentist other than the treating Dentist
 - 5. Intraoral occlusal x-ray
 - 6. One cephalometric x-ray
- 2.4 Class II Basic Services
 - A. Direct placement fillings limited to:
 - 1. Silver amalgam, resin-based composite, compomer, glass-ionomer or equivalent

material accepted by the American Dental Association and/or the United States Food and Drug Administration

- 2. Direct pulp caps and indirect pulp caps
- B. Non-Surgical periodontic services limited to:
 - 1. Periodontal scaling and root planing once per 24 months per quadrant
 - 2. Full mouth debridement to enable comprehensive periodontal procedure one per lifetime
 - 3. Periodontal maintenance procedures four per 12 months
- C. Simple extractions performed without general anesthesia once per tooth per lifetime
- 2.5 Class III Major Services Surgical
 - A. Surgical periodontic services
 - 1. Gingivectomy or gingivoplasty limited to one treatment per 36 months per Member per quadrant or per tooth
 - 2. Osseous Surgery (including flap entry and closure) limited to one treatment per 36 months per Member per quadrant
 - 3. Limited or complete occlusal adjustments in connection with periodontal treatment when services are received on a different date than restorative services
 - 4. Mucogingival Surgery limited to grafts and plastic procedures; one treatment per site limited to one site or quadrant every 36 months

B. Endodontics

- 1. Apicoectomy
- 2. Pulpotomy for deciduous teeth once per tooth per lifetime per Member
- 3. Root canal for permanent teeth once per tooth per lifetime per Member
- 4. Root canal retreatment performed on permanent teeth limited to once per tooth per lifetime per Member
- 5. Root resection once per tooth per lifetime per Member
- 6. Pulpal therapy once per tooth per lifetime per Member
- 7. Endodontic therapy once per tooth per lifetime per Member
- C. Oral Surgical services as required
 - 1. Simple and Surgical extractions, including impactions once per tooth per lifetime per Member
 - 2. Oral Surgery, including treatment for cysts, tumors and abscesses
 - 3. Biopsies of oral tissue if a biopsy report is submitted

- 4. General anesthesia, intravenous (IV) sedation/analgesia, analgesia, and non-intravenous conscious sedation when Medically Necessary and administered by a Dentist who has a license, permit, or certificate to administer conscious sedation or general anesthesia or board certified anesthesiologist (MD, DO, DDS, DMD).
- 5. Hemi-section
- 6. Tooth reimplantation and/or stabilization of accidentally evulsed or displaced tooth
- 7. Vestibuloplasty
- 8. Services limited to once per lifetime per tooth:
 - a) Coronectomy
 - b) Tooth transplantation
 - c) Surgical repositioning of teeth
 - d) Alveoloplasty
 - e) Frenulectomy
 - f) Excision of pericoronal gingiva

2.6 Class IV - Major Services - Restorative

A. Crowns

- 1. Metal and/or porcelain/ceramic crowns and crown build-ups limited to one per 60 months per tooth
- 2. Metal and/or porcelain/ceramic inlays and onlays limited to one per 60 months per tooth
- 3. Stainless steel crowns
- 4. Recementation of crowns and/or inlays limited to once in any twelve (12) month period
- 5. Metal and/or porcelain/ceramic pontics limited to one per 60 months per tooth
- B. Dental Implants are covered procedures only if determined to be Medically Necessary. If CareFirst BlueChoice determines an arch can be restored with a standard prosthesis or restoration, no benefits will be allowed for the individual implant or implant procedures, and only the second phase of treatment (the prosthodontic phase of placing of the implant crown or partial denture) will be a Covered Dental Service.
 - 1. Endosteal implant limited to one per 60 months
 - 2. Surgical placement of interim implant body limited to one per 60 months
 - 3. Eposteal implant limited to one per 60 months
 - 4. Transosteal implant limited to one per 60 months
 - 5. Implant supported complete denture

6. Implant supported partial denture

C. Dentures

- 1. Partial removable dentures, upper or lower, limited to one per 60 months
- 2. Complete removable dentures, upper or lower, limited to one per 60 months
- 3. Pre-operative radiographs required
- 4. Pre-treatment estimate, as described in the Estimate of Eligible Benefits section is recommended for Members
- 5. Tissue conditioning prior to denture impression
- 6. Repairs to denture as required including: repair resin denture base, repair cast framework, addition of tooth or clasp to existing partial denture, replacement of broken tooth, repairs or replacement of clasp, recement fixed partial denture
- D. Fixed prosthetics, including bridges and crowns, inlays and onlays used as abutments for or as a unit of the bridge limited to one per tooth per 60 months
- E. Denture adjustments and relining limited to: Full or partial removable (upper or lower) dentures: once per 24 months, but not within six months of initial placement
- F. Repair of prosthetic appliances, including fixed bridges, and removable dentures, full and/or partial. No limitations for Members under age 19. For Members over age 19, benefit limited to once in any twelve (12) month period per specific area of the appliance.
- G. Repair of prosthetic appliances and removable dentures, full and/or partial.
- H. Occlusal guard, by report, limited to one per 12 months for Members age 13 and older
- 2.7 Class V Orthodontic Services
 - A. Benefits for orthodontic services will only be available until the end of the calendar year in which the Member turns age 19 if the Member:
 - 1. Has fully erupted permanent teeth with at least 1/2 to 3/4 of the clinical crown being exposed (unless the tooth is impacted or congenitally missing); and
 - 2. Has a severe, dysfunctional, handicapping malocclusion and is determined to be Medically Necessary.
 - B. All comprehensive orthodontic services require a pre-treatment estimate (PTE) by CareFirst, as described in the Estimate of Eligible Benefits section. The following documentation must be submitted with the request for a PTE:
 - 1. ADA 2006 or newer claim form with service code requested;
 - 2. A complete series of intra-oral photographs;
 - 3. Diagnostic study models (trimmed) with waxbites or OrthoCad electronic equivalent, and
 - 4. Treatment plan including anticipated duration of active treatment.
 - C. Covered benefits if a PTE is approved

- 1. Retainers
 - a) One set (included in comprehensive orthodontics)
 - b) Replacement allowed one per arch per lifetime within 12 months of date of debanding, if necessary
 - c) Rebonding or recementing fixed retainer
- 2. Pre-orthodontic treatment visit
- 3. Braces once per lifetime
- 4. Periodic treatment visits; not to exceed 24 months (the Member must be eligible for Covered Dental Services on each date of service).
- D. Payment policy: one initial payment for comprehensive orthodontic treatment, a preorthodontic treatment visit and periodic orthodontic treatment visits (not to exceed 24 periodic orthodontic treatment visits).
 - When a Preferred Dentist or Participating Dentist provides the comprehensive orthodontic treatment, additional periodic orthodontic treatment visits beyond 24 will be the orthodontist's financial responsibility and not the Subscriber's. Subscribers may not be billed for broken, repaired, or replacement of brackets or wires. Visits to repair or replace brackets or wires are not separately reimbursable from periodic visits.
 - 2. When a Non-Participating Dentist provides the comprehensive orthodontic treatment, additional periodic orthodontic treatment visits beyond 24 will not be Covered Dental Services. The Member is responsible for the difference between the CareFirst BlueChoice payment for Covered Dental Services and the Non-Participating Dentist's charge.
- E. In cases where the Member has been approved for comprehensive orthodontic benefits, and the parent has decided they do not wish to have the child begin treatment at this time or any time in the near future, the provider may bill for their records, to include the treatment plan, radiographs, models, photos, etc. and explaining the situation on the claim for payment. The reimbursement for these records is the same as if the orthodontic services had been rendered.
- F. If the case is denied, the provider will be informed that CareFirst BlueChoice will not cover the orthodontic treatment. However, Covered Dental Services will include the preorthodontic visit which included treatment plan, radiographs, and/or photos, records and diagnostic models for full treatment cases only.

SECTION 3 PEDIATRIC VISION SERVICES

3.1 <u>Covered Services</u>

Coverage will be provided for pediatric vision benefits for children up to age 19 in accordance with the Federal Employee Program Blue Vision high plan. Benefits include:

- A. One routine eye examination, including dilation, if professionally indicated, each Benefit Period. A vision examination may include, but is not limited to:
 - 1. Case history;
 - 2. External examination of the eye and adnexa;
 - 3. Ophthalmoscopic examination;
 - 4. Determination of refractive status;
 - 5. Binocular balance testing;
 - 6. Tonometry test for glaucoma;
 - 7. Gross visual field testing;
 - 8. Color vision testing;
 - 9. Summary finding; and
 - 10. Recommendation, including prescription of corrective lenses.
- B. Frames and Spectacle Lenses or Contact Lenses
 - 1. Prescribed frames and spectacle lenses or contact lenses, including directly related provider services such as:
 - a) Measurement of face and interpupillary distance;
 - b) Quality assurance; and
 - c) Reasonable aftercare to fit, adjust and maintain comfort and effectiveness.
 - 2. One pair of frames per Benefit Period; and
 - 3. One pair of prescription spectacle lenses per Benefit Period
 - a) Spectacle lenses include choice of glass or plastic lenses, all lens powers (single vision, bifocal, trifocal, lenticular), fashion and gradient tinting, oversized glass-grey #3 prescription sunglass lenses, ultraviolet protective coating, standard progressives, and plastic photosensitive lenses (Transitions®).
 - b) Polycarbonate lenses are covered in full for monocular patients and patients with prescriptions > +/- 6.00 diopters.
 - All spectacle lenses include scratch resistant coating with no additional Copayment. There may be an additional charge at Walmart and Sam's Club

4. Contact Lenses

- a) Contact lens evaluation, fitting, and follow-up care.
- b) Elective contact lenses (in place of frames and spectacle lenses):
 - (1) One pair of elective prescription contact lenses per Benefit Period; or,
 - (2) Multiple pairs of disposable prescription contact lenses per Benefit Period.
- c) One pair of Medically Necessary prescription contact lenses per Benefit Period in lieu of other eyewear.
 - (1) Prior authorization is required. Contracting Vision Providers will obtain the necessary prior authorization for these services.
 - (2) Contact lenses may be determined to be Medically Necessary and appropriate in the treatment of patients affected by certain conditions. Contact lenses may be Medically Necessary and appropriate when the use of contact lenses, in lieu of eyeglasses, will result in significantly better visual and/or improved binocular function, including avoidance of diplopia or suppression. Contact lenses may be determined to be Medically Necessary in the treatment of the following conditions: keratoconus, pathological myopia, aphakia, anisometropia, aniseikonia, aniridia, corneal disorders, post-traumatic disorders, and/or irregular astigmatism.
- C. Low vision services, including one comprehensive Low Vision evaluation every 5 years, 4 follow-up visits in any 5-year period and prescribed low vision aid optical devices, such as high-powered spectacles, magnifiers and telescopes.
 - 1. Ophthalmologists and optometrists specializing in low vision care will evaluate and prescribe optical devices, and provide training and instruction to maximize the remaining usable vision for Members with low vision.
 - 2. Prior authorization is required for low vision services. Contracting Vision Providers will obtain the necessary prior authorization for these services.
- D. Covered Vision Services and benefits for services provided by Non-Contracting Vision Providers are limited. See the In-Network Schedule of Benefits.

3.2 Warranty

The Vision Care Designee's collection frames and all eyeglass lenses manufactured in the Vision Care Designee laboratories are guaranteed for one year from the original date of dispensing. Warranty limitations may apply to provider-supplied or retailer-supplied frames and/or eyeglass lenses. The Contracting Vision Provider can provide the details of the warranty that is available to the Member.

3.3 Limitations

Benefit limited to Members up to age 19. If Member is under age 19 at the start of the Benefit Period but turns 19 during the Benefit Period, then the Member will receive Covered Vision Services through the rest of that Calendar Year. Benefits for the treatment of medical conditions of the eye are covered under Section 1.

SECTION 4 INPATIENT HOSPITAL SERVICES

HOSPITAL ADMISSIONS MUST BE AUTHORIZED OR APPROVED BY CAREFIRST BLUECHOICE, UNLESS EXCEPTIONS ARE STATED.

CareFirst BlueChoice provides coverage for the services listed below in a Contracting Provider Hospital when admitted under the care of a Contracting Provider.

4.1 Covered Inpatient Hospital Services

A Member will receive benefits for the Covered Services listed below when admitted to a Contracting Provider hospital under the care of a Primary Care Physician or other Contracting Physician. Coverage of inpatient hospital services is subject to certification by utilization management for Medical Necessity. Benefits are provided for:

- A. Room and Board
 - Room and board in a semiprivate room (or in a private room when Medically Necessary as determined by CareFirst BlueChoice).
- B. Physician, Medical, and Surgical Services
 Medically Necessary inpatient physician, medical, and surgical services provided by or
 under the direction of the attending Contracting Physician and ordinarily furnished to a
 patient while hospitalized.
- C. Services and Supplies

Related inpatient services and supplies that are not Experimental/Investigational, as determined by CareFirst BlueChoice, and ordinarily furnished by the hospital to its patients, including:

- 1. The use of:
 - a) Operating rooms;
 - b) Treatment rooms; and
 - c) Special equipment in the hospital.
- 2. Drugs, medications, solutions, biological preparations, anesthesia, and services associated with the administration of the same.
- 3. Medical and surgical supplies.
- 4. Blood, blood plasma, and blood products, and related donor processing fees that are not replaced by or on behalf of the Member. Administrations of infusions and transfusions are covered.
- 5. Surgically implanted Prosthetic Devices that replace an internal part of the body. This includes hip joints, skull plates, cochlear implants, and pacemakers. Available benefits under this provision do not include items such as dental implants, fixed or removable dental Prosthetics, artificial limbs, or other external Prosthetics, which may be provided under other provisions of this In-Network Description of Covered Services.
- 6. Medical social services.

4.2 Number of Hospital Days Covered

Provided the conditions, including the requirements below, are met and continue to be met, as determined by CareFirst BlueChoice, hospital benefits for inpatient hospital services will be provided as follows:

A. Hospitalization for Rehabilitation

Benefits are provided for an admission or transfer to a CareFirst BlueChoice approved facility for rehabilitation. Benefits provided during any admission will not exceed any applicable benefit limitation. The limit, if any, on hospitalization for rehabilitation applies to any portion of an admission that:

- 1. Is required primarily for Physical Therapy or other rehabilitative care; and
- 2. Would not be Medically Necessary based solely on the Member's need for inpatient acute care services other than for rehabilitation.

B. Inpatient Coverage Following a Mastectomy Coverage will be provided for a minimum hospital stay of not less than:

- 1. Forty-eight (48) hours following a radical or modified radical Mastectomy; and
- Twenty-four (24) hours following a partial Mastectomy with lymph node dissection.

In consultation with the Member's attending Contracting Physician, the Member may elect to stay less than the minimum prescribed above when appropriate.

C. Hysterectomies

Coverage will be provided for vaginal hysterectomies and abdominal hysterectomies. Coverage includes a minimum stay in the hospital of:

- 1. Not less than twenty-three (23) hours for a laparoscopy-assisted vaginal hysterectomy; and
- 2. Not less than forty-eight (48) hours for a vaginal hysterectomy.

In consultation with the Member's attending Contracting Physician, the Member may elect to stay less than the minimum prescribed above when appropriate.

D. Childbirth

Coverage will be provided for a minimum hospital stay of not less than:

- 1. Forty-eight (48) hours for both the mother and newborn following a routine vaginal delivery;
- 2. Ninety-six (96) hours for both the mother and newborn following a routine cesarean section.

Prior authorization is not required for maternity admissions.

Whenever a mother is required to remain hospitalized after childbirth for medical reasons and the mother requests that the newborn remain in the hospital, coverage includes additional hospitalization for the newborn for up to four (4) days.

If the delivery occurs in the hospital, the length of stay begins at the time of the delivery. If the delivery occurs outside of the hospital, the length of stay begins upon admission to the hospital. The Member and the Contracting Provider may agree to an early discharge.

Non-routine care of the newborn, either during or following the mother's covered hospitalization, requires that the newborn be covered as a Member in the newborn's own right. The In-Network Evidence of Coverage describes the steps, if any, necessary to enroll a newborn Dependent child.

4.3 Other Inpatient Services

Benefits are available for all other care in the nature of usual hospital services that are Medically Necessary for the care and treatment of the patient, provided that those services cannot be rendered in an outpatient setting and are not otherwise specifically excluded.

SECTION 5 SKILLED NURSING FACILITY SERVICES

CareFirst BlueChoice provides coverage for the services listed below in a Contracting Provider Skilled Nursing Facility when admitted under the care of a Primary Care Physician or other Contracting Physician.

See Section 14, General Provisions, for Covered Services that require prior authorization.

- 5.1 <u>Covered Skilled Nursing Facility Services</u> When the Member meets the conditions for coverage listed in Section 5.2, the services listed below are available to Members in a Skilled Nursing Facility:
 - A. Room and board in a semiprivate room;
 - B. Inpatient physician and medical services provided by or under the direction of the attending Contracting Provider; and
 - C. Services, and supplies that are not Experimental/Investigational, as determined by CareFirst BlueChoice, and ordinarily furnished by the facility to inpatients for diagnosis or treatment.
- 5.2 <u>Conditions for Coverage</u> Skilled Nursing Facility care must be authorized or approved by CareFirst BlueChoice as meeting the following conditions for coverage:
 - A. The Member must be under the care of his or her Primary Care Physician or other Contracting Physician.
 - B. The admission to the Skilled Nursing Facility must be a substitute for hospital care (i.e., if the Member were not admitted to a Skilled Nursing Facility, he or she would have to be admitted to a hospital).
 - C. Skilled Nursing Facility benefits will not be provided in a facility that is used primarily as a rest home or a home for the aged, or in a facility for the care of drug addiction or alcoholism.
 - D. The Member must require Skilled Nursing Care or skilled rehabilitation services which are:
 - 1. Required on a daily basis;
 - 2. Not Custodial; and
 - 3. Only provided on an inpatient basis.
 - E. Prior authorization has been obtained from CareFirst BlueChoice.
- 5.3 <u>Custodial Care is Not Provided</u> Benefits will not be provided for any day in a Skilled Nursing Facility that CareFirst BlueChoice determines is primarily for Custodial Care. Services may be deemed Custodial Care even if:
 - A. A Member cannot self-administer the care;
 - B. No one in the Member's household can perform the services;
 - C. Ordered by a physician;
 - D. Necessary to maintain the Member's present condition; or
 - E. Covered by Medicare.

SECTION 6 HOME HEALTH CARE SERVICES

CareFirst BlueChoice provides coverage for the services listed below in a Member's home by a Contracting Provider Home Health Agency when authorized or approved by CareFirst BlueChoice.

See Section 14, General Provisions, for Covered Services that require prior authorization.

6.1 Covered Home Health Care Services

Services must be provided within the Service Area when requested by a Primary Care Physician or other Contracting Physician. Benefits are provided for:

- A. Continued care and treatment, provided by or under the supervision of a Registered Nurse (RN) or Licensed Practical Nurse (LPN). Services of a home health aide, medical social worker, or registered dietician may be provided, but must be performed under the supervision of a licensed professional (RN or LPN) nurse.
- B. Drugs and medications

 Drugs and medications directly administered to the patient during a covered home health care visit and incidental Medical Supplies directly expended in the course of a covered home health care visit are covered.
- C. Home Health Care Services authorized or approved by CareFirst BlueChoice as Medically Necessary.
- D. Prior authorization for Home Health Care Services is required.

Purchase or rental of Durable Medical Equipment is not covered under this provision. See Section 10.3.A, Durable Medical Equipment, for benefit information.

6.2 <u>Conditions for Coverage</u>

Benefits are provided when:

- A. The Member must be confined to home due to a medical, non-psychiatric condition. "Home" cannot be an institution, convalescent home, or any facility which is primarily engaged in rendering medical or rehabilitative services to sick, disabled, or injured persons.
- B. The Home Health Care visits are a substitute for hospital care or for care in a Skilled Nursing Facility (i.e., if Home Health Care visits were not provided, the Member would have to be admitted to a hospital or Skilled Nursing Facility).
- C. The Member requires and continues to require Skilled Nursing Care or rehabilitation services in order to qualify for home health aide services or other types of Home Health Care Services.
- D. The need for Home Health Care Services is not Custodial in nature.
- E. Services of a home health aide, medical social worker, or registered dietician must be performed under the supervision of a licensed professional nurse (RN or LPN).
- F. All services must be arranged and billed by the Qualified Home Health Agency. Providers may not be retained directly by the Member.
- G. The Member must be under the care of a Primary Care Physician or other Contracting Physician.
- H. Prior authorization has been obtained from CareFirst BlueChoice.

6.3 Additional Home Health Care Benefits

A. Home Visits Following Surgical Removal of a Testicle
For a Member who receives less than 48 hours of inpatient hospitalization following the surgical removal of a testicle, or who undergoes the surgical removal of a testicle on an outpatient basis, benefits will be provided for:

- 1. One home visit scheduled to occur within twenty-four (24) hours after discharge from the hospital or outpatient health care facility; and
- 2. An additional home visit if prescribed by the Member's attending Contracting Physician.
- 3. Benefits provided under this provision do not count toward any Home Health Care visit maximum.
- 4. Prior authorization is not required.
- B. Home Visits Following a Mastectomy
 - 1. For a Member who has a shorter hospital stay than that provided under Section 4.2.B, Inpatient Coverage Following a Mastectomy, or who undergoes a Mastectomy on an outpatient basis, benefits will be provided for:
 - a) One home visit scheduled to occur within twenty-four (24) hours after discharge from the hospital or outpatient health care facility; and
 - b) An additional home visit if prescribed by the Member's attending Contracting Physician.
 - 2. For a Member who remains in the hospital for at least the length of time provided in Section 4.2.B, Inpatient Coverage Following a Mastectomy, coverage will be provided for a home visit if prescribed by the Member's attending Contracting Physician.
 - 3. Benefits provided under this provision do not count toward any Home Health Care visit maximum.
 - 4. Prior authorization is not required.
- C. Postpartum Home Visits

Home visits following delivery are covered in accordance with the most current standards published by the American College of Obstetricians and Gynecologists.

- 1. For a mother and newborn child who have a shorter hospital stay than that provided under Section 4.2.D, Childbirth, benefits will be provided for:
 - a) One home visit scheduled to occur within 24 hours after hospital discharge; and
 - b) An additional home visit if prescribed by the attending Contracting Physician.
- 2. For a mother and newborn child who remain in the hospital for at least the length of time provided under Section 4.2.D, Childbirth, benefits will be provided for a home visit if prescribed by the attending Contracting Physician.

- 3. Benefits provided under this provision do not count toward any Home Health Care visit maximum.
- 4. Prior authorization is not required.

SECTION 7 HOSPICE CARE SERVICES

See Section 14, General Provisions, for Covered Services that require prior authorization.

- 7.1 <u>Covered Hospice Care Services</u> Benefits will be provided for the services listed below when provided by a Qualified Hospice Care Program. Coverage for hospice care services is subject to certification of the need and continued appropriateness of such services in accordance with CareFirst BlueChoice utilization management requirements.
 - A. Inpatient and outpatient care;
 - B. Intermittent Skilled Nursing Care;
 - C. Medical social services for the terminally ill patient and his or her Immediate Family;
 - D. Counseling, including dietary counseling, for the terminally ill Member;
 - E. Non-Custodial home health visits:
 - F. Services, visits, medical/surgical equipment, or supplies, including equipment and medication required to maintain the comfort and manage the pain of the terminally ill Member;
 - G. Laboratory test and x-ray services;
 - H. Medically Necessary ground ambulance, as determined by CareFirst BlueChoice;
 - I. Family Counseling will be provided to the Immediate Family and the Family Caregiver before the death of the terminally ill Member, when authorized or approved by CareFirst BlueChoice; and
 - J. Bereavement Counseling.
- 7.2 <u>Conditions for Coverage</u> Hospice care services must be provided by a Qualified Hospice Care Program, and meet the following conditions for coverage:
 - A. The Member must have a life expectancy of six (6) months or less;
 - B. The Member's attending Primary Care Physician or other Contracting Physician must submit a written hospice care services plan of treatment to CareFirst BlueChoice;
 - C. The Member must meet the criteria of the Qualified Hospice Care Program;
 - D. The need and continued appropriateness of hospice care services must be certified by CareFirst BlueChoice as meeting the criteria for coverage in accordance with CareFirst BlueChoice utilization management requirements; and
 - E. Prior authorization has been obtained from CareFirst BlueChoice.
- 7.3 <u>Hospice Eligibility Period</u>

The hospice eligibility period begins on the first date hospice care services are rendered and terminates one hundred eighty (180) days later or upon the death of the terminally ill Member, if sooner. If the Member requires an extension of the eligibility period, the Member or the Member's representative must notify CareFirst BlueChoice in advance to request an extension of benefits. CareFirst BlueChoice reserves the right to extend the eligibility period on an individual case basis if CareFirst BlueChoice determines that the Member's prognosis and continued need for services are consistent with a program of hospice care services.

SECTION 8 MENTAL HEALTH AND SUBSTANCE ABUSE SERVICES

HOSPITAL ADMISSIONS MUST BE AUTHORIZED OR APPROVED BY THE MENTAL HEALTH AND SUBSTANCE ABUSE MANAGEMENT PROGRAM. PRIOR AUTHORIZATION WILL BE OBTAINED BY CONTRACTING PROVIDERS.

See Section 14, General Provisions, for Covered Services that require prior authorization.

8.1 Definitions

Mental Illness and Emotional Disorders are broadly defined as including any mental disorder, mental illness, psychiatric illness, mental condition, or psychiatric condition (whether organic or non-organic, whether of biological, non-biological, chemical or non-chemical origin, and irrespective of cause, basis, or inducement). This includes, but is not limited to, psychoses, neurotic disorders, schizophrenic disorders, affective disorders, personality disorders, and psychological or behavioral abnormalities associated with transient or permanent dysfunction of the brain or related neurohormonal systems. (This is intended to include disorders, conditions and illnesses classified on Axes I and II in the current edition of the Diagnostic and Statistical Manual of Mental Disorders of the American Psychiatric Association, Washington, D.C.)

Mental Health and Substance Abuse Management Program refers to utilization management, benefits administration, and provider network activities administered by or on behalf of CareFirst BlueChoice to ensure that mental health and Substance Abuse services are Medically Necessary and provided in a cost-effective manner.

<u>Partial Hospitalization</u> means the provision of medically directed intensive or intermediate short-term treatment in a licensed or certified facility or program for treatment of Mental Illnesses, Emotional Disorders, and Drug and Alcohol Abuse.

<u>Qualified Partial Hospitalization Program</u> means a licensed or certified facility or program that provides medically directed intensive or intermediate short-term treatment for Mental Illness, Emotional Disorder, Drug Abuse or Alcohol Abuse for a period of less than twenty-four (24) hours, but more than four (4) hours in a day.

Qualified Treatment Facility means a non-hospital residential facility certified by the District of Columbia or by any jurisdiction in which it is located, as a qualified non-hospital provider of treatment for Drug Abuse, Alcohol Abuse, Mental Illness, or any combination of these, in a residential setting. A non-hospital residential facility includes any facility operated by the District of Columbia, any state or territory or the federal government to provide these services in a residential setting. It is not a facility licensed as a general or special hospital. A non-hospital residential facility also must meet or exceed guidelines established for such a facility by CareFirst BlueChoice.

Substance Abuse means:

- A. <u>Alcohol Abuse</u> means any pattern of pathological use of alcohol that causes impairment in social or occupational functioning, or that produces physiological dependency evidenced by physical tolerance or by physical symptoms when it is withdrawn.
- B. <u>Drug Abuse</u> means any pattern of pathological use of drugs that causes impairment in social or occupational functioning, or that produces physiological dependency evidenced by physical tolerance or by physical symptoms when it is withdrawn.

8.2 Outpatient Mental Health and Substance Abuse Services

Benefits are covered for outpatient services provided in a Contracting Physician's office or in other Contracting Provider facilities. Covered Services include the following:

- A. Diagnosis and treatment for Mental Illness and Emotional Disorders at Contracting Physician offices, other outpatient Contracting Provider medical offices and facilities, and in Qualified Partial Hospitalization Programs.
- B. Diagnosis and treatment for Substance Abuse, including detoxification and rehabilitation services as an outpatient in a covered alcohol or drug rehabilitation program or Qualified Partial Hospitalization Program designated by CareFirst BlueChoice.
- C. Other covered medical services and medical ancillary services for conditions related to Mental Illness, Emotional Disorders, and Substance Abuse.
- D. Office visits for medication management in connection with Mental Illness, Emotional Disorders, and Substance Abuse.
- E. Methadone maintenance treatment.
- F. Partial Hospitalization in a Qualified Partial Hospitalization Program.
- G. Electroconvulsive therapy.

8.3 Inpatient Mental Health and Substance Abuse Services

Benefits are provided when the Member is admitted under the care of a Contracting Physician as an inpatient in a hospital or other CareFirst BlueChoice-approved health care facility for treatment of Mental Illness, Emotional Disorders, and Substance Abuse as follows:

- A. Hospital benefits will be provided, as described in Section 4, Inpatient Hospital Services, of this In-Network Description of Covered Services, on the same basis as a medical (non-Mental Health or Substance Abuse) admission.
- B. Contracting Provider services provided to a hospitalized Member, including physician visits, charges for intensive care, or consultative services, only if CareFirst BlueChoice determines that the Contracting Provider rendered services to the Member and that such services were medically required to diagnose or treat the Member's condition.

The following benefits apply if the Member is an inpatient in a hospital covered under inpatient hospitalization benefits following CareFirst BlueChoice certification of the need and continued appropriateness of such services in accordance with CareFirst BlueChoice utilization management requirements:

- 1. Contracting Provider visits during the Member's hospital stay;
- 2. Intensive care that requires a Contracting Provider's attendance;
- 3. Consultation by another Contracting Provider when additional skilled care is required because of the complexity of the Member's condition; and
- C. Benefits are available for diagnosis and treatment for Substance Abuse, including inpatient detoxification and rehabilitation services in an acute care hospital or Qualified Treatment Facility. Members must meet the applicable criteria for acceptance into, and continued participation in, treatment facilities/programs, as determined by CareFirst BlueChoice.
- D. Electroconvulsive therapy.

SECTION 9 EMERGENCY SERVICES AND URGENT CARE

CareFirst BlueChoice provides coverage for the services listed below at hospital emergency rooms and Urgent Care facilities inside or outside of the Service Area.

9.1 <u>Emergency Services and Urgent Care</u>

- A. Benefits are available to a Member for Emergency Services and Urgent Care twenty-four (24) hours per day. Benefits for Emergency Services, Urgent Care, and follow-up care after emergency surgery are provided regardless of where Covered Services are provided.
- B. In the case of a hospital that has an emergency department, benefits include:
 - 1. Appropriate medical screening;
 - 2. Assessment and stabilization services;
 - 3. Ancillary services routinely available to the emergency department to determine whether or not an Emergency Medical Condition exists; and
 - 4. Medically Necessary observation to determine whether the Member's condition requires inpatient hospitalization.
- C. Benefits are available for the costs of a voluntary HIV test, performed during a Member's visit to a hospital emergency room, regardless of the reason for the hospital emergency room visit.

9.2 Notice to CareFirst BlueChoice in the Event of an Emergency

- A. If the Member is admitted to a hospital as a result of an Emergency Medical Condition, CareFirst BlueChoice must be notified the earlier of:
 - 1. The end of the first business day after first receiving the care; or
 - 2. Within forty-eight (48) hours after first receiving the care.
- B. If it was not reasonably possible to give notice, this requirement will be met if notice was given as soon as reasonably possible. The Member must provide information about the Emergency Medical Condition and the care received.

9.3 <u>Ambulance Services</u>

- A. Benefits are available for Medically Necessary air transportation and ground ambulance services. Prior authorization is required for air ambulance services only, except for Medically Necessary air ambulance services in an emergency.
- B. If a Member is outside of the United States and requires treatment for Emergency Services, benefits are provided for Medically Necessary air and ground transportation to the nearest facility where appropriate medical care is available.

9.4 Filing a Claim for a Non-Contracting Provider

When a Member receives Emergency Services, Urgent Care, or follow-up care after emergency surgery from a Non-Contracting Provider, the Member must follow the proof of loss requirements of the In-Network Evidence of Coverage.

9.5 Follow-up Care After Emergency Surgery

If CareFirst BlueChoice authorizes, directs, refers, or otherwise allows a Member to access a hospital emergency facility or other Urgent Care facility for a medical condition that meets the

criteria for Emergency Services, as defined in the In-Network Evidence of Coverage, and the Member requires emergency surgery:

- A. Coverage is provided for services rendered by the physician, surgeon, oral surgeon, periodontist, or podiatrist who performed the surgical procedure for follow-up care that is Medically Necessary and directly related to the condition for which the surgical procedure was performed; and
- B. The Member will be responsible for the same Copayment for each follow-up visit as would be required for a visit to a Contracting Physician for specialty care.

SECTION 10 MEDICAL DEVICES AND SUPPLIES

CareFirst BlueChoice provides benefits for Medical Devices and Supplies obtained through designated Contracting Providers. See Section 14, General Provisions, for Medical Devices and Supplies that require prior authorization.

10.1 Definitions

Durable Medical Equipment means equipment which:

- A. Is primarily and customarily used to serve a medical purpose;
- B. Is not useful to a person in the absence of illness or injury;
- C. Is ordered or prescribed by a Contracting Provider;
- D. Is consistent with the diagnosis;
- E. Is appropriate for use in the home;
- F. Is reusable; and
- G. Can withstand repeated use.

<u>Inherited Metabolic Disease</u> means a disease caused by an inherited abnormality of body chemistry, including a disease for which the state screens newborn babies.

Low Protein Modified Food Product means a food product that is:

- A. Specially formulated to have less than 1 gram of protein per serving; and
- B. Intended to be used under the direction of a physician for the dietary treatment of an Inherited Metabolic Disease.

Low Protein Modified Food Product does not include a natural food that is naturally low in protein.

<u>Medical Devices</u> means Durable Medical Equipment, medical formulas, Medical Supplies, Orthotic Devices and Prosthetic Devices.

Medical Food means a food that is:

- A. Intended for the dietary treatment of a disease or condition for which nutritional requirements are established by medical evaluation; and
- B. Formulated to be consumed or administered under the direction of a physician.

Medical Supplies means items that:

- A. Are primarily and customarily used to serve a medical purpose;
- B. Are not useful to a person in the absence of illness or injury;
- C. Are ordered or prescribed by a Contracting Provider;
- D. Are consistent with the diagnosis;

- E. Are appropriate for use in the home;
- F. Cannot withstand repeated use; and
- G. Are usually disposable in nature.

Orthotic Devices means orthoses and braces which:

- A. Are primarily and customarily used to serve a therapeutic medical purpose;
- B. Are prescribed by a Contracting Provider;
- C. Are corrective appliances that are applied externally to the body to limit or encourage its activity, to aid in correcting or preventing deformity, or to provide mechanical support;
- D. May be purely passive support or may make use of spring devices; and
- E. Include devices necessary for post-operative healing.

Prosthetic Devices means devices which:

- A. Are primarily intended to replace all or part of an organ or body part that has been lost due to disease or injury; or
- B. Are primarily intended to replace all or part of an organ or body part that was absent from birth; or
- C. Are intended to anatomically replace all or part of a bodily function which is permanently inoperative or malfunctioning; and
- D. Are prescribed by a Contracting Provider; and
- E. Are removable and attached externally to the body.

10.2 Covered Services

A. Durable Medical Equipment

Rental, or, (at CareFirst BlueChoice's option), purchase and replacements or repairs of Medically Necessary Durable Medical Equipment prescribed by a Contracting Provider for therapeutic use for a Member's medical condition.

CareFirst BlueChoice's payment for rental will not exceed the total cost of purchase. CareFirst BlueChoice's payment is limited to the least expensive Medically Necessary Durable Medical Equipment adequate to meet the Member's medical needs. CareFirst BlueChoice's payment for Durable Medical Equipment includes related charges for handling, delivery, mailing, shipping, and taxes.

- B. Medical Supplies
- C. Medical Foods and Low Protein Modified Food Products Medical Foods and Low Protein Modified Food Products for the treatment of Inherited Metabolic Diseases are covered if the Medical Foods or Low Protein Modified Food Products are:
 - 1. Prescribed as Medically Necessary for the therapeutic treatment of Inherited Metabolic Diseases; and;
 - 2. Administered under the direction of a physician.

D. Nutritional Substances

Enteral and elemental nutrition when Medically Necessary as determined by CareFirst BlueChoice.

E. Diabetes Equipment and Supplies

- 1. Coverage will be provided for all Medically Necessary and medically appropriate equipment and diabetic supplies necessary for the treatment of diabetes (Types I and II), or elevated blood glucose levels induced by pregnancy.
- 2. Coverage includes Medically Necessary and medically appropriate equipment and diabetic supplies necessary for the treatment of insulin-dependent diabetes, insulin-using diabetes, gestational diabetes and noninsulin-using diabetes.
- 3. Benefits for insulin syringes and other diabetic supplies described herein are covered in Section 11, Prescription Drugs. All other diabetic equipment is covered as a Medical Device or Supply.

F. Hair Prosthesis

Benefits are available for a hair prosthesis when prescribed by a treating oncologist and the hair loss is a result of chemotherapy or radiation treatment for cancer.

G. Orthotic Devices and Prosthetic Devices Benefits include:

- 1. Supplies and accessories necessary for effective functioning of a Covered Service;
- 2. Repairs or adjustments to Medically Necessary devices that are required due to bone growth or change in medical condition, reasonable weight loss or reasonable weight gain, and normal wear and tear during normal usage of the device; and
- 3. Replacement of Medically Necessary devices when repairs or adjustments fail and/or are not possible.

10.3 Repairs

Benefits for the repair, maintenance, or replacement of covered Durable Medical Equipment are limited as follows:

- A. Coverage of maintenance costs is limited to routine servicing such as testing, cleaning, regulating, and checking of equipment.
- B. Coverage of repairs costs is limited to adjustment required by normal wear or by a change in the Member's condition, and repairs necessary to make the equipment/appliance serviceable. Repair will not be authorized if the repair costs exceed the market value of the appliance, prosthetic, or equipment.
- C. Replacement coverage is limited to once every two (2) years due to irreparable damage and/or normal wear, or a significant change in medical condition. Replacement costs necessitated as a result of malicious damage, culpable neglect, or wrongful disposition of the equipment or device on the part of the Member or of a family member are not covered.

10.4 Benefit Limits

Benefits are limited to the least expensive Medically Necessary Durable Medical Equipment, Medical Supply, Orthotic Device or Prosthetic adequate to meet the patient's medical needs.

Purchase or rental of any Medical Device is at the discretion of CareFirst BlueChoice. Benefits will be limited to the lower cost of purchase or rental, taking into account the length of time the

Member requires, or is reasonably expected to require the equipment, and the durability of the equipment, etc. The purchase price or rental cost must be the least expensive of its type adequate to meet the medical needs of the Member. If the Member selects a deluxe version of the appliance, device, or equipment not determined by CareFirst BlueChoice to be Medically Necessary, CareFirst BlueChoice will pay an amount that does not exceed CareFirst's payment for the basic device (minus any Member Copayment or Coinsurance) and the Member will be fully responsible for paying the remaining balance.

10.5 Responsibility of CareFirst BlueChoice

CareFirst BlueChoice will not be liable for any claim, injury, demand, or judgment based on tort or other grounds (including warranty of equipment), arising out of or in connection with the rental, sale, use, maintenance, or repair of Prosthetic Devices, corrective appliances or Durable Medical Equipment, whether or not covered under this In-Network Description of Covered Services.

SECTION 11 PRESCRIPTION DRUGS

11.1 <u>Covered Services</u>

Benefits will be provided for Prescription Drugs, including but not limited to:

- A. Any self-administered contraceptive drug or device, including a contraceptive drug and device on the Preventive Drug List, that is approved by the FDA for use as a contraceptive and is obtained under a prescription written by an authorized prescriber. See Section 1.5.B, Contraceptive Methods and Counseling, for additional coverage of contraceptive drugs and devices.
- B. Human growth hormones. Prior authorization is required.
- C. Any drug that is approved by the FDA as an aid for the cessation of the use of tobacco products and is obtained under a prescription written by an authorized prescriber, including drugs listed in the Preventive Drug List.

<u>Nicotine Replacement Therapy</u>. Nicotine Replacement Therapy means a product, including a product on the Preventive Drug List that is used to deliver nicotine to an individual attempting to cease the use of tobacco products, approved by the FDA as an aid for the cessation of the use of tobacco products and obtained under a prescription written by an authorized prescriber. Coverage for Nicotine Replacement Therapy will be provided on an unlimited yearly basis.

- D. Injectable medications that are self-administered and the prescribed syringes.
- E. Standard covered items such as insulin, glucagon and anaphylaxis kits.
- F. Fluoride products.
- G. Diabetic Supplies.
- H. Oral chemotherapy drugs.
- I. Hormone replacement therapy drugs.
- Mail Order Program. Except as provided in Section 12.3, all Members have the option of ordering Prescription Drugs via mail order. Members ordering Prescription Drugs through the mail order program will be entitled to a thirty (30) day supply for non-Maintenance Drugs and a ninety (90) day supply for Maintenance Drugs. A Member may obtain up to a twelve (12) month supply of contraceptives at one time.
- 11.3 Benefits for Specialty Pharmacy Prescription Drugs. Benefits will be provided for Specialty Drugs only when obtained from a Pharmacy that is part of the Exclusive Specialty Pharmacy Network.

SECTION 12 PATIENT-CENTERED MEDICAL HOME

12.1 Definitions.

<u>Care Coordination Team</u> means the health care providers involved in the collaborative process of assessment, planning, facilitation, and advocacy for options and services to meet the Member's health needs through communication and available resources to promote quality cost-effective outcomes.

<u>Care Plan</u> means the plan directed by a health care provider, and coordinated by a nurse coordinator and Care Coordination Team, with engagement by the Qualifying Individual. The Care Plan is created in accordance with the PCMH goals and objectives.

<u>Patient-Centered Medical Home Program ("PCMH")</u> means medical and associated services directed by the PCMH team of medical professionals to:

- A. Foster the health care provider's partnership with a Qualifying Individual and, where appropriate, the Qualifying Individual's primary caregiver;
- B. Coordinate ongoing, comprehensive health care services for a Qualifying Individual; and
- C. Exchange medical information with CareFirst BlueChoice, other providers and Qualifying Individuals to create better access to health care, increase satisfaction with medical care, and improve the health of the Qualifying Individual.

<u>Qualifying Individual</u> means a Member with a chronic condition, serious illness or complex health care needs, as determined by CareFirst BlueChoice, requiring coordination of health services and who agrees to participate in the PCMH.

12.2 Covered Benefits

Benefits will be provided for the costs associated with the coordination of care for the Qualifying Individual's medical conditions, including:

- A. Assess the Qualifying Individual's medical needs;
- B. Provide liaison services between the Qualifying Individual and the health care provider(s) and the Care Coordination Team;
- C. Create and supervise the Care Plan;
- D. Educate the Qualifying Individual and family regarding the Qualifying Individual's disease and self-care techniques;
- E. Arrange consultations with Specialists and assist with obtaining Medically Necessary supplies and services, including community resources, for the Member; and
- F. Assess treatment compliance.

12.3 <u>Limitations</u>

Benefits provided through the Patient-Centered Medical Home Program are available only when provided by a CareFirst BlueChoice-approved health care provider who has elected to participate in the PCMH.

SECTION 13 COMPLEX CHRONIC OR HIGH RISK ACUTE DISEASE MANAGEMENT

13.1 Definitions.

<u>Chronic Care Coordination Program (CCC Program)</u> means the assessment and coordination of primary care services to a Qualified Member with multiple chronic and severe health conditions.

<u>Complex Case Management Program (CCM Program)</u> means the assessment and coordination of specialty services provided to a Qualified Member with advanced or critical illnesses.

<u>Designated Provider</u> means a provider contracted with CareFirst BlueChoice to provide services under CareFirst BlueChoice's Total Care and Cost Improvement Program, which includes the following components: PCMH Program, CCM Program, CCC Program, Comprehensive Medication Review, Enhanced Monitoring Program, Expert Consultation Program, Home-Based Services Program, Hospice and Palliative Care Program, Pharmacy Coordination Program, Substance Abuse and Behavioral Health Program, or other community-based programs outlined in this Section (collectively, the "TCCI Programs") and who has agreed to participate in care coordination activities in cooperation with CareFirst BlueChoice for Qualified Members with complex chronic disease or high risk acute conditions.

Qualified Member means a Member who:

- A. Is accepted by CareFirst BlueChoice into one or more of the TCCI Programs described in this Section. CareFirst BlueChoice will consult with the Member's treating physician or nurse practitioner to determine whether the Member has a medical condition that meets the parameters for participation in one or more of the TCCI Programs.
- B. Consents to participate and complies with all elements of the TCCI Program(s) in which he/she qualifies including use of a Designated Provider.
- C. Continues to meet the criteria for participation in the TCCI Program(s) and participates fully with any applicable plan of care or other requirements, including compliance with direction from a PCP or Specialist while under a plan of care.
- D. CareFirst BlueChoice and the Qualified Member's treating physician or nurse practitioner determine is cooperating with, and satisfying the requirements of the TCCI Program(s). CareFirst BlueChoice retains final authority to determine whether a Member is a Qualified Member.

13.2. Benefits and Cost Sharing Waiver.

- A. Qualified Members are eligible for a waiver of their cost sharing responsibility for benefits provided under this Section when:
 - 1. While in an active plan of care, the Qualified Member participates in either (a) a CCC Program coordinated by the Qualified Member's PCP who participates in CareFirst BlueChoice's Patient-Centered Medical Home Program or (b) a CCM Program coordinated by the Qualified Member's Specialist, or
 - 2. At CareFirst BlueChoice's initiation, and in consultation with and direction from the Qualified Member's treating provider or nurse practitioner, the Qualified Member participates in one or more of the TCCI Program elements outside of a plan of care and without participating in CCM Program or CCC Program.

- B. Qualified Members participating in a CCM Program or CCC Program as set forth in Section 13.2A.1 are eligible for the following CCM Program and CCC Program benefits while in an active plan of care:
 - Assessment of Qualified Member/family needs related to understanding health care status and physician treatment plans, self-care, compliance capability, and continuum of care;
 - 2. Education of Qualified Member/family regarding illness, physician treatment plans, self-care techniques, treatment compliance, and continuum of care;
 - 3. Assistance in navigating and coordinating health care services and understanding benefits;
 - 4. Assistance in arranging for a primary care physician to deliver and coordinate the Qualified Member's care;
 - 5. Assistance in arranging consultation(s) with Specialists;
 - 6. Identification of and connection to community resources, and other organizations/support services to supplement the Qualified Member's plan of care;
 - 7. Implementation of a plan of care under the direction of the Qualified Member's treating physician or nurse practitioner.
 - 8. Coordination of care, either telephonically or otherwise, between a Designated Provider and a Qualified Member and his/her treating physician.
 - 9. Other Medically Necessary services provided to a Qualified Member while in an active plan of care.
- C. Qualified Members participating in a CCM Program or CCC Program while in an active plan of care under Section 13.2A.1 or, pursuant to CareFirst BlueChoice initiation under Section 13.2A.2, are eligible for benefits under following TCCI Program elements:
 - 1. Comprehensive Medication Review (CMR). Benefits will be provided for a pharmacist's review of medications and consultation with the Qualified Member to improve the effectiveness of pharmaceutical therapy.
 - 2. Enhanced Monitoring Program (EMP). Benefits will be provided for the medical equipment and monitoring services provided to a Qualified Member with a chronic condition or disease in conjunction with the EMP for maintenance of the Qualified Member's chronic condition or disease.
 - 3. Expert Consultation Program (ECP). Benefits will be provided for a review by a team of specialists of a Qualified Member's medical records where the Qualified Member has a complex or rare condition or multiple conditions or diseases for which the course of treatment requires unique expertise.
 - 4. Home-Based Services Program (HBS). Benefits will be provided for medical and associated services specifically outlined in a home-based care management plan. Covered Services provided to a Qualified Member pursuant to a home-based care management plan under this section will not count toward any visit limits stated in the Schedule of Benefits.
 - 5. Hospice and Palliative Care Program. Benefits will be provided for medical and associated services specifically outlined in a hospice/palliative plan of care.

- 6. Pharmacy Coordination Program. Benefits will be provided for care coordination services related to a Qualified Member's use of Specialty Drugs.
- 7. Substance Abuse and Behavioral Health Program. Benefits will be provided for care coordination services related to a Qualified Member's use of mental health and substance abuse services, including behavioral health treatment benefits.
- D. Qualified Member Cost Sharing Responsibilities.
 - 1. Under this section, any applicable cost-sharing responsibilities will be waived for (i) TCCI Program services provided by a Designated Provider and (ii) In-Network services provided to Qualified Members in an active plan of care.
 - Cost-sharing responsibilities are not waived for any (i) prescription drug or other drug benefits; (ii) services provided in an inpatient institution or facility; or (iii) services provided in a hospital.
 - 2. If the Qualified Member's Evidence of Coverage is compatible with a federally-qualified Health Savings Account:
 - a) If the Qualified Member has funded his/her HSA account during the Benefit Period, then the Qualified Member will be responsible for any associated costs for services under this Section until the annual Deductible has been met, unless the Covered Services appear on the list of preventive services maintained by the Internal Revenue Service.
 - b) If the Qualified Member has not funded his/her HSA account during the Benefit Period, then if the Qualified Member agrees not to fund his/her HSA account and provides a signed agreement not to fund his/her HSA account, then the Qualified Member will be eligible for the waiver described in 13.2D.1.

E. Termination.

- 1. The Qualified Member's participation in the TCCI Program(s) and receipt of benefits and cost-sharing waivers under this Section will be terminated under the following circumstances:
 - a) The Qualified Member completes the stated goals of the TCCI Program(s) set forth in the Qualified Member's plan of care and confirmed by the Qualified Member's treating physician or nurse practitioner or, if the TCCI Program(s) benefits are provided to Members not in an active plan of care, when confirmed by the Qualified Member's treating physician or nurse practitioner.
 - b) The CareFirst BlueChoice designated nurse, provider, or care coordinator and the Qualified Member's treating physician or nurse practitioner determine that the Qualified Member failed to comply with the TCCI Program(s) and/or any related plan of care or treatment under this Section. The Qualified Member will be given thirty (30) days prior written notice of termination under this subsection.
 - c) The Qualified Member's coverage under the Evidence of Coverage is terminated.
- 2. If termination is the result of the Qualified Member's failure to comply with the TCCI Program(s) under Section 13.2E.1.(b), the Qualified Member will be provided the opportunity to comply with the TCCI Program(s) during the thirty

- (30) day notice period. If after consultation between the Qualified Member's treating physician or nurse practitioner and the CareFirst BlueChoice designated nurse, provider, or care coordinator a determination is made that the Qualified Member is not and will not be compliant with the applicable TCCI Program(s), the Qualified Member will receive a final written notice of termination of benefits under this Section.
- 3. Upon termination of the Qualified Member's participation in the TCCI Program(s), all cost-sharing waivers and benefits shall be null and void on and after the effective date of the termination of the waiver and the Qualified Member will be responsible for any and all cost-sharing responsibilities as stated in the In-Network Schedule of Benefits on and after the effective date of the termination of the waiver.

SECTION 14 GENERAL PROVISIONS

14.1 How the Plan Works

A. <u>General Coverage</u>

Covered Services must be provided by Contracting Providers, except for Emergency Services, Urgent Care, and follow-up care after emergency surgery.

Contracting Providers will submit claims for Covered Services rendered directly to CareFirst BlueChoice. However, the Member must advise the provider that coverage exists under the In-Network Evidence of Coverage and provide all information necessary to verify coverage within not fewer than 21 business days before the deadline for the Contracting Provider to submit the claim to CareFirst BlueChoice. The Contracting Provider will advise the Member of this deadline upon requesting the coverage verification information. If the Member has provided all information necessary to verify coverage in accordance with this section and the Contracting Provider fails to submit the claim to CareFirst BlueChoice, the Member will be held harmless.

Members do not need a written referral from a Primary Care Physician for Covered Services provided by Contracting Providers. However, services may require a written or oral order from the provider. Certain Covered Services require prior authorization. See Section 14.2.D., Services Subject to Utilization Management.

It is the Member's responsibility to obtain prior authorization for any Covered Services rendered outside of the CareFirst BlueChoice Service Area, except for Emergency Services, Urgent Care, and follow-up care after emergency surgery.

B. Pediatric Vision Coverage

- 1. When the Member receives a vision examination from a Contracting Vision Provider, the benefit payment is accepted as payment in full.
- 2. When a Member receives frames and spectacle lenses or contact lenses from a Contracting Vision Provider, the Member's responsibility is as stated below. The benefit payment is as stated in the attached Schedule of Benefits.
 - a) When the Member receives frames from the display of collection frames (the collection designated by the Vision Care Designee) and basic spectacle lenses from a Contracting Vision Provider, the benefit payment is accepted as payment in full.
 - b) When the Member receives other frames, non-basic spectacle lenses or contact lenses from a Contracting Vision Provider, the Member is responsible for the cost difference between the Vision Care Designee's payment and the Contracting Vision Provider's actual charge.
- 3. When the Member receives Covered Vision Services from a Non-Contracting Vision Provider, the Member is responsible for the cost difference between the Vision Care Designee's payment and the Non-Contracting Vision Provider's actual charge. The Vision Care Designee's payment is stated in the Schedule of Benefits.
- 4. Limited Access Area. If the Member resides in an area that does not have adequate access to a Contracting Vision Provider and the Member receives

Vision Care from a Non-Contracting Vision Provider, the Vision Care Designee will pay up to 100% of the Allowed Benefit. The Member is responsible for any difference between the amount billed and the Vision Care Designee's payment. To determine if the Member resides in a limited access area, the Member must call the Vision Care Designee at the telephone number on the Member's identification card.

C. Pediatric Dental Services

- 1. The pediatric dental coverage offers a choice of Dentists. Payment depends on the Dentist chosen as described in the applicable Schedule of Benefits.
- 2. If a conflict arises regarding the quality and extent of work, the case in question will be submitted to the CareFirst BlueChoice Dental Director for resolution as described in the Benefit Determination and Appeals section of the In-Network Evidence of Coverage.
- 3. Benefits for Covered Dental Services will be provided by CareFirst BlueChoice as stated in the Schedules of Benefits. The date a service is received or the date supplies are purchased will be the date such expenses are incurred.

4. Member/Provider Relationship

- a) The Member has the exclusive right to choose a Dentist. Whether a Dentist is either a Preferred or Participating Dentist relates only to method of payment, and does not imply that any Dentist is more or less qualified than another.
- b) CareFirst BlueChoice makes payment for Covered Dental Services, but does not provide these services. CareFirst BlueChoice is not liable for any act or omission of any Dentist.

5. Estimate of Eligible Benefits

A Dentist may propose a planned dental treatment or series of dental procedures. A Member may choose to obtain a written estimate of the benefits available for such procedure(s).

CareFirst BlueChoice encourages a Member to obtain a written Estimate of Eligible Benefits (CareFirst BlueChoice's written estimate of benefits before a service is rendered) also known as a pre-treatment estimate (PTE) for major dental procedures, thereby alerting a Member of the out-of-pocket expenses that may be associated with the treatment plan, related deductibles, co-insurance and/or procedures that are not Covered Dental Services. Based on an Estimate of Eligible Benefits or PTE from CareFirst BlueChoice, a Member can decide whether or not to incur the expense that may be associated with a particular treatment plan. Failure to obtain an Estimate of Eligible Benefits or PTE has no effect on the benefits to which a Member is entitled under this In-Network Evidence of Coverage, except for orthodontic services. A Member may choose to forgo the Estimate of Eligible Benefits or PTE and proceed with treatment, unless orthodontic services are planned. The process for orthodontic services is described below.

After the services are rendered, the claim will be reviewed by CareFirst BlueChoice. Should the review determine that the service(s) rendered meet CareFirst BlueChoice's criteria for benefits, the benefits will be provided as described in this In-Network Evidence of Coverage. However, should the review of the claim determine that the treatment or procedure(s) did not meet CareFirst

BlueChoice's criteria for benefits, benefits will not be provided.

To request an Estimate of Eligible Benefits or PTE prior to receiving dental treatment or dental procedures, a Member should contact his or her Dentist who will coordinate the request on the Member's behalf. If the Dentist has any questions about the process, he or she may contact the CareFirst BlueChoice Provider Services Department or go to the CareFirst BlueChoice website at www.carefirst.com, which lists information in the Physicians and Providers section, under the subsection for Dental, and list of Resources. The Estimate of Eligible Benefits or PTE is merely an estimate, and it cannot be considered a guarantee of the Member's benefits or enrollment under this In-Network Evidence of Coverage.

The process is different for orthodontic services. The Affordable Care Act requires that orthodontics must be Medically Necessary to be Covered Dental Services. To request a PTE for orthodontic services, the Member must see an orthodontist who will do an exam and orthodontic assessment that may include taking orthodontic records (study models and certain x-rays). The orthodontist will then complete a case assessment using a scoring tool required by the state. Then the orthodontic records and case assessment will be sent to CareFirst BlueChoice for evaluation and confirmation of the assessment score. If the score meets or exceeds the baseline requirement, the orthodontics will be approved for the Member. If the score is less than the minimal required score, then the request for orthodontic benefits will be denied.

A decision by CareFirst BlueChoice to deny benefits as described in this section constitutes an Adverse Decision as defined in the In-Network Evidence of Coverage if the decision is based on a finding that the proposed service is not Medically Necessary, appropriate, or efficient.

D. <u>Prescription Drug Coverage</u>

- 1. Accessing the Prescription Drug Benefit Card Program
 - a) Members may use his/her identification card to purchase Prescription Drugs from Contracting Pharmacy Providers. If the Prescription Drug coverage includes a Deductible, the Member must pay the entire cost of the Prescription Drug(s) until the Deductible is satisfied. Once the Deductible, if applicable, has been satisfied, the Member pays the appropriate Copayment or Coinsurance as stated in the Schedule of Benefits.
 - b) For Prescription Drugs or diabetic supplies purchased from a non-Contracting Pharmacy Provider, the Member is responsible for paying the total charge and submitting a claim to CareFirst BlueChoice or its designee for reimbursement. Members will be entitled to reimbursement from CareFirst BlueChoice or its designee up to the amount of the Prescription Drug Allowed Benefit, minus any applicable Deductible, Copayment, or Coinsurance.
 - c) Members have the option of ordering Prescription Drugs via mail order. The mail order program provides its Member's with a Pharmacy that has an agreement with CareFirst BlueChoice or its designee, to provide mail service Prescription Drugs in accordance with the terms of this provision. The Member is responsible for any applicable Deductible, Copayment or Coinsurance.
- 2. Additional Terms and Conditions

- a) Providers must obtain prior authorization by providing information to support Medical Necessity before prescribing any Prescription Drug contained in the Prescription Guidelines. A copy of the Prescription Guidelines is available to the Member or provider upon request.
- b) Providers may substitute a Generic Drug for a Brand Name Drug. If there is no Generic Drug for the Brand Name Drug, the Member shall pay the applicable Copayment as stated in the In-Network Schedule of Benefits for Non-Preferred Brand Name Drugs.
- c) If a provider prescribes a non-Preferred Brand Name Drug, and the Member selects the non-Preferred Brand Name Drug when a Generic Drug is available, the Member shall pay the applicable Copayment or Coinsurance as stated in the Schedule of Benefits plus the difference between the price of the non-Preferred Brand Name Drug and the Generic Drug. A Member will be allowed to obtain a non-Preferred Brand Name Drug in place of an available Generic Drug and pay only the non-Preferred Brand Name Drug Copayment or Coinsurance when Medically Necessary, as determined by CareFirst BlueChoice.

14.2 Utilization Management

- A. Benefits are subject to review and approval under utilization management requirements established by CareFirst BlueChoice. Through utilization management, CareFirst BlueChoice will:
 - 1. Review Member care and evaluate requests for approval of coverage in order to determine the Medical Necessity for the services;
 - 2. Review the appropriateness of the hospital or facility requested; and,
 - 3. Determine the approved length of confinement or course of treatment in accordance with CareFirst BlueChoice established criteria.

In addition, utilization management may include additional aspects such as prior authorization, and/or preadmission testing requirements, concurrent review, and discharge planning. Failure or refusal to comply with notice requirements and other utilization management authorization and approval procedures may result in a significant reduction in or exclusion of benefits. If coverage is reduced or excluded for failure to comply with utilization management requirements, the reduction or exclusion may be applied to all services related to the treatment, admission, or portion of the admission for which utilization management requirements were not met.

B. Contracting Provider Responsibility

Contracting Providers are responsible for providing utilization management notices and obtaining necessary utilization management approvals on the Member's behalf for services that require prior authorization. However, the Member must advise these providers that coverage exists under the In-Network Evidence of Coverage. If the Contracting Provider fails to obtain prior authorization, the Member will be held harmless.

C. Member Responsibility

If the Member receives Covered Services outside of the CareFirst BlueChoice Service Area, the Member is responsible for all utilization management requirements. It is the Member's responsibility to ensure that providers associated with the Member's care cooperate with utilization management requirements. This includes initial notification in a timely manner, responding to CareFirst BlueChoice inquiries and, if requested, allowing CareFirst BlueChoice representatives to review medical records on-site or in CareFirst

BlueChoice offices. If CareFirst BlueChoice is unable to conduct utilization reviews, Member benefits may be reduced or excluded from coverage.

To initiate utilization management review, the Member may directly contact CareFirst BlueChoice or may arrange to have notification given by a family member or by the provider that is involved in the Member's care. However, these individuals will be deemed to be acting on the Member's behalf. If the Member and/or the Member's representatives fail to contact CareFirst BlueChoice as required or provide inaccurate or incomplete information, benefits may be reduced or excluded.

Members should share the utilization management requirements with family members and other responsible persons who could arrange for care on the Member's behalf in accordance with these provisions in case the Member is unable to do so when necessary. CareFirst BlueChoice will provide additional information regarding utilization management requirements and procedures, including telephone numbers and hours of operation, at the time of enrollment and at any time upon the Member's request.

D. Services Subject to Utilization Management

Prior authorization from CareFirst BlueChoice will be obtained by Contracting Providers. It is the Member's responsibility to obtain prior authorization for any Covered Services provided outside of the CareFirst BlueChoice Service Area.

1. Hospital Inpatient Services

All hospitalizations (except for maternity and Emergency admissions as specified) require prior authorization. The Member must contact CareFirst BlueChoice (or have the provider contact CareFirst BlueChoice) at least five (5) business days prior to an elective or scheduled admission to the hospital. If the admission cannot be scheduled in advance because it is not feasible to delay the admission for five (5) business days due to the Member's medical condition, CareFirst BlueChoice must receive notification of the admission as soon as possible, but in any event within forty-eight (48) hours following the beginning of the admission or by the end of the first business day following the beginning of the admission, whichever is later.

Emergency Admissions

CareFirst BlueChoice must be notified of a Member's admission to a hospital on an emergency basis in accordance with Section 9.2. CareFirst BlueChoice may not render an adverse decision solely because CareFirst BlueChoice was not notified of the emergency admission within the prescribed period of time after that admission if the Member's condition prevented the hospital from determining the Member's insurance status or CareFirst BlueChoice's emergency admission requirements.

2. Inpatient Mental Illness and Substance Abuse Services

CareFirst BlueChoice must be contacted for prior authorization at least five business days prior to an elective or scheduled admission. If the admission cannot be scheduled in advance because care is required immediately due to the Member's condition, CareFirst BlueChoice must receive notification of the admission as soon as possible but in any event within 48 hours of or by the end of the first business day following the beginning of the admission, whichever is later.

Emergency Admissions

CareFirst BlueChoice must be notified of a Member's admission to a hospital on an emergency basis in accordance with Section 9.2. CareFirst BlueChoice may not render an adverse decision solely because CareFirst BlueChoice was not notified of the emergency admission within the prescribed period of time after that admission if the Member's condition

prevented the hospital from determining the Member's insurance status or CareFirst BlueChoice's emergency admission requirements.

In the case of an inpatient mental health and/or Substance Use Disorder admission of a Member who is determined by the Member's physician or psychologist, in conjunction with a member of the hospital staff who has admitting privileges, to be in imminent danger to self or others, CareFirst BlueChoice may not render an adverse authorization determination until the later of 24 hours after the admission or the next business day after the admission.

3. Organ and Tissue Transplants

Transplants and related services must be coordinated and prior authorization must be obtained from CareFirst BlueChoice. Prior authorization is not required for cornea transplants and kidney transplants. Coverage for related medications is available under Section 11, Prescription Drugs.

- 5. Habilitative Services for Adults;
- 6. Ambulance Services
 Prior authorization is required for air ambulance services only, except for Medically Necessary air ambulance services in an emergency;
- 7. Home Health Services;
- 8. Skilled Nursing Facility Services;
- 9. Inpatient and Outpatient Hospice Care Services;
- 11. Pediatric Vision Services. Prior authorization is required for Low Vision Services and Medically Necessary Contact Lenses.
- 12. General Anesthesia for Dental Services; and,
- 13. Prescription Drugs in the Prescription Guidelines and human growth hormones.

Covered Services not listed in this provision do not require prior authorization. CareFirst BlueChoice reserves the right to make changes to the categories of services that are subject to utilization management requirements or to the procedures the Member and/or the providers must follow. CareFirst BlueChoice will notify the Member of these changes at least forty-five (45) days in advance.

Prior authorization is not required for any Covered Services when Medicare is the primary insurer.

E. Appealing a Utilization Management Decision

If the Member, the Member's representative or the Member's provider disagrees with a utilization management decision, CareFirst BlueChoice will review the decision upon request. A utilization management appeal will be reviewed and decided upon by the CareFirst BlueChoice Medical Director or Associate Medical Director not involved in the initial denial decision. If necessary, the Medical Director or Associate Medical Director will discuss the Member's case with the Member's physician and/or request the opinion of a specialist board certified in the same specialty as the treatment under review. Any noncertification or penalty may be appealed. Additional information is provided in the Benefit Determination and Appeals section of the In-Network Evidence of Coverage regarding how to appeal a utilization management decision.

- 14.3 <u>CareFirst BlueChoice Personnel Availability for Prior Authorization</u>
 CareFirst BlueChoice requires prior authorization for certain medical treatment as stated.
 CareFirst BlueChoice will have personnel available to provide prior authorization at all times when such prior authorization is required.
- 14.4 <u>Concurrent Review and Discharge Planning</u>. Following timely notification, CareFirst BlueChoice will instruct the Member or the Member's representative, as applicable, about the procedures to follow, including the need to submit additional information and any requirements for re-notification during the course of treatment.

14.5 Continuing Care with Terminated Providers

- A. When a Contracting Provider terminates their agreement with CareFirst BlueChoice, for any reason except for cause, benefits will be provided for continuing care rendered by the terminated provider as described in this section. CareFirst BlueChoice will send a notice to the Member that the Contracting Provider is no longer available.
- B. The Member may, upon request, continue to receive Covered Services from his/her Primary Care Physician for up to 90 days after the date of the notice of the Primary Care Physician's termination from CareFirst BlueChoice's provider panel, if termination was for reasons unrelated to fraud, patient abuse, incompetence, or loss of licensure status. In addition, a Member may continue treatment with a terminated provider if:
 - 1. A Member was in an active course of treatment with the terminated Contracting Provider prior to the date the Member was notified. The Member needs to request, from CareFirst BlueChoice, to continue receiving care from the terminated Contracting Provider. Benefits will be provided for a period of 90 days from the date the Member is notified by CareFirst BlueChoice that the terminated Contracting Provider is no longer available.
 - 2. A Member who has entered her second trimester of pregnancy may continue to receive Covered Services from the terminated Contracting Provider through postpartum care directly related to the delivery.
 - 3. A Member that was terminally ill (as defined by § 1861(dd)(3)(A) of the Social Security Act) at the time the Contracting Provider's agreement terminated may continue to receive Covered Services directly related to the treatment of the terminal illness until the Member dies.

14.6 <u>Limitation on Provider Coverage</u>

Services are covered only if the provider is licensed in the jurisdiction in which the services are rendered and if the services are within the lawful scope of the services for which that provider is licensed. Coverage does not include services rendered to the Member by any individual who:

- A. is the Member's Spouse, domestic partner, mother, father, daughter, son, brother or sister; or
- B. resides in the Member's home.
- 14.7 Coverage for Services Rendered Outside of the Service Area

Benefits will be provided for Covered Services rendered outside of the Service Area. All Covered Services rendered by participating providers under an Inter-Plan Program outside of the Service Area will be covered under this In-Network Evidence of Coverage. Emergency Services, Urgent Care, and follow-up care after emergency surgery rendered outside of the Service Area by non-participating providers will also be covered under this In-Network Evidence of Coverage. See the Inter-Plan Arrangements Disclosure section of the In-Network Evidence of Coverage for additional information.

When Covered Services are rendered outside of the Service Area, the Member will be responsible for making arrangements with CareFirst BlueChoice to obtain Utilization Management authorizations and approvals required for coverage. Refer to Section 14.2.D. for services that require prior authorization.

SECTION 15 EXCLUSIONS AND LIMITATIONS

15.1 General Exclusions

Coverage is not provided for:

- A. Any services, tests, procedures, or supplies which CareFirst BlueChoice determines are not necessary for the prevention, diagnosis, or treatment of the Member's illness, injury, or condition. Although a service or supply may be listed as covered, benefits will be provided only if it is Medically Necessary and appropriate in the Member's particular case.
- B. Any treatment, procedure, facility, equipment, drug, drug usage, device, or supply which is Experimental/Investigational, or not in accordance with accepted medical or psychiatric practices and standards in effect at the time of treatment, except for covered benefits for clinical trials.
- C. The cost of services that:
 - 1. Are furnished without charge; or
 - 2. Are normally furnished without charge to persons without health insurance coverage; or
 - 3. Would have been furnished without charge if a Member were not covered under the Evidence of Coverage or under any health insurance.

This exclusion does not apply to:

- a) Medicaid;
- b) Care received in a Veteran's hospital unless the care is rendered for a condition that is a result of a Member's military service.
- D. Any service, supply, drug or procedure that is not specifically listed in the Member's Evidence of Coverage as a covered benefit or that do not meet all other conditions and criteria for coverage at the discretion of CareFirst BlueChoice. The provision of services by a Contracting Provider does not, by itself, entitle a Member to benefits if the services are not covered or do not otherwise meet the conditions and criteria for coverage.
- E. Except for Emergency Services, Urgent Care, and follow-up care after emergency surgery, benefits will not be provided for any service(s) provided to a Member by Non-Contracting Physicians or Non-Contracting Providers, unless written prior authorization is specifically obtained from CareFirst BlueChoice.
- F. Routine, palliative, or Cosmetic foot care (except for conditions determined to be Medically Necessary at the discretion of CareFirst BlueChoice), including flat foot conditions, supportive devices for the foot, treatment of subluxations of the foot, care of corns, bunions (except capsular or bone surgery), calluses, toe nails, fallen arches, weak feet, chronic foot strain, and symptomatic complaints of the feet.
- G. Routine eye examinations and vision services. This exclusion does not apply to evidence-informed preventive care and screenings provided for in the comprehensive guidelines supported by the Health Resources and Services Administration for infants, children, and adolescents and as stated in Section 3.
- H. Any type of dental care (except treatment of accidental bodily injuries, oral surgery, cleft lip or cleft palate or both, and pediatric dental services), including extractions, treatment of

cavities, care of the gums or bones supporting the teeth, treatment of periodontal abscess and periodontal disease, removal of teeth, orthodontics, replacement of teeth, or any other dental services or supplies. Benefits for accidental bodily injury are described in Section 1.16. Benefits for oral surgery are described in Section 1.17. Benefits for treatment of cleft lip, cleft palate or both are described in Section 1.18. Benefits for pediatric dental services are described in Section 2. All other procedures involving the teeth or areas and structures surrounding and/or supporting the teeth, including surgically altering the mandible or maxillae (orthognathic surgery) for Cosmetic purposes or for correction of malocclusion unrelated to a documented functional impairment are excluded.

- I. Cosmetic surgery (except benefits for reconstructive breast surgery or reconstructive surgery) or other services primarily intended to correct, change, or improve appearances. Cosmetic means a service or supply which is provided with the primary intent of improving appearances and not for the purpose of restoring bodily function or correcting deformity resulting from disease, trauma, or previous therapeutic intervention as determined by CareFirst BlueChoice.
- J. Treatment rendered by a health care provider who is the Member's Spouse, parent, child, grandparent, grandchild, sister, brother, great grandparent, great grandchild, aunt, uncle, niece, or nephew, or resides in the Member's home.
- K. All non-Prescription Drugs, medications, biologicals, and Over-the-Counter disposable supplies, routinely obtained without a prescription and self-administered by the Member, except as listed as a Covered Service above, including but not limited to: cosmetics or health and beauty aids, support devices, non-medical items, foot care items, first aid and miscellaneous medical supplies (whether disposable or durable), personal hygiene supplies, incontinence supplies, and Over-the-Counter medications and solutions, except for Over-the-Counter medication or supplies dispensed under a written prescription by a health care provider that is identified in the current recommendations of the United States Preventive Services Task Force that have in effect a rating of "A" or "B".
- L. Foods or formulas consumed as a sole source of supplemental nutrition, except as listed as a Covered Service in this In-Network Description of Covered Services.
- M. All assisted reproductive technologies including artificial insemination and intrauterine insemination, in vitro fertilization, gamete intra-fallopian tube transfer, zygote intra-fallopian transfer cryogenic preservation or storage of eggs and embryo and related evaluative procedures, drugs, diagnostic services and medical preparations related to the same.
- N. Treatment of sexual dysfunctions or inadequacies including, but not limited to, surgical implants for impotence, medical therapy, and psychiatric treatment.
- O. Fees and charges relating to fitness programs, weight loss, or weight control programs, physical or other programs involving such aspects as exercise, physical conditioning, use of passive or patient-activated exercise equipment or facilities and self-care or self-help training or education, except for diabetes outpatient self-management training and educational services. Cardiac rehabilitation and pulmonary rehabilitation programs are covered as described in Section 1.
- P. Maintenance programs for Physical Therapy, Speech Therapy, and Occupational Therapy for those services as stated in Section 1.7; and Cardiac Rehabilitation and pulmonary rehabilitation as stated in Section 1.10D and E.
- Q. Medical or surgical treatment for obesity, weight reduction, dietary control or commercial weight loss programs, including morbid obesity. This exclusion does not apply to:
 - 1. Well child care visits for obesity evaluation and management;

- 2. Evidence-based items or services for preventive care and screening for obesity that have in effect a rating of "A" or "B" in the current recommendations of the United States Preventive Services Task Force (USPSTF);
- 3. For infants, children, and adolescents, evidence-informed preventive care and screening for obesity provided for in the comprehensive guidelines supported by the Health Resources and Services Administration;
- 4. Office visits for the treatment of childhood obesity; and
- 5. Professional Nutritional Counseling and Medical Nutrition Therapy as described in this In-Network Description of Covered Services.
- R. Medical or surgical treatment of myopia or hyperopia, including radial keratotomy and other forms of refractive keratoplasty or any complications thereof.
- S. Services that are beyond the scope of the license of the provider performing the service.
- T. Services furnished as a result of a referral prohibited by law.
- U. Services that are solely based on court order or as a condition of parole or probation, unless approved by CareFirst BlueChoice.
- V. Health education classes and self-help programs, other than birthing classes or those for the treatment of diabetes.
- W. Acupuncture services, except when approved or authorized by CareFirst BlueChoice when used for anesthesia.
- X. Any service related to recreational activities. This includes, but is not limited to, sports, games, equestrian, and athletic training. These services are not covered unless authorized or approved by CareFirst BlueChoice even though they may have therapeutic value or be provided by a health care provider.
- Y. Private duty nursing.
- Z. Services or supplies for injuries or diseases related to a covered person's job to the extent the covered person is required to be covered by a workers compensation law.
- AA. Non-medical services. including, but is not limited to:
 - 1. Telephone consultations, failure to keep a scheduled visit, completion of forms (except for forms that may be required by CareFirst BlueChoice), copying charges or other administrative services provided by the health care provider or the health care provider's staff.
 - 2. Administrative fees charged by a physician or medical practice to a Member to retain the physician's or medical practices services, e.g., "concierge fees" or boutique medical practice membership fees. Benefits under the In-Network Evidence of Coverage are available for Covered Services rendered to the Member by a health care provider.
- BB. Rehabilitation services, including Speech Therapy, Occupational Therapy, or Physical Therapy, for conditions not subject to improvement.
- CC. Non-medical ancillary services such as vocational rehabilitation, employment counseling, or educational therapy.

- DD. Services or supplies resulting from accidental bodily injuries arising out of a motor vehicle accident to the extent the services are payable under a medical expense payment provision of an automobile insurance policy, excluding no fault insurance.
- EE. Transportation and travel expenses (except for Medically Necessary air and ground ambulance services, as determined by CareFirst BlueChoice, and services listed under Section 1.12, Organ and Tissue Transplants, of this In-Network Description of Covered Services), whether or not recommended by a health care provider.
- FF. Services or supplies received from a dental or medical department maintained by or on behalf of an employer, mutual benefit association, labor union, trust, or similar persons or groups.
- GG. Services, drugs, or supplies the Member receives without charge while in active military service.
- HH. Habilitative Services delivered through early intervention and school services.
- II. Custodial Care.
- JJ. Services or supplies received before the Effective Date of the Member's coverage under the Evidence of Coverage.
- KK. Durable Medical Equipment or Medical Supplies associated or used in conjunction with non-covered items or services.
- LL. Services required solely for administrative purposes, including but not limited to employment, insurance, foreign travel, school, camp admissions or participation in sports activities.
- MM. Work Hardening Programs. Work Hardening Program means a highly specialized rehabilitation program designed to simulate workplace activities and surroundings in a monitored environment with the goal of conditioning the participant for a return to work.
- NN. Chiropractic services or spinal manipulation treatment other than spinal manipulation treatment for musculoskeletal conditions of the spine.
- OO. Benefits will not be provided for Specialty Pharmacy Prescription Drugs obtained from Pharmacy that is not part of the Exclusive Specialty Pharmacy Network.

15.2 Pediatric Dental Services

A. Limitations

- 1. Covered Dental Services must be performed by or under the supervision of a Dentist with an active and unrestricted license, within the scope of practice for which licensure or certification has been obtained.
- 2. Benefits will be limited to standard procedures and will not be provided for personalized restorations or specialized techniques in the construction of dentures, including precision attachments and custom denture teeth.
- 3. If a Member switches from one Dentist to another during a course of treatment, or if more than one Dentist renders services for one dental procedure, CareFirst BlueChoice shall pay as if only one Dentist rendered the service.

- 4. CareFirst BlueChoice will reimburse only after all dental procedures for the condition being treated have been completed (this provision does not apply to orthodontic services).
- 5. In the event there are alternative dental procedures that meet generally accepted standards of professional dental care for a Member's condition, benefits will be based upon the lowest cost alternative procedure.

B. Exclusions

Benefits will not be provided for:

- 1. Replacement of a denture or crown as a result of loss or theft.
- 2. Replacement of an existing denture or crown that is determined by CareFirst BlueChoice to be satisfactory or repairable.
- 3. Replacement of dentures, implants, metal and/or porcelain crowns, inlays, onlays, pontics and crown build-ups within 60 months from the date of placement or replacement for which benefits were paid in whole or in part under the terms of the In-Network Evidence of Coverage and are judged by CareFirst BlueChoice to be adequate and functional.
- 4. Gold foil fillings.
- 5. Periodontal appliances.
- 6. Oral orthotic appliances, unless specifically listed as a Covered Dental Service.
- 7. Bacteriologic studies, histopathologic exams, accession of tissue, caries susceptibility tests, diagnostic radiographs, and other pathology procedures, unless specifically listed as a Covered Dental Service.
- 8. Intentional tooth reimplantation or transplantation.
- 9. Interim prosthetic devices, fixed or removable and not part of a permanent or restorative prosthetic service.
- 10. Additional fees charged for visits by a Dentist to the Member's home, to a hospital, to a nursing home, or for office visits after the Dentist's standard office hours. CareFirst BlueChoice shall provide the benefits for the dental service as if the visit was rendered in the Dentist's office during normal office hours.
- 11. Transseptal fiberotomy.
- 12. Orthognathic Surgery.
- 13. The repair or replacement of any orthodontic appliance, unless specifically listed as a Covered Dental Service.
- 14. Any orthodontic services after the last day of the month in which Covered Dental Services ended.
- 15. Separate billings for dental care services or supplies furnished by an employee of a Dentist which are normally included in the Dentist's charges and billed for by them.
- 16. Transitional orthodontic appliance, including a lower lingual holding arch placed where there is not premature loss of the primary molar.

- 17. Limited or complete occlusal adjustments in connection with periodontal surgical treatment when received in conjunction with restorative service on the same date of service.
- 18. Provision splinting, intracoronal and extracoronal.
- 19. Endodontic implant.
- 20. Fabrication of athletic mouthguard.
- 21. Services to alter vertical dimension and/or restore or maintain the occlusion. Such procedures include, but are not limited to, equilibration, periodontal splinting, full mouth rehabilitation, and restoration for misalignment of teeth.
- 22. Adjustments to maxillofacial prosthetic appliance.
- 23. Maintenance and cleaning of a maxillofacial prosthesis (extra or intraoral).
- 24. Any orthodontic services after the last day of the calendar year in which the Member turned age 19.

15.3 Pediatric Vision Services

Benefits will not be provided for the following:

- A. Any pediatric vision service stated in Section 3 for Members over age 19. If Member is under age 19 at the start of the Benefit Period but turns 19 during the Benefit Period, then the Member will receive covered pediatric vision services through the rest of that Calendar Year.
- B. Diagnostic services, except as listed in Section 3.
- C. Services or supplies not specifically approved by the Vision Care Designee where required in this In-Network Description of Covered Services.
- D. Orthoptics, vision training, and low vision aids.
- E. Non-prescription (Plano) lenses and/or glasses, sunglasses or contact lenses.
- F. Except as otherwise provided, Vision Care services that are strictly Cosmetic in nature including, but not limited to, charges for personalization or characterization of prosthetic appliances;
- G. Services and materials not meeting accepted standards of optometric practice.
- H. Services and materials resulting from the Member's failure to comply with professionally prescribed treatment.
- I. Office infection control charges.
- J. State or territorial taxes on vision services performed.
- K. Special lens designs or coatings other than those described herein.
- L. Replacement of lost and/or stolen eyewear.
- M. Two pairs of eyeglasses in lieu of bifocals.

N. Insurance of contact lenses.

15.4 Organ and Tissue Transplants

Benefits will not be provided for the following:

- A. Non-human organs and their implantation. This exclusion will not be used to deny Medically Necessary non-Experimental/Investigational skin grafts that are covered under this In-Network Description of Covered Services.
- B. Any hospital or professional charges related to any accidental injury or medical condition of the donor of the transplant material.
- C. Any charges related to transportation, lodging, and meals unless authorized or approved by CareFirst BlueChoice.
- D. Services for a Member who is an organ donor when the recipient is not a Member.
- E. Donor search services.
- F. Any service, supply, or device related to a transplant that is not listed as a benefit in this In-Network Description of Covered Services.

15.5 Inpatient Hospital Services

Coverage is not provided (or benefits are reduced, if applicable) for the following:

- A. Private room, unless Medically Necessary and/or authorized or approved by CareFirst BlueChoice. If a private room is not authorized or approved, the difference between the charge for the private room and the charge for a semiprivate room will not be covered.
- B. Non-medical items and Convenience Items, such as television and phone rentals, guest trays, and laundry charges.
- C. Except for covered Emergency Services and maternity care, a health care facility admission or any portion of a health care facility admission that had not been approved by CareFirst BlueChoice, whether or not services are Medically Necessary and/or meet all other conditions for coverage.
- D. Private duty nursing.

15.6 <u>Home Health Care Services</u>

Coverage is not provided for:

- Custodial Care.
- B. Services in the Member's home if it is outside the Service Area.
- C. Private duty nursing.

15.7 <u>Hospice Care Services</u>

Benefits will not be provided for the following:

- A. Services, visits, medical equipment, or supplies not authorized by CareFirst BlueChoice.
- B. Financial and legal counseling.
- C. Any services for which a Qualified Hospice Care Program does not customarily charge the patient or his or her family.

- D. Reimbursement for volunteer services.
- E. Chemotherapy or radiation therapy, unless used for symptom control.
- F. Services, visits, medical equipment, or supplies not required to maintain the comfort and manage the pain of the terminally ill Member.
- G. Custodial Care, domestic, or housekeeping services.
- H. Meals on Wheels or other similar food service arrangements.
- I. Rental or purchase of renal dialysis equipment and supplies. Benefits for dialysis equipment and supplies are available in Section 10, Medical Devices and Supplies.

15.8 Outpatient Mental Health and Substance Abuse

Coverage is not provided for:

- A. Services solely on court order or as a condition of parole or probation unless approved or authorized by the CareFirst BlueChoice Medical Director.
- B. Intellectual disability, after diagnosis.
- C. Psychoanalysis.

15.9 Inpatient Mental Health and Substance Abuse

Coverage is not provided for:

- A. Admissions as a result of a court order or as a condition of parole or probation unless approved or authorized by the CareFirst BlueChoice Medical Director.
- B. Custodial Care.
- C. Admissions solely for observation or isolation.

15.10 Emergency Services and Urgent Care

Coverage is not provided for:

- A. Charges for services when the claims filing and notice procedures stated in Section 9 have not been followed by the Member.
- B. Except for covered ambulance services, travel, whether or not recommended by a Contracting Provider.

15.11 Medical Devices and Supplies

Benefits will not be provided for purchase, rental, or repair of the following:

A. Convenience Items

Equipment that basically serves comfort or convenience functions or is primarily for the convenience of a person caring for a Member (e.g., an exercycle or other physical fitness equipment, elevators, hoyer lifts, and shower/bath bench).

B. Furniture items, movable objects or accessories that serve as a place upon which to rest (people or things) or in which things are placed or stored (e.g., chair or dresser).

C. Exercise equipment

Any device or object that serves as a means for energetic physical action or exertion in order to train, strengthen, or condition all or part of the human body, (e.g., exercycle or other physical fitness equipment).

- D. Institutional equipment
 Any device or appliance that is appropriate for use in a medical facility and not appropriate for use in the home (e.g., parallel bars).
- E. Environmental control equipment Equipment that can be used for non-medical purposes, such as air conditioners, humidifiers, or electric air cleaners. These items are not covered even though they may be prescribed, in the individual's case, for a medical reason.
- F. Eyeglasses or contact lenses, dental prostheses, appliances, or hearing aids (except as otherwise provided herein for cleft lip or cleft palate or both or as stated in Section 2 and Section 3).
- G. Corrective shoes (unless required to be attached to a leg brace), shoe lifts, or special shoe accessories or inserts.
- H. Medical equipment/supplies of an expendable nature, except those specifically listed as covered Medical Devices and Supplies in this In-Network Description of Covered Services. Non-covered supplies include incontinence pads or ace bandages.
- I. Tinnitus maskers.

840 First Street, NE Washington, DC 20065 (202) 479-8000

An independent licensee of the BlueCross and Blue Shield Association

ATTACHMENT C IN-NETWORK SCHEDULE OF BENEFITS

The benefits and limitations described in this schedule are subject to all terms and conditions stated in the Evidence of Coverage.

Benefits for Covered Services, Covered Dental Services, and Covered Vision Services may be provided either under the In-Network Evidence of Coverage or Out-of-Network Evidence of Coverage. Benefits will not be provided for the same service or supply under both this In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. However, for certain services there are visit or other limitations. Where there is a benefit limitation, the benefit limitation is combined for both the In-Network Evidence of Coverage and Out-of-Network Evidence of Coverage.

CareFirst BlueChoice pays only for Covered Services, Covered Vision Services and Covered Dental Services. The Member pays for services, supplies or care, which are not covered. The Member pays any applicable Deductible, Copayment or Coinsurance. Services that are not listed in the Description of Covered Services, or are listed in the Exclusions and Limitations, are not Covered Services, Covered Vision Services or Covered Dental Services.

When determining the benefits a Member may receive, CareFirst BlueChoice considers all provisions and limitations in the Evidence of Coverage as well as its medical policies. When these conditions of coverage are not met or followed, payments for benefits may be denied. Certain utilization management requirements will also apply. When these requirements are not met, payments may be reduced or denied.

IN-NETWORK DEDUCTIBLE

The In-Network Individual Benefit Period Deductible is \$3,000.

The In-Network Family Benefit Period Deductible is \$6,000.

Individual Coverage: The Member must satisfy the In-Network Individual Deductible.

Family Coverage: Each Member can satisfy his/her own Deductible by meeting the Individual Deductible. In addition, eligible expenses for all covered Members can be combined to satisfy the Family Deductible. An individual family Member may not contribute more than the Individual Deductible toward meeting the Family Deductible. Once the Family Deductible has been met, this will satisfy the Deductible for all covered family Members.

The benefit chart below states whether a Covered Service is subject to the In-Network Deductible.

The following amounts may not be used to satisfy the Benefit Period Deductible:

- Difference between the price of a Non-Preferred Brand Name Drug and Generic Drug when a Member selects a Non-Preferred Brand Name Drug when a Generic Drug is available.
- Discounts, coupons, or other amounts from third parties, including manufacturer coupons and discount prescription card programs.
- Charges in excess of the Allowed Benefit.
- Charges for services which are not covered under the Evidence of Coverage or which exceed the maximum number of covered visits/days listed below.
- Charges for Covered Services not subject to the Deductible.
- Charges for Prescription Drugs.
- Charges for Pediatric Vision Services or Pediatric Dental Services.
- Charges incurred under the Out-of-Network Evidence of Coverage.

IN-NETWORK OUT-OF-POCKET MAXIMUM

The In-Network Individual Benefit Period Out-of-Pocket Maximum is \$7,000.

The In-Network Family Benefit Period Out-of-Pocket Maximum is \$14,000.

Individual Coverage: The Member must meet the Individual Out-of-Pocket Maximum.

Family Coverage: Each Member can satisfy his/her own Individual Out-of-Pocket Maximum by meeting the Individual Out-of-Pocket Maximum. In addition, eligible expenses of all covered family members can be combined to satisfy the Family Out-of-Pocket Maximum. An individual family member cannot contribute more than the Individual Out-of-Pocket Maximum toward meeting the Family Out-of-Pocket Maximum. Once the Family Out-of-Pocket Maximum has been met, this will satisfy the Out-of-Pocket Maximum for all family members.

These amounts apply to the Benefit Period In-Network Out-of-Pocket Maximum:

- In-Network Copayments and Coinsurance for all Covered Services.
- · In-Network Benefit Period Deductible.
- Prescription Drug Deductible
- In-Network Pediatric Dental Deductible and In-Network Coinsurance for Covered Dental Services.

When the Member has reached the Out-of-Pocket Maximum, no further Copayments, Coinsurance or Deductible will be required in that Benefit Period for Covered Services, Covered Dental Services and Covered Vision Services.

The following amounts may <u>not</u> be used to satisfy the Benefit Period Out-of-Pocket Maximum:

- Difference between the price of a Non-Preferred Brand Name Drug and Generic Drug when a Member selects a Non-Preferred Brand Name Drug when a Generic Drug is available.
- Discounts, coupons, or other amounts from third parties, including manufacturer coupons and discount prescription card programs.
- Charges in excess of the Allowed Benefit, Prescription Drug Allowed Benefit, Vision Allowed Benefit and Pediatric Dental Allowed Benefit.
- Charges for services which are not covered under the Evidence of Coverage or which exceed the maximum number of covered visits/days listed below.
- Charges incurred under the Out-of-Network Evidence of Coverage.

BENEFITS					
		SUBJECT TO			
SERVICE	LIMITATIONS	DEDUCTIBLE?	MEMBER PAYS		

The Member is responsible for any applicable Deductible, Copayment or Coinsurance listed in this schedule. When the Allowed Benefit for any Covered Service is less than the Copayment listed, the Member payment will be the Allowed Benefit.

When multiple services are rendered on the same day by more than one provider, Member payments are required for each provider.

The benefit chart below states whether a Covered Service is subject to the Member Copayment for Clinic Visits/ Outpatient Services rendered in a hospital, hospital clinic, or health care provider's office on a hospital campus ("Clinic Visit").

These providers <u>may</u> bill individually resulting in claims from both the hospital/facility and the physician or health care provider rendering care in the hospital/facility/clinic setting. It is the Member's responsibility to determine whether separate claims will be assessed.

OUTPATIENT FACILITY, OFFICE AND PROFESSIONAL SERVICES				
Physician's	Services rendered by Specialists in the	PCP: No	PCP: \$15 per visit	
Office	disciplines listed below will be treated as PCP visits for Member payment purposes.	Specialist: No	Specialist: \$30 per visit	
	 General internal medicine; Family practice medicine; General pediatric medicine; or Geriatric medicine. 	Clinic Visit: Yes	and \$50 per visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic	
Outpatient Non-		PCP: No	PCP: \$15 per visit	
Surgical Services		Specialist: No	Specialist: \$30 per visit	
		Clinic Visit: Yes	and \$50 per visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic	
Laboratory Tests.	X-Ray/Radiology Services, Specialty Imag	ing and Diagnostic	Procedures	
Non-Preventive Laboratory Tests (independent non-hospital laboratory)		No	\$15 per visit	
Non-Preventive Laboratory Tests (outpatient department of a hospital)		Yes	\$30 per visit	
Non-Preventive X-Ray/Radiology Services (independent non-hospital facility)		No	\$30 per visit	
Non-Preventive X-Ray/Radiology		Yes	\$60 per visit	

	BEI	NEFITS	
SERVICE	LIMITATIONS	SUBJECT TO DEDUCTIBLE?	MEMBER PAYS
Services			
outpatient			
lepartment of a			
ospital)			
Non-Preventive		No	\$100 per visit
Specialty			
maging			
independent			
on-hospital			
acility)			
Non-Preventive		Yes	\$200 per visit
Specialty			
maging			
outpatient			
lepartment of a			
ospital)			
Non-Preventive		No	\$30 per visit
Diagnostic			
Testing except as			
otherwise			
pecified (in an			
ndependent non-			
ospital facility)			
Non-Preventive		Yes	\$60 per visit
Diagnostic			
Testing except as			
otherwise			
pecified (in an			
outpatient			
lepartment of a			
ospital)			
Sleep Studies		No	\$20 per study
Member's			
ome)			
1 0, 1	D: 4 : 4: 1	X.Y	φ100 · 1
Sleep Studies	Prior authorization is required.	No	\$100 per study
office or			
reestanding			
acility)	D: d : d : d	***	Φ200
Sleep Studies	Prior authorization is required.	Yes	\$200 per study
outpatient			
lepartment of a			
ospital)	1	i	İ

recommendations of the United States Preventive Services Task Force (USPSTF). With respect to women, such additional preventive care and screenings as provided for in comprehensive guidelines supported by the Health Resources and Services Administration. At a minimum, benefits will be provided for breast cancer screening in accordance with the latest screening guidelines issued by the American Cancer Society or required by the Patient Protection and Affordable Care Act (PPACA).

Prostate Cancer	No	No Copayment or Coinsurance
Screening		2 - 2
Colorectal	No	No Copayment or Coinsurance
Cancer Screening		
Breast Cancer	No	No Copayment or Coinsurance
Screening		

BENEFITS				
CEDVICE	I INDICATIONS	SUBJECT TO	MEMBED DAYS	
SERVICE Don Smaan	LIMITATIONS	DEDUCTIBLE?	MEMBER PAYS	
Pap Smear Human		No No	No Copayment or Coinsurance No Copayment or Coinsurance	
Papillomavirus		INO	No Copayment of Comsurance	
Screening Test				
Preventive		NT.	N. C	
		No	No Copayment or Coinsurance	
Laboratory Tests Preventive X-		N.T.	N. C.	
		No	No Copayment or Coinsurance	
Ray/Radiology Services				
Preventive		Ma	No Consument on Coincinn	
		No	No Copayment or Coinsurance	
Specialty				
Imaging		N.Y.	N. C.	
Preventive		No	No Copayment or Coinsurance	
Diagnostic				
Testing (except				
as otherwise				
specified)				
Immunizations		No	No Copayment or Coinsurance	
Well Child Care		No	No Copayment or Coinsurance	
Adult Preventive		No	No Copayment or Coinsurance	
Care		NO	No Copayment of Comsurance	
Women's		No	No Copayment or Coinsurance	
Preventive		INO	No Copayment of Comsurance	
Services				
Office Visits for		No	No Copayment or Coinsurance	
Treatment of		INO	No Copayment of Comsulance	
Obesity				
Professional		No	No Copayment or Coinsurance	
Nutritional		110	No Copayment of Comsurance	
Counseling and				
Medical				
Nutrition				
Therapy				
Treatment Service	l es			
Family Planning				
Non-Preventive		Professional: No	\$30 per visit and \$50 per visit if	
Gynecological			rendered in the outpatient	
Office Visits		Clinic Visit: Yes	department of a hospital/hospital	
			clinic or provider's office located	
			in a hospital or hospital clinic	
Contraceptive		No	No Copayment or Coinsurance	
Counseling				
Contraceptive	Coverage of self-administered	No	No Copayment or Coinsurance	
Drugs and	contraceptive drugs and devices is	1.0		
Devices	provided under the Prescription Drugs			
	benefit.			
Insertion or	Drug or device must be approved by the	No	No Copayment or Coinsurance	
removal, and any	FDA as a contraceptive.	1.0		
Medically				
Necessary				
examination				
CAMITHATION		l		

	BENEFITS					
		SUBJECT TO				
SERVICE	LIMITATIONS	DEDUCTIBLE?	MEMBER PAYS			
associated with the use of any contraceptive						
devices or drugs Elective Sterilization	Benefits available to female Members with reproductive capacity only.	No	No Copayment or Coinsurance			
Services – Female Members						
Maternity and Re	lated Services					
Preventive Visit		No	No Copayment or Coinsurance			
Non-Preventive Visit		Professional: No	\$30 per visit and \$50 per visit if rendered in the outpatient			
		Clinic Visit: Yes	department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic			
Professional Services for Delivery		Yes	\$30 per visit			
Allergy Services	<u> </u>					
Allergy Testing and Allergy		Professional: No	\$30 per visit and \$50 per visit if rendered in the outpatient			
Treatment		Clinic Visit: Yes	department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic			
Allergy Shots		Professional: No	\$30 per visit and \$50 per visit if rendered in the outpatient			
		Clinic Visit: Yes	department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic			
Rehabilitation Ser	vices					
Rehabilitative Physical Therapy		Professional: No	\$30 per visit and \$50 per visit if rendered in the outpatient			
3 13		Clinic Visit: Yes	department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic			
Rehabilitative Occupational		Professional: No	\$30 per visit and \$50 per visit if rendered in the outpatient			
Therapy		Clinic Visit: Yes	department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic			
Rehabilitative Speech Therapy		Professional: No	\$30 per visit and \$50 per visit if rendered in the outpatient			
		Clinic Visit: Yes	department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic			
Spinal Manipulation		Professional: No	\$30 per visit and \$50 per visit if rendered in the outpatient			
Services		Clinic Visit: Yes	department of a hospital/hospital clinic or provider's office located			

Habilitative Services for Children Cardiac Rehabilitation of Coverage and the Out-of-Network Evidence of Coverage and the Observork Evidence of Coverage Evidence of Coverage and the Observork Evidence of Coverage		BENEFITS					
Habilitative Services for Children Clinic Visit: Yes Soper visit and \$50 per visit if rendered in the outpatient department of a hospital clinic or provider's office located in a hospital or hospital clinic or provider of hospital clinic or	CEDIMOE	Y YAMED A ECHONIC	SUBJECT TO	MEMBER DAVG			
Habilitative Services for Children Habilitative Services for Children Habilitative Services for Adults Benefits available for Member age 21 and older. Cardiac Rehabilitation of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation program per lifetime combined under the In-Network Evidence of Coverage. Professional: No Clinic Visit: Yes Clinic	SERVICE	LIMITATIONS	DEDUCTIBLE?				
Services for Children Benefits available for Member age 21 and older. Professional: No Services for Adults Prior authorization is required for Habilitative services for Adults Prior authorization is required for Habilitative services for Adults Prior authorization is required for Habilitative services for Adults Professional: No Clinic Visit: Yes Sid per visit and \$50 per visit if rendered in the outpatient department of a hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hos	Habilitativa	Limited to Members under the age of 21	Professional: No	110 a nospital or nospital clinic			
Clinic Visit: Yes Department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic		Elimited to Members under the age of 21.	Fiolessional, No				
Babilitative Services for Adults Benefits available for Member age 21 and older. Professional: No older. Soper visit and \$50 per visit if rendered in the outpatient of the optimization is required for Habilitative services for Adults. Professional: No Clinic Visit: Yes Clinic Visit: Yes Clinic Visit: Yes Clinic Visit: Yes Soper visit and \$50 per visit if rendered in the outpatient of a hospital clinic or provider's office located in a hospital or hospital clinic or provid			Clinic Visit: Yes				
Habilitative Services for Adults Services for Adults Professional: No older. Clinic Visit: Yes So per visit and \$50 per visit if rendered in the outpatient department of aboptital clinic or provider's office located in a hospital or hospital or hospital or hospital or hospital or hospital or hospi	Cimurum		- CIIII				
Services for Adults				in a hospital or hospital clinic			
Adults Prior authorization is required for Habilitative services for Adults. Cardiac Rehabilitation of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services Outpatient Therapeutic Treatment Services Outpatient Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Products Clinical Trials Benefits are available to the same extent as benefits provided for other services Retail Health Clinic Clinical Trials Retail Health Clinic Office Professional: No Clinic Visit: Yes Clinic V			Professional: No				
Prior authorization is required for Habilitative services for Adults. Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services Outpatient Treatment Services Outpatient Professional: No Clinic Visit: Yes Clinic		older.	C1' ' X7' ' X7				
Prior authorization is required for Habilitative services for Adults. Cardiac Rehabilitation and Infusion Services Blood Products Clinic Treatment Services Blood Products Clinic Trials Benefits are available to the same extent as benefits provided for other services Retail Health Clinic Trials Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Free-Standing Infusion Center Habilitation of Coverage and the Out-of-Network Evidence of Coverage. Professional: No Clinic Visit: Yes Clinic Visit: Yes Sa0 per visit and \$50 per visit if rendered in the outpatient of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or	Adults		Clinic Visit: Yes				
Cardiac Rehabilitation		Prior authorization is required for					
Rehabilitation of Coverage and the Out-of-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services Other Treatment Services Clinic Visit: Yes Clinic Visi		Habilitative services for Adults.		in a nospital of nospital crime			
Rehabilitation of Coverage and the Out-of-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services Other Treatment Services Clinic Visit: Yes Clinic Visi							
Clinic Visit: Yes Clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital/hospital clinic or provider's office located in a hospital or hospital/hospital clinic or provider's office located in a hospital or hospital/hospital clinic or provider's office located i		Limited to 90 days per Benefit Period	Professional: No				
Evidence of Coverage. Clinic or provider's office located in a hospital or hospital clinic or provider so flice located in a hospital or hospital clinic or provider so flice located in a hospital or hospital clinic or provider so flice located in a hospital or hospital clinic or provider so flice located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located clinic or provider's office located in a hospital or hospital clinic or Professional: No Clinic Visit: Yes of located in a hospital or hospital clinic or provider's office located in a h	Rehabilitation		O1: 1 37: 1, 37				
Pulmonary Rehabilitation Pulmonary Rehabilitation Pulmonary Rehabilitation Program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Professional: No Clinic Visit: Yes Clin			Clinic Visit: Yes	department of a hospital/hospital			
Pulmonary Rehabilitation Rehabilitat		Evidence of Coverage.					
Rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Clinic Visit: Yes Clinic Visit: Yes Clinic Visit: Yes Outpatient Services Outpatient Treatment Services Outpatient Treatment Services Clinic Visit: Yes Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Products Clinical Trials Retail Health Clinic Sene Services Retail Health Clinic Services Benefits are available to the same extent as benefits provided for other services Retail Health Clinic Services Benefits are available to the same extent as benefits provided for other services Professional: No Services Benefits are available to the same extent as benefits provided for other infusion services Retail Health Clinic Services Professional: No Services No Services No Services Professional: No Services Benefits are available to the same extent as benefits provided for other services No Services Professional: No Services No Services No Services No Services Professional: No Services No Services No Services No Services Professional: No Services No Services No Services No Services Professional: No Services No Services No Services No Services No Services Professional: No Services Professional: No Services				in a nospital of nospital crime			
In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.		Limited to one pulmonary rehabilitation	Professional: No				
Out-of-Network Evidence of Coverage. Other Treatment Services Outpatient Therapeutic Treatment Services Clinic Visit: Yes Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Products Clinic Trials Clinic Trials Retail Health Clinic Telemedicine Services Retail Health Services Professional: No Services Clinic Visit: Yes Clinic Visit: Yes Clinic Visit: Yes Clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or h	Rehabilitation	program per lifetime combined under the		rendered in the outpatient			
Other Treatment Services Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation Services) Blood and Blood Products Clinical Trials Retail Health Clinic Telemedicine Services Retail Health Clinic Telemedicine Services Benefits are available to the same extent as benefits provided for other services Retail Health Clinic Telemedicine Services Breties are available to the same extent as benefits provided for other services Retail Health Clinic Telemedicine Services Benefits are available to the same extent as benefits provided for other services Retail Health Clinic Telemedicine Services Benefits are available to the same extent as benefits provided for other services Services Infusion Therapy Physician's Office Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines.			Clinic Visit: Yes	department of a hospital/hospital			
Outpatient Treatment Services Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Products Clinical Trials Clinical Trials Retail Health Clinic Clinic Clinic Clinic Visit: Yes Benefits are available to the same extent as benefits provided for other infusion services Retail Health Clinic Clinic Clinical Trials Clinical Tri		Out-of-Network Evidence of Coverage.					
Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Products Clinical Trials Benefits are available to the same extent as benefits provided for other infusion services Retail Health Clinic Telemedicine Services Benefits are available to the same extent as benefits provided for other services Retail Health Clinic Telemedicine Services Benefits are available to the same extent as benefits provided for other services Services Infusion Therapy Physician's Office Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines.	Other Treatment	Services		in a nospital of nospital chinic			
Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Products Clinical Trials Benefits are available to the same extent as benefits provided for other services Retail Health Clinic Telemedicine Services Benefits are available to the same extent as benefits provided for other services Benefits are available to the same extent as benefits provided for other services Retail Health Clinic Telemedicine Services Infusion Therapy Physician's Office Specialty Drugs in the Prescription Guidelines. Free-Standing Infusion Center Hospital Outpatient Outpatient Department Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Hospital Outpatient Specialty Drugs in the Prescription Guidelines. Prior authorization is required for No \$20 per session		SCI VICES	Professional: No	\$30 per visit and \$50 per visit if			
Services (excluding Cardiae Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Products Clinical Trials Clinic Retail Health Clinic Telemedicine Services Benefits are available to the same extent as benefits provided for other infusion services Retail Health Clinic Telemedicine Services Benefits are available to the same extent as benefits provided for other services Retail Health Clinic Telemedicine Services Infusion Therapy Physician's Office Specialty Drugs in the Prescription Guidelines. Free-Standing Infusion Center Guidelines. Hospital No \$20 per session S20 per session No \$20 per session No S20 per session Specialty Drugs in the Prescription Guidelines. No S20 per session Specialty Drugs in the Prescription Guidelines.							
(excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Products Clinical Trials Retail Health Clinic Telemedicine Services Benefits are available to the same extent as benefits provided for other services Retail Health Clinic Telemedicine Services Infusion Therapy Physician's Office Free-Standing Infusion Center Hospital Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Hospital Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Hospital Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Hospital Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Hospital Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Hospital Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Hospital Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Hospital Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Hospital Prior authorization is required for Specialty Drugs in the Prescription Guidelines.	Treatment		Clinic Visit: Yes	department of a hospital/hospital			
Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Products Clinical Trials Retail Health Clinic Benefits are available to the same extent as benefits provided for other infusion services Benefits are available to the same extent as benefits provided for other services Retail Health Clinic Retail Health Clinic Benefits are available to the same extent as benefits provided for other services Retail Health Clinic Retail Health Clinic Prelemedicine Services Infusion Therapy Physician's Office Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Center Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. No \$20 per session \$20 per session Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines.							
Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Products Clinical Trials Retail Health Clinic Telemedicine Services Infusion Therapy Physician's Office Free-Standing Infusion Center Hospital Outpatient Department Department Pior authorization is required for Outpatient Specialty Drugs in the Prescription Guidelines. Penefits are available to the same extent as benefits provided for other services Benefits are available to the same extent as benefits provided for other services No \$15 per visit No \$20 per session				in a hospital or hospital clinic			
pulmonary rehabilitation and Infusion Services) Blood and Blood Products Clinical Trials Retail Health Clinic Telemedicine Services Infusion Therapy Physician's Office Prior authorization is required for Specialty Drugs in the Prescription Center Guidelines. Free-Standing Infusion Center Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines.							
rehabilitation and Infusion Services) Blood and Blood Products Clinical Trials Retail Health Clinic Telemedicine Services Infusion Therapy Physician's Office Free-Standing Infusion Guidelines. Free-Standing Infusion Center Hospital Outpatient Department Department Benefits are available to the same extent as benefits provided for other services Benefits are available to the same extent as benefits provided for other services No \$15 per visit No \$15 per visit No \$20 per session Specialty Drugs in the Prescription Guidelines. No \$20 per session Specialty Drugs in the Prescription Outpatient Specialty Drugs in the Prescription Guidelines.							
Services Blood and Blood Products Benefits are available to the same extent as benefits provided for other infusion services							
Blood and Blood Products Clinical Trials Clinical Trials Retail Health Clinic Telemedicine Services Infusion Therapy Physician's Office Free-Standing Infusion Center Hospital Hospital Outpatient Outpatient Department Benefits are available to the same extent as benefits provided for other services Benefits are available to the same extent as benefits provided for other services No \$15 per visit No \$15 per visit No \$20 per session Specialty Drugs in the Prescription Guidelines. No \$20 per session Specialty Drugs in the Prescription Guidelines. No \$20 per session Specialty Drugs in the Prescription Guidelines. No \$20 per session Specialty Drugs in the Prescription Guidelines. No \$20 per session							
Blood Products Clinical Trials Clinical Trials Retail Health Clinic Telemedicine Services Infusion Therapy Physician's Office Free-Standing Infusion Center Hospital Outpatient Department Benefits are available to the same extent as benefits provided for other services Provided for other services No \$15 per visit No \$15 per visit No \$20 per session No \$20 per session \$20 per session No \$20 per session No \$20 per session Specialty Drugs in the Prescription Guidelines. No \$20 per session No \$20 per session Specialty Drugs in the Prescription Guidelines. No \$20 per session Specialty Drugs in the Prescription Guidelines.							
Clinical Trials Retail Health Clinic Telemedicine Services Infusion Therapy Physician's Office Free-Standing Infusion Center Center Cuidelines. Prior authorization is required for Specialty Drugs in the Prescription Center Guidelines. Benefits are available to the same extent as benefits provided for other services No \$15 per visit No \$15 per visit No \$20 per session Specialty Drugs in the Prescription Center Guidelines.							
Retail Health Clinic Telemedicine Services Infusion Therapy Physician's Office Specialty Drugs in the Prescription Guidelines. Free-Standing Infusion Center Hospital Outpatient Department Retail Health No \$15 per visit No \$15 per visit No services No services No \$20 per session	Clinical Trials						
Retail Health Clinic Telemedicine Services Infusion Therapy Physician's Office Free-Standing Infusion Center Hospital Outpatient Department Penefits are available to the same extent as benefits provided for other services No \$15 per visit No \$15 per visit No \$15 per visit No \$20 per services No \$20 per session Specialty Drugs in the Prescription Guidelines. No \$20 per session No \$20 per session No \$20 per session No \$20 per session Specialty Drugs in the Prescription Guidelines. No \$20 per session							
Telemedicine Services Infusion Therapy Physician's Office Free-Standing Infusion Center Hospital Outpatient Department Penefits are available to the same extent as benefits provided for other services Services No \$20 per session Outpatient							
Services Infusion Therapy Physician's Office Prior authorization is required for Office Specialty Drugs in the Prescription Guidelines. No \$20 per session		D 6: 11.11					
Infusion Therapy Physician's Office Specialty Drugs in the Prescription Guidelines. Free-Standing Infusion Specialty Drugs in the Prescription Guidelines. Hospital Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Hospital Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Hospital Prior authorization is required for Specialty Drugs in the Prescription Guidelines. Hospital Specialty Drugs in the Prescription Guidelines.		Benefits are available to the same extent as b	penetits provided for	other services			
Physician's Office Specialty Drugs in the Prescription Guidelines. Free-Standing Infusion Specialty Drugs in the Prescription Guidelines. Hospital Outpatient Specialty Drugs in the Prescription Guidelines. Prior authorization is required for Specialty Drugs in the Prescription Guidelines. No \$20 per session No \$20 per session No \$20 per session No \$20 per session Specialty Drugs in the Prescription No \$200 per session Specialty Drugs in the Prescription Guidelines.							
Office Specialty Drugs in the Prescription Guidelines. Free-Standing Infusion Specialty Drugs in the Prescription Center Guidelines. Hospital Prior authorization is required for Outpatient Specialty Drugs in the Prescription Guidelines. Outpatient Specialty Drugs in the Prescription Guidelines.		Prior authorization is required for	No	\$20 per session			
Guidelines. Free-Standing Infusion Specialty Drugs in the Prescription Center Guidelines. Hospital Prior authorization is required for Outpatient Specialty Drugs in the Prescription Guidelines. Outpatient Specialty Drugs in the Prescription Guidelines. Guidelines. No \$20 per session \$200 p				F			
Infusion Specialty Drugs in the Prescription Center Guidelines. Hospital Prior authorization is required for Outpatient Specialty Drugs in the Prescription Department Guidelines. Specialty Drugs in the Prescription Guidelines.							
CenterGuidelines.No\$200 per sessionHospitalPrior authorization is required for Outpatient Specialty Drugs in the Prescription DepartmentNo\$200 per session			No	\$20 per session			
Hospital Prior authorization is required for Outpatient Specialty Drugs in the Prescription Guidelines.							
Outpatient Specialty Drugs in the Prescription Guidelines.			No	\$200 per session			
Department Guidelines.			INU	ψ200 per session			
	Department						
	Member's	Prior authorization is required for	No	\$20 per session			

BENEFITS				
SERVICE	LIMITATIONS	SUBJECT TO DEDUCTIBLE?	MEMBER PAYS	
Home	Specialty Drugs in the Prescription	DEDUCTIBLE:	MENIDER FATS	
	Guidelines.			
	cal Facility and Professional Services		T + 4 0 0	
Surgical Care at		No	\$100 per visit	
an Ambulatory Care Facility				
Outpatient	Routine/Screening Colonoscopy is not	No	\$30 per visit	
Surgical	subject to the Copayment.	110	que o per visit	
Professional				
Services				
Provided at an Ambulatory Care				
Facility				
i demity				
Surgical Care at		Yes	\$200 per visit	
an Outpatient				
Hospital Facility	B :: /G : G1	17	\$20 · · ·	
Outpatient Surgical	Routine/Screening Colonoscopy is <u>not</u> subject to the Copayment or Deductible.	Yes	\$30 per visit	
Professional	subject to the Copayment of Deductible.			
Services				
Provided at an				
Outpatient				
Hospital	SDITAL CEDVICES			
Inpatient Facility	SPITAL SERVICES Hospitalization solely for rehabilitation	Yes	\$200 per admission	
(medical or	limited to ninety (90) days per Benefit	103	\$200 per admission	
surgical	Period combined under the In-Network			
condition,	Evidence of Coverage and the Out-of-			
including	Network Evidence of Coverage.			
maternity and rehabilitation)	Prior authorization is required except for			
ichaointation)	emergency admissions and all maternity			
	admissions.			
Inpatient		Yes	\$30 per visit	
Professional				
Services				
Organ and Tissue	Except for corneal transplants and kidney	Benefits are availa	ble to the same extent as benefits	
Transplants	transplants, prior authorization is required.	provided for other		
•		•		
	ING FACILITY SERVICES	N/	φ20 · · · · · · · · · · · · ·	
Skilled Nursing Facility Services	Limited to 60 days per Benefit Period combined under the In-Network Evidence	Yes	\$30 per admission	
racinty Scrvices	of Coverage and the Out-of-Network			
	Evidence of Coverage.			
	Prior authorization is required.			
HOME HEALTH		Ma	No Consument on Cainage	
Home Health Services	Prior authorization is required.	No	No Copayment or Coinsurance	
Del vices	Limited to ninety (90) visits per "episode			
	of care" combined under the In-Network			
	Evidence of Coverage and the Out-of-			
	Network Evidence of Coverage. A new			

BENEFITS				
CEDVICE	I IN AUTO A TOLONIC	SUBJECT TO	MEMBER BANG	
SERVICE	LIMITATIONS	DEDUCTIBLE?	MEMBER PAYS	
	episode of care begins if the Member does not receive Home Health Care for the same			
	or a different condition for sixty (60)			
	consecutive days.			
Postpartum	Benefits are available to all Members.	No	No Copayment or Coinsurance	
Home Visits		1,0	The depayment of demisurance	
HOSPICE SERV	TCES			
Inpatient Care	Prior authorization is required.	No	No Copayment or Coinsurance	
	Services limited to a maximum one hundred eighty (180) day hospice eligibility period			
	Limited to sixty (60) days per hospice eligibility period combined under the In- Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.			
Outpatient Care	Prior authorization is required.	No	No Copayment or Coinsurance	
	Services limited to a maximum one hundred eighty (180) day hospice eligibility period.			
Respite Care	Prior authorization is required.	No	No Copayment or Coinsurance	
	Services limited to a maximum one hundred eighty (180) day hospice eligibility period.			
Bereavement	Prior authorization is required.	No	No Copayment or Coinsurance	
Services MENTAL HEAL	Covered only if provided within ninety (90) days following death of the deceased. TH AND SUBSTANCE USE DISORDER S	SERVICES		
Outpatient Servi	ces			
Office Visits		No	No Copayment or Coinsurance	
Outpatient Hospital Facility Services		No	No Copayment or Coinsurance	
Outpatient Professional Services		No	No Copayment or Coinsurance	
Provided at an Outpatient Hospital Facility				
Outpatient Psychological and Neuro- psychological Testing for Diagnostic Purposes		No	No Copayment or Coinsurance	
Methadone Maintenance		No	No Copayment or Coinsurance	
Partial Hospitalization		No	No Copayment or Coinsurance	

	BENEFITS				
		SUBJECT TO	1.55		
SERVICE	LIMITATIONS	DEDUCTIBLE?	MEMBER PAYS		
Professional Services at a		No	No Copayment or Coinsurance		
Partial					
Hospitalization					
Facility					
Inpatient Services	5				
Inpatient Facility	Prior authorization is required except for	Yes	\$200 per admission		
Services	emergency admissions.				
Inpatient		Yes	\$30 per visit		
Professional Services					
	ERVICES AND URGENT CARE				
Urgent Care	Limited to unexpected, urgently required	No	\$50 per visit		
Facility	services.	110	φου per visit		
Hospital	Limited to Emergency Services or	Yes	\$150 per visit, waived if admitted		
Emergency	unexpected, urgently required services.				
Room - Facility					
Services					
II anital	Limited to Emparage Compiles on	Yes	\$20 man wisit		
Hospital Emergency	Limited to Emergency Services or unexpected, urgently required services.	res	\$30 per visit		
Room –	unexpected, digently required services.				
Professional					
Services					
Follow-Up Care	Limited to Emergency Services or	Benefits are availa	ble to the same extent as benefits		
after Emergency	unexpected, urgently required services.	provided for other			
Surgery		•			
Ambulance	Prior authorization is required for air	Yes	\$30 per service		
Service	ambulance services, except for Medically				
	Necessary air ambulance services in an				
MEDICAL DEVI	emergency. CES AND SUPPLIES				
Durable Medical	CES AND SUPPLIES	Yes	25% of the Allowed Benefit		
Equipment		1 68	25% of the Allowed Bellefit		
Hair Prosthesis	Limited to one per Benefit Period	Yes	25% of the Allowed Benefit		
Train 1 Tostilesis	combined under the In-Network Evidence	105	25 % of the fine wear Benefit		
	of Coverage and the Out-of-Network				
	Evidence of Coverage.				
Breastfeeding		No	No Copayment or Coinsurance		
Equipment and					
Supplies					
Diabetes	Coverage for Diabetes Supplies will be	Yes	25% of the Allowed Benefit		
Equipment	provided under the Prescription Drug				
benefit. TOTAL CARE AND COST IMPROVEMENT, HEALTH PROMOTION, WELLNESS AND DISEASE					
MANAGEMENT					
Associated Costs	Benefits will be provided as described in	No	No Copayment or Coinsurance		
for the Patient-	the Description of Covered Services for				
Centered Medical	Patient-Centered Medical Home Program				
Home Program	and the Total Care and Cost Improvement,				
(PCMH)	Health Promotion, Wellness and Disease				
	Management Program Amendment.				

	BENEFITS				
SERVICE	LIMITATIONS	SUBJECT TO DEDUCTIBLE?	MEMBER PAYS		
TCCI Program Elements	Benefits will be provided as described in the Total Care and Cost Improvement, Health Promotion, Wellness and Disease Management Program Amendment.	No	No Copayment or Coinsurance		
Services Provided Pursuant to an Active Plan of Care under the BHCC Program, CCC Program, or SUD Program	Benefits will be provided as described in the Total Care and Cost Improvement, Health Promotion, Wellness and Disease Management Program Amendment. Members may simultaneously participate in the CCC Program and either the SUD Program or the BHCC Program, but no Member may simultaneously participate in all three Programs.	No	No Copayment or Coinsurance		
Health Promotion and Wellness	Benefits will be provided as described in the Total Care and Cost Improvement, Health Promotion, Wellness and Disease Management Program Amendment.	No	No Copayment or Coinsurance		
Disease Management	Benefits will be provided as described in the Total Care and Cost Improvement, Health Promotion, Wellness and Disease Management Program Amendment.	No	No Copayment or Coinsurance		

			MEMBER PAYS			
		SUBJECT TO		NON-		
		PRESCRIPTION	CONTRACTING	CONTRACTING		
		DRUG	PHARMACY	PHARMACY		
SERVICE	LIMITATIONS	DEDUCTIBLE	PROVIDER	PROVIDER		
DDESCRIPTION DDICS						

- Except for Emergency Services and Urgent Care outside the Service Area, when Prescription Drugs are purchased from a non-Contracting Pharmacy Provider, charges above the Prescription Drug Allowed Benefit are a non-Covered Service.
- If a Generic Drug is not available, a Brand Name Drug shall be dispensed.
- If a provider prescribes a Non-Preferred Brand Name Drug, and the Member selects the Non-Preferred Brand Name Drug when a Generic Drug is available, the Member shall pay the applicable Copayment as stated in this Schedule of Benefits plus the difference between the price of the Non-Preferred Brand Name Drug and the Generic Drug. A Member will be allowed to obtain a Non-Preferred Brand Name Drug in place of an available Generic Drug and pay only the Non-Preferred Brand Name Drug Copayment when Medically Necessary, as determined by CareFirst BlueChoice.
- A Member may request a Non-Preferred Brand Name Drug be covered for the Preferred Brand Name Drug Copayment if the provider determines that the Preferred Brand Name Drug would not be effective or would result in adverse effects.
- Except as otherwise provided, Covered Services rendered by Contracting Pharmacy Providers and Non-Contracting Pharmacy Providers are subject to the Prescription Drug Deductible.
- The Member shall pay the lesser of the cost of the prescription or the applicable Copayment.
- Prior authorization is required for human growth hormones and all Prescription Drugs contained in the Prescription Guidelines.

Prescription D	Prescription Drug Deductible - The Prescription Drug Deductible is \$250 per Member per Benefit Period.				
Prescription	Limited to a 30-day supply per	Preventive Drugs,	Preventive Drugs, Diabetic Supplies and		
Drugs	prescription or refill.	Diabetic Supplies, oral	Oral Chemotherapy Drugs:		
		chemotherapy and	No Copayment or Coinsurance		
		Generic Drugs: No			
			Generic Drugs:		
		Preferred Brand Name	\$10 per prescription or refill		
		Drug and Non-			
		Preferred Brand Name	Preferred Brand Name Drugs: \$40 per		
		Drug: Yes	prescription or refill		
			Non Duefermed Duesed Name Dunger \$70		
			Non-Preferred Brand Name Drugs: \$70		
Maintananaa	Limited to a OO day armsly son	Duamantina Dunga	per prescription or refill		
Maintenance	Limited to a 90-day supply per		Preventive Drugs, Diabetic Supplies and		
Drugs	prescription or refill.	Diabetic Supplies, oral	Oral Chemotherapy Drugs:		
	A Member may obtain up to a	chemotherapy and Generic Drugs: No	No Copayment or Coinsurance		
	twelve (12) month supply of	Generic Drugs. No	Generic Drugs:		
	contraceptives at one time.	Preferred Brand Name	\$20 per prescription or refill		
	contraceptives at one time.	Drug and Non-	\$20 per prescription of ferm		
	Maintenance Drug means a	Preferred Brand Name	Preferred Brand Name Drugs: \$80 per		
	Prescription Drug anticipated	Drug: Yes	prescription or refill		
	being required for six (6)	Diug. 10s	prescription of term		
	months or more to treat a		Non-Preferred Brand Name Drugs: \$140		
	chronic condition.		per prescription or refill		

			MEMBER PAYS	
		SUBJECT TO		NON-
		PRESCRIPTION	CONTRACTING	CONTRACTING
		DRUG	PHARMACY	PHARMACY
SERVICE	LIMITATIONS	DEDUCTIBLE	PROVIDER	PROVIDER
DDECCDIDION DDUCC				

- Except for Emergency Services and Urgent Care outside the Service Area, when Prescription Drugs are purchased from a non-Contracting Pharmacy Provider, charges above the Prescription Drug Allowed Benefit are a non-Covered Service.
- If a Generic Drug is not available, a Brand Name Drug shall be dispensed.
- If a provider prescribes a Non-Preferred Brand Name Drug, and the Member selects the Non-Preferred Brand Name Drug when a Generic Drug is available, the Member shall pay the applicable Copayment as stated in this Schedule of Benefits plus the difference between the price of the Non-Preferred Brand Name Drug and the Generic Drug. A Member will be allowed to obtain a Non-Preferred Brand Name Drug in place of an available Generic Drug and pay only the Non-Preferred Brand Name Drug Copayment when Medically Necessary, as determined by CareFirst BlueChoice.
- A Member may request a Non-Preferred Brand Name Drug be covered for the Preferred Brand Name Drug Copayment if the provider determines that the Preferred Brand Name Drug would not be effective or would result in adverse effects.
- Except as otherwise provided, Covered Services rendered by Contracting Pharmacy Providers and Non-Contracting Pharmacy Providers are subject to the Prescription Drug Deductible.
- The Member shall pay the lesser of the cost of the prescription or the applicable Copayment.
- Prior authorization is required for human growth hormones and all Prescription Drugs contained in the Prior Authorization List

	Authorization List.			
Specialty	Benefits for Specialty Drugs	Yes	Preferred Specialty Drugs:	
Drugs	are only available when		\$100 per prescription or refill for up to a	
	Specialty Drugs are purchased		30-day supply of a non-Maintenance Drug	
	from and dispensed by a			
	specialty Pharmacy in the		\$200 per prescription or refill for up to a	
	Exclusive Specialty Pharmacy		90-day supply of a Maintenance Drug	
	Network. Coverage for			
	Specialty Drugs will not be		Non-Preferred Specialty Drugs:	
	provided when a Member		\$150 per prescription or refill for up to a	
	purchases Specialty Drugs		30-day supply of a non-Maintenance Drug	
	from a Pharmacy outside of			
	the Exclusive Specialty		\$300 per prescription or refill for up to a	
	Pharmacy Network.		90-day supply of a Maintenance Drug	

Pediatric Vision – Benefit limited to Members up to age 19. If Member is under age 19 at the start of the Benefit Period but turns 19 during the Benefit Period, then the Member will receive Covered Vision Services up to the end of the Calendar Year in which the Member turns age 19.			
		SUBJECT TO	MEMBER PAYS
SERVICE	LIMITATIONS	DEDUCTIBLE?	CONTRACTING VISION PROVIDER
Eye Examination	Limited to one per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of- Network Evidence of Coverage.	No	No Copayment or Coinsurance.
Lenses - Importan	t note regarding Member Pay	ments: "Basic" me	ans spectacle lenses with no "add-ons" such
		gressive lenses, tran	sitional lenses and others which may result
in additional costs t		No	No Consument on Coincipana
Basic Single vision	Limited to one pair per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	NO	No Copayment or Coinsurance
Basic Bifocals	Limited to one pair per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	No	No Copayment or Coinsurance
Basic Trifocals	Limited to one pair per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	No	No Copayment or Coinsurance
Basic Lenticular	Limited to one pair per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	No	No Copayment or Coinsurance
Frames			
Frames	Limited to one frame per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Limited to frames contained in the Vision Care Designee's collection.	No	No Copayment or Coinsurance
Low Vision			Ly a
Low Vision Eye Examination	Prior authorization is required.	No	No Copayment or Coinsurance.
	Limited to one		

Pediatric Vision – Benefit limited to Members up to age 19. If Member is under age 19 at the start of the Benefit Period but turns 19 during the Benefit Period, then the Member will receive Covered Vision Services up to the end of the Calendar Year in which the Member turns age 19.

up to the end of th	<u>ne Calendar Year in which the</u>		
		SUBJECT TO	MEMBER PAYS
SERVICE	LIMITATIONS	DEDUCTIBLE?	CONTRACTING VISION PROVIDER
	comprehensive low vision evaluation every 5 years and 4 follow-up visits in any 5-year period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.		
Follow-up care	Prior authorization required.	No	No Copayment or Coinsurance.
	Limited to four visits in any five-year period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.		
High-power Spectacles, Magnifiers and Telescopes	Prior authorization is required.	No	No Copayment or Coinsurance.
Contact Lenses			
Elective	Includes evaluation, fitting and follow-up fees. Limited to one pair per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Limited to contact lenses contained in the Vision Care Designee's collection. Prior authorization is	No	No Copayment or Coincurance
Medically Necessary	Prior authorization is required. Limited to one pair per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	INO	No Copayment or Coinsurance

Pediatric Dental – Limited to Members up to age 19. If Member is under age 19 at the start of the Benefit Period but turns 19 during the Benefit Period, then the Member will receive Covered Dental Services up to the end of the Calendar Year in which the Member turns age 19.

Pediatric Dental Deductible

The In-Network Deductible of \$25 per Member per Benefit Period applies to all Class II, III, and IV Covered Dental Services.

Pediatric Dental Out-of-Pocket Maximum

Amounts paid by the Member for Covered Pediatric Dental Services will be applied to the In-Network Out-of-Pocket Maximum stated above. Once the In-Network Out-of-Pocket Maximum has been reached, the Member will no longer be required to pay any Deductible or Coinsurance.

		SUBJECT TO	MEMBER PAYS
SERVICE	LIMITATIONS	PEDIATRIC DENTAL DEDUCTIBLE?	PREFERRED DENTIST
Class I Preventive & Diagnostic Services		No	No Copayment or Coinsurance
Class II Basic Services		Yes	20% of the Pediatric Dental Allowed Benefit
Class III Major Services – Surgical		Yes	20% of the Pediatric Dental Allowed Benefit
Class IV Major Services – Restorative		Yes	50% of the Pediatric Dental Allowed Benefit
Class V Orthodontic Services	Limited to Medically Necessary Orthodontia	No	50% of the Pediatric Dental Allowed Benefit

CareFirst BlueChoice, Inc. [Signature]	
[Name] [Title]	

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

IN-NETWORK INTER-PLAN ARRANGEMENTS DISCLOSURE AMENDMENT

This amendment is effective on the effective date of the Evidence of Coverage to which this amendment is attached.

Out-of-Area Services

Overview

CareFirst has a variety of relationships with other Blue Cross and/or Blue Shield Licensees. Generally, these relationships are called "Inter-Plan Arrangements." These Inter-Plan Arrangements work based on rules and procedures issued by the Blue Cross Blue Shield Association ("Association"). Whenever you access healthcare services outside the geographic area CareFirst serves, the claim for those services may be processed through one of these Inter-Plan Arrangements. The Inter-Plan Arrangements are described below.

When you receive care outside of CareFirst service area, you will receive it from one of two kinds of providers. Most providers ("participating providers") contract with the local Blue Cross and/or Blue Shield Plan in that geographic area ("Host Blue"). Some providers ("nonparticipating providers") don't contract with the Host Blue. CareFirst explains below how we pay both kinds of providers.

Inter-Plan Arrangements Eligibility – Claim Types

All claim types are eligible to be processed through Inter-Plan Arrangements, as described above, except for all Dental Care Benefits except when paid as medical claims/benefits, and those Prescription Drug Benefits or Vision Care Benefits that may be administered by a third party contracted by CareFirst to provide the specific service or services.

A. BlueCard® Program

Under the BlueCard® Program, when you receive covered healthcare services within the geographic area served by a Host Blue, CareFirst will remain responsible for doing what we agreed to in the contract. However, the Host Blue is responsible for contracting with and generally handling all interactions with its participating providers.

When you receive covered healthcare services outside the CareFirst service area and the claim is processed through the BlueCard Program, the amount you pay for covered healthcare services is calculated based on the lower of:

- The billed charges for covered services; or
- The negotiated price that the Host Blue makes available to CareFirst.

Often, this "negotiated price" will be a simple discount that reflects an actual price that the Host Blue pays to your healthcare provider. Sometimes, it is an estimated price that takes into account special

arrangements with your healthcare provider or provider group that may include types of settlements, incentive payments and/or other credits or charges. Occasionally, it may be an average price, based on a discount that results in expected average savings for similar types of healthcare providers after taking into account the same types of transactions as with an estimated price.

Estimated pricing and average pricing also take into account adjustments to correct for over- or underestimation of past pricing of claims, as noted above. However, such adjustments will not affect the price CareFirst has used for your claim because they will not be applied after a claim has already been paid.

B. Inter-Plan Programs: Federal/State Taxes/Surcharges/Fees

Federal or state laws or regulations may require a surcharge, tax or other fee that applies to insured accounts. If applicable, CareFirst will include any such surcharge, tax or other fee as part of the claim charge passed on to you.

C. Nonparticipating Providers Outside CareFirst Service Area

1. Member Liability Calculation

When covered healthcare services are provided outside of CareFirst service area by nonparticipating providers, the amount you pay for such services will normally be based on either the Host Blue's nonparticipating provider local payment or the pricing arrangements required by applicable state law. In these situations, you may be responsible for the difference between the amount that the nonparticipating provider bills and the payment CareFirst will make for the covered healthcare services as set forth in this paragraph. Federal or state law, as applicable, will govern payments for out-of-network emergency services.

2. Exceptions

In certain situations, CareFirst may use other payment methods, such as billed charges for covered services, the payment we would make if the healthcare services had been obtained within our service area, or a special negotiated payment to determine the amount CareFirst will pay for services provided by nonparticipating providers. In these situations, you may be liable for the difference between the amount that the nonparticipating provider bills and the payment CareFirst will make for the covered services as set forth in this paragraph.

D. Blue Cross Blue Shield Global® Core

If you are outside the United States (hereinafter "BlueCard service area"), you may be able to take advantage of the Blue Cross Blue Shield Global® Core when accessing covered services. The Blue Cross Blue Shield Global® Core is unlike the BlueCard Program available in the BlueCard service area in certain ways. For instance, although the Blue Cross Blue Shield Global® Core assists you with accessing a network of inpatient, outpatient and professional providers, the network is not served by a Host Blue. As such, when you receive care from providers outside the BlueCard service area, you will typically have to pay the providers and submit the claims yourself to obtain reimbursement for these services.

If you need medical assistance services (including locating a doctor or hospital) outside the BlueCard service area, you should call the Blue Cross Blue Shield Global® Core Service Center at 1.800.810.BLUE

(2583) or call collect at 1.804.673.1177, 24 hours a day, seven days a week. An assistance coordinator, working with a medical professional, can arrange a physician appointment or hospitalization, if necessary.

• Inpatient Services

In most cases, if you contact the service center for assistance, hospitals will not require you to pay for covered inpatient services, except for your deductibles, coinsurance, etc. In such cases, the hospital will submit your claims to the service center to begin claims processing. However, if you paid in full at the time of service, you must submit a claim to receive reimbursement for covered healthcare services. You must contact CareFirst to obtain precertification for non-emergency inpatient services.

Outpatient Services

Physicians, urgent care centers and other outpatient providers located outside the BlueCard service area will typically require you to pay in full at the time of service. You must submit a claim to obtain reimbursement for covered healthcare services/covered services.

• Submitting a Blue Cross Blue Shield Global® Core Claim

When you pay for covered healthcare services outside the BlueCard service area, you must submit a claim to obtain reimbursement. For institutional and professional claims, you should complete a Blue Cross Blue Shield Global® Core claim form and send the claim form with the provider's itemized bill(s) to the service center (the address is on the form) to initiate claims processing. Following the instructions on the claim form will help ensure timely processing of your claim. The claim form is available from CareFirst, the service center or online at www.bcbsglobalcore.com. If you need assistance with your claim submission, you should call the service center at 1.800.810.BLUE (2583) or call collect at 1.804.673.1177, 24 hours a day, seven days a week.

This amendment is subject to all of the terms and conditions of the Evidence of Coverage to which it is attached and does not change any terms or conditions, except as specifically stated herein.

[Signature]	
[Name] [Title]	

CareFirst BlueChoice, Inc.

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

INTER-PLAN PROGRAM ANCILLARY SERVICES AMENDMENT

This amendment is effective on the effective date or renewal date of the Evidence of Coverage to which this amendment is attached.

As used in this amendment, "Out-of-Area Covered Ancillary Services" mean:

- 1. Independent Clinical Laboratory Tests (performed at non-hospital based labs)
- 2. Medical Devices and Supplies
- 3. Specialty Prescription Drugs (non-routine, biological therapeutics such as injectables, infusion therapies, high-cost therapies, and therapies that require complex care)

Under the BlueCard® Program, Members are able to obtain Covered Ancillary Services outside the geographic area that CareFirst BlueChoice services. This program allows Members to obtain Out-of-Area Covered Ancillary Services from providers that have a contractual agreement (i.e., are "participating providers" or "contracted providers") with the local Blue Cross and/or Blue Shield Licensee in another geographic area, as well as non-participating providers in some instances.

As used in this amendment, the "Local Plan" means the plan that is responsible for processing Out-of-Area Covered Ancillary Services claims under the BlueCard® Program.

Member payment for Out-of-Area Covered Ancillary Services at the participating or non-participating provider payment level is determined by the relationship between the provider and the Local Plan. If the provider of Covered Ancillary Services has a contract with the Local Plan (a participating provider), the Member is responsible for the participating provider member payment as stated in the Inter-Plan Arrangements Disclosure Amendment.

If the provider of Covered Ancillary Services does not have a contract with the Local Plan (a non-participating provider), the Member is responsible for the non-participating provider member payment as stated in the Inter-Plan Arrangements Disclosure Amendment.

For Out-of-Area Covered Ancillary Services, the Local Plan is determined as follows:

Independent Clinical Laboratory Tests - if the referring provider is located in the same service area where the specimen was drawn, the plan of the service area where the specimen was drawn is the Local Plan; if the referring provider is not located in the same service area where the specimen was drawn, the plan of the service area where the referring provider is located is the Local Plan.

Medical Devices and Supplies - the plan of the service area where the equipment was shipped to or purchased at a retail store is the Local Plan.

Specialty Prescription Drugs - the plan of the service area where the ordering physician is located is the Local Plan.

This amendment is subject to all of the terms and conditions of the Evidence of Coverage to which it is attached and does not change any terms or conditions, except as specifically stated herein.

CareFirst BlueChoice, Inc.

840 First Street, NE Washington, DC 20065 (202) 479-8000

An independent licensee of the BlueCross and Blue Shield Association

ELIGIBILITY SCHEDULE AMENDMENT

The "Termination of Coverage" section in the In-Network Eligibility Schedule is deleted and replaced with the following:

TERMINATION OF COVERAGE			
Subscriber no longer eligible as a Qualified Employee	When the Subscriber is no longer eligible as a Qualified Employee, enrollment of the Subscriber and any Dependents terminates on the last day of the month that the Subscriber and/or Dependent no longer meet the eligibility requirements stated in the Evidence of Coverage or as instructed by the SHOP Exchange. Extension of benefits and continuation of coverage rights are stated in the In-Network Evidence of Coverage.		
Dependent Child Limiting Age (other than a Dependent Child who is incapacitated as provided in the In-Network Evidence of Coverage)	When a Dependent Child reaches the Limiting Age (other than a Dependent Child who is incapacitated as provided in the Evidence of Coverage) enrollment terminates on December 31 of the Calendar year in which the Dependent Child reached the Limiting Age, unless otherwise instructed by the SHOP Exchange. Extension of benefits rights are stated in the In-Network Evidence of Coverage.		
Dependent no longer eligible (on grounds other than Limiting Age)	When a Member no longer meets the eligibility requirements stated in the Evidence of Coverage (on grounds other than reaching the Limiting Age) enrollment terminates on the last day of the month in which the Member no longer meets the eligibility requirements stated in the In-Network Evidence of Coverage or as instructed by the SHOP Exchange. Extension of benefits and continuation of coverage rights, if any, are stated in the In-Network Evidence of Coverage.		
Death of Subscriber	Upon the death of the Subscriber, enrollment of the Subscriber terminates on the date of death or as instructed by the SHOP Exchange. The enrollment of any Dependents terminates on the last day of the month after the Subscriber's death. Extension of benefits and continuation of coverage rights are stated in the In-Network Evidence of Coverage.		

This amendment is issued to be attached to the In-Network Evidence of Coverage. This amendment does not change the terms and conditions of the In-Network Evidence of Coverage, unless specifically stated herein.

[Signature]	
[Name] [Title]	

CareFirst BlueChoice, Inc.

840 First Street, NE Washington, DC 20065 (202) 479-8000

An independent licensee of the Blue Cross and Blue Shield Association

IN-NETWORK 2019 AMENDMENT

This amendment is effective on the effective date or renewal date of the In-Network Evidence of Coverage to which this amendment is attached.

TABLE OF CONTENTS

SECTION A – DEFINITIONS

SECTION B – SPECIAL ENROLLMENT PERIODS

SECTION C - RESCISSION

SECTION D - SUBROGATION

SECTION E - PRESCRIPTION DRUGS

SECTION F – GENERAL PROVISIONS

SECTION G - CLINICAL TRIAL PATIENT COST COVERAGE

SECTION H – MATERNITY SERVICES

SECTION I – EXCLUSIONS AND LIMITATIONS

SECTION A – DEFINITIONS

1. The definition of "Contracting Pharmacy Provider" in Section 1 of the Evidence of Coverage is deleted and replaced with the following:

<u>Contracting Pharmacy Provider</u> means the separate independent Pharmacist or Pharmacy, including a pharmacy in the Exclusive Specialty Pharmacy Network that has contracted with CareFirst BlueChoice or its designee to provide Prescription Drugs in accordance with the terms of this Evidence of Coverage.

2. The definition of "Preferred Drug List" in Section 1 of the Evidence of Coverage is deleted and replaced with the following:

<u>Preferred Drug List</u> means the list of Preferred Drugs issued by CareFirst BlueChoice and used by health care providers when writing, and Pharmacists, when filling, prescriptions. All Generic Drugs are included in the Preferred Drug List. Not all Brand Name Drugs listed in the Formulary are included in the Preferred Drug List. CareFirst BlueChoice may change this list periodically without notice to Members. A copy of the Preferred Drug List is available to the Member upon request.

3. The definition of "Substance Abuse" and all references to "Substance Abuse" in the Evidence of Coverage are deleted and replaced with the following:

<u>Substance Use Disorder means:</u>

- A. <u>Alcohol Use Disorder</u> means a disease that is characterized by a pattern of pathological use of alcohol with repeated attempts to control its use, and with negative consequences in at least one of the following areas of life: medical, legal, financial, or psycho-social; or
- B. <u>Drug Use Disorder</u> means a disease that is characterized by a pattern of pathological use of a drug with repeated attempts to control the use, and with negative consequences in at least one of the following areas of life: medical, legal, financial, or psycho-social.

4. The following definitions are added to Section 1 of the Evidence of Coverage:

<u>Covered Prescription Drug</u> means a Prescription Drug included in the CareFirst BlueChoice Formulary.

<u>Covered Specialty Drug</u> means a Specialty Drug included in the CareFirst BlueChoice Formulary.

<u>Formulary</u> means the means the list of Prescription Drugs issued by CareFirst BlueChoice and used by health care providers when writing, and Pharmacists, when filling, prescriptions for which coverage will be provided under this Evidence of Coverage. CareFirst BlueChoice may change this list periodically without notice to Members. A copy of the Formulary is available to the Member upon request.

Non-Preferred Brand Name Drug means a Brand Name Drug included in the Formulary as a Covered Prescription Drug but not included on the Preferred Drug List.

Non-Preferred Specialty Drug means a Specialty Drug included in the Formulary as a Covered Prescription Drug but not included on the Preferred Drug List.

Preferred Specialty Drug means a Specialty Drug included in the Preferred Drug List.

SECTION B - SPECIAL ENROLLMENT PERIODS

- 1. Section 2.5C.1.a)(2) in the Evidence of Coverage is deleted. All subsequent provisions are renumbered accordingly.
- 2. Section 2.5C.1.f) in the Evidence of Coverage is deleted and replaced with the following:
 - f) The Qualified Employee or his or her Dependent gains access to new Qualified Health Plans as a result of a permanent move and either:
 - (1) had minimum essential coverage as described in 26 CFR 1.5000A-1(b) for one or more days during the sixty (60) days preceding the date of the permanent move; or.
 - (2) was living in a foreign country or in a United States territory for one or more days during the sixty (60) days preceding the date of the permanent move.
- 3. Section 2.5C.1.g) in the Evidence of Coverage is deleted and replaced with the following:
 - g) The Qualified Employee is an Indian, as defined in section 4 of the Indian Health Care Improvement Act, who may enroll in a Qualified Health Plan or change coverage from one Qualified Health Plan to another one time per month or who is or becomes a Dependent of an Indian, as defined by section 4 of the Indian Health Care Improvement Act and is enrolled or is enrolling in a Qualified Health Plan through an Exchange on the same application as the Indian, may change from one Qualified Health Plan to another one time per month, at the same time as the Indian.
- 4. Section 2.5C.1 in the Evidence of Coverage is amended to add the following:
 - i) The Qualified Employee or his or her Dependent:

- (1) is a victim of domestic abuse or spousal abandonment, as defined by 26 CFR 1.36B-2T, as amended, including a dependent or unmarried victim within a household, is enrolled in minimum essential coverage and seeks to enroll in coverage separate from the perpetrator of the abuse or abandonment; or,
- (2) is a dependent of a victim of domestic abuse or spousal abandonment, on the same application as the victim, may enroll in coverage at the same time as the victim.
- j) The Qualified Employee or his or her Dependent:
 - (1) applies for coverage on the Exchange during the annual open enrollment period or due to a qualifying event, is assessed by the SHOP Exchange as potentially eligible for Medicaid or the Children's Health Insurance Program (CHIP), and is determined ineligible for Medicaid or CHIP by the State Medicaid or CHIP agency either after open enrollment has ended or more than sixty (60) days after the qualifying event; or
 - applies for coverage at the State Medicaid or CHIP agency during the annual open enrollment period, and is determined ineligible for Medicaid or CHIP after open enrollment has ended;
- k) The Qualified Employee, or his or her Dependent, adequately demonstrates to the SHOP Exchange that a material error related to plan benefits, Service Area, or Premium influenced the eligible individual's decision to purchase a QHP through the SHOP Exchange.
- 5. Section 2.5C in the Evidence of Coverage is amended to add the following:
 - 5. A Dependent of a Qualified Employee is not eligible for a Special Enrollment Period if the Group does not extend coverage the offer of coverage to Dependents.
- 6. Section 2.5D.7 in the Evidence of Coverage is deleted and replaced with the following:
 - 7. The Effective Date for a Qualified Employee or Dependent who timely enrolls due to a qualifying event stated in Section 2.5C.1.d) (enrollment or non-enrollment was unintentional, inadvertent, or erroneous and is the result of an error by the SHOP Exchange or the United States Department of Health and Human Services), Section 2.5C.1.e) (a Qualified Health Plan substantially violated a material provision of its contract), Section 2.5C.1.h) (other exceptional circumstances as determined by the SHOP Exchange), Section 2.5C.1.j) (applies for a QHP during annual open enrollment, is assessed by the SHOP-Exchange as potentially eligible for Medicaid or CHIP and then determined ineligible or applied for Medicaid or CHIP during annual open enrollment and determined to be ineligible) and 2.5C.1.k) (demonstration to the Exchange that a material error related to plan benefits, service area, or premium influenced the individual's decision to purchase a QHP) shall be the appropriate date based on the circumstances of the Special Enrollment Period as determined by the Exchange.

SECTION C - RESCISSION

Section 3.3 in the Evidence of Coverage is amended to add the following:

D. CareFirst BlueChoice demonstrates, to the reasonable satisfaction of the SHOP Exchange, if required by the SHOP Exchange, that the rescission is appropriate

SECTION D – SUBROGATION

Section 5.4, Subrogation, in the Evidence of Coverage is deleted and replaced with the following:

5.4 <u>Subrogation</u>

- A. CareFirst BlueChoice has subrogation and reimbursement rights. Subrogation requires the Member to turn over to CareFirst BlueChoice any rights the Member may have against a third party. A third party is any person, corporation, insurer or other entity that may be liable to a Member for an injury or illness. Subrogation applies to any illness or injury which is:
 - 1. Caused by an act or omission of a third party; or
 - 2. Covered under a member's uninsured or underinsured policy issued to or otherwise covering the Member; or
 - 3. Covered by No Fault Insurance. <u>No Fault Insurance</u> means motor vehicle casualty insurance. This term also refers to motor vehicle insurance issued under any other state or federal legislation of similar purpose.
- B. If the Member receives or is entitled to receive payment from any person, organization or entity in connection with an injury, illness or need for care for which benefits were provided or will be provided under this Evidence of Coverage, the payment will be treated as having been paid to the Member as a recovery for the medical, hospital and other expenses for which CareFirst BlueChoice provided or will provide benefits. CareFirst BlueChoice may recover the amounts paid or will pay in benefits up to the amount received from or on behalf of the third party or applicable first party coverage.
- C. CareFirst BlueChoice's right of recovery is not subject to reduction for attorney's fees and costs under the "common fund" or any other doctrine. If required by law, CareFirst BlueChoice will reduce the amount owed by the Member to CareFirst BlueChoice in accordance with applicable law.
- D. CareFirst BlueChoice will have a lien on all funds the Member recovers up to the total amount of benefits provided. We are entitled under our right of recovery to be reimbursed for our benefit payments even if you are not "made whole" for all of your damages in the recoveries that you receive. CareFirst BlueChoice may give notice of that lien to any party who may have contributed to the Member's loss, or who may be liable for payment as a result of that loss.
- E. CareFirst BlueChoice has the option to be subrogated to the Member's rights to the extent of the benefits provided under this Evidence of Coverage. This includes CareFirst BlueChoice right to bring suit or file claims against the third party in the Member's name.
- F. Members agree to take action, furnish information and assistance, and execute such instruments that CareFirst BlueChoice may require while enforcing CareFirst BlueChoice rights under this Section. The Member agrees to not take any action which prejudices CareFirst BlueChoice's rights and interests under this provision.

SECTION E - PRESCRIPTION DRUGS

Section 11, Prescription Drugs, in the Description of Covered Services, is deleted and replaced with the following:

SECTION 11 PRESCRIPTION DRUGS

11.1 Covered Services

Benefits will be provided for Prescription Drugs, including but not limited to:

- A. Any self-administered contraceptive drug or device, including a contraceptive drug and device on the Preventive Drug List, that is approved by the FDA for use as a contraceptive and is obtained under a prescription written by an authorized prescriber. See Section 1.5.B, Contraceptive Methods and Counseling, for additional coverage of contraceptive drugs and devices.
- B. Human growth hormones. Prior authorization is required.
- C. Any drug that is approved by the FDA as an aid for the cessation of the use of tobacco products and is obtained under a prescription written by an authorized prescriber, including drugs listed in the Preventive Drug List.

<u>Nicotine Replacement Therapy</u>. Nicotine Replacement Therapy means a product, including a product on the Preventive Drug List that is used to deliver nicotine to an individual attempting to cease the use of tobacco products, approved by the FDA as an aid for the cessation of the use of tobacco products and obtained under a prescription written by an authorized prescriber. Coverage for Nicotine Replacement Therapy will be provided on an unlimited yearly basis.

- D. Injectable medications that are self-administered and the prescribed syringes and needles.
- E. Standard covered items such as insulin, glucagon and anaphylaxis kits.
- F. Fluoride products.
- G. Diabetic Supplies.
- H. Oral chemotherapy drugs.
- I. Hormone replacement therapy drugs.

11.2 <u>Dispensing</u>.

- A. Non-Maintenance Drugs are limited to up to a thirty (30)-day supply.
- B. Maintenance Drugs are limited to up to a ninety (90)-day supply.
- C. A Member may obtain up to a twelve (12) month supply of contraceptives at one time.
- 11.3 <u>Mail Order Program</u>. Except as provided in Section 12.3, all Members have the option of ordering Covered Prescription Drugs via mail order. Members ordering Covered Prescription Drugs through the mail order program will be entitled to a thirty (30) day supply for non-Maintenance Drugs and a ninety (90) day supply for Maintenance Drugs. A Member may obtain up to a twelve (12) month supply of contraceptives at one time.
- 11.4 <u>Benefits for Specialty Pharmacy Prescription Drugs</u>. Benefits will be provided for Covered Specialty Drugs only when obtained from a Pharmacy that is part of the Exclusive Specialty Pharmacy Network.

SECTION F – GENERAL PROVISIONS

Section 14.1.D General Provisions, Prescription Drug Coverage, in the Description of Covered Services, is deleted and replaced with the following:

D. Prescription Drug Coverage.

- 1. Accessing the Prescription Drug Benefit Card Program.
 - a) Members may use his/her identification card to purchase Covered Prescription Drugs from Contracting Pharmacy Providers. If the Prescription Drug coverage includes a Deductible, the Member must pay the entire cost of the Covered Prescription Drug(s) until the Deductible is satisfied. Once the Deductible, if applicable, has been satisfied, the Member pays the appropriate Copayment or Coinsurance as stated in the Schedule of Benefits.
 - b) For Covered Prescription Drugs or diabetic supplies purchased from a non-Contracting Pharmacy Provider, the Member is responsible for paying the total charge and submitting a claim to CareFirst BlueChoice or its designee for reimbursement. In cases of Emergency Services or Urgent Care received outside of the Service Area, Members will be entitled to reimbursement from CareFirst BlueChoice or its designee up to the amount of the total charge, minus any applicable Deductible, Copayment or Coinsurance. In all other cases, Members will be entitled to reimbursement from CareFirst BlueChoice or its designee for the amount up to the Prescription Drug Allowed Benefit, minus any applicable Deductible, or Copayment or Coinsurance.
 - c) Except for Specialty Drugs, Members have the option of ordering Covered Prescription Drugs via mail order. The mail order program provides Members with a Pharmacy that has an agreement with CareFirst BlueChoice or its designee, to provide mail service for Covered Prescription Drugs in accordance with the terms of this provision. The Member is responsible for any applicable Deductible, Copayment, or Coinsurance.

2. Additional Terms and Conditions

- a) Members or health care providers must obtain prior authorization by providing information to support Medical Necessity before prescribing any Covered Prescription Drug in the Prescription Guidelines. A copy of the Prescription Guidelines is available to the Member or provider upon request.
- b) Providers may substitute a Generic Drug for a Brand Name Drug. If there is no Generic Drug for the Brand Name Drug the Member shall pay the applicable Copayment or Coinsurance as stated in the Schedule of Benefits for Preferred Brand Name Drugs or Non-Preferred Brand Name Drugs.
- c) If a provider prescribes a Non-Preferred Brand Name Drug, and the Member selects the Non-Preferred Brand Name Drug when a Generic Drug is available, the Member shall pay the applicable Copayment or Coinsurance as stated in the Schedule of Benefits plus the difference between the price of the Non-Preferred Brand Name Drug and the Generic Drug. A Member will be allowed to obtain a Non-Preferred Brand Name Drug in place of an available Generic Drug and pay only the Non-Preferred Brand Name Drug Copayment or Coinsurance when Medically Necessary, as determined by CareFirst BlueChoice.
- d) A Member may request a Non-Preferred Brand Name Drug be covered for the Preferred Brand Name Drug Copayment or Coinsurance if the provider

- determines that the Preferred Brand Name Drug would not be effective or would result in adverse effects.
- e) When a Generic version of a Prescription Drug becomes available, the Brand Name Drug may be removed from the Formulary or moved to the Non-Preferred level.
- f) Members must use 80% of a dispensed non-Maintenance Drug or Maintenance Drug in the manner prescribed before a refill of that prescription can be obtained.
- g) The Member is responsible for obtaining prior authorization for Covered Prescription Drugs in the Prescription Guidelines when obtained from a non-Contracting Pharmacy Provider by calling the customer service telephone number listed on the identification card.
- 3. How to Obtain Prescription Drugs Not Included in the CareFirst BlueChoice Formulary.

The Member may request an exception for coverage of a Prescription Drug not contained on the CareFirst BlueChoice Formulary.

- a) The Member, the Member's authorized representative or the Member's provider may request an exception based upon Medical Necessity by contacting the CareFirst BlueChoice at the telephone number located on the back of the Member's identification card.
- b) An exception form should be submitted by the prescribing provider and returned to CareFirst BlueChoice. The prescribing provider may submit a letter of Medical Necessity for dispensing of the non-Covered Prescription Drug.
- c) Upon review by the CareFirst BlueChoice, the prescribing provider and the Member or Member's representative will be notified.
 - 1) If the request is approved then the Prescription Drug will be dispensed and the Member will be responsible for the Non-Preferred Brand Name Drug Copayment. If the Prescription Drug exception request is for a non-Formulary Specialty Drug and the exception is granted, then the Member will be responsible for the Non-Preferred Specialty Drug Copayment.
 - 2) If the exception request is denied, the denial shall be considered an Adverse Decision and may be appealed in accordance with the process outlined in the Benefit Determination and Appeals Amendment.

In addition, if the exception request is denied, the Member, the Member's representative or the prescribing provider may submit an external exception request to CareFirst BlueChoice requiring that the original exception request and subsequent denial be reviewed by an independent review organization.

- d) Timeframe for review and notification of outcome of exception request:
 - 1) Urgent requests based on exigent circumstances from the Member's prescribing provider will be completed within twenty-four (24) hours.

For purposes of this provision, exigent circumstances exist when a Member is suffering from a health condition that may seriously jeopardize the Member's life, health, or ability to regain maximum

- function or when a Member is undergoing a current course of treatment using a non-Formulary Prescription Drug.
- 2) Non-urgent requests will be completed within seventy-two (72) hours.
- 3) A request for an external review of the original exception request will be completed no later than twenty-four (24) hours after receipt of the request if the original exception request was urgent and seventy-two (72) hours following receipt of the request if the original exception request was non-urgent.
- 4) CareFirst BlueChoice shall provide coverage for the non-Formulary drug for the duration of the prescription (including refills) if coverage is granted under a standard exception request, or for the duration of the exigency if coverage is granted under an expedited exception request.

SECTION G – CLINICAL TRIAL PATIENT COST COVERAGE

The definition of "Qualified Individual" in Section 1.14, Clinical Trial Patient Cost Coverage, in the Description of Covered Services, is deleted and replaced with the following:

Qualified Individual, as used in this section, means a Member who is eligible to participate in an approved clinical trial according to the trial protocol, with respect to the prevention, early detection, treatment or monitoring of cancer, chronic disease, or other life-threatening illness, and the provider who recommended the Member for the clinical trial has concluded that the Member's participation in such trial is appropriate to prevent, detect early, treat or monitor cancer, chronic disease, or life-threatening illness, or the Member's participation is based on medical and scientific information.

The definition of "Routine Patient Costs" in Section 1.14, Clinical Trial Patient Cost Coverage, in the Description of Covered Services, is deleted and replaced with the following:

Routine Patient Costs means the costs of all Medically Necessary items and health care services consistent with the Covered Services that are typically provided for a Qualified Individual who is not enrolled in a clinical trial that are incurred as a result of the item, device, or service being provided to the Qualified Individual for purposes of the clinical trial. Routine Patient Costs do not include the investigational item, device, or service itself; items or services that are provided solely to satisfy data collection and analysis needs and that are not used in the direct clinical management of the patient; or a service that is clearly inconsistent with widely accepted and established standards of care for a particular diagnosis.

Section B, Covered Services, is deleted and replaced with the following:

B. Covered Services

- 1. Benefits for Routine Patient Costs to a Qualified Individual in a clinical trial will be provided if the Qualified Individual's participation in the clinical trial is the result of prevention, early detection, treatment or monitoring of cancer, chronic disease, or other life-threatening illness
- 2. Coverage for Routine Patient Costs will be provided only if:
 - a) The item device or service is being provided or the studies are being conducted in a Phase I, Phase II, Phase III, or Phase IV clinical trial for cancer; or,
 - b) The item device or service is being provided in a Phase I, Phase II, Phase III, or Phase IV clinical trial for any other life-threatening disease or condition, or chronic disease;

- c) The item device or service is being provided in a federally funded or approved clinical trial approved by one of the National Institutes of Health, the Centers for Disease Control and Prevention, the Agency for Health Care Research and quality, the Centers for Medicare and Medicaid Services, an NIH Cooperative Group, an NIH Center, the FDA in the form of an investigational new drug or device application, the federal Department of Defense, the federal Department of Veterans Affairs, the federal Department of Energy or a qualified nongovernmental research entity to which the National Cancer Institute has awarded a support grant,, or an institutional review board of an institution in a state that has a Multiple Project Assurance Contract approved by the Office of Protection from Research Risks of the NIH;
- d) The item device or service is being provided under a drug trial that is exempt from the requirement of an investigational new drug application.
- e) The facility and personnel providing the item device or service are capable of doing so by virtue of their experience, training, and volume of patients treated to maintain expertise;
- 3. Coverage is provided for the Routine Patient Costs incurred for drugs and devices that have been approved for sale by the FDA whether or not the FDA has approved the drug or device for use in treating the Qualified Individual's particular condition, to the extent that the drugs or devices are not paid for by the manufacturer, distributor, or provider of that drug or device.

SECTION H—MATERNITY SERVICES

Description of Covered Services, Section 1.5 C.2.c), Non-Preventive Services, is deleted and replaced with the following:

c) Non-preventive routine professional services rendered to the newborn during a covered hospitalization for delivery. A newborn Dependent child will be automatically covered for the first thirty-one (31) days following the child's birth. The Agreement describes the steps, if any, necessary to enroll a newborn Dependent child.

SECTION I – EXCLUSIONS AND LIMITATIONS

Section 15.1 LL, Exclusions and Limitations, is deleted and replaced with the following exclusion:

LL. Services required solely for administrative purposes, for example, employment, insurance, foreign travel, school, camp admissions or participation in sports activities.

Section 15.1, Exclusions and Limitations, is amended to add the following exclusion:

PP. Except as otherwise provided, Prescription Drugs not contained in the CareFirst BlueChoice Formulary.

This amendment is issued to be attached to the In-Network Evidence of Coverage. This amendment does not change the terms and conditions of the Evidence of Coverage, unless specifically stated herein.

CareFirst BlueChoice, Inc.

[Signature]

[Name] [Title]

840 First Street, NE Washington, DC 20065 (202) 479-8000

An independent licensee of the Blue Cross and Blue Shield Association

TOTAL CARE AND COST IMPROVEMENT, HEALTH PROMOTION, WELLNESS AND DISEASE MANAGEMENT PROGRAM AMENDMENT

This amendment is effective on the effective date or renewal date of the In-Network Evidence of Coverage to which this amendment is attached.

Section 13, Complex Chronic or High Risk Acute Disease Management, is deleted and replaced with the following:

SECTION 13 TOTAL CARE AND COST IMPROVEMENT, HEALTH PROMOTION, WELLNESS AND DISEASE MANAGEMENT PROGRAM

13.1 Definitions.

Behavioral Health Care Coordination Program (BHCC Program) means the assessment and coordination of behavioral health care services to a Member.

<u>Biometric Screening</u> means an onsite event during which Member screenings are provided for various measurements, such as: height, weight, BMI, waist circumference, blood pressure, LDL, HDL, total cholesterol, the total cholesterol to HDL ratio, triglycerides, and glucose.

<u>Complex Care Coordination Program (CCC Program)</u> means the assessment and coordination of services provided to a Member with multiple chronic and severe health conditions or advanced or critical illnesses.

<u>Designated Provider</u> means a provider contracted with CareFirst BlueChoice to provide services under CareFirst BlueChoice's Total Care and Cost Improvement Program, which includes the following components: PCMH Program, BHCC Program, CCC Program, Comprehensive Medication Review, Enhanced Monitoring Program, Expert Consultation Program, Home-Based Services Program, Pharmacy Coordination Program, Substance Use Disorder Program, or other community-based programs outlined in this Section (collectively, the "TCCI Programs") and who has agreed to participate in care coordination activities in cooperation with CareFirst BlueChoice for Members with complex chronic disease, high risk acute conditions or lifestyle behavior change.

<u>Disease Management Coaching Session</u> means an interactive coaching session provided to the Member with the Member's consent that furthers the Member's Disease Management Program.

<u>Disease Management Program</u> means a coordinated, confidential program designed to manage a Member's chronic disease so that action can be taken to improve outcomes in the future.

<u>Health Promotion and Wellness Program</u> means a coordinated program, including Lifestyle Management Coaching Sessions, designed to prevent disease, identify a Member's risk factors for disease or detect early stages of a Member's disease so that action can be taken to prevent poor outcomes in the future.

<u>Lifestyle Management Coaching Session</u> means an interactive wellness coaching session provided to the Member with the Member's consent that furthers the Member's Health Promotion and Lifestyle Management Program.

Qualified Member means a Member who:

- A. Is participating in and complies with all elements of one or more of the TCCI Programs described in this Section including use of a Designated Provider.
- B. Continues to meet the criteria for participation in the TCCI Program(s) and participates fully with any applicable plan of care or other TCCI Program requirements, including compliance with direction from a Designated Provider when in an active plan of care; and
- C. CareFirst BlueChoice and the Member's Designated Provider determine is cooperating with, and satisfying the requirements of the TCCI Program(s).

CareFirst BlueChoice retains final authority to determine whether a Member is a Qualified Member.

<u>Substance Use Disorder</u> means a disease that is characterized by a pattern of pathological use of alcohol or a drug with repeated attempts to control the use, and with negative consequences in at least one of the following areas of life: medical, legal, financial, or psycho-social.

<u>Substance Use Disorder Program (SUD Program)</u> means the CareFirst BlueChoice program for Members with a diagnosed Substance Use Disorder. The program includes a maximum of twelve (12) consecutive months of ambulatory/outpatient detoxification, individual therapy, group therapy, and medication assisted therapy.

<u>Weight Loss Services</u> means CareFirst BlueChoice approved services available to clinically overweight or obese (BMI > 25) Members for the purpose of achieving measurable weight loss and sustainable weight maintenance, as part of the Health Promotion, Wellness, and Disease Management Program.

13.2 Benefits and Cost Sharing Waiver.

- A. To participate in any TCCI Program:
 - 1. CareFirst BlueChoice will consult with a provider to determine whether the Member meets the parameters for and would benefit from participation in one or more of the TCCI Programs; and
 - 2. The Member must consent to participate in and comply with all elements of the TCCI Program(s) in which he/she qualifies including use of a Designated Provider.
- B. Members in an active plan of care participating in a BHCC Program, CCC Program, or SUD Program are eligible for the following benefits:
 - 1. Assessment of Member/family needs related to understanding health care status and provider treatment plans, self-care, compliance capability, and continuum of care;
 - 2. Education of Member/family regarding illness, provider treatment plans, self-care techniques, treatment compliance, and continuum of care;
 - 3. Assistance in navigating and coordinating health care services and understanding benefits:

- 4. Assistance in arranging for a Designated Provider to deliver and coordinate the Member's care;
- 5. Assistance in arranging consultation(s) with Specialists or other applicable and medically necessary in-network providers;
- 6. Identification of and connection to community resources, and other organizations/support services to supplement the Member's plan of care;
- 7. Implementation of a plan of care under the direction of the Member's provider;
- 8. Coordination of care, either telephonically or otherwise, between a Designated Provider and a Member and his/her provider; and
- 9. Other Medically Necessary services provided to a Member while in an active plan of care.
- C. Members may simultaneously participate in the CCC Program and either the SUD Program or the BHCC Program, but no Member may simultaneously participate in all three Programs (the BHCC, CCC and SUD Programs).
- D. Members are eligible for medically necessary benefits under the following TCCI Program elements in addition to any benefits in Section 13.2.B, so long as the Member meets the requirements of the applicable TCCI Program element:
 - 1. Comprehensive Medication Review (CMR). Benefits will be provided for a pharmacist's review of medications and consultation with the Qualified Member to improve the effectiveness of pharmaceutical therapy.
 - 2. Enhanced Monitoring Program (EMP). Benefits will be provided for the medical equipment and monitoring services provided to a Qualified Member with a chronic condition or disease in conjunction with the EMP for maintenance of the Qualified Member's chronic condition or disease.
 - 3. Expert Consultation Program (ECP). Benefits will be provided for a review by a team of Specialists of a Qualified Member's medical records where the Qualified Member has a complex or rare condition or multiple conditions or diseases for which the course of treatment requires unique expertise.
 - 4. Home-Based Services Program (HBS). Benefits will be provided for medical and associated services specifically outlined in a home-based care management plan. The Member does not need to be confined to the home to participate in HBS.
- E. All TCCI Program benefits are subject to applicable contract limits.
- F. Qualified Member Cost Sharing Responsibilities.
 - 1. Except as provided in paragraphs 3, 4, and 5 of this Section F, for Qualified Members in an active plan of care in the BHCC Program, CCC Program, or SUD Program, CareFirst BlueChoice shall waive all cost-sharing responsibilities for all medically necessary, in-network covered services.
 - 2. Except as provided in paragraphs 3, 4 and 5 of this Section F, for Qualified Members not in an active plan of care but in a TCCI Program element, CareFirst BlueChoice shall waive all cost-sharing responsibilities for services in such TCCI Program when provided by a Designated Provider.

- 3. CareFirst BlueChoice will not waive cost-sharing responsibilities for any (i) prescription or other drug benefits; (ii) institutional claims or facility charges (iii) services provided on an inpatient basis; (iv) services provided in an emergency department or ambulatory surgical center; (v) ambulance benefits; or (vi) durable medical equipment benefits. Subsection (ii) of this paragraph does not apply to services provided under the HBS Program or to Qualified Members in an intensive outpatient program as part of the SUD Program. Subsection (vi) of this paragraph does not apply to services provided under the EMP Program.
- 4. CareFirst BlueChoice will not waive cost-sharing responsibilities for more than
 (i) twelve (12) consecutive months for a Member in the SUD Program and (ii) six
 (6) consecutive months for a Member in any other TCCI Program including the
 BHCC Program and CCC Program. The duration of any waiver of cost-sharing
 responsibility will run concurrent with the benefit plan and terminate in
 accordance with Section G.
- 5. If the Qualified Member's In-Network Evidence of Coverage is compatible with a federally-qualified Health Savings Account:
 - a) If the Qualified Member has funded his/her HSA account during the Benefit Period, then the Qualified Member will be responsible for any associated costs for services under this Section until the annual Deductible has been met, unless the Covered Services appear on the list of preventive services maintained by the Internal Revenue Service.
 - b) If the Qualified Member has not funded his/her HSA account during the Benefit Period, then if the Qualified Member agrees not to fund his/her HSA account and provides a signed agreement not to fund his/her HSA account, then the Qualified Member will be eligible for the waiver described in Section 13.2.F.1.

G. Termination.

- 1. The Qualified Member's participation in the CCC Program, BHCC Program, or SUD Program and/or any TCCI Program element and receipt of benefits and cost-sharing waivers under this Section will be terminated under the following circumstances:
 - a) The Qualified Member completes the stated goals of the TCCI Program(s) set forth in the Qualified Member's plan of care or other TCCI Program requirements as confirmed by the Qualified Member's Designated Provider.
 - b) The Designated Provider determines that the Qualified Member failed to comply with the TCCI Program(s) and/or any related plan of care or treatment under this Section. The Qualified Member will be given thirty (30) days prior written notice of termination under this subsection.
 - c) The Qualified Member's coverage under this In-Network Evidence of Coverage is terminated.

- 2. If termination is the result of the Qualified Member's failure to comply with the TCCI Program(s) under Section 13.G.1.(b), the Qualified Member will be provided the opportunity to comply with the TCCI Program(s) during the thirty (30) day notice period. If after consultation between the Qualified Member's provider and the Designated Provider a determination is made that the Qualified Member is not and will not be compliant with the applicable TCCI Program(s), the Qualified Member will receive a final written notice of termination of benefits under this Section.
- 3. Upon termination of the Qualified Member's participation in the TCCI Program(s), all cost-sharing waivers and benefits shall be null and void on and after the effective date of the termination of the waiver and the Qualified Member will be responsible for any and all cost-sharing responsibilities as stated in the Schedule of Benefits on and after the date of termination of the waiver.

13.3 Health Promotion and Wellness.

- A. Health Assessments are available for all adult Members.
- B. Benefits are available for Biometric Screening of Members, as defined above.
- C. Lifestyle Management Coaching Session services are available as follows:
 - 1. With the Member's consent, an initial discussion with a lifestyle coach to establish defined goal(s) for wellness coaching, and to determine the frequency of future coaching sessions in order to best meet the goal(s) established.
 - 2. After the initial discussion, Coaching Sessions to track, support, and advance the Member's wellness/lifestyle goal(s).
- D. Other Wellness Program benefits are available, and shall include tobacco-cessation, well-being challenges, and financial well-being improvement programs.
- E. Weight Loss Services are available to clinically overweight or obese Members, as follows:
 - 1. A clinically overweight Member is a Member whose Body Mass Index (BMI) score is equal to or greater than twenty-five (25). A clinically obese Member is a Member whose Body Mass Index (BMI) score is equal to or greater than thirty (30).
 - 2. A dedicated, CareFirst BlueChoice approved coach is assigned to the Member to assist the Member in the development of healthy eating habits, physical activity habits, and to address the emotional, social, and environmental aspects shown to be important for sustained weight loss.
 - 3. The Members receive one-on-one telephonic interventions with the coach and online educational resources, robust food, exercise trackers, recipes, peer-to-peer communication, and group community features for complete social support and accountability.

13.4 <u>Disease Management</u>.

A. Disease Management services, which may include a Disease Management Program to help the Member understand his/her disease and health status and physician treatment plans, individual and family education regarding the disease, treatment compliance and self-care techniques, and help to organize care for the disease, including arranging for needed services and supplies.

- B. Disease Management Coaching Session services are available as follows:
 - 1. With the Member's consent, an initial discussion with a coach to establish defined goal(s) for disease management coaching, and to determine the frequency of future coaching sessions in order to best meet the established goal(s) and manage the disease.
 - 2. After the initial discussion, Disease Management Coaching Sessions to track, support, and advance the Member's disease management goal(s).

This amendment is issued to be attached to the In-Network Evidence of Coverage. This amendment does not change the terms and conditions of the In-Network Evidence of Coverage unless specifically stated herein.

CareFirst BlueChoice, Inc.

	[Signature]
-	[Name] [Title]

CareFirst BlueChoice, Inc.

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

PATIENT PROTECTION DISCLOSURE NOTICE

Primary Care Provider Designation

CareFirst BlueChoice generally requires the designation of a primary care provider. You have the right to designate any primary care provider who participates in our network and who is available to accept you or your family members. Until you make this designation, CareFirst BlueChoice designates one for you. For information on how to select a primary care provider, and for a list of the participating primary care providers, contact the CareFirst BlueChoice at the customer service telephone number listed on your identification card.

For children, you may designate a CareFirst BlueChoice pediatrician as the primary care provider.

Obstetrics and Gynecological Care

You do not need prior authorization from CareFirst BlueChoice or from any other person (including a primary care provider) in order to obtain access to obstetrical or gynecological care from a health care professional in our network who specializes in obstetrics or gynecology. The health care professional, however, may be required to comply with certain procedures, including obtaining prior authorization for certain services, following a pre-approved treatment plan, or procedures for making referrals. For a list of CareFirst BlueChoice health care professionals who specialize in obstetrics or gynecology, contact CareFirst BlueChoice at customer service telephone number listed on your identification card.

CareFirst BlueChoice, Inc.

CareFirst BlueChoice, Inc.

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

IN-NETWORK GROUP INCENTIVE PROGRAM RIDER

This rider is issued by CareFirst BlueChoice to be attached to and become part of the In-Network Evidence of Coverage. A Member's effective date of coverage under this rider and termination date of coverage under this rider are the same as the Member's effective date and termination date under the In-Network Evidence of Coverage.

This rider adds an incentive program to the In-Network Evidence of Coverage that rewards Members for: 1) selecting specific health care providers and completing a Health Screening 2) permitting the receipt of wellness-related communications and completing a Health Assessment; 3) updating/confirming the responses to the Health Assessment after a defined time period; and/or 4) participating in health coaching sessions for Coach-Directed Members.

Members receive incentives in the form of a credit to a medical expense debit card.

TABLE OF CONTENTS

SECTION A – DEFINITIONS

SECTION B – INCENTIVE ACTIVITIES AND REQUIREMENTS

SECTION C – INCENTIVE AMOUNTS

SECTION D – CONDITIONS AND LIMITATIONS

A. **DEFINITIONS**:

In addition to the definitions contained in the In-Network Evidence of Coverage to which this rider is attached, the underlined terms below, when capitalized, have the following meanings:

<u>Annual Incentive</u> means the amount of the reward provided to Eligible Members for satisfaction of the incentive requirements set forth in Section B of this rider.

<u>Coach-Directed Eligible Members</u> means a Member who either (1) has an Illness Burden Score greater than or equal to 1.00; (2) has a Body Mass Index (BMI) greater than or equal to 30 and/or (3) tests positive for or self-identifies as being a tobacco user.

<u>Health Assessment</u> means a questionnaire that asks about the Member's age, habits, recent test results and medical history.

<u>Health Screening</u> means a diagnostic screening that captures biometrics to identify potential health risks.

<u>Illness Burden Score</u> means the calculation, specific to the Member, using a methodology that considers demographic information (age and gender) and data on all Member medical claims (including inpatient, outpatient, and pharmacy claims).

<u>PCMH Primary Care Provider (PCMH PCP)</u> means a Primary Care Provider that participates in the CareFirst BlueChoice Patient-Centered Medical Home Program.

B. INCENTIVE ACTIVITIES AND REQUIREMENTS:

Members who successfully complete the following requirements ("Eligible Members") will be provided an Annual Incentive as set forth in Section C, and in the form of a medical expense

debit card, which can be used to pay any Copayments, Coinsurance, or Deductibles. Dependent Children are not eligible for any incentive amounts under this rider.

1. Select a PCMH PCP and Complete a Health Screening.

- a) A Member must select a PCMH PCP within 120 days of enrollment or renewal.
- b) A Member must complete a Health Screening and submit the results of the Health Screening to CareFirst BlueChoice within 120 days of enrollment or renewal.

2. Consent to Receive Wellness-Related Communications and Complete a Health Assessment.

- a) A Member must provide consent to receive communications related to healthy lifestyles, well-being, wellness, and disease management information and activities within 120 days of enrollment or renewal.
- b) A Member must complete the Health Assessment within 120 days of enrollment or renewal.

3. Update/Confirm the Health Assessment Responses.

- a) To achieve this activity, a Member must complete the activities required in item B.2.
- b) A Member must update their responses, which will be prepopulated with information provided during the Health Assessment activity completed in activity B.2.b.no earlier than 180 days after taking the original Health Assessment.

4. Participate in Health Coaching Sessions.

Coach-Directed Eligible Members must consent to participate in and complete coaching sessions. Coaching sessions provide voluntary and confidential one-on-one support with a primary coach to collaborate and help Members reach their health goals. Sessions are personalized to each Member and focused on developing health-related goals, helping members understand their health risks and conditions, and preventing the progression of disease. Members can work with their coach on a variety of lifestyle improvements including: physical activity, weight management, nutrition and tobacco cessation support.

Members can earn rewards for completing one (1), two (2), or three (3) coaching sessions. Completing coaching sessions are based on Member participation and not dependent on achieving an outcome and/or health-related goals. Only one (1) coaching session per 30 days will count towards an incentive. A maximum of three (3) coaching sessions per Benefit Period will count towards the incentive.

C. INCENTIVE AMOUNTS

1. Select a PCMH PCP and Complete a Health Screening. Members who complete the participation requirements in Section B.1 will receive the Annual Incentive in the form of a medical expense debit card equal to \$100 per Benefit Period.

Eligible Members will be issued the Annual Incentive on an individual basis as the incentive is earned.

2. Consent to Receive Wellness-Related Communications and Complete a Health Assessment. Members who complete the participation requirements in Section B.2 will

receive the Annual Incentive in the form of a medical expense debit card equal to \$50 per Benefit Period.

Eligible Members will be issued the Annual Incentive on an individual basis as the incentive is earned.

3. Update/Confirm the Health Assessment Responses. Members who complete the participation requirements in Section B.3 will receive the Annual Incentive in the form of a medical expense debit card equal to \$25 per Benefit Period.

Eligible Members will be issued the Annual Incentive on an individual basis as the incentive is earned.

4. Participate in Health Coaching Sessions. Coach-Directed Eligible members who complete the participation requirements in Section B.4 will receive the annual incentive in the form of a medical expense debit card equal to \$30 for coaching session one (1), \$70 for coaching session two (2), \$100 for coaching session three (3).

A maximum incentive of \$200 per Benefit Period, per Coach-Directed Eligible Member, can be earned under this provision.

Eligible Members will be issued the Annual Incentive on an individual basis as the incentive is earned.

5. Maximum Annual Incentive. The total Maximum Annual Incentive may not exceed \$375 per Eligible Member for completion of all participation incentives.

D. CONDITIONS AND LIMITATIONS

- 1. Members are eligible to qualify for each incentive once per Benefit Period. Coach-Directed Eligible Members are eligible for incentive-based coaching sessions once identified by CareFirst BlueChoice.
- 2. Providers may join or leave the PCMH program at any time. To earn the Annual Incentive, a Member must select a Primary Care Provider who is a PCMH PCP at the time the selection is made.
- 3. Only one medical expense debit card will be issued per family. The medical expense debit card may be used by any Member in the family.
- 4. Members may satisfy the Health Screening requirement in Section B.1 through the Member's PCMH PCP, other In-Network provider or any process approved by CareFirst BlueChoice.
- 5. Once the Annual Incentive is awarded in a Benefit Period, it will not be withdrawn nor any amounts recouped during the Benefit Period.
- 6. Members agree to comply with any requirements concerning the use of the medical expense debit card.
- 7. If the Member's In-Network Evidence of Coverage is compatible with a federally-qualified Health Savings Account, the medical expense debit card:
 - a) cannot be used to pay for qualified medical expenses or other cost-sharing responsibilities unless (i) the Member first satisfies his/her minimum deductible as established by the Internal Revenue Service or (ii) the Member provides a signed agreement stating that he/she has not funded and agrees not to fund an HSA account during the Benefit Period; and

- b) can be used to pay for eligible dental and vision expenses that are part of the Member's benefit plan.
- 8. If the coverage allows for out-of-area benefits that extend beyond Emergency Services, Urgent Care, and follow-up care after emergency surgery, Members residing outside of CareFirst BlueChoice's service area will earn the participation incentive by selecting a participating provider in a PCP-like specialty (family practice, general practice, internist, geriatrics, pediatrics) in the Blue Cross and Blue Shield Plan where the Member resides and completing the activities identified in Section B.1.
- 9. Only a Subscriber and Subscriber's Dependent Spouse, or, if applicable, the Subscriber's Dependent Domestic Partner or Civil Union Partner, are eligible for incentives under this rider. Dependent Children are not eligible for any incentives under this rider.

This rider is issued to be attached to the In-Network Evidence of Coverage.

[Signature]	
[Name] [Title]	

CareFirst BlueChoice, Inc.

CareFirst BlueChoice, Inc. 840 First Street, NE

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

ATTACHMENT D IN-NETWORK ELIGIBILITY SCHEDULE

Effective Date	Effective as of the Group Contract Effective Date
	ELIGIBILITY
Subscriber	A Qualified Employee, who works at least 30 hours per week on a regular (not seasonal or temporary) basis, is eligible to enroll.
Spouse	Coverage for a Spouse is available.
Dependent Children	Coverage for Dependent Children is available.
Limiting Age for Dependent Children (other than an incapacitated Dependent Child)	Age 26
	ROLLMENT PERIODS AND EFFECTIVE DATES
Annual Open Enrollment	and Newly Eligible Qualified Employees
Annual Open Enrollment Newly Eligible Qualified Employee	Coverage is effective on the Group Contract Effective Date. A newly eligible Qualified Employee is eligible to enroll him or herself and any eligible Dependents on the date of the Qualified Employee's employment in the Group.
	The enrollment period for a new Qualified Employee and any Dependents is thirty (30) days from the date of the Qualified Employee's employment in the Group.
	The Effective Date of coverage is:
	A. For enrollment received by the SHOP Exchange between the first and the fifteenth day of the month, the first day of the following month; and
	B. For enrollment received by the SHOP Exchange between the sixteenth and the last day of the month, the first day of the second following month.

Special Enrollment Periods		
Newly Eligible Dependent	The Special Enrollment Period is the thirty-one (31) day period from the	
Child: (Newborn	date of the qualifying event.	
Dependent Child, Newly		
Adopted Dependent Child,	The Effective Date of coverage is the Dependent Child's First Eligibility	
a Minor Dependent Child	Date:	
for whom guardianship		
has been granted by court	A. Newly born Dependent Child: the date of birth.	
or testamentary		
appointment, or MCSO/QMSO)	B. Adopted Dependent Child: the date of Adoption, which is the earlier of the date a judicial decree of adoption is signed; or the assumption of custody, pending adoption, of a prospective adoptive child by a prospective adoptive parent.	
	C. Testamentary or court appointed guardianship of a Dependent Child: the date of appointment.	
	D. Dependent child who is the subject of a Medical Child Support Order or Qualified Medical Support Order that creates or recognizes the right of the Dependent Child to receive benefits under a parent's health insurance coverage:	
	Medical Child Support Order: the date specified in the Medical Child Support Order.	
	Qualified Medical Support Order: the date specified in the Qualified Medical Support Order.	
	If the Qualified Employee is enrolled, the newly eligible Dependent Child will be covered automatically, but only for the first thirty-one (31) days following the child's First Eligibility Date. The Subscriber must enroll such a Dependent Child within thirty-one (31) days of the child's First Eligibility Date.	
Newly Eligible Dependent	The Special Enrollment Period is the thirty-one (31) day period from the	
(including a Spouse or a	date of the qualifying event.	
newly eligible Dependent	date of the qualitying events	
Child not described	The Effective Date of coverage is the first day of the month following the	
above)	receipt of enrollment by the SHOP Exchange.	
Eligible Qualified	The Special Enrollment Period is the thirty (30) day period from the date of	
Individual who loses other	the qualifying event.	
Minimum Essential	1	
Coverage (as provided in	The Effective Date of coverage is the first day of the month following the	
the Evidence of Coverage)	receipt of enrollment by the SHOP Exchange.	
Eligible Qualified	The Special Enrollment Period is the sixty (60) day period from the date of	
Individual who loses	the qualifying event.	
coverage under a Medicaid plan or a state child health plan,(as provided in the Evidence of Coverage).	The Effective Date of coverage is the first day of the month following the receipt of enrollment by the SHOP Exchange.	
Eligible Qualified	The Special Enrollment Period shall be sixty (60) days prior to the and of	
Eligible Qualified Individual or is permitted	The Special Enrollment Period shall be sixty (60) days prior to the end of coverage under the employer sponsored plan.	
to terminate coverage	coverage under the employer sponsored plan.	
under a non-qualifying	The Effective Date of coverage is the first day of the month following the	
employer-sponsored	receipt of enrollment by the SHOP Exchange.	
health benefit plan (as	Livering of childring of the brior Exchange.	
provided in the Evidence		
of Coverage)		
DC/CEDC/SHOD/ELIC (1/14)	DCADW/IIDADW IN	

Eligible Qualified Individual who becomes eligible to enroll due to any other qualifying event stated in Section 2.5 of the Evidence of Coverage The Special Enrollment Period is the thirty (30) day period from the date of the qualifying event.

The Effective Date of the Qualified Individual's coverage will be:

- A. For enrollment received by the SHOP Exchange between the first and the fifteenth day of the month, the first day of the following month; and
- B. For enrollment received by the SHOP Exchange between the sixteenth and the last day of the month, the first day of the second following month.

For a Qualified Individual who enrolls during a special enrollment period due to (i) an error by the SHOP Exchange or the United States Department of Health and Human Services; (ii) another Qualified Health Plan substantially violating a material provision of its contract with the Qualified Individual; or (iii) other exceptional circumstances as determined by the SHOP Exchange, the Effective Date of the Qualified Individual's coverage will be established by the SHOP Exchange to be either:

- A. The date of the event that triggered the Special Enrollment Period under these circumstances; or
- B. The Effective Date set forth for all other circumstances stated above.

TERMINATION OF ENROLLMENT

Termination of Enrollment by the Subscriber

Subscriber terminates
enrollment of a Member

Termination is effective:

- A. If notice of termination of enrollment given because a Member will enroll in another Qualified Health Plan: the day before the effective date of coverage under the new Qualified Health Plan.
- B. If notice of termination of enrollment given because a Member is newly eligible for Medicaid, the federal child health insurance plan (CHIP) or a State-funded low-income basic health plan (known as a BHP): the day before coverage under one of these programs begins.
- C. In all other cases:
 - 1. On the date stated by the Subscriber, if the Subscriber has given reasonable notice. For purposes of this provision, reasonable notice is defined as fourteen (14) days from the requested termination date; or
 - 2. Fourteen (14) days after the date the Subscriber requested termination, if the Subscriber does not provide reasonable notice.

Termination of Enrollment by CareFirst BlueChoice or the SHOP Exchange

Subscriber no longer eligible as a Qualified Employee or Qualified Individual

Dependent Child
Limiting Age (other than Enrollment of the Subscriber and any Dependents terminates on the last day of the month following the month in which employment with the Group terminates or, otherwise, in which the Subscriber first no longer meets the eligibility requirements stated in the Evidence of Coverage.

Enrollment terminates on the last day of the month following the month of the Dependent Child's 26th birthday.

Dependent Child Limiting Age (other than a Dependent Child who is incapacitated as provided in the Evidence of Coverage)

Dependent no longer eligible (on grounds other than Limiting Age)	Enrollment terminates on the last day of the month following the month in which the Exchange notifies the Member that he or she no longer meets the eligibility requirements stated in the Evidence of Coverage.	
Death of Subscriber	Enrollment of the Subscriber and any Dependents terminates on the last day of the month after the Subscriber's death.	

Group Hospitalization and Medical Services, Inc.

doing business as

CareFirst BlueCross BlueShield (CareFirst)

840 First Street, NE

Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

OUT-OF-NETWORK EVIDENCE OF COVERAGE FOR A QUALIFIED HEALTH PLAN

This Qualified Health Plan is being offered through the SHOP Exchange.

This is the Out-of-Network Evidence of Coverage for the jointly offered Point-of-Service product with In-Network benefits administered by CareFirst BlueChoice, Inc. and Out-of-Network indemnity benefits administered by Group Hospitalization and Medical Services, Inc., doing business as CareFirst BlueCross BlueShield (CareFirst). Each time that services are sought, the Member may choose to receive either In-Network benefits or to receive Out-of-Network indemnity benefits.

This Out-of-Network Evidence of Coverage, including any attachments, notices, amendments and riders, is a part of the Out-of-Network Group Contract issued to the Group through which Members are enrolled for covered health benefits. In addition, the Out-of-Network Group Contract includes other provisions that explain the duties of CareFirst and the Group. The Group's payment to the SHOP Exchange and CareFirst's issuance of the Out-of-Network Group Contract make the Out-of-Network Group Contract's terms and provisions binding on CareFirst and the Group.

The Group reserves the right to change, modify, or terminate the plan, in whole or in part.

Members should not rely on any oral description of the plan because the written terms in the Group's plan documents always govern.

CareFirst recommends that the Member familiarizes himself or herself with the CareFirst complaint and appeal procedure, and make use of it before taking any other action.

NOTE: It is a crime to provide false or misleading information to an insurer for the purpose of defrauding the insurer or any other person. Penalties include imprisonment and/or fines. In addition, CareFirst may deny insurance benefits if false information materially related to a claim was provided by the applicant.

Group Name:	Sample
Group Number:	Sample
Product Name:	BlueChoice Advantage Gold 3000
Group Effective Date:	<u>January 1, 2019</u>
	Group Hospitalization and Medical Services, Inc.
	[Signature]
	[Name] [Title]

SECTION	TABLE OF CONTENTS	PAGE
1	Definitions	3
2	Eligibility and Enrollment	13
3	Termination of Coverage	22
4	Continuation of Coverage	26
5	Coordination of Benefits (COB); Subrogation	28
6	General Provisions	35
ATTACHMENTS		
A	Benefit Determinations and Appeals	A-1
В	Description of Covered Services	B-1
С	Schedules of Benefits	C-1
	Amendments/Notices/Riders	
D	Eligibility Schedule	D-1

SECTION 1 DEFINITIONS

The underlined terms, when capitalized, are defined as follows:

<u>Adoption</u> means the earlier of a judicial decree of adoption, or the assumption of custody, pending adoption, of a prospective adoptive child by a prospective adoptive parent.

<u>Affordable Care Act</u> means the Patient Protection and Affordable Care Act of 2010 (Pub. L. 111–148), as amended by the Health Care and Education Reconciliation Act of 2010 (Pub. L. 111–152).

Allowed Benefit means:

- A. For a Participating Provider, the Allowed Benefit for a Covered Service is the amount agreed upon between CareFirst and the Participating Provider which, in some cases, will be a rate set by a regulatory agency. The benefit is payable to the provider and is accepted as payment in full, except for any applicable Deductible, Copayment, or Coinsurance amounts, for which the Member is responsible.
- B. For a Non-Participating Provider that is a health care practitioner, the Allowed Benefit for a Covered Service is based upon the lesser of the provider's actual charge or the established fee schedule. The benefit is payable to the Member or to the provider at the discretion of CareFirst. The Member is responsible for any applicable Deductible, Copayment, or Coinsurance amounts stated in the Schedule of Benefits and the difference between the Allowed Benefit and the practitioner's actual charge. The provider may bill the Member directly for such amounts. It is the Member's responsibility to apply any CareFirst payments to the claim from the Non-Participating Provider charge.
- C. For a Non-Participating Provider that is a health care facility, the Allowed Benefit for a Covered Service is based upon either the provider's actual charge or the established fee schedule. The benefit is payable to the Member or to the facility, at the discretion of CareFirst. Benefit payments to Department of Defense and Veteran Affairs providers will be made directly to the provider. The Member is responsible for any applicable Deductible, Copayment, or Coinsurance amounts stated in the Schedule of Benefits and, unless negotiated, for the difference between the Allowed Benefit and the provider's actual charge. It is the Member's responsibility to apply any CareFirst payments to the claim from the Non-Preferred Facility.

In some cases, and on an individual basis, CareFirst is able to negotiate a lower rate with an eligible provider. In that instance, the CareFirst payment will be based on the negotiated fee and the provider agrees to accept the amount as payment in full except for any applicable Deductible, Copayment, or Coinsurance amounts, for which the Member is responsible.

D. For a Covered Service rendered by a Non-Participating Provider of ambulance services, the Allowed Benefit for a Covered Service is based upon the lesser of the provider's actual charge or the established fee schedule. The benefit is payable to the Member or to the Non-Participating Provider of ambulance services, at the discretion of CareFirst. When benefits are paid to the Member, it is the Member's responsibility to apply any CareFirst payments to the claim from the Non-Participating Provider of ambulance services. The Member is responsible for any applicable Deductible, Copayment, or Coinsurance amounts stated in the Schedule of Benefits and the difference between the Allowed Benefit and the provider's actual charge. The provider may bill the Member directly for such amounts.

For Emergency Services provided by a Non-Participating Provider, the Allowed Benefit for a Covered Service is based upon the lesser of the provider's actual charge, or the amount that would be paid to a Participating Provider for the Covered Service. The

Member is responsible for any applicable Deductible, Copayment, or Coinsurance amounts stated in the Schedule of Benefits and the difference between the Allowed Benefit and the practitioner's actual charge. The provider may bill the Member directly for such amounts.

Pediatric Dental Allowed Benefit means:

- A. For Participating Dentists, the Pediatric Dental Allowed Benefit payable to a Participating Dentist for a Covered Dental Service will be the lesser of (1) the Dentist's actual charge; or (2) the benefit amount, according to the CareFirst rate schedule for the Covered Dental Service that applies on the date that the service is rendered. The benefit payment is made directly to the Participating Dentist and is accepted as payment in full except for any applicable Deductible, Copayment or Coinsurance for which the Subscriber is responsible as stated in the Schedule of Benefits.
- B. For Non-Participating Dentists, the Allowed Benefit payable to a Non-Participating Dentist for a Covered Dental Service will be the lesser of (1) the Dentist's actual charge; or (2) the benefit amount, according to the CareFirst rate schedule for the Covered Dental Service that applies on the date that the service is rendered. For any Non-Participating Dentist, the benefit is payable to the Subscriber or to the Non-Participating Dentist at the discretion of CareFirst. The Subscriber is responsible for payment for services to the Non-Participating Dentist, including any applicable Deductible and Coinsurance amounts as stated in the Schedule of Benefits and for any balance bill amounts. The Non-Participating Dentist may bill the Subscriber directly for such amounts. It is the Subscriber's responsibility to apply any CareFirst payments to the claim from the Non-Participating Dentist.

Vision Allowed Benefit means:

For a Non-Contracting Vision Provider, the Vision Allowed Benefit for a Covered Vision Service is the lesser of:

- 1. The actual charge; or
- 2. The benefit amount, according to the Vision Care Designee's rate schedule for the covered service or supply that applies on the date the service is rendered.

Benefits may be paid to the Member or to the Non-Contracting Vision Provider at the discretion of the Vision Care Designee. The Member is responsible for the cost difference between the Vision Care Designee's payment and the Non-Contracting Vision Provider's actual charge. The Non-Contracting Vision Provider may bill the Member directly. It is the Member's responsibility to apply any CareFirst payments to the claim from the Non-Contracting Vision Provider.

Annual Open Enrollment Period means the period of no less than thirty (30) days each year prior to the Group's Contract Renewal Date during which an individual may enroll or change coverage in this Qualified Health Plan through the SHOP Exchange.

<u>Benefit Period</u> means the Contract Year during which coverage is provided for Covered Services, Covered Dental Services and Covered Vision Services.

<u>Bereavement Counseling</u> means counseling provided to the Immediate Family or Family Caregiver of the Member after the Member's death to help the Immediate Family or Family Caregiver cope with the death of the Member.

Calendar Year means January 1 through December 31 of each year.

<u>Cardiac Rehabilitation</u> means inpatient or outpatient services designed to limit the physiologic and psychological effects of cardiac illness, reduce the risk for sudden death or reinfarction, control cardiac

symptoms, stabilize or reverse atherosclerotic process, and enhance the psychosocial and vocational status of eligible Members.

<u>CareFirst</u> means Group Hospitalization and Medical Services, Inc., doing business as CareFirst BlueCross BlueShield.

<u>CareFirst BlueChoice</u> means CareFirst BlueChoice, Inc., the corporate entity which underwrites the In-Network benefits under the In-Network Evidence of Coverage.

<u>Civil Union</u> means a same-sex relationship similar to marriage that is recognized by law. The Subscriber's partner in a Civil Union is eligible for coverage to the same extent as an eligible Spouse.

<u>Coinsurance</u> means the percentage of the Allowed Benefit allocated between CareFirst and the Member, whereby CareFirst and the Member share in the payment for Covered Services, Covered Dental Services, or Covered Vision Services.

<u>Contract Renewal Date</u> means the date on which the Out-of-Network Group Contract renews and each anniversary of such date.

Contract Year means the twelve (12) month period beginning on the Group Effective Date.

<u>Contracting Pharmacy Provider</u> means a separate independent Pharmacist or Pharmacy that has contracted with CareFirst or its designee to provide covered Prescription Drugs.

<u>Convenience Item</u> means any item that increases physical comfort or convenience without serving a Medically Necessary purpose (e.g., elevators, hoyer/stair lifts, ramps, shower/bath benches, and items available without a prescription).

<u>Copayment (Copay)</u> means the fixed dollar amount that a Member must pay for certain Covered Services, Covered Dental Services or Covered Vision Services.

<u>Cosmetic</u> means a service or supply which is provided with the primary intent of improving appearance, not restoring bodily function or correcting deformity resulting from disease, trauma, or previous therapeutic intervention, as determined by CareFirst.

<u>Covered Dental Services</u> means Medically Necessary services or supplies listed in Section 2 of the Description of Covered Services.

<u>Covered Service</u> means Medically Necessary services or supplies provided in accordance with the terms of this Out-of-Network Evidence of Coverage other than Covered Dental Services or Covered Vision Services.

<u>Covered Vision Services</u> means Medically Necessary services or supplies listed in Section 3 of the Description of Covered Services.

<u>Custodial Care</u> means care provided primarily to meet the personal needs of the patient. Custodial Care does not require skilled medical or paramedical personnel. Such care includes help in walking, bathing, or dressing. Custodial Care also includes preparing food or special diets, feeding, administering medicine, or any other care not requiring continuing services of medically trained personnel.

<u>Decertification or Decertified</u> means the termination by the SHOP Exchange of the certification and offering of this Qualified Health Plan.

<u>Deductible</u> means the dollar amount of the Allowed Benefits payable during a Benefit Period for Covered Services or Covered Dental Services that must first be incurred by the Member before CareFirst will make payments for Covered Services or Covered Dental Services.

<u>Dental Director</u> is a Dentist appointed by the Medical Director of CareFirst to perform administrative

duties with regard to the dental services listed in this Out-of-Network Evidence of Coverage.

<u>Dentist</u> means an individual who is licensed to practice dentistry as defined by the respective jurisdiction where the practitioner provides care.

<u>Dependent</u> means a Member who is covered as an eligible Spouse or Dependent Child as defined in Sections 2.2 and 2.3. The eligibility of Dependents to enroll is stated in the Eligibility Schedule.

Dependent Child or Dependent Children means one or more eligible individuals as defined in Section 2.3.

<u>Diabetic Supply or Diabetic Supplies</u> means all Medically Necessary and appropriate supplies prescribed by a health care provider for the treatment of diabetes, including but not limited to lancets, alcohol wipes, test strips (blood and urine), syringes and needles.

<u>Effective Date</u> means the date on which the Member's coverage becomes effective. Covered Services, Covered Dental Services, and Covered Vision Services rendered on or after the Member's Effective Date are eligible for coverage.

Emergency Medical Condition means:

- A. A medical condition manifesting itself by acute symptoms of sufficient severity (including severe pain) such that the absence of immediate medical attention could reasonably be expected to result in placing the health of the individual (or, with respect to a pregnant woman, the health of the woman or her unborn child) in serious jeopardy, serious impairment to bodily functions, or serious dysfunction of any bodily organ or part; or
- B. With respect to a pregnant woman who is having contractions: there is inadequate time to effect a safe transfer to another hospital before delivery, or transfer may pose a threat to the health or safety of the woman or the unborn child.

Emergency Services means, with respect to an Emergency Medical Condition:

- A. A medical screening examination (as required under section 1867 of the Social Security Act, 42 U.S.C. 1395dd) within the capability of the emergency department of a hospital, including ancillary services routinely available to the emergency department to evaluate such Emergency Medical Condition; and
- B. Such further medical examination and treatment, to the extent they are within the capability of the staff and facilities available at the hospital, as are required under section 1867 of the Social Security Act (42 U.S.C. 1395dd(e)(3)) to stabilize the Member. The term to "stabilize" with respect to an Emergency Medical Condition, has the meaning given in section 1867(e)(3) of the Social Security Act (42 U.S.C. 1395dd(e)(3))).

<u>Experimental/Investigational</u> means a service or supply in the developmental stage and in the process of human or animal testing excluding patient costs for clinical trials as stated in the Description of Covered Services. Services or supplies that do not meet all five of the criteria listed below are deemed to be Experimental/Investigational:

- A. The Technology* must have final approval from the appropriate government regulatory bodies:
- B. The scientific evidence must permit conclusions concerning the effect of the Technology on health outcomes;
- C. The Technology must improve the net health outcome;
- D. The Technology must be as beneficial as any established alternatives; and

E. The improvement must be attainable outside the Investigational settings.

* "Technology" includes drugs, devices, processes, systems, or techniques.

FDA means the United States Food and Drug Administration.

<u>Family Caregiver</u> means a relative by blood, marriage, or Adoption who lives with or is the primary Caregiver of the terminally ill Member.

<u>Family Counseling</u> means counseling given to the Immediate Family or Family Caregiver of the terminally ill Member for the purpose of learning to care for the Member and to adjust to the impending death of the Member.

<u>Group</u> means the Qualified Employer to which CareFirst has issued the Out-of-Network Group Contract and the Out-of-Network Evidence of Coverage.

<u>Habilitative Services</u> mean health care services and devices that help a person keep, learn, or improve skills and functioning for daily living. Examples include therapy for a child who is not walking or talking at the expected age. These services may include physical and occupational therapy, speech-language pathology and other services for people with disabilities in a variety of inpatient and/or outpatient settings.

<u>Home Health Care or Home Health Care Services</u> means the continued care and treatment of a Member in the home by a licensed Home Health Agency if:

- A. The institutionalization of the Member in a hospital, related institution, or Skilled Nursing Facility would otherwise have been required if Home Health Care Services were not provided; and
- B. The Plan of Treatment covering the Home Health Care Service is established and approved in writing by the health care provider, and determined to be Medically Necessary by CareFirst.

<u>Immediate Family</u> means the Spouse, parents, siblings, grandparents, and children of the terminally ill Member.

<u>Infusion Services</u> means treatment provided by placing therapeutic agents into the vein, and parenteral administration of medication and nutrients. Infusion services also include enteral nutrition, which is the delivery of nutrients by tube into the gastrointestinal tract. Infusion Services include all medications administered intravenously and/or parenterally.

<u>In-Network</u> means Point-of-Service benefits provided to the Member under the In-Network Evidence of Coverage issued to the Subscriber by CareFirst BlueChoice under the In-Network Group Contract between the Group and CareFirst BlueChoice.

<u>In-Network Evidence of Coverage</u> means the Point-of-Service in-network agreement, including all duly authorized attachments, notices, amendments and riders, if any, issued to the Group and the Subscriber by CareFirst BlueChoice under the In-Network Group Contract between the Group and CareFirst BlueChoice.

<u>In-Network Group Contract</u> means the Point-of-Service in-network contract, including all duly authorized attachments, notices, amendments and riders between the Group and CareFirst BlueChoice.

<u>Limiting Age</u> means the maximum age up to which a Dependent Child may be covered as stated in the Eligibility Schedule.

<u>Low Vision</u> means a significant loss of vision but not total blindness. Ophthalmologists and optometrists specializing in Low Vision care can evaluate and prescribe optical devices, and provide training and instruction to maximize the remaining usable vision for Members with Low Vision.

Mastectomy means the surgical removal of all or part of the breast.

<u>Medical Child Support Order (MCSO)</u> means an order issued in the format prescribed by federal law and issued by an appropriate child support enforcement agency to enforce the health insurance coverage provisions of a child support order. An order means a judgment, decree, or a ruling (including approval of a settlement agreement) that:

- A. Is issued by a court or administrative child support enforcement agency of any state or the District of Columbia; and
- B. Creates or recognizes the right of a child to receive benefits under a parent's health insurance coverage or establishes a parent's obligation to pay child support and provide health insurance coverage for a child.

<u>Medical Director</u> means a board certified physician who is appointed by CareFirst. The duties of the Medical Director may be delegated to qualified persons.

<u>Medically Necessary or Medical Necessity</u> means health care services or supplies that a health care provider, exercising clinical judgment, renders to or recommends for a patient for the purpose of preventing, evaluating, diagnosing or treating an illness, injury, disease, or its symptoms. These health care services or supplies are:

- A. In accordance with generally accepted standards of medical practice;
- B. Clinically appropriate, in terms of type, frequency, extent, site and duration, and considered effective for a patient's illness, injury, or disease;
- C. Not primarily for the convenience of a patient or health care provider; and
- D. Not more costly than an alternative service or sequence of services that are at least as likely to produce equivalent therapeutic or diagnostic results in the diagnosis or treatment of the patient's illness, injury, or disease.

For these purposes, "generally accepted standards of medical practice" means standards based on credible scientific evidence published in peer-reviewed medical literature generally recognized by the relevant medical community, physician specialty society recommendations and views of health care providers practicing in relevant clinical areas, and any other relevant factors.

The fact that a health care provider may prescribe, authorize or direct a service does not of itself make it Medically Necessary or covered by the Out-of-Network Group Contract.

<u>Medical Nutrition Therapy</u> provided by a licensed dietitian-nutritionist involves the assessment of the Member's overall nutritional status followed by the assignment of an individualized diet, counseling, and/or specialized nutrition therapies to treat a chronic illness or condition. The licensed dietitian-nutritionist, working in a coordinated, multidisciplinary team effort with the Primary Care Physician takes into account a Member's condition, food intake, physical activity, course of any medical therapy, including medications and other treatments, individual preferences, and other factors.

<u>Member</u> means an individual who meets all applicable eligibility requirements of Section 2, is enrolled for coverage and for whom the CareFirst receives the premiums and other required payments. A Member can be either a Subscriber or Dependent.

Minimum Essential Coverage has the meaning given in the Affordable Care Act, 26 U.S.C. §5000A(f).

<u>Non-Contracting Vision Provider</u> means any optometrist or ophthalmologist licensed as such by the duly constituted authority in the jurisdiction in which Covered Vision Services are rendered when acting within the scope of such license; and, who does not have an agreement with the Vision Care Designee for the rendering of Covered Vision Services. A Non-Contracting Vision Provider may or may not have contracted with CareFirst. The Member should contact the Vision Care Designee for the current list of Contracting Vision Providers.

<u>Non-Participating Dentist</u> means any Dentist who, at the time of rendering a Covered Dental Service to the Member, does not have a written agreement with CareFirst or CareFirst's designee for the rendering of such service.

Non-Participating or Non-Par Provider means any health care provider that does not contract with CareFirst to provide Covered Services. Neither Participating Dentists or Non-Participating Dentists who provide Covered Dental Services nor Non-Contracting Vision Providers who provide Covered Vision Services are Non-Participating Providers for the purposes of this definition.

<u>Out-of-Network</u> means Point-of-Service benefits provided to the Member under this Out-of-Network Evidence of Coverage.

<u>Out-of-Network Evidence of Coverage</u> means this Point-of-Service out-of-network agreement, including all attachments, notices, amendments and riders, if any, issued to the Group and the Subscriber by CareFirst under the Out-of-Network Group Contract between the Group and CareFirst.

<u>Out-of-Network Group Contract</u> means the Point-of-Service out-of-network contract, including all attachments, notices, amendments and riders between the Group and CareFirst.

Out-of-Pocket Maximum means the maximum amount the Member will have to pay for his/her share of benefits in any Benefit Period. The Out-of-Pocket Maximum does not include premiums, the cost of services that are not Covered Services, or any amounts paid to providers in excess of the Allowed Benefit, the Pediatric Dental Allowed Benefit, the Vision Allowed Benefit or the Prescription Drug Allowed Benefit. Once the Member meets the Out-of-Pocket Maximum, the Member will no longer be required to pay Copayments, Coinsurance or Deductible for the remainder of the Benefit Period.

Over-the-Counter means any item or supply, as determined by CareFirst, available for purchase without a prescription, unless otherwise a Covered Service. This includes, but is not limited to, non-prescription eye wear, family planning and contraception items for men, cosmetics or health and beauty aids, food and nutritional items, support devices, non-medical items, foot care items, first aid and miscellaneous medical supplies (whether disposable or durable), personal hygiene supplies, incontinence supplies, and related over-the-counter medications, solutions, items or supplies.

<u>Participating Dentist</u> means any Dentist who, at the time of rendering a Covered Dental Service to the Member, has a written agreement with CareFirst or CareFirst's designee for the rendering of such service.

<u>Participating Provider or Par Provider</u> means a physician, health care professional or health care facility that contracts with CareFirst to be paid directly for rendering Covered Services to Members.

<u>Point-of-Service</u> means a benefit program that offers two benefit levels: In-Network that provides HMO-level benefits; and an Out-of-Network that provides indemnity level of benefits. Point-of-Service means that each time Members receive Covered Services, Covered Dental Services, or Covered Vision Services they can choose to receive either In-Network benefits or Out-of-Network benefits.

<u>Prescription Drug</u> means:

A. A drug, biological or compounded prescription intended for outpatient use that carries the FDA legend "may not be dispensed without a prescription";

- B. Drugs prescribed for treatments other than those stated in the labeling approved by the FDA, if the drug is recognized for such treatment in standard reference compendia or in the standard medical literature as determined by CareFirst;
- C. A covered Over-the-Counter medication or supply; or
- D. Any Diabetic Supply.
- E. Prescription Drugs do not include:
 - 1 Compounded bulk powders that contain ingredients that:
 - a) Do not have FDA approval for the route of administration being compounded, or
 - b) Have no clinical evidence demonstrating safety and efficacy, or
 - c) Do not require a prescription to be dispensed.
 - 2. Compounded drugs that are available as a similar commercially available Prescription Drug unless:
 - a) There is no commercially available bioequivalent Prescription Drug; or
 - b) The commercially available bioequivalent Prescription Drug has caused or is likely to cause the Member to have an adverse reaction.

<u>Primary Care Dependent</u> means an unmarried grandchild, niece or nephew for whom the Subscriber provides primary care including food, shelter and clothing on a regular and continuous basis during the time the District of Columbia public schools are in regular session.

Primary Care Physician (PCP) means health care practitioners in the following disciplines:

- A. General internal medicine;
- B. Family practice medicine;
- C. General pediatric medicine; or
- D. Geriatric medicine.

Services rendered by Specialists in the disciplines above will be treated as PCP visits for Member payment purposes.

<u>Prior Authorization List</u> means the limited list of Prescription Drugs issued by CareFirst for which providers, when writing, and Pharmacists, when filling prescriptions, must obtain prior authorization from CareFirst. A copy of the Prior Authorization List is available to the Member upon request.

<u>Professional Nutritional Counseling</u> means individualized advice and guidance given to a Member who is at nutritional risk due to nutritional history, current dietary intake, medication use, or chronic illness or condition, about options and methods for improving nutritional status. Professional Nutritional Counseling must be provided by a licensed dietitian-nutritionist, physician, physician assistant, or nurse practitioner.

<u>Qualified Employee</u> means an eligible individual who has been offered health insurance coverage by the Group through the SHOP Exchange. The Group's eligibility requirements for a Qualified Employee are stated in the Eligibility Schedule.

<u>Qualified Employer</u> means the employer that the SHOP Exchange has determined to be qualified to offer Qualified Health Plan(s).

<u>Qualified Health Plan</u> means a health plan certified by the SHOP Exchange as having met the standards established by the U.S. Department of Health and Human Services.

<u>Qualified Home Health Agency</u> means a licensed program approved for participation as a home health agency under Medicare, or certified as a home health agency by the Joint Commission on Accreditation of Healthcare Organizations, its successor, or the applicable state regulatory agency.

Qualified Hospice Care Program means a coordinated, interdisciplinary program provided by a hospital, Qualified Home Health Agency, or other health care facility licensed or certified by the state in which it operates as a hospice program and is designed to meet the special physical, psychological, spiritual, and social needs of terminally ill individuals and their families, by providing palliative and supportive medical, nursing, and other health services through home or inpatient care during the illness and bereavement period. Benefits are available to:

- A. Individuals who have no reasonable prospect of cure as estimated by a physician; and
- B. The immediate families or Family Caregivers of those individuals.

<u>Qualified Medical Support Order (QMSO)</u> means a Medical Child Support Order, issued under state law or the laws of the District of Columbia, that is issued to an employer sponsored health plan that complies with section 609(A) of the Employee Retirement Income Security Act of 1974 (ERISA), as amended.

<u>Rescind</u> or <u>Rescission</u> means a termination, cancellation or discontinuance of coverage that has retroactive effect. For example, a cancellation that treats coverage as void from the time of the individual's or group's enrollment is a Rescission. As another example, a cancellation that voids benefits paid up to a year before the cancellation is also a Rescission for this purpose. Coverage is not Rescinded and a cancellation or discontinuance of coverage is not a Rescission if:

- A. The termination, cancellation or discontinuance of coverage has only a prospective effect; or
- B. The termination, cancellation or discontinuance of coverage is effective retroactively to the extent it is attributable to a failure to timely pay charges when due, by the Group.

<u>Respite Care</u> means temporary care provided to the terminally ill Member to relieve the Caregiver/Family Caregiver from the daily care of the Member.

<u>Service Area</u> means the clearly defined geographic area in which CareFirst has arranged for the provision of health care services to be generally available and readily accessible to Members, except for emergency services. CareFirst may amend the defined Service Area at any time by notifying the Group in writing.

The Service Area is as follows: the District of Columbia; the state of Maryland; in the Commonwealth of Virginia, the cities of Alexandria and Fairfax, Arlington County, the town of Vienna and the areas of Fairfax and Prince Williams Counties in Virginia lying east of Route 123.

SHOP Exchange means the District of Columbia Health Benefit Exchange (DC HBX).

Skilled Nursing Facility means a licensed institution (or a distinct part of a hospital) accredited or approved under Medicare or The Joint Commission and provides continuous Skilled Nursing Care and related services for Members who require medical care, Skilled Nursing Care, or rehabilitation services. Inpatient skilled nursing is for patients who are medically fragile with limited endurance and require a licensed health care professional to provide skilled services in order to ensure the safety of the patient and to achieve the medically desired result. Inpatient skilled nursing services must be provided on a 24 hour basis, 7 days a week.

<u>Sound Natural Teeth</u> means teeth restored with intra- or extra-coronal restorations (fillings, inlays, onlays, veneers, and crowns) that are in good condition; absent decay, fracture, bone loss, periodontal disease, root canal pathology or root canal therapy and excludes any tooth replaced by artificial means (implants, fixed or removable bridges, dentures).

<u>Special Enrollment Period</u> means a period during which an eligible individual who experiences certain qualifying events may enroll in, or change enrollment in, a Qualified Health Plan through the SHOP Exchange outside of any Annual Open Enrollment Periods.

<u>Specialist</u> means a licensed health care provider who is certified or trained in a specified field of medicine.

<u>Spouse</u> means a person of the same or opposite sex who is legally married to the Subscriber under the laws of the state or jurisdiction in which the marriage took place. A marriage legally entered into in another jurisdiction will be recognized as a marriage in the District of Columbia. The Subscriber's partner in a Civil Union or domestic partnership is eligible for coverage to the same extent as an eligible Spouse. Whenever the term "Spouse" appears in this Evidence of Coverage, the provision includes a Civil Union partner or domestic partner.

<u>Subscriber</u> means a Member who is enrolled as a Qualified Employee or eligible former Qualified Employee rather than as a Dependent.

<u>Urgent Care</u> means treatment for a condition that is not a threat to life or limb but does require prompt medical attention. Also, the severity of an urgent condition does not necessitate a trip to the hospital emergency room. An Urgent Care facility is a freestanding facility that is not a physician's office and which provides Urgent Care.

<u>Vision Care Designee</u> means the entity with which CareFirst has contracted to administer Vision Care. CareFirst's Vision Care Designee is Davis Vision, Inc. Davis Vision, Inc. is an independent company and administers the Vision Care benefits on behalf of CareFirst.

<u>Waiting Period</u> means the period of time, stated in the Eligibility Schedule, that must pass before a Qualified Employee or any Dependent is eligible for coverage under the terms of the Out-of-Network Group Contract.

SECTION 2 ELIGIBILITY AND ENROLLMENT

- 2.1 <u>Requirements for Coverage</u>. To be covered, <u>all</u> of the following conditions must be met:
 - A. A Subscriber must be an eligible Qualified Employee of the Group. To enroll as a Subscriber, the individual must, at the time of enrollment, meet the eligibility requirements established by the Group. The Group's eligibility requirements for a Qualified Employee are stated in the Eligibility Schedule.
 - B. Any other Member must be an eligible individual who is a Dependent of a Subscriber. To enroll as a Dependent, the individual must, at the time of enrollment meet the eligibility requirements established by the Group. The eligibility of Dependents to enroll is stated in the Eligibility Schedule.
 - C. The Subscriber or Member must be enrolled under the In-Network Evidence of Coverage.
 - D. For each Subscriber and Member, the SHOP Exchange must receive premium payments as required by the Out-of-Network Group Contract.
- 2.2 <u>Eligibility of Subscriber's Spouse</u>. If the Group has elected to include coverage for the Subscriber's Spouse, then a Subscriber may enroll a Spouse as a Dependent. A Subscriber cannot cover a former Spouse once divorced or if the marriage had been annulled. Premium changes resulting from the enrollment of a Spouse will be effective as of the Effective Date of the Spouse's enrollment.
- 2.3 <u>Eligibility of Dependent Children</u>. If the Group has elected to include coverage for Dependent Children of the Subscriber or a Subscriber's covered Spouse, then a Subscriber may enroll one or more Dependent Children. A Dependent Child means an individual who:
 - A. Is:
 - 1. The natural child, stepchild, or adopted child of the Subscriber;
 - 2. A child placed with the Subscriber or the Subscriber's covered Spouse for legal Adoption;
 - 3. A child under testamentary or court appointed guardianship, other than temporary guardianship for less than twelve (12) months' duration, of the Subscriber or the Subscriber's covered Spouse; or
 - 4. An unmarried grandchild, niece or nephew, who meets the requirements for coverage as the Subscriber's Primary Care Dependent as stated below:
 - a) The child must be the Subscriber's unmarried grandchild, niece, or nephew;
 - b) The child is under the Subscriber's Primary Care. <u>Primary Care</u> means the Subscriber provides food, clothing and shelter for the child on a regular and continuous basis during the time District of Columbia public schools are in regular session; and.
 - c) If the child's legal guardian is someone other than the Subscriber, the child's legal guardian is not covered under any other health insurance policy.

The Subscriber must provide CareFirst with proof upon application that the child meets the requirements for coverage as a Primary Care Dependent, including proof of the child's relationship and primary dependency on the Subscriber and certification that the child's legal guardian does not have other coverage. CareFirst reserves the right to verify whether the child is and continues to qualify as a Primary Care Dependent, and

- 5. A child who becomes a Dependent of the Subscriber through a child support order (MCSO or QMSO) or other court order.
- B. Is under the Limiting Age as stated in the Eligibility Schedule; or
- C. Is a disabled Dependent Child who is older than the Limiting Age and the Subscriber provides proof that: (1) the Dependent Child is incapable of self-support or maintenance because of a mental or physical incapacity; (2) the Dependent Child is primarily dependent upon the Subscriber or the Subscriber's covered Spouse for support and maintenance; and (3) the Dependent Child had been covered under the Subscriber's or the Subscriber's Spouse's prior health insurance coverage since before the onset of the mental or physical incapacity.
- D. Is the subject of a Medical Child Support Order (MCSO) or Qualified Medical Support Order (QMSO) that creates or recognizes the right of the child to receive benefits under the health insurance coverage of the Subscriber or the Subscriber's covered Spouse.
- E. A child whose relationship to the Subscriber is not listed above, including, but not limited to, foster children or children whose only relationship is one of legal guardianship (except as provided above) is not eligible to enroll and is not covered under this Out-of-Network Evidence of Coverage, even though the child may live with the Subscriber and be dependent upon him or her for support.

2.4 <u>Limiting Age for Covered Dependent Children.</u>

- A. All covered Dependent Children are eligible for coverage up to the Limiting Age for Dependent Children as stated in the Eligibility Schedule.
- B. A Dependent Child covered under this Out-of-Network Evidence of Coverage will be eligible for coverage past the Limiting Age if, at the time coverage would otherwise terminate:
 - 1. The Dependent Child is unmarried and is incapable of self-support or maintenance because of a mental or physical incapacity;
 - 2. The Dependent Child is primarily dependent upon the Subscriber or the Subscriber's covered Spouse for support and maintenance;
 - 3. The mental or physical incapacity occurred before the covered Dependent Child reached the Limiting Age; and
 - 4. The Subscriber provides CareFirst with proof of the Dependent Child's mental or physical incapacity within thirty-one (31) days after the Dependent Child reaches the Limiting Age for Dependent Children. CareFirst has the right to verify whether the child is and continues to qualify as an incapacitated Dependent Child.
- C. Dependents' coverage will automatically terminate if there is a change in their age, status, or relationship to the Subscriber, such that they no longer meet the eligibility requirements of this Out-of-Network Evidence of Coverage or the Eligibility Schedule. Coverage of an ineligible Dependent will terminate as stated in the Eligibility Schedule.

- 2.5 <u>Open Enrollment Opportunities and Effective Dates</u>. A Qualified Employee may elect coverage for himself or herself or for an eligible Dependent only during the following times and under the following conditions.
 - A. <u>Annual Open Enrollment</u>. During an Annual Open Enrollment Period, a Qualified Employee or eligible individual may enroll as a Subscriber or Member.
 - B. <u>Newly Eligible Subscriber</u>. If a Qualified Employee is a new employee or a newly eligible employee of the Group, the new Qualified Employee may enroll him or herself and any eligible Dependent within thirty (30) days after a new Qualified Employee first becomes eligible. The eligibility requirements for a new Qualified Employee in the Group are stated in the Eligibility Schedule.
 - C. Special Enrollment. If a Qualified Employee or Dependent does not enroll during an Annual Open Enrollment Period or during the enrollment period for newly eligible Qualified Employees, he or she may only enroll during a Special Enrollment Period
 - 1. Thirty (30) or Thirty-One (31) Day Special Enrollment Period. Except as otherwise specified, an eligible individual may enroll as a Subscriber or Dependent upon the occurrence of one of the following qualifying events:
 - a) The Qualified Employee or a Dependent:
 - (1) Loses Minimum Essential Coverage. A loss of Minimum Essential Coverage includes those circumstances described in 26 CFR 54.9801-6(a)(3)(i) through (iii).

Loss of coverage described herein includes those circumstances described in 26 CFR 54.9801-6(a)(3)(i) through (iii) and in paragraphs (d)(1)(ii) through (iv) of 45 CFR §155.420. Loss of coverage does not include voluntary termination of coverage or other loss due to:

- (a) Failure to pay premiums on a timely basis, including COBRA premiums prior to the expiration of COBRA coverage; or
- (b) Situations allowing for a Rescission.
- (2) Is enrolled in any non-Calendar Year health insurance coverage even if the Qualified Employee or his or her Dependent has the option to renew such coverage. The date of the loss of coverage is the last day of the plan or policy year.
- (3) Loses pregnancy-related coverage described in 1902(a)(10)(A)(i)(IV) and (a)(10)(A)(ii)(IX) of the Social Security Act (42 U.S.C. 1396a(a)(10)(A)(i)(IV), (a)(10)(A)(ii)(IX)). The date of the loss of coverage is the last day the consumer would have pregnancy-related coverage; or
- (4) Loses medically needy coverage as described under section 1902(a)(10)(C) of the Social Security Act only once per calendar year. The date of the loss of coverage is the last day the consumer would have medically needy coverage.
- (5) At the option of the SHOP Exchange, the enrollee who is the Qualified Employee or the Spouse of the Qualified Employee if the enrollee loses a Dependent or is no longer considered to be a

Dependent due to divorce or legal separation; or if the employee or his or her Dependent dies.

- b) The Qualified Employee or a Dependent gains, or becomes, an eligible Dependent through marriage, birth, adoption, placement for adoption, or grant of court or testamentary guardianship or child support order (MCSO or QMSO) or other court order.
- c) The Qualified Employee or a Dependent has a foster child placed by an accredited foster child agency. (Note: The foster child is not eligible for coverage.)
- d) The Qualified Employee's or Dependent's enrollment in another Qualified Health Plan or non-enrollment, as evaluated and determined by the SHOP Exchange, is unintentional, inadvertent, or erroneous and is the result of an error, misrepresentation, misconduct, or inaction of an officer, employee, or agent of the SHOP Exchange or the United States Department of Health and Human Services or its instrumentalities, or a non-Exchange entity providing enrollment assistance or conducting enrollment activities.
- e) The Qualified Employee or Dependent, who is an enrollee in another Qualified Health Plan, demonstrates to the SHOP Exchange that the other Qualified Health Plan in which he or she has enrolled substantially violated a material provision of its contract in relation to the Qualified Employee or Dependent.
- f) The Qualified Employee or Dependent gains access to new Qualified Health Plans as a result of a permanent move.
- g) The Qualified Employee is an Indian, as defined in section 4 of the Indian Health Care Improvement Act, who may enroll in a Qualified Health Plan or change coverage from one Qualified Health Plan to another one time per month.
- h) The Qualified Employee or Dependent demonstrates to the SHOP Exchange, in accordance with guidelines issued by the United States Department of Health and Human Services that he or she meets other exceptional circumstances determined by the SHOP Exchange.
- i) The Qualified Employee or Dependent enrolled in the same Qualified Health Plan is determined to be newly eligible or newly ineligible for advance premium tax credit or has a change in eligibility for cost-sharing reductions

For purposes of this provision, advance premium tax credit means tax credits specified under section 1401 of the Affordable Care Act which are provided on an advance basis to an eligible individual enrolled in a Qualified Health Plan through the District of Columbia Health Benefit Exchange. For purposes of this provision, cost-sharing reduction means an affordability program under Section 1402 of the Affordable Care Act.

- 2. The Special Enrollment Period shall be as follows:
 - a) For the qualifying event listed in Section 2.5C.1.b) and c) (Qualified Employee or Dependent gains, or becomes, an eligible Dependent through marriage, birth, adoption, placement for adoption, placement for foster care, grant of court or testamentary guardianship), or child support

- order (MCSO or QMSO) or other court order: The thirty-one (31) day period from the date of the qualifying event.
- b) If an eligible employee or Dependent meets the requirements for the triggering event described in 2.5C.1.d), the exchange may take any action necessary to correct or eliminate the effects of the error, misrepresentation, or inaction.
- c) For the qualifying event listed in Section 2.5C.1.g) (Qualified Employee is an Indian, as defined in section 4 of the Indian Health Care Improvement Act): One time per month.
- d) For all other qualifying events listed in Section 2.5C.1: The thirty (30) day period from the date of the qualifying event, unless otherwise determined by the SHOP Exchange.
- 3. Sixty (60) Day Special Enrollment Period. In addition, an eligible individual may enroll as a Subscriber or Dependent upon the occurrence of one of the following qualifying events:
 - a) The Qualified Employee or Dependent loses eligibility for coverage under a Medicaid plan under title XIX of the Social Security Act or a state child health plan under title XXI of the Social Security Act;
 - b) The Qualified Employee or Dependent becomes eligible for assistance, with respect to coverage under a Qualified Health Plan offered through the SHOP Exchange, under such Medicaid plan or a state child health plan (including any waiver or demonstration project conducted under or in relation to such a plan); or
 - c) The Qualified Employee or Dependent is enrolled in an employersponsored health benefit plan that is not qualifying coverage in an eligible employer sponsored plan and is allowed to terminate existing coverage.
- 4. With the exception of Section 2.5.C.3.c), the Special Enrollment Period for the qualifying events listed in Section 2.5C.3 shall be the sixty (60) day period from the date of the qualifying event. In the event of a qualifying event under Section 2.5.C.3.c), the Special Enrollment Period shall be sixty (60) days prior to the end of coverage under the employer sponsored plan.

D. Effective Dates

- 1. The Effective Date for a Qualified Employee who enrolls during an Annual Open Enrollment Period shall be the first day of the new policy year, unless otherwise provided by the SHOP Exchange.
- 2. The Effective Date for a newly eligible Qualified Employee and any Dependent who timely enrolls as provided in Section 2.5B shall be as stated in the Eligibility Schedule.
- 3. The Effective Date for a Dependent Child who timely enrolls during a Special Enrollment Period is the Dependent Child's First Eligibility Date (unless otherwise provided by the SHOP Exchange).
 - a) First Eligibility Date means:

- (1) For a newborn Dependent Child, the child's date of birth;
- (2) For a non-newborn Dependent Child who is a stepchild, the date the stepchild became a Dependent of the Subscriber:
- (3) For a newly adopted Dependent Child, the earlier of;
 - (a) A judicial decree of Adoption; or
 - (b) Placement of the child in the Subscriber's home as the legally recognized proposed adoptive parent;
- (4) For a Dependent Child for whom guardianship of at least twelve (12) months' duration has been granted by court or testamentary appointment, the date of the appointment;
- (5) For a Dependent Child who has been placed with the Subscriber through a child support order (MCSO or QMSO) or other court order, the date that the court order is effective;
- (6) For all other Dependent Children, the first day of the month following the receipt of enrollment by the SHOP Exchange.
- b) The Dependent Child will be covered automatically, but only for the first thirty-one (31) days following the child's First Eligibility Date. The Subscriber must enroll such a Dependent Child within thirty-one (31) days of the child's First Eligibility Date when an additional premium is due for the enrollment of the Dependent Child. Otherwise, the Dependent Child will not be covered and cannot be enrolled until the next Annual Open Enrollment Period. (An additional premium will be due unless there are three (3) or more Dependent Children under the age of twenty-one (21) already enrolled by the Subscriber).
- 4. The Effective Date for a Qualified Employee or Dependent who gains, or becomes, an eligible Dependent through marriage and who timely enrolls during a Special Enrollment Period shall be the first day of the month following the receipt of enrollment by the SHOP Exchange.
- 5. The Effective Date for a Qualified Employee (and eligible Dependents) with whom a child was placed through a foster care program shall be the date that the child was placed with the Qualified Employee (and eligible Dependents) in foster care (unless otherwise provided by the SHOP Exchange). Note that foster children are not eligible for coverage.
- 6. The Effective Date for a Qualified Employee or Dependent who loses other coverage, or who gains access to new Qualified Heath plans due to a permanent move, who timely enrolls during a Special Enrollment Period shall be the first day of the month following the receipt of enrollment by the SHOP Exchange, or as stated in Section 2.5D.8. below, at the option of the SHOP Exchange.

- 7. The Effective Date for a Qualified Employee or Dependent who timely enrolls due to a qualifying event stated in Section 2.5C.1.c) (enrollment or non-enrollment was unintentional, inadvertent, or erroneous and is the result of an error by the SHOP Exchange or the United States Department of Health and Human Services), Section 2.5C.1.d) (a Qualified Health Plan substantially violated a material provision of its contract), Section 2.5C.1.g) (other exceptional circumstances as determined by the SHOP Exchange), , shall be the appropriate date based on the circumstances of the Special Enrollment Period as determined by the Exchange.
- 8. If an enrollee or his or her Dependent dies as stated in 2.5C.1.a)(5), the Effective Date of coverage is the first day of the month following plan selection or provided in Section 2.5D.6.b) of this provision as determined by the SHOP Exchange.
- 9. In all other cases, the Effective Date for a Qualified Employee or Dependent who timely enrolls during a Special Enrollment Period will be:
 - a) For enrollment received by the SHOP Exchange between the first (1st) and the fifteenth (15th) day of the month, the first day of the following month; and
 - b) For enrollment received by the SHOP Exchange between the sixteenth (16th) and the last day of the month, the first day of the second following month.
- 10. Premium changes resulting from the enrollment of a Subscriber or a Dependent during a Special Enrollment Period will be effective as of the Effective Date of the Subscriber's or the Dependent's enrollment.

2.6 Child Support Orders (MCSO or QMSO).

A. <u>Eligibility and Termination</u>.

- 1. Upon receipt of an MCSO or QMSO, CareFirst will accept enrollment of a Dependent Child that is the subject of an MCSO or QMSO and the Qualified Employee parent of such child, without regard to enrollment period restrictions, within the time period prescribed by law. Coverage will be effective as of the effective date of the order, and the premium will be adjusted as needed. If the Subscriber does not enroll the child then CareFirst will accept enrollment from the non-Subscriber custodial parent; or, the appropriate child support enforcement agency of any state or the District of Columbia. If the Subscriber is subject to a Waiting Period, the child will not be enrolled until the end of the waiting period.
- 2. Enrollment for such a child will not be denied because the child:
 - a) Was born out of wedlock;
 - b) Is not claimed as a dependent on the Subscriber's federal tax return;
 - c) Does not reside with the Subscriber; or
 - d) Is covered under any Medical Assistance or Medicaid program.

- 3. Coverage required by an MCSO or QMSO will be effective as of the date of the order.
- 4. Termination. Unless coverage is terminated for non-payment of the premium, a covered child subject to an MCSO or QMSO may not be terminated unless written evidence is provided to CareFirst stating:
 - a) The MCSO or QMSO is no longer in effect;
 - b) The child has been or will be enrolled under other comparable health insurance coverage that will take effect not later than the effective date of the termination of coverage;
 - c) The Group has eliminated family members' coverage for all its employees; or
 - d) The employer no longer employs the insuring parent, except if the parent elects to exercise the provisions of the Consolidated Omnibus Budget Reconciliation Act of 1985 (COBRA), coverage shall be provided for the child consistent with the Group's plan for postemployment health insurance coverage for Dependents.
- B. <u>Administration</u>. When the child subject to an MCSO or QMSO does not reside with the Subscriber, CareFirst will:
 - 1. Send to the non-insuring custodial parent the identification cards, claim forms, the applicable Out-of-Network Evidence of Coverage, and any information needed to obtain benefits;
 - 2. Allow the non-insuring custodial parent or a provider of a Covered Service to submit a claim without the approval of the Subscriber; and
 - 3. Provide benefits directly to:
 - a) The non-insuring parent;
 - b) The provider of the Covered Services, Covered Dental Services or Covered Vision Services; or
 - c) The appropriate child support enforcement agency of any state or the District of Columbia.
- 2.7 <u>Clerical or Administrative Error</u>. If an individual is ineligible for coverage, the individual cannot become eligible just because CareFirst, the Group or the SHOP Exchange made a clerical or administrative error in recording or reporting information. Likewise, if a Member is eligible for coverage, the Member will not lose coverage because CareFirst, the Group or the SHOP Exchange made an administrative or clerical error in recording or reporting information.
- 2.8 <u>Cooperation and Submission of Information</u>. CareFirst may require verification from the Group and/or Subscriber pertaining to the eligibility of a Subscriber or Dependent enrolled hereunder. The Group and/or Subscriber agree to cooperate with and assist CareFirst and the SHOP Exchange, including providing CareFirst and the SHOP Exchange with reasonable access to Group records upon request. At any time coverage under this Out-of-Network Evidence of Coverage is in effect, CareFirst reserves the right to request documentation substantiating eligibility as described in this Out-of-Network Evidence of Coverage and to provide any information it receives regarding a Member's eligibility to the Group or the SHOP Exchange.
- 2.9 <u>Proof of Eligibility</u>. CareFirst retains the right to require proof of relationships or facts to establish



SECTION 3 TERMINATION OF COVERAGE

- 3.1 <u>Termination of Enrollment by the Subscriber</u>.
 - A. In the manner permitted by the SHOP Exchange, the Subscriber may terminate his or her enrollment, or (except as provided in Section 3.5) the enrollment of a Dependent, during the Annual Open Enrollment Period by notifying the Group or the SHOP Exchange. Upon termination of the Subscriber's enrollment, the enrollment of all enrolled Dependents will also be terminated.
 - B. Termination of Enrollment by the Subscriber Due to Qualifying Events.
 - 1. If certain life events occur, a Subscriber may be able to make a mid-year change to reduce and/or terminate the coverage of the Subscriber or Dependent. The following is a list of qualifying life events that allow the Subscriber to reduce or terminate coverage. The changes in coverage must satisfy the consistency requirements as described below.
 - a) Qualifying Life Events:
 - (1) Legal marital status. A change in a Member's legal marital status, including marriage, divorce, death of spouse, a legal separation or an annulment.
 - (2) Employment status. A change in a Subscriber's, Spouse's or Dependent's employment status due to termination or commencement of employment, a strike or lockout, an unpaid leave of absence, or a change in worksite.
 - (3) Dependent status. A change in status of a Dependent that results in the Dependent's eligibility or ineligibility for coverage because of age or similar circumstances.
 - (4) Any of the qualifying events listed in Section 2.5C of the Out-of-Network Evidence of Coverage, not otherwise specified in this provision.
 - (5) Any reduction or termination that a Subscriber makes must be consistent with the life event. The life event must affect eligibility for coverage under the plan or under a plan of the spouse or Dependent, which covers the spouse or Dependent as a Subscriber. The change in coverage must correspond with the life event.
 - 2. Under certain circumstances, a Subscriber may make mid-year reduction or termination to coverage for reasons, such as coverage cost or Medicare eligibility as described below.
 - a) Coverage Events:
 - (1) If there is reduction or elimination of coverage during the Benefit Period.
 - (2) If the Spouse has a cafeteria plan which allows a Subscriber and Dependents to make an enrollment change during that plan's

- annual open enrollment period, the Subscriber may make a corresponding mid-year change.
- b) Cost Events: If the cost of coverage increases or decreases significantly during a Benefit Period (including a Subscriber's change from part-time to full-time work or vice versa) and the Group does not offer a similar, but less costly, coverage option.
- c) Entitlement to Medicaid, the federal child health insurance plan (CHIP) or a State-funded low-income basic health plan (known as a BHP). If a Subscriber or Dependent becomes eligible for Medicaid, the federal child health insurance plan (CHIP) or a State-funded low-income basic health plan (known as a BHP) mid-year, a Subscriber or Dependent *may* (but is not required) terminate coverage.
- C. The effective date of the termination will be as follows:
 - 1. On the date stated by the Subscriber, if the Subscriber has given reasonable notice. For purposes of this provision, reasonable notice is defined as fourteen (14) days from the requested termination date.
 - 2. Fourteen (14) days after the date the Subscriber requested termination if the Subscriber does not provide reasonable notice.
 - 3. If the Subscriber and Dependents give notice of termination of enrollment in order to enroll in another Qualified Health Plan, the day before the effective date of coverage under the new Qualified Health Plan.
 - 4. If the Subscriber and Dependents are newly eligible for Medicaid, the federal child health insurance plan (CHIP) or a State-funded low-income basic health plan (known as a BHP), the day before coverage under one of these programs begins.
 - 5. If the Subscriber terminates enrollment during an Annual Open Enrollment Period, on the last day of the Benefit Period.
- 3.2 <u>Termination of Individual Enrollment by the CareFirst or the SHOP Exchange</u>. CareFirst or the SHOP Exchange may terminate the coverage of a Subscriber and/or a Dependent under the following circumstances by providing the Subscriber at least thirty (30) days' notice prior to the last day of coverage:
 - A. Termination of Individual Enrollment for Ineligibility:
 - 1. The Subscriber is no longer eligible for coverage for any reason. In this circumstance, the enrollment of the Subscriber and any Dependents will be terminated.
 - 2. A Dependent is no longer eligible for coverage as a Dependent due to a change in the Dependent's age, status or relationship to the Subscriber, or no longer meets the eligibility requirements established by the Group.
 - 3. The date of termination is stated in the Eligibility Schedule.
 - 4. The Subscriber is responsible for notifying the SHOP Exchange of any changes in the status of a Dependent as an eligible individual; or his or her eligibility for coverage, except when the Dependent Child reaches the Limiting Age. These changes include a divorce and the marriage of a Dependent Child. If the Subscriber knows of a Dependent's ineligibility for coverage and intentionally

fails to notify the SHOP Exchange of these types of changes, CareFirst has the right to seek Rescission of the coverage of the Dependent under Section 3.3 as of the initial date of the Dependent's ineligibility and recover the full value of the services and benefits provided during the period of Dependent's ineligibility. CareFirst can recover these amounts from the Subscriber and/or from the Dependent, less the premium paid during the period of ineligibility, at the option of CareFirst.

- B. Termination of Out-of-Network Group Contract due to the Decertification of the Out-of-Network Evidence of Coverage as a Qualified Health Plan. If this Out-of-Network Evidence of Coverage is Decertified as a Qualified Health Plan, the date of termination shall be the date established by the SHOP Exchange after written notice has been provided to the Subscriber and the Subscriber has been afforded an opportunity to enroll in other coverage.
- C. <u>Accommodation for Persons with Disabilities</u>. Notwithstanding the termination provisions above, CareFirst, when required by the SHOP Exchange, shall make reasonable accommodation of these provisions for all individuals with disabilities (as defined by the Americans with Disabilities Act) before terminating the coverage of such individuals.
- 3.3 <u>Rescission of Individual Enrollment for Fraud or Misrepresentation</u>. Coverage of a Member may be Rescinded if:
 - A. The Member has performed an act, practice, or omission that constitutes fraud;
 - B. The Member has made an intentional misrepresentation of material fact; or
 - C. An act, practice or omission that constitutes fraud includes, but is not limited to, fraudulent use of CareFirst's identification card by the Member, the alteration or sale of prescriptions by the Member, or an attempt by a Subscriber to enroll non-eligible persons.

CareFirst will provide thirty (30) days advance written notice of any Rescission. CareFirst shall have the burden of persuasion that its Rescission complies with applicable local law. The Rescission shall either (i) void the enrollment of the Member as of the Member's Effective Date (for fraudulent acts, practices, or omissions that occur at the time of enrollment); or (ii) in all other cases, void the enrollment of the Member as of the first date the Member performed an act, practice or omission which constituted fraud or made an intentional misrepresentation of material fact. The Subscriber will be responsible for payment of any voided benefits paid by CareFirst, net of applicable premiums paid.

- 3.4 <u>Death of Subscriber</u>. In the event of the Subscriber's death, coverage of any Dependents may continue under the Subscriber's enrollment as stated in this Out-of-Network Evidence of Coverage. The date of termination of the Subscriber's enrollment and, if applicable, the enrollment of any Dependents will be as stated in the Eligibility Schedule.
- 3.5 <u>Medical Child Support Orders or Qualified Medical Support Orders.</u> Unless coverage is Rescinded or terminated for non-payment of the premium, a child subject to an MCSO/QMSO may not be terminated unless written evidence is provided to CareFirst or the SHOP Exchange that:
 - A. The MCSO/QMSO is no longer in effect;
 - B. The child has been or will be enrolled under other comparable health insurance coverage that will take effect not later than the date of the termination of coverage;
 - C. The Group has eliminated family member coverage for all Members; or

- D. The Group no longer employs the Subscriber, except if the Subscriber elects continuation coverage under applicable state or federal law the child will continue in this postemployment coverage.
- 3.6 <u>Termination of Out-of-Network Evidence of Coverage upon Termination of Out-of-Network Group Contract</u>. This Out-of-Network Evidence of Coverage, and the enrollment of the Member(s), will terminate automatically upon the date of the termination of the Out-of-Network Group Contract by the Group, the SHOP Exchange or CareFirst for any reason.
- 3.7 <u>Effect of Termination</u>. No benefits will be provided for any services received on or after the date on which the Member's coverage terminates. This includes services received for an injury or illness that occurred before the date of termination.
- 3.8 <u>No Reinstatement</u>. Upon termination, enrollment will not reinstate automatically under any circumstances.

SECTION 4 CONTINUATION OF COVERAGE

- 4.1 <u>Continuation of Eligibility upon Loss of Group Coverage.</u>
 - A. Federal Continuation of Coverage under COBRA. If the Group health benefit plan provided under this Out-of-Network Evidence of Coverage is subject to the Consolidated Omnibus Budget Reconciliation Act of 1985 (COBRA), as amended from time to time, and a Member's coverage terminates due to a "Qualifying Event" as described under COBRA, continuation of participation in this Group health benefit plan may be possible. The employer offering this Group health benefit plan is the plan administrator. It is the plan administrator's responsibility to notify a Member concerning terms, conditions and rights under COBRA. If a Member has any questions regarding COBRA, the Member should contact the plan administrator.
 - B. <u>Uniformed Services Employment and Reemployment Rights Act</u> (USERRA). USERRA protects the job rights of individuals who voluntarily or involuntarily leave employment positions to undertake military service or certain types of service in the Natural Disaster Medical System. USERRA also prohibits employers, and insurers, from discriminating against past and present members of the uniformed services, and applicants to the uniformed services.

If an eligible employee leaves his or her job to perform military service, the eligible employee has the right to elect to continue his or her Group coverage including any Dependents for up to twenty-four (24) months while in the military. Even if continuation of coverage was not elected during the eligible employee's military service, the eligible employee has the right to be reinstated in their Group coverage when re-employed, without any Waiting Periods or pre-existing condition exclusion periods except for service connected illnesses or injuries. If an eligible employee has any questions regarding USERRA, the eligible employee should contact the plan administrator. The plan administrator determines eligible employees and provides the information to CareFirst.

- 4.2. <u>District of Columbia Continuation of Health Coverage (DCCHC)</u>. This provision applies to Subscribers enrolled in an employer-maintained health benefit plan for less than twenty (20) employees.
 - A. The Subscriber and any Dependents have the right to continue coverage under the Outof-Network Group Contract for a period of three (3) months, or for the period of time during which the Subscriber is eligible for premium assistance under the American Recovery and Reinvestment Act of 2009, as amended, unless:
 - 1. The Subscriber's employment was terminated for gross misconduct;
 - 2. The Member is eligible for any extension of coverage required under COBRA; or
 - The Member fails to complete timely election and payment as provided below.
 - B. <u>Duties of the Group.</u>
 - 1. The Group shall furnish Subscribers whose coverage terminates with written notification of the Subscriber's eligibility to continue coverage under DCCHC. Such notice shall be furnished no later than fifteen (15) days of the date coverage under this Out-of-Network Evidence of Coverage would otherwise terminate. Failure by the Group to furnish the required notification shall not extend the right to continue coverage beyond the three-month period, or for the period of time during which the Subscriber is eligible for premium assistance under the American Recovery and Reinvestment Act of 2009, as amended.

2. The Group shall forward to CareFirst the names of Members who apply for DCCHC Continuation of Coverage within fifteen (15) days from the date of application.

C. <u>Duties of the Subscriber</u>.

- 1. Individuals who elect coverage under this Section shall bear the cost of the continued coverage and such cost shall not exceed one hundred two percent (102%) of the Group's rate.
- 2. An individual who elects to continue coverage shall tender to the Group the amount described above within forty-five (45) days from the date coverage under this Out-of-Network Evidence of Coverage would otherwise terminate.
- D. <u>Termination of Continued Coverage</u>. Coverage under this provision shall continue without interruption for the continued eligibility period and shall not terminate unless:
 - 1. The Member establishes residence outside CareFirst Service Area;
 - 2. The Member fails to make timely payment of the required cost of coverage;
 - 3. The Member violates a material condition of this Out-of-Network Evidence of Coverage;
 - 4. The Member becomes covered under another group health benefits plan that does not contain any exclusion or limitation with respect to any preexisting condition that affects the Member;
 - 5. The Member becomes entitled to Medicare; or
 - 6. The Group no longer offers group coverage to any employee.
- 4.3 Right to Continue Coverage Under Only One Provision. If a Member is eligible to continue coverage under the Out-of-Network Group Contract under more than one continuation provision as described above, the Member will receive only one such continuation coverage. The Group will select the continuation option the member will receive.

SECTION 5 COORDINATION OF BENEFITS (COB); SUBROGATION

5.1 <u>Coordination of Benefits (COB)</u>.

A. Applicability.

- 1. This Coordination of Benefits (COB) provision applies to this CareFirst Plan when a Member has health care coverage under more than one Plan.
- 2. If this COB provision applies, the Order of Benefit Determination Rules should be reviewed first. Those rules determine whether the benefits of this CareFirst Plan are determined before or after those of another Plan. The benefits of this CareFirst Plan:
 - a) Shall not be coordinated when, under the order of determination rules, this CareFirst Plan determines its benefits before another Plan;
 - b) May be coordinated when, under the order of determination rules, another Plan determines its benefits first. The coordination is explained in Section 5.1D.2.
- B. <u>Definitions</u>. For the purpose of this COB section, the following terms are defined. The definitions of other capitalized terms are found in the definitions section of this Out-of-Network Evidence of Coverage.

Allowable Expenses means any health care expense, including deductibles, coinsurance or copayments that are covered in whole or in part by any of the Plans covering the Member. This means any expense or portion of an expense not covered by any of the Plans is not an Allowable Expense. If this CareFirst Plan is advised by a Member that all Plans covering the Member are high-deductible health plans and the Member intends to contribute to a health savings account, the primary Plan's deductible is not an Allowable Expense, except for any health care expense incurred that may not be subject to the deductible, as stated in Section 223(c)(2)(C) of the Internal Revenue Code of 1986.

CareFirst Plan means this Out-of-Network Evidence of Coverage.

<u>Dental Plan</u> means any dental insurance policy, including those of nonprofit health service plans, and those of commercial group, blanket, and individual policies, any subscriber contracts issued by Health Maintenance Organizations (HMOs), and any other established programs under which the insured may make a claim. The term Dental Plan includes coverage under a governmental plan, or coverage required to be provided by law. This does not include a State plan under Medicaid (Title XIX, Grants to States for Medical Assistance Programs, of the United States Social Security Act, as amended from time to time.)

<u>Intensive Care Policy</u> means a health insurance policy that provides benefits only when treatment is received in a specifically designated health care facility of a hospital that provides the highest level of care and which is restricted to those patients who are physically, critically ill or injured.

<u>Plan</u> means any health insurance policy issued on a group basis, including those of a nonprofit health service plan, those of a commercial, group, and blanket policy, any subscriber contracts issued by health maintenance organizations, and any other established programs under which the insured may make a claim. The term Plan includes coverage required or provided by law and coverage under a governmental plan, except a governmental plan which, by law, provides benefits in excess of those of any private insurance plan or other non-governmental plan. This does not include a state plan under

Medicaid (Title XIX, Grants to States for Medical Assistance Programs, of the United States Social Security Act, as amended from time to time).

The term Plan does not include:

- 1. An individually underwritten and issued, guaranteed renewable, specified disease policy;
- 2. An intensive care policy, which does not provide benefits on an expense incurred basis;
- 3. Coverage regulated by a motor vehicle reparation law;
- 4. The first one-hundred dollars (\$100) per day of a hospital indemnity contract; or
- 5. An elementary and/or secondary school insurance program sponsored by a school or school system.

<u>Primary Plan or Secondary Plan</u> means the order of benefit determination rules stating whether this CareFirst Plan is a Primary Plan or Secondary Plan as to another Plan covering the Member.

- 1. When this CareFirst Plan is a Primary Plan, its benefits are determined before those of the other Plan and without considering the other Plan's benefits.
- 2. When this CareFirst Plan is a Secondary Plan, its benefits are determined after those of the other Plan and may be coordinated because of the other Plan's benefits.
- 3. When there are more than two Plans covering the Member, this CareFirst Plan may be a Primary Plan as to one of the other Plans, and may be a Secondary Plan as to a different Plan or Plans.

<u>Specified Disease Policy</u> means a health insurance policy that provides (1) benefits only for a disease specified in the policy or for the treatment unique to a specific disease; or (2) additional benefits for a disease specified in the policy or for treatment unique to a specified disease.

C. Order of Benefit Determination Rules.

- 1. General. When there is a basis for a claim under this CareFirst Plan and another Plan, this CareFirst Plan is a Secondary Plan which has its benefits determined after those of the other Plan, unless;
 - a) The other Plan has rules coordinating benefits with those of this CareFirst Plan; and
 - b) Both those rules and this CareFirst Plan's rules require this CareFirst Plan's benefits be determined before those of the other Plan.
- 2. Rules. This CareFirst Plan determines its order of benefits using the first of the following rules which applies:
 - a) Non-dependent/dependent. The benefits of the Plan which covers the person as an employee, member or subscriber (that is, other than as a dependent) are determined before those of the Plan which covers the person as a dependent; except if the person is also a Medicare

beneficiary, and the result of the rule established by Title XVIII of the Social Security Act and implementing regulations, Medicare is:

- (1) Secondary to the Plan covering the person as a dependent, and
- (2) Primary to the Plan covering the person as other than a dependent (e.g., retired employee),

Then the benefits of the Plan covering the person as a dependent are determined before those of the Plan covering the person as other than a dependent.

- b) Dependent child covered by more than one Plan. Unless there is a court decree stating otherwise, when this CareFirst Plan and another Plan cover the same child as a dependent, the order of benefits shall be determined as follows:
 - (1) For a dependent child whose parents are married or are living together:
 - (a) The benefits of the Plan of the parent whose birthday falls earlier in a year are determined before those of the Plan of the parent whose birthday falls later in the year; but
 - (b) If both parents have the same birthday, the benefits of the Plan that covered one parent longer are determined before those of the Plan that covered the other parent for a shorter period of time.
 - (2) For a dependent child whose parents are separated, divorced, or are not living together:
 - (a) If the specific terms of a court decree state one of the parents is responsible for the health care expenses or health care coverage of the child, and the entity obligated to pay or provide the benefits of the Plan of that parent has actual knowledge of those terms, the benefits of that Plan are determined first. If the parent with responsibility has no health care coverage for the dependent child's health care expenses, but the parent's Spouse does, that parent's Spouse's plan is the primary plan. This paragraph does not apply with respect to any claim for services rendered before the entity has actual knowledge of the terms of the court decree.

The rule described in (1) above also shall apply if: i) a court decree states both parents are responsible for the dependent child's health care expenses or health care coverage, or ii) a court decree states the parents have joint custody without specifying one parent has responsibility for the health care expenses or coverage of the dependent child.

(b) If there is no court decree setting out the responsibility for the child's health care expenses or health care coverage, the order of benefits for the dependent child are as follows:

- i) The Plan of the parent with custody of the child;
- ii) The Plan of the Spouse of the parent with the custody of the child;
- iii) The Plan of the parent not having custody of the child; and then
- iv) The Plan of the Spouse of the parent who does not have custody of the child.
- (3) For a dependent child covered under more than one plan of individuals who are not the parents of the child, the order of benefits shall be determined, as applicable, under the rules stated in (1) and (2) of this paragraph as if those individuals were parents of the child.
- c) Active/inactive employee. The benefit of a Plan which covers a person as an employee who is neither laid off nor retired is determined before those of a Plan that covers that person as a laid off or retired employee. The same would hold true if a person is a dependent of a person covered as an employee who is neither laid off nor retired or a person covered as a laid off or retired employee. If the other Plan does not have this rule, and if, as a result, the Plans do not agree on the order of benefits, this rule is ignored.
- d) Continuation coverage. If a person whose coverage is provided under the right of continuation pursuant to federal, state or local law also is covered under another Plan, the following shall be the order of benefits determination:
 - (1) First, the benefits of a Plan covering the person as an employee, retiree, member or subscriber (or as that person's dependent);
 - (2) Second, the benefits under the continuation coverage.

If the other Plan does not have the rule described above, and if, as a result, the Plans do not agree on the order of benefits, this rule is ignored.

- e) Longer/shorter length of coverage. If none of the above rules determines the order of benefits, the benefits of the Plan that covered the person longer are determined before those of the Plan that covered that person for the shorter term.
- f) Medical and Dental Plan. When a Member has coverage under two Plans, and one of the Plans is a medical plan and the other is a Dental Plan, and a determination cannot be made in accordance with the order of benefit determination rules in this CareFirst Plan, the medical plan should be considered as the Primary Plan.
- D. <u>Effect on the Benefits of this CareFirst Plan.</u>
 - 1. When this Section Applies. This section applies when, in accordance with the prior section, order of benefits determination rules, this CareFirst Plan is a Secondary Plan as to one or more other Plans. In such an event, the benefits of this CareFirst Plan may be coordinated under this section. Any additional other Plan or Plans are referred to as "the other Plans" immediately below.

- 2. Coordination in this CareFirst Plan's Benefits. When this CareFirst Plan is the Secondary Plan, the benefits under this CareFirst Plan *may* be coordinated so that the total benefits would be payable or provided by all the other Plans do not exceed one hundred percent (100%) of the total Allowable Expenses. If the benefits of this CareFirst Plan are coordinated, each benefit is coordinated in proportion. It is then charged against any applicable benefit limit of this CareFirst Plan.
- E. <u>Right to Receive and Release Needed Information</u>. Certain facts are needed to apply these COB rules. CareFirst has the right to decide which facts it needs. It may get the needed facts from or give them to any other organization or person for purposes of treatment, payment, and health care operations. CareFirst need not tell, or get the consent of, any person to do this. Each person claiming benefits under this CareFirst Plan must give this CareFirst Plan any facts it needs to pay the claim.
- F. <u>Facility of Payment</u>. A payment made under another Plan may include an amount that should have been paid under this CareFirst Plan. If it does, this CareFirst Plan may pay the amount to the organization that made that payment. The amount will then be treated as though it were a benefit paid under this CareFirst Plan. This CareFirst Plan will not have to pay the amount again. The term "payment made" includes providing benefits in the form of services, in which case "payment made" means the reasonable cash value of the benefits provided in the form of services.
- G. <u>Right of Recovery</u>. If the amount of the payments made by this CareFirst is more than it should have paid under this COB provision, it may recover the excess from one or more of:
 - 1. The persons it has paid or for whom it has paid;
 - 2. Insurance companies; or
 - 3. Other organizations.

The "amount of the payments made" includes the reasonable cash value of any benefits provided in the form of services.

- 5.2 <u>Medicare Eligibility</u>. This provision applies to Members who are entitled to Part A and/or Part B of Medicare. A Member will not be terminated as a result of reaching the age of sixty-five (65) or becoming eligible for Medicare. Benefits not covered by Medicare will be provided as described in the Out-of-Network Evidence of Coverage. Benefits covered by Medicare are subject to the provisions in this section.
 - A. <u>Coverage Secondary to Medicare</u>. Except where prohibited by law, the benefits under this CareFirst Plan are secondary to Medicare.
 - B. Medicare as Primary.
 - 1. When benefits for Covered Services, Covered Dental Services or Covered Vision Services are paid by Medicare as primary, this CareFirst Plan will not duplicate those payments. CareFirst will coordinate and pay benefits based on Medicare's payment (or the payment Medicare would have paid). When CareFirst coordinates the benefits with Medicare, CareFirst's payments will be based on the Medicare allowance (if the provider is a participating provider in Medicare) or the Medicare maximum limiting charge (if the provider is not a participating provider in Medicare), less any claim reduction or denial due to a Member's failure to comply with Medicare's administrative requirements. CareFirst's right to coordinate is not contingent on any payment actually being made on the claim by Medicare. Members enrolled in Medicare agree to, and shall, complete and

- submit to Medicare, CareFirst, and/or any health care providers all claims, consents, releases, assignments and other documents required to obtain or assure such claim payment by Medicare.
- 2. If a Medicare-eligible Member has not enrolled in Medicare Part A and/or Part B, CareFirst will not "carve-out," coordinate, or reject a claim based on the amount Medicare would have paid had the Member actually applied for, claimed, or received Medicare benefits.
- 5.3 <u>Employer or Governmental Benefits</u>. Coverage does not include the cost of services or payment for services for any illness, injury, or condition for which, or as a result of which, a Benefit (as defined below) is provided or is required to be provided either:
 - A. Under any federal, state, county or municipal workers' compensation or employer's liability law or other similar program; or
 - B. From any federal, state, county or municipal or other government agency, including, in the case of service-connected disabilities, the United States Department of Veterans Affairs, to the extent that benefits are payable by the federal, state, county or municipal or other government agency, but excluding Medicare benefits and Medicaid benefits.

<u>Benefit</u> as used in this provision includes a payment or any other benefit, including amounts received in settlement of a claim for Benefits.

- 5.4 <u>Determination of Benefits under Point-of-Service</u>. The following terms and conditions will govern the applicability of benefits between this Out-of-Network Evidence of Coverage and the In-Network Evidence of Coverage:
 - A. Services and supplies which do not meet the conditions for coverage under the In-Network Evidence of Coverage will be considered for coverage Out-of-Network subject to the terms and conditions of this Out-of-Network Evidence of Coverage.
 - B. Benefits for a given service or supply may be provided either In-Network or Out-of-Network. Benefits will not be provided for the same service or supply under both this Out-of-Network Evidence of Coverage and the In-Network Evidence of Coverage.
 - C. Refer to the Schedule of Benefits to determine if the In-Network and Out-of-Network Deductible and Out-of-Pocket Maximum amounts contribute to one another.
- 5.5 <u>Subrogation</u>. CareFirst has subrogation and reimbursement rights. Subrogation requires the Member to turn over to CareFirst any rights the Member may have against a third party. A third party is any person, corporation, insurer or other entity that may be liable to a Member for an injury or illness. This right applies to the amount of benefits paid by CareFirst for injuries or illnesses where a third party could be liable.

Recovery means to be successful in a lawsuit, to collect or obtain an amount; to obtain a favorable or final judgment; to obtain an amount in any legal manner; an amount finally collected or the amount of judgment as a result of an action brought against a third-party or involving uninsured or underinsured motorist claims. A Recovery does not include payments made to the Member under the Member's personal injury protection policy. CareFirst will not recover medical expenses from a Subscriber unless the Subscriber or Member recovers for medical expenses in a cause of action or settlement.

A. The Member shall notify CareFirst as soon as reasonably possible that a third-party may be liable for the injuries or illnesses for which benefits are being provided or paid.

- B. To the extent actual payments made by CareFirst result from the occurrence that gave rise to the cause of action, CareFirst shall be subrogated and succeed to any right of recovery of the Member against any person or organization.
- C. The Member shall pay CareFirst the amount recovered by suit, settlement, or otherwise from any third-party's insurer, any uninsured or underinsured motorist coverage, or as permitted by law, to the extent any actual payments made by CareFirst result from the occurrence that gave rise to the cause of action.
- D. The Member shall furnish information and assistance, and execute papers that CareFirst may require to facilitate enforcement of these rights. The Member shall not commit any action prejudicing the rights and interests of CareFirst.
- E. In a subrogation claim arising out of a claim for personal injury, the amount recovered by CareFirst may be reduced by:
 - 1. Dividing the total amount of the personal injury recovery into the total amount of the attorney's fees incurred by the injured person for services rendered in connection with the injured person's claim; and
 - 2. Multiplying the result by the amount of CareFirst's subrogation claim. This percentage may not exceed one-third (1/3) of CareFirst's subrogation claim.
- F. On written request by CareFirst, a Member or Member's attorney who demands a reduction of the subrogation claim shall provide CareFirst with a certification by the Member that states the amount of the attorney's fees incurred.
- G. These provisions do not apply to residents of the Commonwealth of Virginia.

SECTION 6 GENERAL PROVISIONS

6.1 <u>Entire Certificate; Changes.</u> The entire Out-of-Network Evidence of Coverage includes: (a) this Out-of-Network Evidence of Coverage; (b) Benefit Determinations and Appeals; (c) the Description of Covered Services; (d) Schedule of Benefits; (e) Eligibility Schedule; and (f) any additional duly authorized notices, amendments and riders.

No amendment or modification of any term or provision of this Out-of-Network Evidence of Coverage is effective unless authorized in writing by an executive officer of CareFirst. Any duly authorized notice, amendment or rider will be issued by CareFirst to be attached to the Out-of-Network Evidence of Coverage. No agent has authority to change this agreement or to waive any of its provisions. Any waiver of an Out-of-Network Evidence of Coverage term or provision shall only be given effect for its stated purpose and shall not constitute or imply any subsequent waiver.

Oral statements cannot be relied upon to modify or otherwise affect the benefits, limitations and/or exclusions of this Out-of-Network Evidence of Coverage, or increase or void any coverage or reduce any benefits. Such oral statements cannot be used in the prosecution or defense of a claim.

6.2 Claims and Payment of Claims.

A. <u>Claim Forms</u>. CareFirst does not require a written notice of claims. A claim form can be requested by calling the Member and Provider Service telephone number on the identification card during regular business hours. CareFirst shall provide claim forms for filing proof of loss to each claimant or to the Group for delivery to the claimant. If CareFirst does not provide the claim forms within fifteen (15) days after notice of claim is received, the claimant is deemed to have complied with the requirements of the policy as to proof of loss if the claimant submits, within the time fixed in the policy for filing proof of loss, written proof of the occurrence, character, and extent of the loss for which the claim is made.

When a child subject to a Medical Child Support Order or a Qualified Medical Support Order does not reside with the Subscriber, CareFirst will:

- 1. Send the non-insuring, custodial parent identification cards, claims forms, the applicable certificate of coverage or member contract, and any information needed to obtain benefits;
- 2. Allow the non-insuring, custodial parent or a provider of a Covered Service to submit a claim without the approval of the Subscriber; and
- 3. Provide benefits directly to:
 - a) The non-insuring, custodial parent;
 - b) The provider of the Covered Services, Covered Dental Services, or Covered Vision Services; or
 - c) The appropriate child support enforcement agency of any state or the District of Columbia.

B. Proof of Loss.

For Covered Services provided by Participating Providers, Participating Dentists, Contracting Vision Providers, and Contracting Pharmacies, Members are not required to submit claims in order to obtain benefits. For Covered Services provided by Non-Participating Providers, Non-Participating Dentists, Non-Contracting Vision Providers, and Non-Contracting Pharmacies, Members must furnish written proof of loss, or have the provider submit proof of loss, to CareFirst within one (1) year after the date of the loss.

Failure to furnish proof within the time required shall not invalidate or reduce any claim if it was not reasonably possible to give proof within the required time, provided proof is furnished as soon as reasonably possible and in no event, except in the absence of legal capacity, later than one (1) year from the time proof is otherwise required.

CareFirst will honor claims submitted for Covered Services, Covered Dental Services, or Covered Vision Services by any agency of the federal, state, or local government that has the statutory authority to submit claims beyond the time limits established under this Out-of-Network Evidence of Coverage. These claims must be submitted to CareFirst before the filing deadline established by the applicable statute on claims forms that provide all of the information CareFirst deems necessary to process the claims. CareFirst provides forms for this purpose.

- C. <u>Time of Payment of Claims</u>. Benefits payable will be paid immediately after receipt of written proof of loss.
- D. <u>Claim Payments Made in Error</u>. If CareFirst makes a claim payment to or on behalf of a Member in error, the Member is required to repay CareFirst the amount paid in error. If the Member has not repaid the full amount owed CareFirst and CareFirst makes a subsequent benefit payment, CareFirst may subtract the amount owed CareFirst from the subsequent payment.
- E. <u>Payment of Claims</u>. Payments for Covered Services will be made by CareFirst directly to Contracting Vision Providers, Participating Dentists and Participating Providers. Direct payments will also be made by CareFirst to providers from the United States Department of Defense and the United States Department of Veteran Affairs. If a Member receives Covered Services from Non-Contracting Vision or Non-Preferred or Non-Participating Providers, CareFirst reserves the right to pay either the Member or the provider. If the Member has paid the health care provider for services rendered, benefits will be payable to the Member. The payment will, in either case, be full and complete satisfaction of CareFirst's obligation, unless an appeal or grievance has been filed by, or on behalf of, the Member.
- 6.3 No Assignment. A Member cannot assign any benefits or payments due under this Out-of-Network Evidence of Coverage to any person, corporation or other organization, except as specifically provided by this Out-of-Network Evidence of Coverage or required by applicable law.
- 6.4 <u>Legal Actions</u>. A Member cannot bring any lawsuit against CareFirst to recover under this Out-of-Network Evidence of Coverage before the expiration of sixty (60) days after written proof of loss has been furnished, and not after three (3) years from the date written proof of loss is required to be submitted to CareFirst.
- 6.5 Events Outside of CareFirst's Control. If CareFirst, for any reason beyond the control of CareFirst, is unable to provide the coverage promised, CareFirst is liable for reimbursement of the expenses necessarily incurred by any Member in procuring the services through other providers, to the extent prescribed by law.
- 6.6 <u>Physical Examinations and Autopsy</u>. CareFirst, at its own expense, has the right and opportunity to examine the Member when and as often as it may reasonably require during the pendency of a claim hereunder and to make an autopsy in case of death where it is not forbidden by law.
- 6.7 <u>Identification Card</u>. Any cards issued to Members are for identification only.

- A. Possession of an identification card confers no right to benefits.
- B. To be entitled to such benefits, the holder of the card must, in fact, be a Member on whose behalf all applicable premiums have actually been paid.
- C. Any person receiving benefits to which he or she is not then entitled will be liable for the actual cost of such benefits.
- 6.8 <u>Member Medical Records</u>. It may be necessary to obtain Member medical records and information from hospitals, Skilled Nursing Facilities, physicians or other practitioners who treat the Member. When a Member becomes covered, the Member (and, if the Member is legally incapable of giving such consent, the representative of such Member) automatically gives CareFirst permission to obtain and use such records and information, including medical records and information requested to assist CareFirst in determining benefits and eligibility of Members.
- Member Privacy. CareFirst shall comply with state, federal and local laws pertaining to the dissemination or distribution of non-public personally identifiable financial, medical or health related data. In that regard, CareFirst will not provide to unauthorized third parties any personally identifiable financial or medical information without the prior written authorization of the Member or parent/guardian of the Member or as otherwise permitted by law. Personal information, including email addresses and phone numbers, may be used and shared with other businesses who work with CareFirst to administer and/or provide benefits under this plan. Personal information may also be used to notify enrollees about treatment options, health-related services, and/or coverage options. Enrollees may contact CareFirst to change the information used to communicate with them.
- Relationship of CareFirst to Health Care Providers. Health care providers, including Participating Providers, Participating Dentists, Contracting Vision Providers, and Contracting Pharmacy Providers, are independent contractors or organizations and are related to CareFirst by contract only. Participating Providers, Participating Dentists, Contracting Vision Providers, and Contracting Pharmacy Providers are not employees or agents of CareFirst and are not authorized to act on behalf of or obligate CareFirst with regard to interpretation of the terms of the Out-of-Network Evidence of Coverage, including eligibility of Members for coverage or entitlement to benefits. Participating Providers, Participating Dentists, Contracting Vision Providers, and Contracting Pharmacy Providers maintain a provider-patient relationship with the Member and are solely responsible for the professional services they provide. CareFirst is not responsible for any acts or omissions, including those involving malpractice or wrongful death of Participating Providers, Participating Dentists, Contracting Vision Providers, Contracting Pharmacy Providers, or any other individual, facility or institution which provides services to Members or any employee, agent or representative of such providers.
- 6.11 <u>Provider and Services Information</u>. Listings of current Participating Providers, Participating Dentists, Contracting Vision Providers, and Contracting Pharmacy Providers will be made available to Members at the time of enrollment. Updated listings are available to Members upon request. The listing of Participating Providers, Participating Dentists and Contracting Vision Providers is updated every fifteen (15) days on the CareFirst website (www.carefirst.com).
- 6.12 <u>Administration of Out-of-Network Evidence of Coverage</u>. CareFirst may adopt reasonable policies, procedures, rules and interpretations to promote the orderly and efficient administration of this Out-of-Network Evidence of Coverage.
- 6.13 <u>CareFirst's Relationship to the Group</u>. The Group is not an agent or representative and is not liable for any acts or omissions by CareFirst or any health care provider. CareFirst is not an agent or representative of the Group and is not liable for any act or omission of the Group.

- 6.14 <u>Delivery of Out-of-Network Evidence of Coverage</u>. Unless CareFirst makes delivery directly to the Member, CareFirst will provide to the Group, for delivery to each Subscriber, a statement that summarizes the essential features of the coverage and states to whom benefits under the Out-of-Network Evidence of Coverage are payable. Only one (1) statement will be issued for each family unit, except in the instance of an eligible child who is covered due to an MCSO/QMSO. In this instance, an additional Out-of-Network Evidence of Coverage will be delivered to the custodial parent upon request.
- 6.15 Out-of-Network Evidence of Coverage Binding on Members. The Out-of-Network Evidence of Coverage can be amended, modified or terminated in accordance with any provision of the Out-of-Network Evidence of Coverage or by mutual agreement between CareFirst and the Group without the consent or concurrence of Members. By electing coverage under this Out-of-Network Evidence of Coverage, or accepting benefits under this Out-of-Network Evidence of Coverage, Members are subject to all terms, conditions, and provisions of the Out-of-Network Group Contract and Out-of-Network Evidence of Coverage.
- 6.16 Payment of Contributions. The Out-of-Network Group Contract is issued to the Group on a contributory basis in accordance with the Group's policies. The Group has agreed to collect from Members any contributory portion of the premium and pay to the SHOP Exchange the premium as specified in the Out-of-Network Group Contract for all Members.
- Retirement Income Security Act of 1974, as amended (ERISA), the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended (COBRA), the Health Insurance Portability and Accountability Act of 1996 (HIPAA), and/or the Affordable Care Act) that relates to the health benefits provided under this Out-of-Network Group Contract. For the purposes of ERISA and/or COBRA, the Group is the "plan administrator." As the plan administrator, it is the Group's responsibility to provide Members with certain information, including access to and copies of plan documents describing Member's benefits and rights to coverage under the Group health plan. Such rights include the right to continue coverage upon the occurrence of certain "Qualifying Events."
 - In any event, the Member should check with the Group to determine the Member's rights under ERISA, COBRA, HIPAA and/or the Affordable Care Act, as applicable.
- 6.18 Representations and not Warranties. All statements made by the Subscriber shall be deemed to be representations and not warranties. No statement made for the purpose of obtaining coverage shall void such coverage or reduce benefits unless contained in a written instrument signed by the Subscriber, a copy of which has been furnished to CareFirst.
- 6.19 <u>Rules for Determining Dates and Times</u>. The following rules will be used when determining dates and times:
 - A. All dates and times of day will be based on the dates and times applicable to the Washington, DC area, i.e., Eastern Standard Time or Eastern Daylight Savings Time, as applicable.
 - B. When reference is made to coverage being effective on a particular date, this means 12:01 a.m. on that date.
 - C. When reference is made to termination being effective on a particular date, this means 12:00 midnight on that date.
 - D. "Days" mean calendar days, including weekends, holidays, etc., unless otherwise noted.
 - E. "Year" refers to Calendar Year, unless a different benefit year basis is specifically stated.
- 6.20 Notices.

- A. To the Member. Notice to Members will be sent via electronic mail, if the Member has consented to receive such notices via electronic mail, or by first class mail to the most recent address or electronic address for the Member in CareFirst's files. It is the Subscriber's responsibility to notify the Group, and the Group's responsibility to notify CareFirst, of an address change. The notice will be effective on the date mailed, whether or not the Member in fact receives the notice or there is a delay in receiving the notice. The notice will be effective on the date mailed, whether or not the Member in fact receives the notice or there is a delay in receiving the notice.
- B. To CareFirst. When notice is sent to CareFirst, it must be sent by first class mail to:

Group Hospitalization and Medical Services, Inc. 840 First Street, NE Washington, DC 20065

- 1. Notice will be effective on the date of receipt by CareFirst, unless the notice is sent by registered mail, in which case the notice is effective on the date of mailing, as certified by the Postal Service.
- 2. CareFirst may change the address at which notice is to be given by giving written notice thereof to the Subscriber.
- 6.21 <u>Amendment Procedure.</u> Amendments must be consistent with federal and state law. CareFirst may amend the Evidence of Coverage with respect to any matter by the means and within the time frame allowed by the procedures established by the SHOP Exchange.
 - A. CareFirst will give notice of any amendment at least sixty (60) days before the effective date that coverage will be renewed. If a material modification required to conform the Evidence of Coverage to changes in applicable state or federal law is made at a time other than renewal and it affects the content of the summary of benefits and coverage, CareFirst will provide at least sixty (60) days advanced notice of the modification.
 - Regardless of when the amendment is received, the Evidence of Coverage is considered to be automatically amended as of the date specified in the contract amendment or the notice (if not stated in the contract amendment), unless otherwise mandated to conform with any applicable changes to state or federal law.
 - B. No agent or other person, except an officer of CareFirst, has authority to waive any conditions or restrictions of the Evidence of Coverage, or to bind CareFirst by making any promise or representation or by giving or receiving any information. No change in the Evidence of Coverage will be binding on CareFirst, unless evidenced by an amendment signed by an authorized representative of CareFirst.
- 6.22 Regulation of CareFirst. CareFirst is subject to regulation in the District of Columbia by the Department of Insurance, Securities and Banking pursuant to Title 31 of the District of Columbia Code and the District of Columbia Department of Health pursuant to Reorganization Plan No. 4 of 1996, as amended.
- 6.23 <u>Conformity to Law.</u> Any provision in this Out-of-Network Evidence of Coverage that is in conflict with the requirements of any state or federal law that applies to this Out-of-Network Evidence of Coverage is automatically changed to satisfy the minimum requirements of such law.
- 6.24 <u>Credit Monitoring.</u> CareFirst is offering credit monitoring to the Subscriber and eligible Dependents at no additional charge through services administered by Experian. Credit monitoring is available on an opt-in basis for all eligible Members during the effective Benefit Period of their CareFirst health insurance policy. Eligible Members may enroll by calling the number on the back of their ID card or visiting www.carefirst.com.

Group Hospitalization and Medical Services, Inc.

doing business as

CareFirst BlueCross BlueShield (CareFirst)

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

ATTACHMENT A

BENEFIT DETERMINATIONS AND APPEALS

This attachment contains certain terms that have a specific meaning as used herein. These terms are capitalized and defined in Section A below, and/or in the Out-of-Network Evidence of Coverage to which this document is attached.

These procedures replace all prior procedures issued by CareFirst, which afford CareFirst Members recourse pertaining to denials and reductions of claims for benefits by CareFirst.

These procedures only apply to claims for benefits. Notification required by these procedures will only be sent when a Member requests a benefit or files a claim in accordance with CareFirst procedures.

An authorized representative may act on behalf of the Member in pursuing a benefit claim or appeal of an Adverse Benefit Determination. CareFirst may require reasonable proof to determine whether an individual has been properly authorized to act on behalf of a Member. In the case of a claim involving Urgent/Emergent Care, a Health Care Provider with knowledge of a Member's medical condition is permitted to act as the authorized representative.

TABLE OF CONTENTS

- A. **DEFINITIONS**
- **B.** BENEFIT DETERMINATIONS
- C. INTERNAL GRIEVANCE PROCEDURE
- D. FAIR AND FULL REVIEW
- E. DEEMED EXHAUSTION OF INTERNAL CLAIMS AND APPEAL PROCESS
- F. EXTERNAL APPEAL PROCEDURE

A. **DEFINITIONS**

Adverse Benefit Determination means, as used in this attachment, the following:

- 1. A denial, reduction, or termination of, or a failure to provide or make payment (in whole or in part) for, a benefit, including any such denial, reduction, termination, or failure to provide or make payment that is based on a determination of a Member's eligibility to participate in this plan. An Adverse Benefit Determination includes a Rescission.
- 2. A denial, reduction, or termination of, or a failure to provide or make payment (in whole or in part) for, a benefit resulting from the application of any utilization review, as well as a failure to cover an item or service for which benefits are otherwise provided because it is determined to be Cosmetic, Experimental or Investigational, or not Medically Necessary or appropriate.

<u>Health Care Provider</u> means a physician or other health care professional licensed, accredited, or certified to perform specified health services consistent with State law.

<u>Pre-Service Claim</u> means any claim for a benefit when the receipt of the benefit, in whole or in part, is conditioned on the prior approval of the service in advance by CareFirst. These are services that must be "preauthorized" or "precertified" by CareFirst under the terms of the Member's contract.

Post Service Claim means any claim for a benefit that is not a Pre-Service Claim.

<u>Rescission</u> means, as used in this attachment, a cancellation or discontinuance of coverage that has retroactive effect, except to the extent it is attributable to a failure to pay required premiums or contributions towards the cost of coverage.

<u>Urgent/Emergent Care</u> means a Pre-Service or Concurrent Care claim for medical care or with respect to which the application of the time periods for making non-Urgent/Emergent Care determinations:

- 1. Could seriously jeopardize the life or health of the Member or the ability of the Member to regain maximum function; or,
- 2. In the opinion of a Health Care Provider with knowledge of the Member's medical condition, would subject the Member to severe pain that cannot be adequately managed without the care or treatment that is the subject of the claim.

Whether a claim involves Urgent/Emergent Care is to be determined by an individual acting on behalf of CareFirst applying the judgment of a prudent layperson who possesses an average knowledge of health and medicine. If a Health Care Provider with knowledge of the Member's medical condition determines that a claim involves Urgent/Emergent Care then CareFirst will treat the claim as one that involves Urgent/Emergent Care.

B. BENEFIT DETERMINATIONS

- 1. Request for Urgent/Emergent Care Coverage. When the Member or authorized representative requests a pre-service determination regarding Urgent/Emergent Care, then CareFirst will notify the Member or authorized representative of the benefit determination (whether adverse or not) as soon as possible, taking into account the medical exigencies, the earlier of:
 - a. 24 hours after CareFirst's receipt of the necessary information to make the benefit determination, or
 - b. 72 hours after receipt of the request for coverage.

If a Member fails to provide sufficient information for CareFirst to determine whether benefits are covered or payable, CareFirst will notify the Member as soon as possible, but not later than 24 hours after receipt of the claim, of the specific information necessary to complete the claims. The Member shall be afforded a reasonable amount of time, taking into account the circumstances, but not less than 48 hours, to provide the specified information. CareFirst will notify the Member of the benefit determination as soon as possible, but in no case later than 48 hours after the earlier of:

- a. CareFirst's receipt of the specified information, or
- b. The end of the period afforded the Member to provide the specified additional information.
- 2. <u>Pre-Service Claims</u>. In the case of a Pre-Service Claim, CareFirst shall notify the Member of the benefit determination (whether adverse or not) within a reasonable period of time appropriate to the medical circumstances, but no later than 15 days after receipt of the claim.

This period may be extended one time by CareFirst for up to 15 days, provided that such an extension is necessary due to matters beyond the control of CareFirst and CareFirst notifies the Member, prior to the expiration of the initial 15-day period, of

the circumstances requiring the extension of time and the date by which CareFirst expects to render a decision. If such an extension is necessary due to a failure of the Member to submit the information necessary to decide the claim, the notice of extension shall specifically describe the required information, and the Member will have at least 45 days from receipt of the notice within which to provide the specified information.

In the case of a failure by a Member or authorized representative to follow CareFirst procedures for filing a Pre-Service Claim, the Member or authorized representative shall be notified of the failure and the proper procedures to be followed in filing a claim for benefits. This notification shall be provided to the Member or authorized representative, as appropriate, as soon as possible, but not later than 5 working days following the failure. Notice will be sent within 24 hours in the case of a failure to file a claim involving Urgent/Emergent Care. Notification may be oral, unless written notification is requested by the Member or authorized representative.

This paragraph shall apply only in the case of a communication:

- a. By a Member or authorized representative that is received by CareFirst or its authorized agent customarily responsible for handling benefit matters; and,
- b. That names a specific Member; a specific medical condition or symptom; and a specific treatment, service, or product for which approval is requested.
- 3. Post-Service Claims. In the case of a Post-Service Claim, CareFirst shall notify the Member of the CareFirst's Adverse Benefit Determination within a reasonable period of time, but not later than 30 days after receipt of the claim. This period may be extended one time by CareFirst for up to 15 days, provided that CareFirst both determines that such an extension is necessary due to matters beyond the control of CareFirst and notifies the Member, prior to the expiration of the initial 30-day period, of the circumstances requiring the extension of time and the date by which CareFirst expects to render a decision. If such an extension is necessary due to a failure of the Member to submit the information necessary to decide the claim, the notice of extension shall specifically describe the required information, and the Member shall be afforded at least 45 days from receipt of the notice within which to provide the specified information.
- 4. <u>Concurrent Care Decisions</u>. If CareFirst has approved an ongoing course of treatment to be provided over a period of time or number of treatments:
 - a. CareFirst will notify the Member of any reduction or termination of such course of treatment (other than by a change in the plan's coverage by amendment or termination of coverage) before the end of such period of time or number of treatments and at a time sufficiently in advance of the reduction or termination to allow the Member to appeal and obtain a determination on review before the benefit is reduced or terminated.
 - b. Any request by a Member to extend the course of treatment beyond the period of time or number of treatments that is a claim involving Urgent/Emergent Care will be decided as soon as possible, taking into account the medical exigencies. CareFirst will notify the Member of the benefit determination, whether adverse or not, within 24 hours after receipt of the claim, provided that any such claim is made to CareFirst at least 24 hours prior to the expiration of the prescribed period of time or number of treatments.
- 5. <u>Rescissions.</u> If CareFirst has made an Adverse Determination that is a Rescission, CareFirst shall provide 30 days advance written notice to any covered person who would be affected by the proposed Rescission.

6. <u>Calculating Time Periods</u>. For purposes of this Part B, the period of time within which an Adverse Benefit Determination is required to be made shall begin at the time a claim is filed in accordance with CareFirst procedures. The time is counted regardless to whether all the information necessary to make a benefit determination accompanies the filing. In the event that a period of time is extended due to a Member's failure to submit information necessary to decide a claim, the period for making the benefit determination shall be tolled from the date on which the notification of the extension is sent to the Member until the date on which the Member responds to the request for additional information.

C. INTERNAL GRIEVANCE PROCEDURE

- 1. A grievance must be filed within 180 days from the date of receipt of the written notice of any Adverse Benefit Determination.
- 2. A Member or authorized representative should first contact CareFirst about a denial of benefits. CareFirst can provide information and assistance on how to file a grievance. All grievances filed should be in writing, except grievances involving Urgent/Emergent Care which may be submitted orally or in writing.
- 3. The Member or authorized representative may submit written comments, documents, records, and other information relating to a claim for benefits.
- 4. The grievance decision for Urgent/Emergent Care claim shall be made as soon as possible but no later than the earlier of 24 hours after CareFirst's receipt of the necessary information to make the decision regarding request for coverage, or 72 hours after receipt of the request for coverage.
- 5. A Member shall be provided, upon request and free of charge, reasonable access to, and copies of, all documents, records, and other information relevant to the Member's claim for benefits. A document, record, or other information shall be considered relevant to a Member's claim if it:
 - a. Was relied upon in making the benefit determination;
 - b. Was submitted, considered, or generated in the course of making the benefit determination, without regard to whether such document, record, or other information was relied upon in making the benefit determination; or,
 - c. Demonstrates compliance with the administrative processes and safeguards designed to ensure and verify that benefit claim determinations are made in accordance with governing plan documents and that, where appropriate, the plan provisions have been applied consistently with respect to similarly situated members.
- 6. A grievance and any applicable documentation should be sent to the correspondence address stated on the reverse of the Member identification card.
- 7. Timing of CareFirst responses. The time limits for responding to a grievance will begin at the time an appeal is filed in accordance with these procedures, without regard to whether all the information necessary to make a decision is initially included. CareFirst will make a grievance decision and written notification will be sent:
 - a. Within 30 days after receipt of the grievance for a case involving a Pre-Service Claim;

b. Within 60 days after receipt of the grievance for a case involving a Post-Service Claim; and

In the case of an expedited appeal regarding a claim relating to a prescription for the alleviation of cancer pain, the appeal decision shall be made as soon as possible but no later than 24 hours after receipt of the appeal.

- 8. When more information is needed for a decision. CareFirst will send notice within 5 working days of the receipt of the appeal that it cannot proceed with its review unless the additional information is provided. CareFirst will assist in gathering the necessary information. The response deadlines described above may be extended one time by CareFirst for up to 15 days, provided that CareFirst both:
 - a. determines that such an extension is necessary due to matters beyond the control of CareFirst; and,
 - b. notifies the Member, prior to the expiration of the initial 15-day period, of the circumstances requiring the extension of time and the date by which CareFirst expects to render a decision.

If such an extension is necessary due to a failure of the Member to submit the information necessary to decide the claim, the notice of extension shall specifically describe the required information, and the Member shall be afforded at least 45 days from receipt of the notice within which to provide the specified information.

In the event that a period of time is extended due to a Member's failure to submit necessary information, the period for responding to a grievance shall be tolled from the date on which the notification of the extension is sent to the Member until the date on which the Member responds to the request for additional information.

The Member must agree to this extension in writing. The Member will be asked to sign a consent form.

D. FAIR AND FULL REVIEW

CareFirst will provide a review that:

- 1. Takes into account all comments, documents, records, and other information submitted by the Member relating to the claim, without regard to whether such information was submitted or considered in the initial benefit determination;
- 2. Does not afford deference to the initial Adverse Benefit Determination and is conducted by an appropriate named fiduciary of CareFirst who is neither the individual who made the Adverse Benefit Determination that is subject to the appeal, nor the subordinate of such individual;
- 3. In deciding an appeal of an Adverse Benefit Determination that is based in whole or in part on a medical judgment, including determinations with regard to whether a particular treatment, drug, or other item is Cosmetic, Experimental/Investigational, or not Medically Necessary, the appropriate named fiduciary shall consult with a Health Care Provider who has appropriate training and experience in the field of medicine involved in the medical judgment;
- 4. Provides for the identification of medical or vocational experts whose advice was obtained on behalf of CareFirst in connection with a Member's Adverse Benefit Determination, without regard to whether the advice was relied upon in making the benefit determination; and,

5. The Health Care Provider engaged for purposes of a consultation is an individual who is neither an individual who was consulted in connection with the Adverse Benefit Determination, nor the subordinate of any such individual.

E. DEEMED EXHAUSTION OF INTERNAL CLAIMS AND APPEAL PROCESS

In the case of a plan that fails to adhere to the minimum requirements for employee benefit plan procedures relating to Claims for Benefits, the Member is deemed to have exhausted the internal claims and appeals processes of paragraph C and D herein. Accordingly the Member may initiate an external review under paragraph F of this section, as applicable. The Member is also entitled to pursue any available remedies under section 502(a) of ERISA or under State law, as applicable, on the basis that the plan has failed to provide a reasonable internal claims and appeals process that would yield a decision on the merits of the Claim for Benefits. If a Member chooses to pursue remedies under section 502(a) of ERISA under such circumstances, the Claim for Benefits or Appeal is deemed denied on review without the exercise of discretion by an appropriate fiduciary.

F. EXTERNAL APPEAL PROCEDURE

A Member who is dissatisfied with a decision rendered in a final internal grievance process shall have the opportunity to pursue an appeal before an external independent review organization if filed within four (4) months of the final grievance decision.

If a Member is dissatisfied with the resolution reached through CareFirst's internal grievance system regarding medical necessity, the Member may contact the Director, Office of Health Care Ombudsman and Bill of Rights, at the following:

District of Columbia Department of Health Care Finance Office of Health Care Ombudsman and Bill of Rights One Judiciary Square 441 4th St. NW, 900 South Washington, DC 20001 ((877) 685-6391; (202) 442-6724 or fax (202) 478-1397)

If a Member is dissatisfied with the resolution reached through CareFirst's internal grievance system regarding all other grievances, the Member may contact the Commissioner at the following:

Commissioner, Department of Insurance, Securities and Banking 1050 First St. N.E., Suite 801 Washington, D.C. 20002 (202) 727-8000 Fax: (202) 354-1085

A Member shall also have the option to contact the District of Columbia Department of Insurance, Securities and Banking to request an investigation or file a complaint with the Department at any time during the internal claims and appeal process.

Group Hospitalization and Medical Services, Inc.

[Signature]	
[Name] [Title]	

Group Hospitalization and Medical Services, Inc.

doing business as

CareFirst BlueCross BlueShield (CareFirst)

840 First Street, NE

Washington, DC 20065

202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

ATTACHMENT B OUT-OF-NETWORK DESCRIPTION OF COVERED SERVICES

The services described herein are eligible for coverage under the Out-of-Network Evidence of Coverage. CareFirst will provide the benefits described in the Out-of-Network Schedule of Benefits for Medically Necessary Covered Services incurred by a Member.

It is important to refer to the Out-of-Network Schedule of Benefits to determine the percentage of the Allowed Benefit that CareFirst will pay and any specific limits on the number of services that will be covered. The Out-of-Network Schedule of Benefits also lists important information about Deductibles, the Out-of-Pocket Maximum, and other features that affect Member coverage, including specific benefit limitations.

Refer to the Out-of-Network Evidence of Coverage for additional definitions of capitalized terms included in this Out-of-Network Description of Covered Services.

[Signature]	
 [Name] [Title]	

Group Hospitalization and Medical Services, Inc.

SECTION	TABLE OF CONTENTS	PAGE
1	Outpatient Facility, Office, and Professional Services	B-3
2	Pediatric Dental Services	B-15
3	Pediatric Vision Services	B-20
4	Inpatient Hospital Services	B-22
5	Skilled Nursing Facility Services	B-25
6	Home Health Care Services	B-26
7	Hospice Care Services	B-28
8	Mental Health and Substance Abuse Services	B-30
9	Medical Devices and Supplies	B-32
10	Prescription Drugs	B-36
11	Patient-Centered Medical Home	B-37
12	Complex Chronic or High Risk Acute Disease Management	B-38
13	General Provisions	B-39
14	Utilization Management	B-43
15	Exclusions and Limitations	B-46

SECTION 1 OUTPATIENT FACILITY, OFFICE, AND PROFESSIONAL SERVICES

See Section 14, Utilization Management, for Covered Services that require prior authorization.

1.1 Office Visits

Benefits are available for office visits for the diagnosis and treatment of a medical condition, including care and consultation provided by primary care providers and Specialists.

- 1.2 <u>Laboratory Tests, X-Ray/Radiology Services, Specialty Imaging, and Diagnostic Procedures</u>
 Coverage is provided for laboratory and pathology services, x-ray/radiology services, specialty imaging, and diagnostic procedures. Covered services include mammograms, ultrasounds, nuclear medicine, CAT Scans, MRIs, EKGs, EEGs, MRAs, MRSs, CTAs, PET scans, SPECT scans, nuclear cardiology, and related professional services for lab interpretation, x-ray reading, and scan reading.
 - A. For purposes of this provision, specialty imaging includes MRI's, MRA's and MRS's, PET scans, CAT scans and nuclear medicine studies.
 - B. Sleep Studies.
 - 1. Coverage is provided for electro-diagnostic tests used to diagnose sleep disorders, including obstructive sleep apnea. These tests may also be used to help adjust a treatment plan for a sleep disorder that has been previously diagnosed. These tests may be done at home, freestanding facilities, outpatient hospital facilities, or at a sleep disorder unit within a hospital.
 - 2. Prior authorization is required for facility-based sleep tests, independent sleep clinic services, and inpatient sleep tests. Prior authorization is not required for home sleep tests.

1.3 <u>Preventive Services</u>

In addition to the benefits listed in this provision, CareFirst will provide benefits for health exams and other services for the prevention and detection of disease, at intervals appropriate to the Member's age, sex, and health status, in accordance with the Patient Protection and Affordable Care Act, as amended, and the Health Care and Education Reconciliation Act of 2010, as amended, as well as CareFirst preventive guidelines. At a minimum, benefits for preventive services listed in this provision will be provided once per Benefit Period.

Benefits will be provided for evidence-based items or services that have in effect a rating of "A" or "B" in the current recommendations of the United States Preventive Services Task Force (USPSTF). This includes benefits for preventive maternity care. CareFirst will update new recommendations to the preventive benefits listed in this provision at the schedule established by the Secretary of Health and Human Services.

Benefits for preventive care include the following:

- A. Cancer Screening Services Benefits include:
 - 1. Prostate Cancer Screening
 Benefits are available when rendered in accordance with the most current
 American Cancer Society's guidelines and include a medically recognized
 diagnostic examination, annual digital rectal examinations, and the prostatespecific antigen (PSA) tests.
 - 2. Colorectal Cancer Screening

Colorectal cancer screening provided in accordance with the latest guidelines issued by the American Cancer Society.

3. Pap Smears

Benefits are available for pap smears, including tests performed using FDA approved gynecological cytology screening technologies, at intervals appropriate to the Member's age and health status, as determined by CareFirst.

4. Breast Cancer Screening

At a minimum, benefits will be provided for breast cancer screening in accordance with the latest screening guidelines issued by the American Cancer Society. The current recommendations of the United States Preventive Services Task Force regarding breast cancer screening, mammography, and prevention of breast cancer will be considered the most current other than those issued in or around November 2009.

B. Human Papillomavirus Screening Test

- 1. Coverage is provided for a Human Papillomavirus Screening Test at the screening intervals supported by the Health Resources and Services Administration of the U.S. Department of Health and Human Services.
- 2. Human Papillomavirus Screening Test means any laboratory test that specifically detects for infection by one or more agents of the human papillomavirus and is approved for this purpose by the FDA.

C. Immunizations

Coverage is provided for immunizations for routine use in children, adolescents, and adults that have in effect a recommendation from the Advisory Committee on Immunization Practices of the Centers for Disease Control and Prevention with respect to the individual involved. Immunizations required solely for travel or work are not covered.

A recommendation from the Advisory Committee on Immunization Practices of the Centers for Disease Control and Prevention is considered to be:

- 1. In effect after it has been adopted by the Director of the Centers for Disease Control and Prevention; and
- 2. For routine use if it is listed on the immunization schedules of the Centers for Disease Control and Prevention.

D. Well Child Care

With respect to infants, children, and adolescents, coverage is provided for evidence-informed preventive care and screenings in the Recommendations for Preventive Pediatric Health by the American Academy of Pediatrics and the Recommended Uniform Screening Panels by the Secretary's Advisory Committee on Heritable Disorders in Newborns and Children.

E. Adult Preventive Care

Benefits include health care services incidental to and rendered during an annual visit at intervals appropriate to the Member's age, sex , and health status, including evidence-based items or services that have in effect a rating of "A" or "B" in the current recommendations of the United States Preventive Services Task Force (USPSTF).

F. Preventive Gynecological Care

Benefits include recommended preventive services that are age and developmentally appropriate as provided for in comprehensive guidelines supported by the Health Resources and Services Administration.

- G. Prevention and Treatment of Obesity Benefits will be provided for:
 - 1. Well child care visits for obesity evaluation and management;
 - 2. Evidence-based items or services for preventive care and screening for obesity that have in effect a rating of "A" or "B" in the current recommendations of the United States Preventive Services Task Force (USPSTF);
 - 3. For infants, children, and adolescents, evidence-informed preventive care and screening for obesity provided for in the comprehensive guidelines supported by the Health Resources and Services Administration; and,
 - 4. Office visits for the treatment of obesity.

H. Osteoporosis Prevention and Treatment Services

1. Definitions

<u>Bone Mass Measurement</u> means a radiologic or other scientifically proven technology for the purpose of identifying bone mass or detecting bone loss.

Qualified Individual means a Member:

- a. Who is estrogen deficient and at clinical risk for osteoporosis;
- b. With a specific sign suggestive of spinal osteoporosis, including roentgenographic osteopenia or roentgenographic evidence suggestive of collapse, wedging, or ballooning of one or more thoracic or lumbar vertebral bodies, who is a candidate for therapeutic intervention or for an extensive diagnostic evaluation for metabolic bone disease;
- c. Receiving long-term glucocorticoid (steroid) therapy;
- d. With primary hyperparathyroidism; or,
- e. Being monitored to assess the response to, or efficacy of, an approved osteoporosis drug therapy.
- 2. Covered Benefits

Benefits for Bone Mass Measurement for the prevention, diagnosis, and treatment of osteoporosis are covered when requested by a Health Care Provider for a Qualified Individual.

1.4 <u>Professional Nutritional Counseling and Medical Nutrition Therapy</u>
Benefits are available for Medically Necessary Professional Nutritional Counseling and Medical Nutrition Therapy as determined by CareFirst.

1.5 <u>Family Planning Services</u>

Benefits will be provided for:

- A. Non-Preventive Gynecological Care
 Benefits are available for Medically Necessary gynecological care. Benefits for preventive gynecological care are described in Section 1.3.F.
- B. Contraceptive Methods and Counseling Covered Benefits:

- 1. Contraceptive patient education and counseling for all Members with reproductive capacity.
- 2. Benefits will be provided for all FDA approved contraceptive drugs and devices for all Members, and sterilization procedures and other contraceptive methods for female Members that must be administered to the Member in the course of a covered outpatient or inpatient treatment.
- 3. Coverage will be provided for the insertion or removal, and any Medically Necessary examination associated with the use of any contraceptive devices or drugs that are approved by the FDA.
- 4. Voluntary sterilization for male Members and surgical reversal of voluntary sterilization for all members.
- 5. Elective abortion.

See Section 10, Prescription Drugs, in the Description of Covered Services in the In-Network Evidence of Coverage for coverage for self-administered FDA-approved contraceptive drugs and devices.

C. Maternity Services

The following maternity services are provided for all female members.

1. Preventive Services

- a) Routine outpatient obstetrical care of an uncomplicated pregnancy, including prenatal evaluation and management office visits and one post-partum office visit;
- b) Prenatal laboratory tests and diagnostic services related to the outpatient care of an uncomplicated pregnancy, including those identified in the current recommendations of the United States Preventive Services Task Force that have in effect a rating of "A" or "B" or provided in the comprehensive guidelines for women's preventive health supported by the Health Resources and Services Administration:
- c) Preventive laboratory tests and services rendered to a newborn during a covered hospitalization for delivery, identified in the current recommendations of the United States Preventive Services Task Force that have in effect a rating of "A" or "B," the Periodicity Schedule of the Bright Futures Recommendations for Pediatric Preventive Health Care, and the Uniform Panel of the Secretary's Advisory Committee on Heritable Disorders in Newborns and Children, including the collection of adequate samples for hereditary and metabolic newborn screening and newborn hearing screening; and
- d) Breastfeeding support, supplies, and consultation.

2. Non-Preventive Services

a) Outpatient obstetrical care and professional services for all prenatal and post-partum complications. Services include prenatal and post-partum office visits and ancillary services provided during those visits, such as Medically Necessary laboratory tests and diagnostic services;

- b) Inpatient care for delivery;
- c) Non-preventive routine professional services rendered to the newborn during a covered hospitalization for delivery. Non-routine care of the newborn, either during or following the mother's covered hospitalization, requires that the newborn be covered as a Member in the newborn's own right. The Out-of-Network Evidence of Coverage describes the steps, if any, necessary to enroll a newborn Dependent child.
- 3. Postpartum Home Visits. See Section 6.3.C., Home Health Care Services.
- D. Newborn Coverage. Coverage includes:
 - 1. Medically Necessary routine newborn visits including admission and discharge exams and visits for the collection of adequate samples for hereditary and metabolic newborn screening;
 - 2. Medically Necessary care and treatment of medically diagnosed congenital defects and birth abnormalities; and
 - 3. Routine hearing screening consisting of one of the following:
 - a. Auditory brain stem response;
 - b. Otoacustic emissions; or
 - c. Other appropriate, nationally recognized, objective physiological screening test.

Additionally, benefits will be provided for infant hearing screenings and all necessary audiological examinations provided using any technology approved by the United States Food and Drug Administration, and as recommended by the most current standards addressing early hearing detection and intervention programs by the National Joint Committee on Infant Hearing. Such coverage will include follow-up audiological examinations as recommended by a physician or audiologist and performed by a licensed audiologist to confirm the existence or absence of hearing loss. Infant as used here is defined according to the most current recommendation of the American Academy of Pediatrics.

E. Infertility Services

Benefits are available for the diagnosis of infertility. Benefits are limited to the following:

- A. Infertility counseling; and
- B. Testing.

1.6 Allergy Services

Benefits are available for allergy testing and treatment, including allergy serum and the administration of injections.

1.7 <u>Rehabilitation Services</u>

A. Definitions

<u>Physical Therapy (PT)</u> includes the short-term treatment that can be expected to result in a significant improvement of a condition. Physical Therapy is the treatment of disease or injury through the use of therapeutic exercise and other interventions that focus on improving a person's ability to go through the functional activities of daily living, to develop and/or restore maximum potential function, and to reduce disability following an

illness, injury, or loss of a body part. These may include improving posture, locomotion, strength, endurance, balance, coordination, joint mobility, flexibility, and alleviating pain.

Occupational Therapy (OT) means the use of purposeful activity or interventions designed to achieve functional outcomes that promote health, prevent injury or disability, and that develop, improve, sustain or restore the highest possible level of independence of an individual who has an injury, illness, cognitive impairment, psychosocial dysfunction, mental illness, developmental or learning disability, physical disability, loss of a body part, or other disorder or condition. Occupational Therapy services do not include the adjustment or manipulation of any of the osseous structures of the body or spine.

<u>Speech Therapy (ST)</u> means the treatment of communication impairment and swallowing disorders. Speech Therapy services facilitate the development and maintenance of human communication and swallowing through assessment, diagnosis, and rehabilitation, including cognitive rehabilitation.

B. Covered Benefits

Coverage includes benefits for rehabilitation services including Physical Therapy, Occupational Therapy, and Speech Therapy for the treatment of individuals who have sustained an illness or injury that CareFirst determines to be subject to improvement.

The goal of rehabilitation services is to return the individual to his/her prior skill and functional level.

1.8 Spinal Manipulation

A. Covered Services

Coverage is provided for Medically Necessary spinal manipulation, evaluation, and treatment for the musculoskeletal conditions of the spine when provided by a licensed chiropractor, doctor of osteopathy (D.O.), or other eligible practitioner.

B. Limitations. Benefits will not be provided for spinal manipulation services other than for musculoskeletal conditions of the spine.

1.9 Habilitative Services

- A. For Members from birth to age 21.
 - 1. Coverage for Habilitative Services include health care services and devices that help a person keep, learn, or improve skills and functioning for daily living. Examples include therapy for a child who is not walking or talking at the expected age. These services may include physical and occupational therapy, speech-language pathology and other services for people with disabilities in a variety of inpatient and/or outpatient settings.
 - 2. Benefits are not available for Habilitative Services delivered through early intervention and school services.
- B. For Members age 21 and over.
 - 1. Coverage for Habilitative Services include health care services and devices that help a person keep, learn, or improve skills and functioning for daily living. These services may include physical and occupational therapy, speech-language pathology and other services for people with disabilities in a variety of inpatient and/or outpatient settings.
 - 2. Prior authorization is required.

1.10 Outpatient Therapeutic Treatment Services. Benefits are available for outpatient services rendered in a health care provider's office, in the outpatient department of a hospital, in an ambulatory surgical facility, or other facility in connection with a medical or surgical procedure covered under Section 1, Outpatient Facility, Office and Professional Services.

Benefits include:

- A. Hemodialysis and peritoneal dialysis;
- B. Radiation therapy, including radiation administration;
- C. Cardiac Rehabilitation benefits for Members who have been diagnosed with significant cardiac disease, or who have suffered a myocardial infarction, or have undergone invasive cardiac treatment immediately preceding referral for Cardiac Rehabilitation. Cardiac Rehabilitation is a comprehensive program involving medical evaluation, prescribed exercise, cardiac risk factor modification, education and counseling. Benefits include:
 - 1. Continuous EKG telemetric monitoring during exercise, EKG rhythm strip with interpretation, physician's revision of exercise prescription, and follow up examination for physician to adjust medication or change regimen; and
 - 2. Increased outpatient rehabilitation services (physical therapy, speech therapy and occupational therapy) for Cardiac Rehabilitation of ninety (90) visits per therapy per Benefit Period.
 - 3. Services must be provided at a place of service equipped and approved to provide Cardiac Rehabilitation.
- D. Pulmonary rehabilitation benefits for Members who have been diagnosed with significant pulmonary disease.
 - 1. Limited to one (1) program per lifetime.
 - 2. Services must be provided at a place of service equipped and approved to provide pulmonary rehabilitation services.
- E. Transfusion services and Infusion Services, including home infusions, infusion of therapeutic agents, medication and nutrients, enteral nutrition into the gastrointestinal tract, chemotherapy and prescription medications;
- F. Radioisotope treatment.

1.11 Blood and Blood Products

Benefits are available for blood and blood products (including derivatives and components) that are not replaced by or on behalf of the Member.

1.12 Organ and Tissue Transplants

- A. Coverage is provided for all Medically Necessary, non-Experimental/Investigational bone marrow, solid organ transplant, and other non-solid organ transplant procedures. Medical Necessity is determined by CareFirst.
- B. Covered Services include the following:
 - 1. The expenses related to registration at transplant facilities. The place of registry is subject to review and determination by CareFirst.
 - 2. Organ procurement charges including harvesting, recovery, preservation, and transportation of the donated organ.

- 3. Cost of hotel lodging and air transportation for the recipient Member and a companion (or the recipient Member and two companions if the recipient Member is under the age of eighteen (18) years) to and from the site of the transplant.
- 4. There is no limit on the number of re-transplants that are covered.
- 5. If the Member is the recipient of a covered organ/tissue transplant, CareFirst will cover the Donor Services (as defined below) to the extent that the services are not covered under any other health insurance plan or contract.

<u>Donor Services</u> means services covered under the Out-of-Network Evidence of Coverage which are related to the transplant surgery, including evaluating and preparing the actual donor, regardless of whether the transplant is attempted or completed, and recovery services after the donor procedure which are directly related to donating the organ or tissue.

6. Immunosuppressant maintenance drugs are covered when prescribed for a covered transplant.

1.13 High Dose Chemotherapy/Bone Marrow or Stem Cell Transplant

Benefits will be provided for high dose chemotherapy bone marrow or stem cell transplant treatment that is not Experimental/ Investigational, when performed pursuant to protocols approved by the institutional review board of any United States medical teaching college including, but not limited to, National Cancer Institute protocols that have been favorably reviewed and utilized by hematologists or oncologists experienced in dose-intensive chemotherapy/autologous bone marrow transplants or stem cell transplants.

1.14 <u>Clinical Trial Patient Cost Coverage</u>

A. Definitions

<u>Cooperative Group</u> means a formal network of facilities that collaborate on research projects and have an established NIH-approved peer review program operating within the Group. Cooperative Group includes the National Cancer Institute Clinical Cooperative Group, National Cancer Institute Community Clinical Oncology Program, AIDS Clinical Trials Group, and Community Programs for Clinical Research in AIDS.

<u>Multiple Project Assurance Contract</u> means a contract between an institution and the federal Department of Health and Human Services that defines the relationship of the institution to the federal Department of Health and Human Services, and sets out the responsibilities of the institution and the procedures that will be used by the institution to protect human subjects.

NIH means the National Institutes of Health.

<u>Qualified Individual</u>, as used in this section, means a Member who is eligible to participate in an approved clinical trial according to the trial protocol, with respect to the treatment of cancer or other life-threatening disease or condition, and the provider who recommended the Member for the clinical trial has concluded that the Member's participation in such trial is appropriate to treat the disease or condition, or the Member's participation is based on medical and scientific information.

Routine Patient Costs means the costs of all Medically Necessary items and health care services consistent with the Covered Services that are typically provided for a Qualified Individual who is not enrolled in a clinical trial that are incurred as a result of the treatment being provided to the Qualified Individual for purposes of the clinical trial. Routine Patient Costs do not include the investigational item, device, or service itself;

items or services that are provided solely to satisfy data collection and analysis needs and that are not used in the direct clinical management of the patient; or a service that is clearly inconsistent with widely accepted and established standards of care for a particular diagnosis.

B. Covered Services

- Benefits for Routine Patient Costs to a Qualified Individual in a clinical trial will be provided if the Qualified Individual's participation in the clinical trial is the result of:
 - Treatment provided for a life-threatening disease or condition or chronic disease; or
 - b) Prevention, early detection, treatment, and monitoring studies on cancer.
- 2. Coverage for Routine Patient Costs will be provided only if:
 - a) The treatment is being provided or the studies are being conducted in a Phase I, Phase II, Phase III, or Phase IV clinical trial for cancer; or,
 - b) The treatment is being provided in a Phase I, Phase II, Phase III, or Phase IV clinical trial for any other life-threatening disease or condition;
 - c) The treatment is being provided in a federally funded or approved clinical trial approved by one of the National Institutes of Health, the Centers for Disease Control and Prevention, the Agency for Health Care Research and quality, the Centers for Medicare and Medicaid Services, an NIH Cooperative Group, an NIH Center, the FDA in the form of an investigational new drug or device application, the federal Department of Defense, the federal Department of Veterans Affairs, the federal Department of Energy or a qualified nongovernmental research entity to which the National Cancer Institute has awarded a support grant,, or an institutional review board of an institution in a state that has a Multiple Project Assurance Contract approved by the Office of Protection from Research Risks of the NIH;
 - d) The treatment is being provided under a drug trial that is exempt from the requirement of an investigational new drug application.
 - e) The facility and personnel providing the treatment are capable of doing so by virtue of their experience, training, and volume of patients treated to maintain expertise;
- 3. Coverage is provided for the Routine Patient Costs incurred for drugs and devices that have been approved for sale by the FDA whether or not the FDA has approved the drug or device for use in treating the Qualified Individual's particular condition, to the extent that the drugs or devices are not paid for by the manufacturer, distributor, or provider of that drug or device.

1.15 Diabetes Equipment and Supplies, and Self-Management Training

A. If deemed necessary, diabetes outpatient self-management training and educational services, including Medical Nutrition Therapy, will be provided through an in-person program supervised by an appropriately licensed, registered, or certified CareFirst-approved facility or health care provider whose scope of practice includes diabetes education or management.

B. Coverage information for diabetic equipment and supplies is located in Section 10, Medical Devices and Supplies and Section 11, Prescription Drugs.

1.16 <u>Dental Services</u>

Pediatric dental benefits for Members up to age 19 are described in Section 2. Benefits will be provided to all Members for the following:

Accidental Injury

A. Covered Benefits

Dental benefits will be provided to repair or replace Sound Natural Teeth that have been damaged or lost due to injury if the injury did not arise while or as a result of biting or chewing, and treatment is commenced within six (6) months of the injury or, if due to the nature of the injury, treatment could not begin within six (6) months of the injury, treatment began within six (6) months of the earliest date that it would be medically appropriate to begin such treatment.

As used in this provision, accidental injury means an injury to Sound Natural Teeth as a result of an external force or trauma resulting in damage to a tooth or teeth, surrounding bone and/or jaw.

B. Conditions and Limitations

Benefits are limited to Medically Necessary dental services as a restoration of the tooth or teeth or the initial placement of a bridge or denture to replace the tooth or teeth injured or lost as a direct and sole result of the accidental bodily injury. Except as listed here, or in Section 1.18, describing benefits for the treatment of cleft lip or cleft palate or both, dental care is excluded from coverage. Benefits for oral surgery are described in Section 1.17.

C. Exclusions

Injuries to teeth that are not Sound Natural Teeth are not covered. Injuries as a result of biting or chewing are not covered.

1.17 Oral Surgery

Benefits include:

- A. Medically Necessary procedures, as determined by CareFirst, to attain functional capacity, correct a congenital anomaly (excluding odontogenic congenital anomalies or anomalies limited to the teeth), reduce a dislocation, repair a fracture, excise tumors, non-odontogenic cysts or exostoses, or drain abscesses involving cellulitis and are performed on the lips, tongue, roof, and floor of the mouth, sinuses, salivary glands or ducts, and jaws.
- B. Medically Necessary procedures, as determined by CareFirst, needed as a result of an accidental injury, when the Member requests oral surgical services or dental services for Sound Natural Teeth and supporting structures or the need for oral surgical services or dental services for Sound Natural Teeth and supporting structures is identified in the patient's medical records within sixty (60) days of the accident. Benefits for such oral surgical services will be provided up to three (3) years from the date of injury.
- C. Surgical treatment for temporomandibular joint syndrome (TMJ) if there is clearly demonstrable radiographic evidence of joint abnormality due to an illness.

All other procedures involving the teeth or areas surrounding the teeth including the shortening of the mandible or maxillae (orthognathic surgery) for Cosmetic or other purposes or for correction of the malocclusion unrelated to a functional impairment that cannot be corrected non-surgically are excluded.

1.18 Treatment for Cleft Lip or Cleft Palate or Both

Benefits will be provided for inpatient or outpatient expenses arising from orthodontics, oral surgery, otologic, audiological, and speech/language treatment for cleft lip or cleft palate or both.

1.19 Outpatient Surgical Procedures

- A. Benefits are available for surgical procedures performed by a health care provider on an outpatient basis including, but not limited to, colonoscopy, sigmoidoscopy, and endoscopy.
- B. Benefits are available for services in a hospital outpatient department or in an ambulatory surgical facility, in connection with a covered surgical procedure, including:
 - 1. Use of operating room and recovery room.
 - 2. Use of special procedure rooms.
 - 3. Diagnostic procedures, laboratory tests, and radiology services.
 - 4. Drugs, medications, solutions, biological preparations, and services associated with the administration of the same.
 - 5. Medical and surgical supplies.
 - 6. Blood, blood plasma and blood products, and related donor processing fees that are not replaced by or on behalf of the Member. Administration of infusions is covered.
- 1.20 <u>Anesthesia Services for Medical or Surgical Procedures</u>. Benefits are available for the administration of general anesthesia in connection with a covered medical or surgical procedure. To be eligible for separate coverage, a health care provider other than the operating surgeon or assistant at surgery must administer the anesthesia. For example, a local anesthetic used while performing a medical or surgical procedure is not generally viewed as a separately covered charge.

1.21 Reconstructive Surgery

Benefits for reconstructive surgery are limited to surgical procedures that are Medically Necessary, as determined by CareFirst, and operative procedures performed on structures of the body to improve or restore bodily function or to correct a deformity resulting from disease, trauma, or previous therapeutic intervention.

1.22 <u>Reconstructive Breast Surgery</u>

Benefits will be provided for reconstructive breast surgery resulting from a Mastectomy.

- A. Reconstructive breast surgery means surgery performed as a result of a Mastectomy to reestablish symmetry between the two breasts. Reconstructive breast surgery includes:
 - 1. Augmentation mammoplasty;
 - 2. Reduction mammoplasty; and
 - 3. Mastopexy.
- B. Benefits are provided for all stages of reconstructive breast surgery performed on the non-diseased breast to establish symmetry with the diseased breast when reconstructive breast surgery on the diseased breast is performed.
- C. Benefits are provided regardless of whether the Mastectomy was performed while the Member was covered under the Out-of-Network Evidence of Coverage.

D. Coverage will be provided for treatment of physical complications at all stages of Mastectomy, including lymphedemas, in a manner determined in consultation with the Member and the Member's attending physician.

1.23 Retail Health Clinics

Coverage is provided for treatment of common conditions or ailments, which require rapid and specific treatment that can be administered in a limited duration of time. Retail Health Clinics are mini-medical office chains typically staffed by nurse practitioners with an on-call physician. Retail Health Clinic services are non-emergency and non-urgent services for common ailments for which a reasonable, prudent layperson who possesses an average knowledge of health and medicine would seek in a Retail Health Clinic, including, but not limited to: ear, bladder, and sinus infections; pink eye; flu, and strep throat.

1.24 Urgent Care Services

Benefits are available for Urgent Care received from an Urgent Care center.

1.25 <u>Emergency Services</u>

Benefits are available for Emergency Services received in or through a hospital emergency room. Benefits include coverage for the costs of a voluntary HIV test, performed during a Member's visit to a hospital emergency room, regardless of the reason for the hospital emergency room visit.

1.26 Ambulance Services

Benefits are available for Medically Necessary air and ground ambulance services as determined by CareFirst.

If the Member is outside the United States and requires treatment by a medical professional, benefits will be provided to transport the Member to the nearest location where more appropriate medical care is available. Benefits include air or ground ambulance services, when Medically Necessary.

1.27 <u>Telemedicine Services</u>

- A. Coverage shall be provided for the use of interactive audio, video, or other electronic media for the purpose of consultation, diagnosis, or treatment of the patient.
- B. Benefits for telemedicine shall be provided by a health care provider to deliver health care services within the scope of the provider's practice at a site other than the site where the patient is located.
- C. Benefits for telemedicine are not subject to any annual dollar maximum or annual visit limitation.
- D. CareFirst shall not exclude a service from coverage solely because the service is provided through telemedicine and is not provided through face-to-face consultation or contact between a health care provider and a patient for services appropriately provided through telemedicine.

Telemedicine does not include an audio-only telephone, electronic mail message, or facsimile transmission between a health care provider and a patient.

SECTION 2 PEDIATRIC DENTAL SERVICES

- 2.1 Subject to the terms and conditions of this Out-of-Network Description of Covered Services, benefits will be provided for the following Covered Dental Services when rendered and billed for by a Dentist as specified in the attached Schedules of Benefits.
- 2.2 Pediatric dental benefits for Members up to the end of the calendar year in which the Member turns age 19 will be provided in accordance with the High Option dental benefits of the Federal Employees Dental and Vision Insurance Program (FEDVIP) as specified in the Out-of-Network Schedule of Benefits.
- 2.3 Class I Preventive and Diagnostic Services
 - A. Services limited to twice per Benefit Period.
 - 1. Oral examination including oral health risk assessment
 - 2. Routine cleaning of teeth (dental prophylaxis)
 - 3. Topical application of fluoride
 - 4. Bitewing x-ray (not taken on the same date as those in B. below)
 - 5. Pulp vitality tests; additional tests may be allowed for accidental injury and trauma, or other emergency
 - B. Services limited to one per 60 months
 - 1. Intraoral complete series x-ray (full mouth x-ray including bitewings)
 - 2. One panoramic x-ray and one additional set of bitewing x-rays
 - C. Services limited to once per tooth per 36 months: sealants on permanent molars
 - D. Space maintainers when Medically Necessary due to the premature loss of a posterior primary tooth
 - E. Services as required
 - 1. Palliative Treatments once per date of service
 - 2. Emergency Oral Exam once per date of service
 - 3. Periapical and occlusal x-rays limited to the site of injury or infection
 - 4. Professional consultation rendered by a Dentist, limited to one consultation per condition per Dentist other than the treating Dentist
 - 5. Intraoral occlusal x-ray
 - 6. One cephalometric x-ray
- 2.4 Class II Basic Services
 - A. Direct placement fillings limited to:

- 1. Silver amalgam, resin-based composite, compomer, glass-ionomer or equivalent material accepted by the American Dental Association and/or the United States Food and Drug Administration
- 2. Direct pulp caps and indirect pulp caps
- B. Non-Surgical periodontic services limited to:
 - 1. Periodontal scaling and root planing once per 24 months per quadrant
 - 2. Full mouth debridement to enable comprehensive periodontal procedure one per lifetime
 - 3. Periodontal maintenance procedures four per 12 months
- C. Simple extractions performed without general anesthesia once per tooth per lifetime
- 2.5 Class III Major Services Surgical
 - A. Surgical periodontic services
 - 1. Gingivectomy or gingivoplasty limited to one treatment per 36 months per Member per quadrant or per tooth
 - 2. Osseous Surgery (including flap entry and closure) limited to one treatment per 36 months per Member per quadrant
 - 3. Limited or complete occlusal adjustments in connection with periodontal treatment when services are received on a different date than restorative services
 - 4. Mucogingival Surgery limited to grafts and plastic procedures; one treatment per site limited to one site or quadrant every 36 months

B. Endodontics

- 1. Apicoectomy
- 2. Pulpotomy for deciduous teeth once per tooth per lifetime per Member
- 3. Root canal for permanent teeth once per tooth per lifetime per Member
- 4. Root canal retreatment performed on permanent teeth limited to once per tooth per lifetime per Member
- 5. Root resection once per tooth per lifetime per Member
- 6. Pulpal therapy once per tooth per lifetime per Member
- 7. Endodontic therapy once per tooth per lifetime per Member
- C. Oral Surgical services as required
 - 1. Simple and Surgical extractions, including impactions once per tooth per lifetime per Member
 - 2. Oral Surgery, including treatment for cysts, tumors and abscesses
 - 3. Biopsies of oral tissue if a biopsy report is submitted

- 4. General anesthesia, intravenous (IV) sedation/analgesia, analgesia, and non-intravenous conscious sedation when Medically Necessary and administered by a Dentist who has a license, permit, or certificate to administer conscious sedation or general anesthesia or board certified anesthesiologist (MD, DO, DDS, DMD).
- 6. Hemi-section
- 7. Tooth reimplantation and/or stabilization of accidentally evulsed or displaced tooth
- 8. Vestibuloplasty
- 9. Services limited to once per lifetime per tooth:
 - a) Coronectomy
 - b) Tooth transplantation
 - c) Surgical repositioning of teeth
 - d) Alveoloplasty
 - e) Frenulectomy
 - f) Excision of pericoronal gingiva

2.6 Class IV - Major Services - Restorative

A. Crowns

- 1. Metal and/or porcelain/ceramic crowns and crown build-ups limited to one per 60 months per tooth
- 2. Metal and/or porcelain/ceramic inlays and onlays limited to one per 60 months per tooth
- 3. Stainless steel crowns
- 4. Recementation of crowns and/or inlays limited to once in any twelve (12) month period
- 5. Metal and/or porcelain/ceramic pontics limited to one per 60 months per tooth
- B. Dental Implants are covered procedures only if determined to be Medically Necessary. If CareFirst determines an arch can be restored with a standard prosthesis or restoration, no benefits will be allowed for the individual implant or implant procedures, and only the second phase of treatment (the prosthodontic phase of placing of the implant crown or partial denture) will be a Covered Dental Service.
 - 1. Endosteal implant limited to one per 60 months
 - 2. Surgical placement of interim implant body limited to one per 60 months
 - 3. Eposteal implant limited to one per 60 months
 - 4. Transosteal implant limited to one per 60 months
 - 5. Implant supported complete denture

6. Implant supported partial denture

C. Dentures

- 1. Partial removable dentures, upper or lower, limited to one per 60 months
- 2. Complete removable dentures, upper or lower, limited to one per 60 months
- 3. Pre-operative radiographs required
- 4. Pre-treatment estimate, as described in the Estimate of Eligible Benefits section is recommended for Members
- 5. Tissue conditioning prior to denture impression
- 6. Repairs to denture as required including: repair resin denture base, repair cast framework, addition of tooth or clasp to existing partial denture, replacement of broken tooth, repairs or replacement of clasp, recement fixed partial denture
- D. Denture adjustments and relining limited to: Full or partial removable (upper or lower) dentures: once per 24 months, but not within six months of initial placement
- E. Repair of prosthetic appliances and removable dentures, full and/or partial.
- F. Occlusal guard, by report, limited to one per 12 months for Members age 13 and older

2.7 Class V - Orthodontic Services

- A. Benefits for orthodontic services will only be available until the end of the calendar year in which the Member turns age 19 if the Member:
 - 1. Has fully erupted permanent teeth with at least 1/2 to 3/4 of the clinical crown being exposed (unless the tooth is impacted or congenitally missing); and
 - 2. Has a severe, dysfunctional, handicapping malocclusion and is determined to be Medically Necessary.
- B. All comprehensive orthodontic services require a pre-treatment estimate (PTE) by CareFirst, as described in the Estimate of Eligible Benefits section. The following documentation must be submitted with the request for a PTE:
 - 1. ADA 2006 or newer claim form with service code requested;
 - 2. A complete series of intra-oral photographs;
 - 3. Diagnostic study models (trimmed) with waxbites or OrthoCad electronic equivalent, and
 - 4. Treatment plan including anticipated duration of active treatment.
- C. Covered benefits if a PTE is approved
 - 1. Retainers
 - a) One set (included in comprehensive orthodontics)

- b) Replacement allowed one per arch per lifetime within 12 months of date of debanding, if necessary
- c) Rebonding or recementing fixed retainer
- 2. Pre-orthodontic treatment visit
- 3. Braces once per lifetime
- 4. Periodic treatment visits; not to exceed 24 months (the Member must be eligible for Covered Dental Services on each date of service).
- D. Payment policy: one initial payment for comprehensive orthodontic treatment, a preorthodontic treatment visit and periodic orthodontic treatment visits (not to exceed 24 periodic orthodontic treatment visits).
 - When a Preferred Dentist or Participating Dentist provides the comprehensive orthodontic treatment, additional periodic orthodontic treatment visits beyond 24 will be the orthodontist's financial responsibility and not the Subscriber's. Subscribers may not be billed for broken, repaired, or replacement of brackets or wires. Visits to repair or replace brackets or wires are not separately reimbursable from periodic visits.
 - 2. When a Non-Participating Dentist provides the comprehensive orthodontic treatment, additional periodic orthodontic treatment visits beyond 24 will not be Covered Dental Services. The Member is responsible for the difference between the CareFirst payment for Covered Dental Services and the Non-Participating Dentist's charge.
- E. In cases where the Member has been approved for comprehensive orthodontic benefits, and the parent has decided they do not wish to have the child begin treatment at this time or any time in the near future, the provider may bill for their records, to include the treatment plan, radiographs, models, photos, etc. and explaining the situation on the claim for payment. The reimbursement for these records is the same as if the orthodontic services had been rendered.
- F. If the case is denied, the provider will be informed that CareFirst will not cover the orthodontic treatment. However, Covered Dental Services will include the preorthodontic visit which included treatment plan, radiographs, and/or photos, records and diagnostic models for full treatment cases only.

SECTION 3 PEDIATRIC VISION SERVICES

3.1 <u>Covered Services</u>

Coverage will be provided for pediatric vision benefits for children up to age 19 in accordance with the Federal Employee Program Blue Vision high plan. Benefits include:

- A. One routine eye examination, including dilation, if professionally indicated, each Benefit Period. A vision examination may include, but is not limited to:
 - 1. Case history;
 - 2. External examination of the eye and adnexa;
 - 3. Ophthalmoscopic examination;
 - 4. Determination of refractive status;
 - 5. Binocular balance testing;
 - 6. Tonometry test for glaucoma;
 - 7. Gross visual field testing;
 - 8. Color vision testing;
 - 9. Summary finding; and
 - 10. Recommendation, including prescription of corrective lenses.
- B. Frames and Spectacle Lenses or Contact Lenses
 - 1. Prescribed frames and spectacle lenses or contact lenses, including directly related provider services such as:
 - a) Measurement of face and interpupillary distance;
 - b) Ouality assurance; and
 - c) Reasonable aftercare to fit, adjust and maintain comfort and effectiveness.
 - 2. One pair of frames per Benefit Period; and
 - 3. One pair of prescription spectacle lenses per Benefit Period
 - a) Spectacle lenses include choice of glass or plastic lenses, all lens powers (single vision, bifocal, trifocal, lenticular), fashion and gradient tinting, oversized glass-grey #3 prescription sunglass lenses, ultraviolet protective coating, standard progressives, and plastic photosensitive lenses (Transitions®).
 - b) Polycarbonate lenses are covered in full for monocular patients and patients with prescriptions > +/- 6.00 diopters.
 - c) All spectacle lenses include scratch resistant coating with no additional Copayment. There may be an additional charge at Walmart and Sam's Club

4. Contact Lenses

- a) Contact lens evaluation, fitting, and follow-up care.
- b) Elective contact lenses (in place of frames and spectacle lenses):
 - (1) One pair of elective prescription contact lenses per Benefit Period; or,
 - (2) Multiple pairs of disposable prescription contact lenses per Benefit Period.
- c) One pair of Medically Necessary prescription contact lenses per Benefit Period in lieu of other eyewear.
 - (1) Prior authorization must be obtained from the Vision Care Designee by calling the Vision Care Designee at the telephone number on the Member's identification card.
 - (2) Contact lenses may be determined to be Medically Necessary and appropriate in the treatment of patients affected by certain conditions. Contact lenses may be Medically Necessary and appropriate when the use of contact lenses, in lieu of eyeglasses, will result in significantly better visual and/or improved binocular function, including avoidance of diplopia or suppression. Contact lenses may be determined to be Medically Necessary in the treatment of the following conditions: keratoconus, pathological myopia, aphakia, anisometropia, aniseikonia, aniridia, corneal disorders, post-traumatic disorders, and/or irregular astigmatism.
- C. Low vision services, including one comprehensive Low Vision evaluation every 5 years, 4 follow-up visits in any 5-year period and prescribed low vision aid optical devices, such as high-powered spectacles, magnifiers and telescopes.
 - 1. Ophthalmologists and optometrists specializing in low vision care will evaluate and prescribe optical devices, and provide training and instruction to maximize the remaining usable vision for Members with low vision.
 - 2. Prior authorization is required for low vision services. Prior authorization must be obtained from the Vision Care Designee by calling the Vision Care Designee at the telephone number on the Member's identification card.
- D. Covered Vision Services and benefits for services provided by Non-Contracting Vision Providers are limited. See the Out-of-Network Schedule of Benefits.

3.2 Limitations

Benefit limited to Members up to age 19. If Member is under age 19 at the start of the Benefit Period but turns 19 during the Benefit Period, then the Member will receive Covered Vision Services through the rest of that Calendar Year. Benefits for the treatment of medical conditions of the eye are covered under Section 1.

SECTION 4 INPATIENT HOSPITAL SERVICES

HOSPITAL ADMISSIONS MUST BE AUTHORIZED OR APPROVED BY CAREFIRST UNLESS EXCEPTIONS ARE STATED.

4.1 <u>Covered Inpatient Hospital Services</u>

A Member will receive benefits for the Covered Services listed below when admitted to a hospital. Coverage of inpatient hospital services is subject to certification by utilization management for Medical Necessity. Benefits are provided for:

A. Room and Board

Room and board in a semiprivate room (or in a private room when Medically Necessary as determined by CareFirst).

B. Physician, Medical, and Surgical Services

Medically Necessary inpatient physician, medical, and surgical services provided by or under the direction of the attending physician and ordinarily furnished to a patient while hospitalized.

C. Services and Supplies

Related inpatient services and supplies that are not Experimental/Investigational, as determined by CareFirst, and ordinarily furnished by the hospital to its patients, including:

- 1. The use of:
 - a) Operating rooms;
 - b) Treatment rooms; and
 - c) Special equipment in the hospital.
- 2. Drugs, medications, solutions, biological preparations, anesthesia, and services associated with the administration of the same.
- 3. Medical and surgical supplies.
- 4. Blood, blood plasma, and blood products, and related donor processing fees that are not replaced by or on behalf of the Member. Administrations of infusions and transfusions are covered.
- 5. Surgically implanted Prosthetic Devices that replace an internal part of the body. This includes hip joints, skull plates, cochlear implants, and pacemakers. Available benefits under this provision do not include items such as dental implants, fixed or removable dental Prosthetics, artificial limbs, or other external Prosthetics, which may be provided under other provisions of this Out-of-Network Description of Covered Services.
- 6. Medical social services.

4.2 <u>Number of Hospital Days Covered</u>

Provided the conditions, including the requirements below, are met and continue to be met, as determined by CareFirst, hospital benefits for inpatient hospital services will be provided as follows:

A. Hospitalization for Rehabilitation

Benefits are provided for an admission or transfer to a CareFirst approved facility for

rehabilitation. Benefits provided during any admission will not exceed any applicable benefit limitation. The limit, if any, on hospitalization for rehabilitation applies to any portion of an admission that:

- 1. Is required primarily for Physical Therapy or other rehabilitative care; and
- 2. Would not be Medically Necessary based solely on the Member's need for inpatient acute care services other than for rehabilitation.

B. Inpatient Coverage Following a Mastectomy Coverage will be provided for a minimum hospital stay of not less than:

- 1. Forty-eight (48) hours following a radical or modified radical Mastectomy; and
- 2. Twenty-four (24) hours following a partial Mastectomy with lymph node dissection.

In consultation with the Member's attending physician, the Member may elect to stay less than the minimum prescribed above when appropriate.

C. Hysterectomies

Coverage will be provided for vaginal hysterectomies and abdominal hysterectomies. Coverage includes a minimum stay in the hospital of:

- 1. Not less than twenty-three (23) hours for a laparoscopy-assisted vaginal hysterectomy; and
- 2. Not less than forty-eight (48) hours for a vaginal hysterectomy.

In consultation with the Member's attending physician, the Member may elect to stay less than the minimum prescribed above when appropriate.

D. Childbirth

Coverage will be provided for a minimum hospital stay of not less than:

- 1. Forty-eight (48) hours for both the mother and newborn following a routine vaginal delivery;
- 2. Ninety-six (96) hours for both the mother and newborn following a routine cesarean section.

Prior authorization is not required for maternity admissions.

Whenever a mother is required to remain hospitalized after childbirth for medical reasons and the mother requests that the newborn remain in the hospital, coverage includes additional hospitalization for the newborn for up to four (4) days.

If the delivery occurs in the hospital, the length of stay begins at the time of the delivery. If the delivery occurs outside of the hospital, the length of stay begins upon admission to the hospital. The Member and the attending physician may agree to an early discharge.

Non-routine care of the newborn, either during or following the mother's covered hospitalization, requires that the newborn be covered as a Member in the newborn's own right. Section 2.7 of the Out-of-Network Evidence of Coverage describes the steps, if any, necessary to enroll a newborn Dependent child.

4.3

Other Inpatient Services
Benefits are available for all other care in the nature of usual hospital services that are Medically Necessary for the care and treatment of the patient, provided that those services cannot be rendered in an outpatient setting and are not otherwise specifically excluded.

SECTION 5 SKILLED NURSING FACILITY SERVICES

See Section 14, Utilization Management, for Covered Services that require prior authorization.

5.1 Covered Skilled Nursing Facility Services

When the Member meets the conditions for coverage listed in Section 5.2, the services listed below are available to Members in a Skilled Nursing Facility:

- A. Room and board in a semiprivate room;
- B. Inpatient physician and medical services provided by or under the direction of the attending physician; and
- C. Services and supplies that are not Experimental/Investigational, as determined by CareFirst, and ordinarily furnished by the facility to inpatients for diagnosis or treatment.

5.2 Conditions for Coverage

Skilled Nursing Facility care must be authorized or approved by CareFirst as meeting the following conditions for coverage:

- A. The admission to the Skilled Nursing Facility must be a substitute for hospital care (i.e., if the Member were not admitted to a Skilled Nursing Facility, he or she would have to be admitted to a hospital).
- B. Skilled Nursing Facility benefits will not be provided in a facility that is used primarily as a rest home or a home for the aged, or in a facility for the care of drug addiction or alcoholism.
- C. The Member must require Skilled Nursing Care or skilled rehabilitation services which are:
 - 1. Required on a daily basis;
 - 2. Not Custodial; and
 - 3. Only provided on an inpatient basis.

5.3 Custodial Care is Not Provided

Benefits will not be provided for any day in a Skilled Nursing Facility that CareFirst determines is primarily for Custodial Care. Services may be deemed Custodial Care even if:

- A. A Member cannot self-administer the care:
- B. No one in the Member's household can perform the services;
- C. Ordered by a physician;
- D. Necessary to maintain the Member's present condition; or
- E. Covered by Medicare.

SECTION 6 HOME HEALTH CARE SERVICES

See Section 14, Utilization Management, for Covered Services that require prior authorization.

6.1 <u>Covered Home Health Care Services</u>

Benefits are provided for:

A. Continued care and treatment provided by or under the supervision of a Registered Nurse (RN) or Licensed Practical Nurse (LPN). Services of a home health aide, medical social worker, or registered dietician may be provided, but must be performed under the supervision of a licensed professional (RN or LPN) nurse.

B. Drugs and medications

Drugs and medications directly administered to the patient during a covered home health care visit and incidental Medical Supplies directly expended in the course of a covered home health care visit are covered.

C. Home Health Care Services authorized or approved by CareFirst as Medically Necessary under the utilization management requirements as meeting the conditions for coverage.

Purchase or rental of Durable Medical Equipment is not covered under this provision. See Section 9.3.A, Durable Medical Equipment, for benefit information.

6.2 Conditions for Coverage

Benefits are provided when:

- A. The Member must be confined to home due to a medical, non-psychiatric condition. "Home" cannot be an institution, convalescent home, or any facility which is primarily engaged in rendering medical or rehabilitative services to sick, disabled, or injured persons.
- B. The Home Health Care visits are a substitute for hospital care or for care in a Skilled Nursing Facility (i.e., if Home Health Care visits were not provided, the Member would have to be admitted to a hospital or Skilled Nursing Facility).
- C. The Member requires and continues to require Skilled Nursing Care or rehabilitation services in order to qualify for home health aide services or other types of Home Health Care Services.
- D. The need for Home Health Care Services is not Custodial in nature.
- E. Services of a home health aide, medical social worker, or registered dietician must be performed under the supervision of a licensed professional nurse (RN or LPN).
- F. All services must be arranged and billed by the Qualified Home Health Agency. Providers may not be retained directly by the Member.

6.3 Additional Home Health Care Benefits

- A. Home Visits Following Surgical Removal of a Testicle
 For a Member who receives less than 48 hours of inpatient hospitalization following the surgical removal of a testicle, or who undergoes the surgical removal of a testicle on an outpatient basis, benefits will be provided for:
 - 1. One home visit scheduled to occur within twenty-four (24) hours after discharge from the hospital or outpatient health care facility; and
 - 2. An additional home visit if prescribed by the Member's attending physician.

3. Benefits provided under this provision do not count toward any Home Health Care visit maximum.

B. Home Visits Following a Mastectomy

- 1. For a Member who has a shorter hospital stay than that provided under Section 4.2.B, Inpatient Coverage Following a Mastectomy, or who undergoes a Mastectomy on an outpatient basis, benefits will be provided for:
 - a) One home visit scheduled to occur within twenty-four (24) hours after discharge from the hospital or outpatient health care facility; and
 - b) An additional home visit if prescribed by the Member's attending physician.
- 2. For a Member who remains in the hospital for at least the length of time provided in Section 4.2.B, Inpatient Coverage Following a Mastectomy, coverage will be provided for a home visit if prescribed by the Member's attending physician
- 3. Benefits provided under this provision do not count toward any Home Health Care visit maximum.

C. Postpartum Home Visits

Home visits following delivery are covered in accordance with the most current standards published by the American College of Obstetricians and Gynecologists.

- 1. For a mother and newborn child who have a shorter hospital stay than that provided under Section 4.2.D, Childbirth, benefits will be provided for:
 - a) One home visit scheduled to occur within 24 hours after hospital discharge; and
 - b) An additional home visit if prescribed by the attending physician.
- 2. For a mother and newborn child who remain in the hospital for at least the length of time provided under Section 4.2.D, Childbirth, benefits will be provided for a home visit if prescribed by the attending physician.
- 3. Benefits provided under this provision do not count toward any Home Health Care visit maximum.

SECTION 7 HOSPICE CARE SERVICES

See Section 14, Utilization Management, for Covered Services that require prior authorization.

7.1 Covered Hospice Care Services

Benefits will be provided for the services listed below when provided by a Qualified Hospice Care Program. Coverage for hospice care services is subject to certification of the need and continued appropriateness of such services in accordance with CareFirst utilization management requirements.

- A. Inpatient and outpatient care;
- B. Intermittent Skilled Nursing Care;
- C. Medical social services for the terminally ill patient and his or her Immediate Family;
- D. Counseling, including dietary counseling, for the terminally ill Member;
- E. Non-Custodial home health visits.
- F. Services, visits, medical/surgical equipment, or supplies, including equipment and medication required to maintain the comfort and manage the pain of the terminally ill Member;
- G. Laboratory test and x-ray services;
- H. Medically Necessary ground ambulance, as determined by CareFirst;
- I. Family Counseling will be provided to the Immediate Family and the Family Caregiver before the death of the terminally ill Member, when authorized or approved by CareFirst; and
- J. Bereavement Counseling will be provided for the Immediate Family or Family Caregiver of the Member for the six (6) month period following the Member's death or fifteen (15) visits, whichever occurs first.

7.2 Conditions for Coverage

Hospice care services must be certified by CareFirst, provided by a Qualified Hospice Care Program, and meet the following conditions for coverage:

- A. The Member must have a life expectancy of six (6) months or less;
- B. The Member's attending physician must submit a written hospice care services plan of treatment to CareFirst:
- C. The Member must meet the criteria of the Qualified Hospice Care Program;
- D. The need and continued appropriateness of hospice care services must be certified by CareFirst as meeting the criteria for coverage in accordance with CareFirst utilization management requirements.

7.3 Hospice Eligibility Period

The hospice eligibility period begins on the first date hospice care services are rendered and terminates one hundred eighty (180) days later or upon the death of the terminally ill Member, if sooner. If the Member requires an extension of the eligibility period, the Member or the Member's representative must notify CareFirst in advance to request an extension of benefits. CareFirst reserves the right to extend the eligibility period on an individual case basis if CareFirst

determines the program of h	nat the Member's prog nospice care services.	gnosis and continue	ed need for services	are consistent with a

SECTION 8 MENTAL HEALTH AND SUBSTANCE ABUSE SERVICES

HOSPITAL ADMISSIONS MUST BE AUTHORIZED OR APPROVED BY THE MENTAL HEALTH AND SUBSTANCE ABUSE MANAGEMENT PROGRAM

8.1 Definitions

Mental Illness and Emotional Disorders are broadly defined as including any mental disorder, mental illness, psychiatric illness, mental condition, or psychiatric condition (whether organic or non-organic, whether of biological, non-biological, chemical or non-chemical origin, and irrespective of cause, basis, or inducement). This includes, but is not limited to, psychoses, neurotic disorders, schizophrenic disorders, affective disorders, personality disorders, and psychological or behavioral abnormalities associated with transient or permanent dysfunction of the brain or related neurohormonal systems. (This is intended to include disorders, conditions and illnesses classified on Axes I and II in the current edition of the Diagnostic and Statistical Manual of Mental Disorders of the American Psychiatric Association, Washington, D.C.)

Mental Health and Substance Abuse Management Program refers to utilization management, benefits administration, and provider network activities administered by or on behalf of CareFirst to ensure that mental health and Substance Abuse services are Medically Necessary and provided in a cost-effective manner.

<u>Partial Hospitalization</u> means the provision of medically directed intensive or intermediate short-term treatment in a licensed or certified facility or program for treatment of Mental Illnesses, Emotional Disorders, and Drug and Alcohol Abuse.

<u>Qualified Partial Hospitalization Program</u> means a licensed or certified facility or program that provides medically directed intensive or intermediate short-term treatment for Mental Illness, Emotional Disorder, Drug Abuse or Alcohol Abuse for a period of less than twenty-four (24) hours, but more than four (4) hours in a day.

Qualified Treatment Facility means a non-hospital residential facility certified by the District of Columbia or by any jurisdiction in which it is located, as a qualified non-hospital provider of treatment for Drug Abuse, Alcohol Abuse, Mental Illness, or any combination of these, in a residential setting. A non-hospital residential facility includes any facility operated by the District of Columbia, any state or territory or the federal government to provide these services in a residential setting. It is not a facility licensed as a general or special hospital. A non-hospital residential facility also must meet or exceed guidelines established for such a facility by CareFirst.

Substance Abuse means:

- A. <u>Alcohol Abuse</u> means any pattern of pathological use of alcohol that causes impairment in social or occupational functioning, or that produces physiological dependency evidenced by physical tolerance or by physical symptoms when it is withdrawn.
- B. <u>Drug Abuse</u> means any pattern of pathological use of drugs that causes impairment in social or occupational functioning, or that produces physiological dependency evidenced by physical tolerance or by physical symptoms when it is withdrawn.

8.2 <u>Outpatient Mental Health and Substance Abuse Services</u> Covered Services include the following:

A. Diagnosis and treatment for Mental Illness and Emotional Disorders at health care provider offices, other outpatient health care provider medical offices and facilities, and in Qualified Partial Hospitalization Programs.

- B. Diagnosis and treatment for Substance Abuse, including detoxification and rehabilitation services as an outpatient in a covered alcohol or drug rehabilitation program or Qualified Partial Hospitalization Program designated by CareFirst.
- C. Other covered medical services and medical ancillary services for conditions related to Mental Illness, Emotional Disorders, and Substance Abuse.
- D. Office visits for medication management in connection with Mental Illness, Emotional Disorders, and Substance Abuse.
- E. Methadone maintenance treatment.
- F. Partial Hospitalization in a Qualified Partial Hospitalization Program.
- G. Electroconvulsive therapy.

8.3 <u>Inpatient Mental Health and Substance Abuse Services</u>

Benefits are provided when the Member is admitted as an inpatient in a hospital or other CareFirst-approved health care facility for treatment of Mental Illness, Emotional Disorders, and Substance Abuse as follows:

- A. Hospital benefits will be provided, as described in Section 4, Inpatient Hospital Services, of this Out-of-Network Description of Covered Services, on the same basis as a medical (non-Mental Health or Substance Abuse) admission.
- B. Services provided to a hospitalized Member, including physician visits, charges for intensive care, or consultative services, only if CareFirst determines that the health care provider rendered services to the Member and that such services were medically required to diagnose or treat the Member's condition.

The following benefits apply if the Member is an inpatient in a hospital covered under inpatient hospitalization benefits following CareFirst certification of the need and continued appropriateness of such services in accordance with CareFirst utilization management requirements:

- 1. Health care provider visits during the Member's hospital stay;
- 2. Intensive care that requires a health care provider's attendance;
- 3. Consultation by another health care provider when additional skilled care is required because of the complexity of the Member's condition; and
- C. Benefits are available for diagnosis and treatment for Substance Abuse, including inpatient detoxification and rehabilitation services in an acute care hospital or Qualified Treatment Facility. Members must meet the applicable criteria for acceptance into, and continued participation in, treatment facilities/programs, as determined by CareFirst.
- D. Electroconvulsive therapy.

SECTION 9 MEDICAL DEVICES AND SUPPLIES

9.1 Definitions

<u>Durable Medical Equipment</u> means equipment which:

- A. Is primarily and customarily used to serve a medical purpose;
- B. Is not useful to a person in the absence of illness or injury;
- C. Is ordered or prescribed by a health care provider;
- D. Is consistent with the diagnosis;
- E. Is appropriate for use in the home;
- F. Is reusable; and
- G. Can withstand repeated use.

<u>Inherited Metabolic Disease</u> means a disease caused by an inherited abnormality of body chemistry, including a disease for which the state screens newborn babies.

Low Protein Modified Food Product means a food product that is:

- A. Specially formulated to have less than 1 gram of protein per serving; and
- B. Intended to be used under the direction of a physician for the dietary treatment of an Inherited Metabolic Disease.

Low Protein Modified Food Product does not include a natural food that is naturally low in protein.

<u>Medical Devices</u> means Durable Medical Equipment, medical formulas, Medical Supplies, Orthotic Devices and Prosthetic Devices.

Medical Food means a food that is:

- A. Intended for the dietary treatment of a disease or condition for which nutritional requirements are established by medical evaluation; and
- B. Formulated to be consumed or administered under the direction of a physician.

Medical Supplies means items that:

- A. Are primarily and customarily used to serve a medical purpose;
- B. Are not useful to a person in the absence of illness or injury;
- C. Are ordered or prescribed by a health care provider;
- D. Are consistent with the diagnosis;
- E. Are appropriate for use in the home;

- F. Cannot withstand repeated use; and
- G. Are usually disposable in nature.

Orthotic Devices means orthoses and braces which:

- A. Are primarily and customarily used to serve a therapeutic medical purpose;
- B. Are prescribed by a health care provider;
- C. Are corrective appliances that are applied externally to the body to limit or encourage its activity, to aid in correcting or preventing deformity, or to provide mechanical support;
- D. May be purely passive support or may make use of spring devices; and
- E. Include devices necessary for post-operative healing.

Prosthetic Devices means devices which:

- A. Are primarily intended to replace all or part of an organ or body part that has been lost due to disease or injury; or
- B. Are primarily intended to replace all or part of an organ or body part that was absent from birth; or
- C. Are intended to anatomically replace all or part of a bodily function which is permanently inoperative or malfunctioning; and
- D. Are prescribed by a health care provider; and
- E. Are removable and attached externally to the body.

9.2 <u>Covered Services</u>

A. Durable Medical Equipment

Rental, or, (at CareFirst's option), purchase and replacements or repairs of Medically Necessary Durable Medical Equipment prescribed by a health care provider for therapeutic use for a Member's medical condition.

CareFirst's payment for rental will not exceed the total cost of purchase. CareFirst's payment is limited to the least expensive Medically Necessary Durable Medical Equipment adequate to meet the Member's medical needs. CareFirst's payment for Durable Medical Equipment includes related charges for handling, delivery, mailing, shipping, and taxes.

- B. Medical Supplies
- C. Medical Foods and Low Protein Modified Food Products Medical Foods and Low Protein Modified Food Products for the treatment of Inherited Metabolic Diseases are covered if the Medical Foods or Low Protein Modified Food Products are:
 - 1. Prescribed as Medically Necessary for the therapeutic treatment of Inherited Metabolic Diseases; and;
 - 2. Administered under the direction of a physician.

D. Nutritional Substances
Enteral and elemental nutrition when Medically Necessary as determined by CareFirst.

E. Diabetes Equipment and Supplies

- 1. Coverage will be provided for all Medically Necessary and medically appropriate equipment and diabetic supplies necessary for the treatment of diabetes (Types I and II), or elevated blood glucose levels induced by pregnancy.
- 2. Coverage includes Medically Necessary and medically appropriate equipment and diabetic supplies necessary for the treatment of insulin-dependent diabetes, insulin-using diabetes, gestational diabetes and noninsulin-using diabetes.
- 3. Benefits for insulin syringes and other diabetic supplies described herein are covered in Section 11, Prescription Drugs. All other diabetic equipment is covered as a medical device or supply.

F. Hair Prosthesis

Benefits are available for a hair prosthesis when prescribed by a treating oncologist and the hair loss is a result of chemotherapy or radiation treatment for cancer.

G. Orthotic Devices and Prosthetic Devices Benefits include:

- Supplies and accessories necessary for effective functioning of a Covered Service;
- 2. Repairs or adjustments to Medically Necessary devices that are required due to bone growth or change in medical condition, reasonable weight loss or reasonable weight gain, and normal wear and tear during normal usage of the device; and
- 3. Replacement of Medically Necessary devices when repairs or adjustments fail and/or are not possible.

9.3 Repairs

Benefits for the repair, maintenance, or replacement of covered Durable Medical Equipment are limited as follows:

- A. Coverage of maintenance costs is limited to routine servicing such as testing, cleaning, regulating, and checking of equipment.
- B. Coverage of repairs costs is limited to adjustment required by normal wear or by a change in the Member's condition, and repairs necessary to make the equipment/appliance serviceable. Repair will not be authorized if the repair costs exceed the market value of the appliance, prosthetic, or equipment.
- C. Replacement coverage is limited to once every two (2) years due to irreparable damage and/or normal wear, or a significant change in medical condition. Replacement costs necessitated as a result of malicious damage, culpable neglect, or wrongful disposition of the equipment or device on the part of the Member or of a family member are not covered.

9.4 Benefit Limits

Benefits are limited to the least expensive Medically Necessary Durable Medical Equipment, Medical Supply, Orthotic Device or Prosthetic adequate to meet the patient's medical needs.

Purchase or rental of any Medical Device is at the discretion of CareFirst. Benefits will be limited to the lower cost of purchase or rental, taking into account the length of time the Member

requires, or is reasonably expected to require the equipment, and the durability of the equipment, etc. The purchase price or rental cost must be the least expensive of its type adequate to meet the medical needs of the Member. If the Member selects a deluxe version of the appliance, device, or equipment not determined by CareFirst to be Medically Necessary, CareFirst will pay an amount that does not exceed CareFirst's payment for the basic device (minus any Member Copayment or Coinsurance) and the Member will be fully responsible for paying the remaining balance.

9.5 Responsibility of CareFirst

CareFirst will not be liable for any claim, injury, demand, or judgment based on tort or other grounds (including warranty of equipment), arising out of or in connection with the rental, sale, use, maintenance, or repair of Prosthetic Devices, corrective appliances or Durable Medical Equipment, whether or not covered under this Out-of-Network Description of Covered Services.

SECTION 10 PRESCRIPTION DRUGS

Benefits for Prescription Drugs rendered by pharmacies not contracted with CareFirst or its designee will be provided under the In-Network Evidence of Coverage.

SECTION 11 PATIENT-CENTERED MEDICAL HOME

Benefits provided through the Patient-Centered Medical Home Program are available only when provided by a CareFirst-approved health care provider who has elected to participate in the PCMH. Benefits will be provided only under the In-Network Evidence of Coverage.

SECTION 12
COMPLEX CHRONIC OR HIGH RISK ACUTE DISEASE MANAGEMENT
Benefits for complex chronic or high risk acute disease management are provided only under the In-Network Evidence of Coverage.

SECTION 13 GENERAL PROVISIONS

13.1 How the Plan Works

Benefits will be provided when Covered Services are provided by a Participating or Non-Participating Provider. When the Member uses a Participating or Non-Participating Provider, benefits are based on the appropriate Allowed Benefit. The Allowed Benefit for a Non-Participating Provider may be substantially less than the provider's actual charge to the Member. Non-Participating Providers are not required to accept the Allowed Benefit as full payment and will collect additional amounts from the Member up to the provider's actual charge. Non-Participating Providers will also collect from the Member any applicable Deductibles, Copayment and Coinsurance as well as any charges for non-Covered Services.

A. <u>Participating Provider</u>

Participating Providers will bill CareFirst directly for Covered Services. The Member will not be responsible for amounts in excess of the Allowed Benefit for Covered Services, except any applicable Copayments, Coinsurance and Deductibles. When Covered Services are rendered inside the Service Area, the Participating Provider will make arrangements with CareFirst to obtain utilization management authorizations and approvals required for coverage. Outside of the CareFirst Service Area, an In-Network Provider under the Inter-Plan Arrangements Disclosure Amendment will obtain prior authorization for hospital inpatient services, inpatient mental health and Substance Abuse services, Skilled Nursing Facility Services and inpatient hospice care services only. In all other cases, when Covered Services are rendered outside of the Service Area by In-Network Providers, the Member will be responsible for making arrangements with CareFirst to obtain utilization management authorizations and approvals required for coverage.

B. Non-Participating Provider

Non-Participating Providers are not required to accept the Allowed Benefit as full payment and will collect additional amounts from the Member up to the provider's actual charge. The Allowed Benefit may be substantially less than the provider's actual charge to the Member. Therefore, when Covered Services are provided by Non-Participating Providers, Members should expect to pay additional amounts to providers that exceed the Allowed Benefit.

A Non-Participating Provider may contract with the Member for a different payment amount. However, such provider may not seek payment from CareFirst for these services. In this instance, the Member must submit the claim and payment, not to exceed the Allowed Benefit, will be made to the Member. When benefits are paid to the Member, it is the Member's responsibility to apply any CareFirst payments to the claim from the Non-Participating Provider.

13.2 <u>Member Responsibilities</u>

- A. Members are required to submit claims for Covered Services rendered by Non-Participating Providers to receive benefits. Members may have claims submitted by the Non-Participating Provider on their behalf. A claim submitted by a Non-Participating Provider on behalf of a Member must be submitted within the time frame granted to the Member to file the claim. Refer to the Out-of-Network Evidence of Coverage for claims submission requirements.
- B. Members are responsible for providing all information requested by CareFirst with respect to claims for Covered Services provided by Non-Participating Providers, including, but not limited to, medical records.
- C. Except as provided in Section 13.1A, if the Member receives Covered Services outside of the Service Area, or care is rendered by a Non-Participating Provider, the Member is

responsible for all utilization management requirements. See Section 14, Utilization Management, for the services that require prior authorization.

13.3 Coverage for Services Rendered Outside of the Service Area

Benefits will be provided for Covered Services rendered outside of the Service Area. All Covered Services rendered outside of the service area will be covered, except Emergency Services, Urgent Care, and follow-up care after emergency surgery which will be covered as In-Network services. See the Inter-Plan Arrangements Disclosure Amendment attached to the Out-of-Network Evidence of Coverage for additional information.

Except as provided in Section 13.1A, when Covered Services are rendered outside of the service area, the Member will be responsible for making arrangements with CareFirst to obtain utilization management authorizations and approvals required for coverage. See Section 14, Utilization Management, for the services that require prior authorization.

13.4 <u>Limitation on Provider Coverage</u>

The provider must be licensed, or otherwise authorized by law, in the jurisdiction where the services are rendered. In addition, to be covered, the services must be within the lawful scope of the services for which that provider is licensed or otherwise authorized by law. Coverage does not include services rendered to Members by:

- A. The Member him/herself, or by the Member's Spouse, mother, father, daughter, son, brother, or sister; or,
- B. Anyone who resides in the Member's home.

13.5 Pediatric Vision Coverage.

- A. When the Member receives a vision examination from a Contracting Vision Provider, the benefit payment is accepted as payment in full.
- B. When a Member receives frames and spectacle lenses or contact lenses from a Contracting Vision Provider, the Member's responsibility is as stated below. The benefit payment is as stated in the attached Out-of-Network Schedule of Benefits.
 - 1. When the Member receives frames from the display of collection frames (the collection designated by the Vision Care Designee) and basic spectacle lenses from a Contracting Vision Provider, the benefit payment is accepted as payment in full.
 - 2. When the Member receives other frames, non-basic spectacle lenses or contact lenses from a Contracting Vision Provider, the Member is responsible for the cost difference between the Vision Care Designee's payment and the Contracting Vision Provider's actual charge.
- C. When the Member receives Covered Vision Services from a Non-Contracting Vision Provider, the Member is responsible for the cost difference between the Vision Care Designee's payment and the Non-Contracting Vision Provider's actual charge. The Vision Care Designee's payment is stated in the Schedule of Benefits.
- D. Limited Access Area: If the Member resides in an area that does not have adequate access to a Contracting Vision Provider and the Member receives Vision Care from a Non-Contracting Vision Provider, the Vision Care Designee will pay up to 100% of the Allowed Benefit. The Member is responsible for any difference between the amount billed and the Vision Care Designee's payment. To determine if the Member resides in a limited access area, the Member must call the Vision Care Designee at the telephone number on the Member's identification card.

13.6 Pediatric Dental Coverage

- A. The Member has the exclusive right to choose a Dentist. Whether a Dentist is a Preferred or Participating Dentist or not relates only to method of payment, and does not imply that any Dentist is more or less qualified than another.
- B. CareFirst makes payment for Covered Dental Services, but does not provide these services. CareFirst is not liable for any act or omission of any Dentist.
- C. Services of Participating Dentists
 - 1. Claims will be submitted directly to CareFirst by the Dentist.
 - 2. CareFirst will pay benefits directly to the Dentist.
 - 3. The Member is responsible for only the Deductible and Coinsurance.

D. Services of Non-Participating Dentists

- 1. Claims may be submitted directly to CareFirst by the Non-Participating Dentist or the Member. In either case, it is the responsibility of the Member to make sure that proof of loss is filed on time as stated in the Proof of Loss section of the Evidence of Coverage.
- 2. All benefits for Covered Dental Services rendered by a Non-Participating Dentist will be payable to the Subscriber or to the Non-Participating Dentist, at the discretion of CareFirst.
- 3. The Member is responsible for the difference between the CareFirst payment and the Non-Participating Dentist's charge.

E. Services of Preferred Dentists

- 1. Many Participating Dentists have special agreements with CareFirst and are part of a network of Preferred Dentists. In general, if a Member chooses a Preferred Dentist, the cost to the Member is lower than if the Member chooses a Non-Preferred Dentist. In the Schedule of Benefits, the Coinsurance percentages are listed as either "In-Network" (for a Preferred Dentist) or "Out-of-Network" (for a Non-Preferred Dentist).
- 2. If a Preferred Dentist is not reasonably available when a Member requires emergency care (Palliative Treatment and/or Emergency Oral Exam), benefits will be paid based on the "In-Network" Coinsurance percentage listed in the Schedules of Benefits. Participating Dentists will accept the Allowed Benefit as payment in full, except for any applicable Deductible and Coinsurance amounts for which the Member is responsible. Non-Participating Dentists may bill the Member for the difference between the CareFirst payment and the Non-Participating Dentist's charge.

F. Estimate of Eligible Benefits

A Dentist may propose a planned dental treatment or series of dental procedures. A Member may choose to obtain a written estimate of the benefits available for such procedures.

CareFirst encourages a Member to obtain a written Estimate of Eligible Benefits (CareFirst's written estimate of benefits before a service is rendered) also known as a pre-

treatment estimate (PTE) for major dental procedures, thereby alerting a Member of the out-of-pocket expenses that may be associated with the treatment plan, related deductibles, co-insurance and/or procedures that are not Covered Dental Services. Based on an Estimate of Eligible Benefits or PTE from CareFirst, a Member can decide whether or not to incur the expense that may be associated with a particular treatment plan.

Failure to obtain an Estimate of Eligible Benefits or PTE has no effect on the benefits to which a Member is entitled. A Member may choose to forgo the Estimate of Eligible Benefits or PTE and proceed with treatment.

After the services are rendered, the claim will be reviewed by CareFirst. Should the review determine that the service(s) rendered meet CareFirst's criteria for benefits, the benefits will be provided as described in this Description of Covered Services. However, should the review of the claim determine that the treatment or procedures did not meet CareFirst's criteria for benefits, benefits will <u>not</u> be provided.

To request an Estimate of Eligible Benefits or PTE prior to receiving dental treatment or dental procedures, a Member should contact his or her Dentist who will coordinate the request on the Member's behalf. If the Dentist has any questions about the process, he or she may contact the CareFirst Provider Services Department or go to the CareFirst website at www.carefirst.com, which lists information in the Physicians and Providers section, under the subsection for Dental, and list of Resources. The Estimate of Eligible Benefits or PTE is merely an estimate, and it cannot be considered a guarantee of the Member's benefits or enrollment.

The process is different for orthodontic services. The Affordable Care Act requires that orthodontics must be Medically Necessary to be Covered Dental Services. To request a PTE for orthodontic services, the Member must see an orthodontist who will do an exam and orthodontic assessment that may include taking orthodontic records (study models and certain x-rays). The orthodontist will then complete a case assessment using a scoring tool required by the state. Then the orthodontic records and case assessment will be sent to CareFirst for evaluation and confirmation of the assessment score. If the score meets or exceeds the baseline requirement, the orthodontics will be approved for the Member. If the score is less than the minimal required score, then the request for orthodontic benefits will be denied.

A decision by CareFirst to deny benefits as described in this section constitutes an Adverse Decision if the decision is based on a finding that the proposed service is not Medically Necessary, appropriate, or efficient.

SECTION 14 UTILIZATION MANAGMENT

Failure to meet the requirements of the utilization management program may result in a reduction or denial of benefits even if the services are Medically Necessary.

14.1 <u>Utilization Management</u>

Benefits are subject to review and approval under utilization management requirements established by CareFirst. Through utilization management, CareFirst will:

- A. Review Member care and evaluate requests for approval of coverage in order to determine the Medical Necessity for the services;
- B. Review the appropriateness of the hospital or facility requested; and,
- C. Determine the approved length of confinement or course of treatment in accordance with CareFirst established criteria.

In addition, utilization management may include additional aspects such as prior authorization, and/or preadmission testing requirements, concurrent review, and discharge planning.

If coverage is reduced or excluded for failure to comply with utilization management requirements, the reduction or exclusion may be applied to all services related to the treatment, admission, or portion of the admission for which utilization management requirements were not met. The terms that apply to a Member's coverage for failure to comply with utilization management requirements are stated in the Schedule of Benefits.

14.2 <u>Member Responsibility</u>

If the Member receives Covered Services outside of the service area, or care is rendered by a Non-Participating Provider, the Member is responsible for all utilization management requirements.

It is the Member's responsibility to ensure that providers associated with the Member's care cooperate with utilization management requirements. This includes initial notification in a timely manner, responding to CareFirst inquiries and, if requested, allowing CareFirst representatives to review medical records on-site or in CareFirst offices. If CareFirst is unable to conduct utilization reviews, Member benefits may be reduced or excluded from coverage.

14.3 Procedures

To initiate utilization management review, the Member may directly contact CareFirst or may arrange to have notification given by a family member or by the provider that is involved in the Member's care. However, these individuals will be deemed to be acting on the Member's behalf. If the Member and/or the Member's representatives fail to contact CareFirst as required, or provide inaccurate or incomplete information, benefits may be reduced or excluded.

Members should share the requirements of this section with family members and other responsible persons who could arrange for care on the Member's behalf in accordance with this section in case the Member is unable to do so when necessary. CareFirst will provide additional information regarding utilization management requirements and procedures, including telephone numbers and hours of operation, at the time of enrollment and at any time upon the Member's request.

14.4 <u>Services Subject to Utilization Management</u>

It is the Member's responsibility to obtain prior authorization for the following services when Covered Services are rendered by Non-Participating Providers, and for any Covered Services provided outside of the CareFirst service area.

A. Hospital Inpatient Services

All hospitalizations require prior authorization (except for maternity and Emergency admissions as specified). The Member must contact (or have the provider contact CareFirst) at least five (5) business days prior to an elective or scheduled admission to the hospital. If the admission cannot be scheduled in advance because it is not feasible to delay the admission for five (5) business days due to the Member's medical condition, CareFirst must receive notification of the admission as soon as possible but in any event within forty-eight (48) hours following the beginning of the admission or by the end of the first business day following the beginning of the admission, whichever is later.

Emergency Admissions

CareFirst may not render an adverse decision solely because CareFirst was not notified of the emergency admission within the prescribed period of time after that admission if the Member's condition prevented the hospital from determining the Member's insurance status or CareFirst's emergency admission requirements.

B. Inpatient Mental Illness and Alcohol and/or Substance Abuse Services
The Member must contact CareFirst (or have the provider contact CareFirst) at least five
(5) business days prior to an elective or scheduled admission. If the admission cannot be
scheduled in advance because care is required immediately due to the Member's
condition, CareFirst must receive notification of the admission as soon as possible but in
any event within forty-eight (48) hours following the beginning of the admission or by
the end of the first business day following the beginning of the admission, whichever is
later

For emergency admissions, CareFirst may not render an adverse decision solely because CareFirst was not notified of the emergency admission within the prescribed period of time after that admission if the Member's condition prevented the hospital from determining the Member's insurance status or CareFirst's emergency admission requirements.

- C. Organ and Tissue Transplants Prior authorization must be obtained from CareFirst for all transplants, except cornea and kidney transplants.
 - D. Habilitative Services for Adults.
 - E. Ambulance Services. Prior authorization is required for air ambulance services only, except for Medically Necessary air ambulance services in an emergency.
 - F. Skilled Nursing Facility Services.
 - G. Home Health Services.
 - H. Hospice Care Services.
 - I. Pediatric Vision Services.

Covered Services not listed in this provision do not require prior authorization. CareFirst reserves the right to make changes to the categories of services that are subject to utilization management requirements or to the procedures the Member and/or the providers must follow. CareFirst will notify the Member of these changes at least forty-five (45) days in advance.

Prior authorization is not required for any Covered Services when Medicare is the primary insurer.

- 14.5 <u>CareFirst Personnel Availability for Prior Authorization</u>
 CareFirst will have personnel available to provide prior authorization at all times when prior authorization is required.
- 14.6 <u>Concurrent Review and Discharge Planning</u>. Following timely notification, CareFirst will instruct the Member or the Member's representative, as applicable, about the procedures to follow,

including the need to submit additional information and any requirements for re-notification during the course of treatment.

14.7 Appealing a Utilization Management Decision

If the Member, the Member's representative or Member's provider disagrees with a utilization management decision, CareFirst will review the decision upon request. A utilization management appeal will be reviewed and decided upon by the CareFirst Medical Director or Associate Medical Director not involved in the initial denial decision. If necessary, the Medical Director or Associate Medical Director will discuss the Member's case with the Member's physician and/or request the opinion of a specialist board certified in the same specialty as the treatment under review. Any noncertification or penalty may be appealed. Additional information is provided in the Benefit Determination and Appeals section of the Out-of-Network Evidence of Coverage on how to appeal a utilization management decision.

SECTION 15 EXCLUSIONS AND LIMITATIONS

15.1 General Exclusions

Coverage is not provided for:

- A. Any services, tests, procedures, or supplies which CareFirst determines are not necessary for the prevention, diagnosis, or treatment of the Member's illness, injury, or condition. Although a service or supply may be listed as covered, benefits will be provided only if it is Medically Necessary and appropriate in the Member's particular case.
- B. Any treatment, procedure, facility, equipment, drug, drug usage, device, or supply which is Experimental/Investigational, or not in accordance with accepted medical or psychiatric practices and standards in effect at the time of treatment, except for covered benefits for clinical trials.
- C. The cost of services that:
 - 1. Are furnished without charge; or
 - 2. Are normally furnished without charge to persons without health insurance coverage; or
 - 3. Would have been furnished without charge if a Member were not covered under the Out-of-Network Evidence of Coverage or under any health insurance.

This exclusion does not apply to:

- a) Medicaid:
- b) Care received in a Veteran's hospital unless the care is rendered for a condition that is a result of a Member's military service.
- D. Any service, supply, drug or procedure that is not specifically listed in the Member's Outof-Network Evidence of Coverage as a covered benefit or that do not meet all other conditions and criteria for coverage at the discretion of CareFirst. Provision of services by a health care provider does not, by itself, entitle a Member to benefits if the services are not covered or do not otherwise meet the conditions and criteria for coverage.
- E. Routine, palliative, or Cosmetic foot care (except for conditions determined to be Medically Necessary at the discretion of CareFirst), including flat foot conditions, supportive devices for the foot, treatment of subluxations of the foot, care of corns, bunions (except capsular or bone surgery), calluses, toe nails, fallen arches, weak feet, chronic foot strain, and symptomatic complaints of the feet.
- F. Routine eye examinations and vision services. This exclusion does not apply to evidence-informed preventive care and screenings, including vision care, provided for in the comprehensive guidelines supported by the Health Resources and Services Administration for infants, children, and adolescents and as stated in Section 3.
- G. Any type of dental care (except treatment of accidental bodily injuries, oral surgery, cleft lip or cleft palate or both, and pediatric dental services), including extractions, treatment of cavities, care of the gums or bones supporting the teeth, treatment of periodontal abscess and periodontal disease, removal of teeth, orthodontics, replacement of teeth, or any other dental services or supplies. Benefits for accidental bodily injury are described in Section 1.16. Benefits for oral surgery are described in Section 1.17. Benefits for treatment of cleft lip, cleft palate or both are described in Section 1.18. Benefits for pediatric dental services are described in Section 2. All other procedures involving the teeth or areas and structures

- surrounding and/or supporting the teeth, including surgically altering the mandible or maxillae (orthognathic surgery) for Cosmetic purposes or for correction of malocclusion unrelated to a documented functional impairment are excluded.
- H. Cosmetic surgery (except benefits for reconstructive breast surgery or reconstructive surgery) or other services primarily intended to correct, change, or improve appearances. Cosmetic means a service or supply which is provided with the primary intent of improving appearances and not for the purpose of restoring bodily function or correcting deformity resulting from disease, trauma, or previous therapeutic intervention as determined by CareFirst.
- I. Treatment rendered by a health care provider who is the Member's Spouse, parent, child, grandparent, grandchild, sister, brother, great grandparent, great grandchild, aunt, uncle, niece, or nephew, or resides in the Member's home.
- J. All non-Prescription Drugs, medications, biologicals, and Over-the-Counter disposable supplies, routinely obtained without a prescription and self-administered by the Member, except as listed as a Covered Service above, including but not limited to: cosmetics or health and beauty aids, support devices, non-medical items, foot care items, first aid and miscellaneous medical supplies (whether disposable or durable), personal hygiene supplies, incontinence supplies, and Over-the-Counter medications and solutions, except for Over-the-Counter medication or supplies dispensed under a written prescription by a health care provider that is identified in the current recommendations of the United States Preventive Services Task Force that have in effect a rating of "A" or "B".
- K. Foods or formulas consumed as a sole source of supplemental nutrition, except as listed as a Covered Service in this Out-of-Network Description of Covered Services.
- L. All assisted reproductive technologies including artificial insemination and intrauterine insemination, in vitro fertilization, gamete intra-fallopian tube transfer, zygote intra-fallopian transfer cryogenic preservation or storage of eggs and embryo and related evaluative procedures, drugs, diagnostic services and medical preparations related to the same.
- M. Treatment of sexual dysfunctions or inadequacies including, but not limited to, surgical implants for impotence, medical therapy, and psychiatric treatment.
- N. Fees and charges relating to fitness programs, weight loss, or weight control programs, physical or other programs involving such aspects as exercise, physical conditioning, use of passive or patient-activated exercise equipment or facilities and self-care or self-help training or education, except for diabetes outpatient self-management training and educational services. Cardiac Rehabilitation and pulmonary rehabilitation programs are covered as described in Section 1.
- O. Maintenance programs for Physical Therapy, Speech Therapy, and Occupational Therapy for those services as stated in Section 1.7; and Cardiac Rehabilitation and pulmonary rehabilitation as stated in Section 1.10D and E.
- P. Medical or surgical treatment for obesity, weight reduction, dietary control or commercial weight loss programs, including morbid obesity. This exclusion does not apply to:
 - 1. Well child care visits for obesity evaluation and management;
 - 2. Evidence-based items or services for preventive care and screening for obesity that have in effect a rating of "A" or "B" in the current recommendations of the United States Preventive Services Task Force (USPSTF);

- 3. For infants, children, and adolescents, evidence-informed preventive care and screening for obesity provided for in the comprehensive guidelines supported by the Health Resources and Services Administration;
- 4. Office visits for the treatment of childhood obesity; and
- 5. Professional Nutritional Counseling and Medical Nutrition Therapy as described in this Out-of-Network Description of Covered Services.
- Q. Medical or surgical treatment of myopia or hyperopia, including radial keratotomy and other forms of refractive keratoplasty or any complications thereof.
- R. Services that are beyond the scope of the license of the provider performing the service.
- S. Services that are solely based on court order or as a condition of parole or probation, unless approved by CareFirst.
- T. Health education classes and self-help programs, other than birthing classes or those for the treatment of diabetes.
- U. Acupuncture services, except when approved or authorized by CareFirst when used for anesthesia.
- V. Any service related to recreational activities. This includes, but is not limited to, sports, games, equestrian, and athletic training. These services are not covered unless authorized or approved by CareFirst even though they may have therapeutic value or be provided by a health care provider.
- W. Services or supplies for injuries or diseases related to a covered person's job to the extent the covered person is required to be covered by a workers compensation law.
- X. Private duty nursing.
- Y. Non-medical services. including, but is not limited to:
 - 1. Telephone consultations, failure to keep a scheduled visit, completion of forms (except for forms that may be required by CareFirst), copying charges or other administrative services provided by the health care provider or the health care provider's staff.
 - 2. Administrative fees charged by a physician or medical practice to a Member to retain the physician's or medical practices services, e.g., "concierge fees" or boutique medical practice membership fees. Benefits under the Out-of-Network Evidence of Coverage are available for Covered Services rendered to the Member by a health care provider.
- Z. Rehabilitation services, including Speech Therapy, Occupational Therapy, or Physical Therapy, for conditions not subject to improvement.
- AA. Non-medical ancillary services such as vocational rehabilitation, employment counseling, or educational therapy.
- BB. Services or supplies resulting from accidental bodily injuries arising out of a motor vehicle accident to the extent the services are payable under a medical expense payment provision of an automobile insurance policy, excluding no fault insurance.
- CC. Transportation and travel expenses (except for Medically Necessary air and ground ambulance services, at the discretion of CareFirst, and services listed under Section 1.12,

- Organ and Tissue Transplants, of this Out-of-Network Description of Covered Services), whether or not recommended by a health care provider.
- DD. Services or supplies received from a dental or medical department maintained by or on behalf of an employer, mutual benefit association, labor union, trust, or similar persons or groups.
- EE. Services, drugs, or supplies the Member receives without charge while in active military service.
- FF. Habilitative Services delivered through early intervention and school services.
- GG. Custodial Care.
- HH. Services or supplies received before the Effective Date of the Member's coverage under the Out-of-Network Evidence of Coverage.
- II. Durable Medical Equipment or Medical Supplies associated or used in conjunction with non-covered items or services.
- JJ. Services required solely for administrative purposes, including but not limited to employment, insurance, foreign travel, school, camp admissions or participation in sports activities.
- KK. Work Hardening Programs. Work Hardening Program means a highly specialized rehabilitation program designed to simulate workplace activities and surroundings in a monitored environment with the goal of conditioning the participant for a return to work.
- LL. Chiropractic services or spinal manipulation treatment other than spinal manipulation treatment for musculoskeletal conditions of the spine.
- MM. Any illness or injury caused by war (a conflict between nation states), declared or undeclared, including armed aggression.

15.2 Pediatric Dental Services

A. Limitations

- 1. Covered Dental Services must be performed by or under the supervision of a Dentist with an active and unrestricted license, within the scope of practice for which licensure or certification has been obtained.
- 2. Benefits will be limited to standard procedures and will not be provided for personalized restorations or specialized techniques in the construction of dentures including precision attachments and custom denture teeth.
- 3. If a Member switches from one Dentist to another during a course of treatment, or if more than one Dentist renders services for one dental procedure, CareFirst shall pay as if only one Dentist rendered the service.
- 4. CareFirst will reimburse only after all dental procedures for the condition being treated have been completed (this provision does not apply to orthodontic services).
- 5. In the event there are alternative dental procedures that meet generally accepted standards of professional dental care for a Member's condition, benefits will be based upon the lowest cost alternative procedure.

B. Exclusions

Benefits will not be provided for:

- 1. Replacement of a denture or crown as a result of loss or theft.
- 2. Replacement of an existing denture or crown that is determined by CareFirst to be satisfactory or repairable.
- 3. Replacement of dentures, implants, metal and/or porcelain crowns, inlays, onlays, pontics and crown build-ups within 60 months from the date of placement or replacement for which benefits were paid in whole or in part under the terms of this Out-of-Network Description of Covered Services and are judged by CareFirst to be adequate and functional.
- 4. Gold foil fillings.
- 5. Periodontal appliances.
- 6. Oral orthotic appliances, unless specifically listed as a Covered Dental Service.
- 7. Bacteriologic studies, histopathologic exams, accession of tissue, caries susceptibility tests, diagnostic radiographs, and other pathology procedures, unless specifically listed as a Covered Dental Service.
- 8. Intentional tooth reimplantation or transplantation.
- 9. Interim prosthetic devices, fixed or removable and not part of a permanent or restorative prosthetic service.
- 10. Additional fees charged for visits by a Dentist to the Member's home, to a hospital, to a nursing home, or for office visits after the Dentist's standard office hours. CareFirst shall provide the benefits for the dental service as if the visit was rendered in the Dentist's office during normal office hours.
- 11. Transseptal fiberotomy.
- 12. Orthognathic Surgery.
- 13. The repair or replacement of any orthodontic appliance, unless specifically listed as a Covered Dental Service.
- 14. Any orthodontic services after the last day of the month in which Covered Dental Services ended.
- 15. Separate billings for dental care services or supplies furnished by an employee of a Dentist which are normally included in the Dentist's charges and billed for by them.
- 16. Transitional orthodontic appliance, including a lower lingual holding arch placed where there is not premature loss of the primary molar.
- 17. Limited or complete occlusal adjustments in connection with periodontal surgical treatment when received in conjunction with restorative service on the same date of service.
- 18. Provision splinting, intracoronal and extracoronal.

- 19. Endodontic implant.
- 20. Fabrication of athletic mouthguard.
- 21. Services to alter vertical dimension and/or restore or maintain the occlusion. Such procedures include, but are not limited to, equilibration, periodontal splinting, full mouth rehabilitation, and restoration for misalignment of teeth.
- 22. Adjustments to maxillofacial prosthetic appliance.
- 23. Maintenance and cleaning of a maxillofacial prosthesis (extra or intraoral).
- 24. Any orthodontic services after the last day of the calendar year in which the Member turned age 19.

15.3 Pediatric Vision Services

Benefits will not be provided for the following:

- A. Any pediatric vision service stated in Section 3 for Members over age 19. If Member is under age 19 at the start of the Benefit Period but turns 19 during the Benefit Period, then the Member will receive covered pediatric vision services through the rest of that Calendar Year.
- B. Diagnostic services, except as listed in Section 3.
- C. Services or supplies not specifically approved by the Vision Care Designee where required in this Out-of-Network Description of Covered Services.
- D. Orthoptics, vision training, and low vision aids.
- E. Non-prescription (Plano) lenses and/or glasses, sunglasses or contact lenses.
- F. Except as otherwise provided, Vision Care services that are strictly Cosmetic in nature including, but not limited to, charges for personalization or characterization of prosthetic appliances;
- G. Services and materials not meeting accepted standards of optometric practice.
- H. Services and materials resulting from the Member's failure to comply with professionally prescribed treatment.
- I. Office infection control charges.
- J. State or territorial taxes on vision services performed.
- K. Special lens designs or coatings other than those described herein.
- L. Replacement of lost and/or stolen eyewear.
- M. Two pairs of eyeglasses in lieu of bifocals.
- N. Insurance of contact lenses.

15.4 Organ and Tissue Transplants

Benefits will not be provided for the following:

- A. Non-human organs and their implantation. This exclusion will not be used to deny Medically Necessary non-Experimental/Investigational skin grafts that are covered under this Out-of-Network Description of Covered Services.
- B. Any hospital or professional charges related to any accidental injury or medical condition of the donor of the transplant material.
- C. Any charges related to transportation, lodging, and meals unless authorized or approved by CareFirst.
- D. Services for a Member who is an organ donor when the recipient is not a Member.
- E. Donor search services.
- F. Any service, supply, or device related to a transplant that is not listed as a benefit in this Out-of-Network Description of Covered Services.

15.5 <u>Inpatient Hospital Services</u>

Coverage is not provided (or benefits are reduced, if applicable) for the following:

- A. Private room, unless Medically Necessary and/or authorized or approved by CareFirst. If a private room is not authorized or approved, the difference between the charge for the private room and the charge for a semiprivate room will not be covered.
- B. Non-medical items and Convenience Items, such as television and phone rentals, guest trays, and laundry charges.
- C. Except for covered Emergency Services and maternity care, a health care facility admission or any portion of a health care facility admission that had not been approved by CareFirst, whether or not services are Medically Necessary and/or meet all other conditions for coverage.
- D. Private duty nursing.

15.6 <u>Home Health Care Services</u>

Coverage is not provided for:

- A. Custodial Care.
- C. Private duty nursing.

15.7 <u>Hospice Care Services</u>

Benefits will not be provided for the following:

- A. Services, visits, medical equipment, or supplies not authorized by CareFirst.
- B. Financial and legal counseling.
- C. Any services for which a Qualified Hospice Care Program does not customarily charge the patient or his or her family.
- D. Reimbursement for volunteer services.
- E. Chemotherapy or radiation therapy, unless used for symptom control.
- F. Services, visits, medical equipment, or supplies not required to maintain the comfort and manage the pain of the terminally ill Member.

- G. Custodial Care, domestic, or housekeeping services.
- H. Meals on Wheels or other similar food service arrangements.
- I. Rental or purchase of renal dialysis equipment and supplies. Benefits for dialysis equipment and supplies are available in Section 10, Medical Devices and Supplies.

15.8 Outpatient Mental Health and Substance Abuse

Coverage is not provided for:

- A. Services solely on court order or as a condition of parole or probation unless approved or authorized by the CareFirst Medical Director.
- B. Intellectual disability, after diagnosis.
- C. Psychoanalysis.

15.9 <u>Inpatient Mental Health and Substance Abuse</u>

Coverage is not provided for:

- A. Admissions as a result of a court order or as a condition of parole or probation unless approved or authorized by the CareFirst Medical Director.
- B. Custodial Care.
- C. Admissions solely for observation or isolation.

15.10 Medical Devices and Supplies

Benefits will not be provided for purchase, rental, or repair of the following:

- A. Convenience Items
 - Equipment that basically serves comfort or convenience functions or is primarily for the convenience of a person caring for a Member (e.g., an exercycle or other physical fitness equipment, elevators, hoyer lifts, and shower/bath bench).
- B. Furniture items, movable objects or accessories that serve as a place upon which to rest (people or things) or in which things are placed or stored (e.g., chair or dresser).
- C. Exercise equipment

Any device or object that serves as a means for energetic physical action or exertion in order to train, strengthen, or condition all or part of the human body, (e.g., exercycle or other physical fitness equipment).

- D. Institutional equipment
 - Any device or appliance that is appropriate for use in a medical facility and not appropriate for use in the home (e.g., parallel bars).
- E. Environmental control equipment
 - Equipment that can be used for non-medical purposes, such as air conditioners, humidifiers, or electric air cleaners. These items are not covered even though they may be prescribed, in the individual's case, for a medical reason.
- F. Eyeglasses or contact lenses, dental prostheses, appliances, or hearing aids (except as otherwise provided herein for cleft lip or cleft palate or both or as stated in Section 2 and Section 3).

- G. Corrective shoes (unless required to be attached to a leg brace), shoe lifts, or special shoe accessories or inserts.
- H. Medical equipment/supplies of an expendable nature, except those specifically listed as covered Medical Devices and Supplies in this Out-of-Network Description of Covered Services. Non-covered supplies include incontinence pads or ace bandages.
- I. Tinnitus maskers.

doing business as

CareFirst BlueCross BlueShield
840 First Street, NE
Washington, DC 20065
202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

ATTACHMENT C OUT-OF-NETWORK SCHEDULE OF BENEFITS

The benefits and limitations described in this schedule are subject to all terms and conditions stated in the Evidence of Coverage.

Except as otherwise stated, benefits for Covered Services, Covered Dental Services, and Covered Vision Services may be provided either under the In-Network Evidence of Coverage or Out-of-Network Evidence of Coverage. Benefits will not be provided for the same service or supply under both this Out-of-Network Evidence of Coverage and the In-Network Evidence of Coverage. However, for certain services there are visit or other limitations. Where there is a benefit limitation, the benefit limitation is combined for both the In-Network and Out-of-Network Evidences of Coverage.

CareFirst pays only for Covered Services, Covered Dental Services and Covered Vision Services. The Member pays for services, supplies or care, which are not covered. The Member pays any applicable Deductible, Copayment or Coinsurance. Services that are not listed in the Description of Covered Services, or are listed in the Exclusions and Limitations, are not Covered Services, Covered Dental Services or Covered Vision Services.

When determining the benefits a Member may receive, CareFirst considers all provisions and limitations in the Evidence of Coverage as well as its medical policies. When these conditions of coverage are not met or followed, payments for benefits may be denied. Certain utilization management requirements will also apply. When these requirements are not met, payments may be reduced or denied.

OUT-OF-NETWORK DEDUCTIBLE

The Individual Benefit Period Deductible is \$6,000.

The Family Benefit Period Deductible is \$12,000.

Individual Coverage: The Member must satisfy the Individual Deductible.

Family Coverage: Each Member can satisfy his/her own Deductible by meeting the Individual Deductible. In addition, eligible expenses for all covered Members can be combined to satisfy the Family Deductible. An individual family Member may not contribute more than the Individual Deductible toward meeting the Family Deductible. Once the Family Deductible has been met, this will satisfy the Deductible for all covered family Members.

The benefit chart below states whether a Covered Service is subject to the Out-of-Network Deductible.

The following amounts may <u>not</u> be used to satisfy the Out-of-Network Deductible:

- Amounts incurred for failure to comply with the utilization management program requirements.
- Charges in excess of the Allowed Benefit.
- Charges for services which are not covered under the Evidence of Coverage or which exceed the maximum number of covered visits/days listed below.
- Charges for Covered Services not subject to the Deductible.
- Charges for Pediatric Vision Services or Pediatric Dental Services.
- Charges incurred under the In-Network Evidence of Coverage.

OUT-OF-NETWORK OUT-OF-POCKET MAXIMUM

The Individual Out-of-Network Out-of-Pocket Maximum is \$14,000.

The Family Out-of-Network Out-of-Pocket Maximum is \$28,000.

Individual Coverage: The Member must meet the Individual Out-of-Pocket Maximum.

Family Coverage: Each Member can satisfy his/her own Individual Out-of-Pocket Maximum by meeting the Individual Out-of-Pocket Maximum. In addition, eligible expenses of all covered family members can be combined to satisfy the Family Out-of-Pocket Maximum. An individual family member cannot contribute more than the Individual Out-of-Pocket Maximum toward meeting the Family Out-of-Pocket Maximum. Once the Family Out-of-Pocket Maximum has been met, this will satisfy the Out-of-Pocket Maximum for all family members.

These amounts apply to the Out-of-Network Out-of-Pocket Maximum:

- Out-of-Network Copayments and Coinsurance for all Covered Services.
- Out-of-Network Deductible
- Out-of-Network Pediatric Dental Deductible and Coinsurance for Out-of-Network Covered Dental Services.

When the Member has reached the Out-of-Pocket Maximum, no further Copayments, Coinsurance or Deductible will be required in that Benefit Period for Covered Services, Covered Dental Services and Covered Vision Services.

The following amounts may <u>not</u> be used to satisfy the Out-of-Network Out-of-Pocket Maximum:

- Amounts incurred for failure to comply with the utilization management program requirements.
- Charges in excess of the Allowed Benefit.
- Charges for services which are not covered under the Evidence of Coverage or which exceed the maximum number of covered visits/days listed below.
- Charges for Out-of-Network Covered Vision Services.
- Charges incurred under the In-Network Evidence of Coverage.

UTILIZATION MANAGEMENT

Failure or refusal to comply with utilization management program requirements will result in a 50% reduction in benefits for services associated with the Member's care or treatment (other than Prescription Drug, Pediatric Vision and Pediatric Dental benefits).

BENEFITS			
SUBJECT TO			
SERVICE	LIMITATIONS	DEDUCTIBLE?	MEMBER PAYS

The Member is responsible for any applicable Deductible, Copayment or Coinsurance listed in this schedule. When the Allowed Benefit for any Covered Service is less than the Copayment listed, the Member payment will be the Allowed Benefit.

When multiple services are rendered on the same day by more than one provider, Member payments are required for each provider.

The benefit chart below states whether a Covered Service is subject to the Member Copayment for Clinic Visits/ Outpatient Services rendered in a hospital, hospital clinic, or health care provider's office on a hospital campus ("Clinic Visit").

These providers <u>may</u> bill individually resulting in claims from both the hospital/facility and the physician or health care provider rendering care in the hospital/facility/clinic setting. It is the Member's responsibility to determine whether separate claims will be assessed.

OUTPATIENT FACILITY, OFFICE AND PRO	FESSIONAL SERVICES
Physician's Office	Yes \$50 per visit and \$150 per visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic
Outpatient Non-Surgical Services	Yes \$50 per visit and \$150 per visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic
Laboratory Tests, X-Ray/Radiology Services, Sp	
Non-Preventive Laboratory Tests (independent non-hospital laboratory)	Yes \$65 per visit
Non-Preventive Laboratory Tests (outpatient department of a hospital)	Yes \$110 per visit
Non-Preventive X- Ray/Radiology Services (independent non-hospital facility)	Yes \$80 per visit
Non-Preventive X- Ray/Radiology Services (outpatient department of a hospital)	Yes \$110 per visit
Non-Preventive Specialty Imaging (independent non-hospital facility)	Yes \$150 per visit
Non-Preventive Specialty Imaging (outpatient department of a hospital)	Yes \$250 per visit
Non-Preventive Diagnostic Testing (except as otherwise specified in an independent non-hospital facility)	Yes \$80 per visit
Non-Preventive Diagnostic	Yes \$110 per visit

BENEFITS			
SERVICE	LIMITATIONS	SUBJECT TO DEDUCTIBLE?	MEMBER PAYS
Testing (except as otherwise specified in an			
outpatient department of a			
hospital)			
Sleep Studies (Member's		Yes	\$70 per study
home)			
Sleep Studies (office or	Prior authorization is required.	Yes	\$200 per study
freestanding facility)	_		
Sleep Studies (outpatient	Prior authorization is required.	Yes	\$300 per study
department of a hospital)			

Preventive Care – Evidence-based items or services that have in effect a rating of "A" or "B" in the current recommendations of the United States Preventive Services Task Force (USPSTF). With respect to women, such additional preventive care and screenings as provided for in comprehensive guidelines supported by the Health Resources and Services Administration. At a minimum, benefits will be provided for breast cancer screening in accordance with the latest screening guidelines issued by the American Cancer Society or required by the

Patient Protection and Affordable Care Act (PPACA).

Prostate Cancer Screening	Yes	No Copayment or Coinsurance
Colorectal Cancer	Yes	No Copayment or
Screening	103	Coinsurance
Breast Cancer Screening	No	No Copayment or
		Coinsurance
Pap Smear	No	No Copayment or
		Coinsurance
Human Papillomavirus	Yes	No Copayment or
Screening Test		Coinsurance
Preventive Laboratory	Yes	No Copayment or
Tests		Coinsurance
Preventive X-	Yes	No Copayment or
Ray/Radiology Services		Coinsurance
Preventive Specialty	Yes	No Copayment or
Imaging		Coinsurance
Preventive Diagnostic	Yes	No Copayment or
Testing (except as		Coinsurance
otherwise specified)		
Immunizations	Yes	No Copayment or
*** 11 61 11 1 6		Coinsurance
Well Child Care	No	No Copayment or
4.1.1.D	**	Coinsurance
Adult Preventive Care	Yes	No Copayment or
W 1 D	***	Coinsurance
Women's Preventive	Yes	No Copayment or
Services	37	Coinsurance
Office Visits for Treatment	Yes	No Copayment or
of Obesity	****	Coinsurance
Professional Nutritional	Yes	No Copayment or
Counseling and Medical Nutrition Therapy		Coinsurance
Treatment Services		
Family Planning		
Non-Preventive	Yes	\$50 per visit and \$150 per
Gynecological Office Visits	1 68	visit if rendered in the
Gynecological Office Visits		outpatient department of a
		hospital/hospital clinic or
		nospitai/nospitai ciinic 0i

BENEFITS			
CEDIMOE	LIMITATIONS	SUBJECT TO	MEMBER DAYS
SERVICE	LIMITATIONS	DEDUCTIBLE?	mEMBER PAYS provider's office located in a
			hospital or hospital clinic
Contraceptive Counseling		Yes	\$50 per visit and \$150 per
community community		100	visit if rendered in the
			outpatient department of a
			hospital/hospital clinic or
			provider's office located in a
Contraceptive Drugs and	Coverage of self-administered	Yes	hospital or hospital clinic \$50 per device
Devices	contraceptive drugs and devices is	1 65	\$30 per device
201100	provided under the Prescription		
	Drugs benefit.		
Insertion or removal, and	Drug or device must be approved	Yes	\$50 per visit and \$150 per
any Medically Necessary	by the FDA as a contraceptive.		visit if rendered in the
examination associated			outpatient department of a
with the use of any contraceptive devices or			hospital/hospital clinic or provider's office located in a
drugs			hospital or hospital clinic
Elective Sterilization	Benefits available to female	Yes	\$50 per visit and \$150 per
Services – Female	Members with reproductive		visit if rendered in the
Members	capacity only.		outpatient department of a
			hospital/hospital clinic or
			provider's office located in a
Maternity and Related Ser	wiees		hospital or hospital clinic
Preventive Visit	VICES	Yes	\$50 per visit
Treventive visit		103	φου per visit
Non-Preventive Visit		Yes	\$50 per visit and \$150 per
			visit if rendered in the
			outpatient department of a
			hospital/hospital clinic or provider's office located in a
			hospital or hospital clinic
Professional Services for		Yes	\$50 per visit
Delivery		105	φου per visit
Allergy Services			
Allergy Testing and		Yes	\$50 per visit and \$150 per
Allergy Treatment			visit if rendered in the
			outpatient department of a
			hospital/hospital clinic or provider's office located in a
			hospital or hospital clinic
Allergy Shots		Yes	\$50 per visit and \$150 per
			visit if rendered in the
			outpatient department of a
			hospital/hospital clinic or
			provider's office located in a
Rehabilitation Services			hospital or hospital clinic
Rehabilitative Physical		Yes	\$50 per visit and \$150 per
Therapy		105	visit if rendered in the
1 3			outpatient department of a
			hospital/hospital clinic or
			provider's office located in a
			hospital or hospital clinic

Rehabilitative Occupational Therapy So per visit and \$150 per visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital		BENEFITS			
Rehabilitative Occupational Therapy Rehabilitative Speech Therapy Therapy Rehabilitative Speech Therapy Rehabilitative Speech Therapy Th	SEDVICE	LIMITATIONS		MEMRED DAVS	
Therapy outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or		LIMITATIONS			
Rehabilitative Speech Therapy T			103	visit if rendered in the	
Rehabilitative Speech Therapy Yes So per visit and \$150 per visit in content of a hospital clinic or provider's office located in a hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office loca					
Rehabilitative Speech Therapy Per Sopial or hospital clinic or sopial and S150 per visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital				hospital/hospital clinic or	
Rehabilitative Speech Therapy					
Therapy Visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a ho	Dahahilitatiya Craash		Vac	hospital or hospital clinic	
Spinal Manipulation Services Spinal Manipulation Services Spinal Manipulation Services Particles Limited to Members under the age of 21. Limited to Members under the age of 21. Limited to Members under the age of 21. Sport visit and \$150 per visit and the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office loca			1 68	visit if rendered in the	
Spinal Manipulation Services Solidad India or nospital clinic or provider's office located in a hospital or hospital clinic or services Solidad India or nospital clinic or provider's office located in a hospital or hospital clinic or visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's office located in a hospital hospital clinic or provider's offi	Therapy				
Spinal Manipulation Services Spinal Manipulation Services Spinal Manipulation Services Services Spinal Manipulation Services Services Spinal Manipulation Limited to Members under the age of 21. Limited to Members under the age of 21. Spinal Manipulation Limited to Members under the age of 21. Spinal Manipulation Member age of 21. Spinal Manipulation or provider's office located in the outpatient department of a hospital on hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's off				hospital/hospital clinic or	
Spinal Manipulation Services So per visit and \$150 per visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in					
Habilitative Services for Children	Spinal Manipulation		Vac	hospital or hospital clinic	
Habilitative Services for Children Habilitative Services for Children Habilitative Services for Children Habilitative Services for Children Habilitative Services for Adults Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage Evidence Of Evide			ies		
Habilitative Services for Children Habilitative Services for Adults Habilitative Services for Adults Cardiac Rehabilitation Cardiac Rehabilitation Pulmonary Rehabilitation Pulmonary Rehabilitation Limited to one pulmonary rehabilitation Coverage. Limited to one pulmonary Rehabilitation Coverage and the Out-of-Network Evidence of Coverage and the Out-	Services				
Habilitative Services for Children Habilitative Services for Children Benefits available for Member age 21 and older. Prior authorization is required for Habilitative services for Adults Cardiac Rehabilitation Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Pulmonary Rehabilitation Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and Evidence of Coverage and the Out-of-Network Evidence of Cover				hospital/hospital clinic or	
Habilitative Services for Children Limited to Members under the age of 21. Benefits available for Member age 21 and older. Habilitative Services for Adults Benefits available for Member age 21 and older. Prior authorization is required for Habilitative services for Adults. Cardiac Rehabilitation Cardiac Rehabilitation Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Dulmonary Rehabilitation Chreatment Services Other Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Limited to Members under the age visit if rendered in the outpatient department of a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clin					
Children of 21. Habilitative Services for Adults Prior authorization is required for Habilitative services for Adults Cardiac Rehabilitation Prior authorization is required for Habilitative services for Adults Cardiac Rehabilitation Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Limited to one pulmonary rehabilitation redained under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Other Treatment Services Outpatient Therapeutic Treatment Services Outpatient Therapeutic (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Benefits available for Member age Aboptical clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office l				hospital or hospital clinic	
Children of 21. Habilitative Services for Adults Prior authorization is required for Habilitative services for Adults Cardiac Rehabilitation Prior authorization is required for Habilitative services for Adults Cardiac Rehabilitation Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Limited to one pulmonary rehabilitation redained under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Other Treatment Services Outpatient Therapeutic Treatment Services Outpatient Therapeutic (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Benefits available for Member age Aboptical clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office l					
Children of 21. Habilitative Services for Adults Benefits available for Member age 21 and older. Cardiac Rehabilitation Cardiac Rehabilitation Pulmonary Rehabilitation Pulmonary Rehabilitation Pulmonary Rehabilitation Other Treatment Services Outpatient Department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or pr	Habilitative Services for	Limited to Members under the age	Yes	\$50 per visit and \$150 per	
Habilitative Services for Adults Benefits available for Member age 21 and older. Prior authorization is required for Habilitative services for Adults Cardiac Rehabilitation Cimited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage. Pulmonary Rehabilitation Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage. Pulmonary Rehabilitation Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage. Pulmonary Rehabilitation Pulmonary Rehabilitation Pulmonary Rehabilitation Other Treatment Services Outpatient Therapeutic Treatment Services Outpatient Therapeutic Treatment Services Outpatient Therapeutic Treatment Services Page 1 Adults Services Yes \$50 per visit and \$150 per visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in	Children			visit if rendered in the	
Habilitative Services for Adults Benefits available for Member age 21 and older. Prior authorization is required for Habilitative services for Adults. Cardiac Rehabilitation Cardiac Rehabilitation Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Limited to one pulmonary rehabilitation Pulmonary Rehabilitation Cardiac Rehabilitation Limited to one pulmonary rehabilitation Pulmonary Rehabilitation Cardiac Rehabilitation Descrices Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation pulmonary rehab					
Habilitative Services for Adults Benefits available for Member age 21 and older. Yes \$50 per visit and \$150 per visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital/hospital clinic or provider's office located in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located					
Habilitative Services for Adults Benefits available for Member age 21 and older. Prior authorization is required for Habilitative services for Adults. Cardiac Rehabilitation Cardiac Rehabilitation Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Limited to one pulmonary rehabilitation reduce the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and the Out-of-N					
Adults 21 and older. 21 and older. Prior authorization is required for Habilitative services for Adults. Cardiac Rehabilitation Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Energite for Adults. Yes Ves S50 per visit and \$150 per visit if rendered in the outpatient department of a hospital or hospital clinic or provider's office located in a hospital/hospital clinic or provider's office located in a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic				nospital of nospital elline	
Adults 21 and older. 21 and older. Prior authorization is required for Habilitative services for Adults. Cardiac Rehabilitation Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Energite of Coverage in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic	TI 1:1: C	D C. 111 C M 1	**	φσο : :. 1 φ1σο	
Prior authorization is required for Habilitative services for Adults. Cardiac Rehabilitation Cardiac Rehabilitation Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Limited to one pulmonary rehabilitation reprivation of a hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in the outpatient department of a hospital or hospital clinic or provider's office located in the outpatient department of a hospital/hospital clinic or provider's office located in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital or hospital clinic or provider's office located in a hospital or hospital or hospital clinic or provider's office located in a hosp			Yes		
Prior authorization is required for Habilitative services for Adults. Cardiac Rehabilitation Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Limited to one pulmonary rehabilitation Pulmonary Rehabilitation Presented and the Out-of-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and the In-Network Evidence of Coverage. Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Benefits are available to the same extent as	Adults	21 and older.			
Cardiac Rehabilitation Cardiac Rehabilitation Limited to 90 days per Benefit Period combined under the In- Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Pulmonary Rehabilitation Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Habilitative services for Adults. Yes \$50 per visit and \$150 per visit if rendered in the outpatient department of a hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or ho					
Cardiac Rehabilitation Limited to 90 days per Benefit Period combined under the In- Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Limited to 90 days per Benefit Period combined under the In-Network Evidence of Coverage and the In-Network Evidence of Coverage and the Out-oprovider's office located in a hospital clinic or provider's office located in a hospital clinic or provider's office located in a hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or ho					
Period combined under the In- Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Pulmonary Rehabilitation Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation and Infusion Blood and Blood Period combined under the In-Network Evidence of Coverage and hospital clinic or provider's office located in a hospital or hospital or hospital or hospital or hospital or hospital or hospit	Continu Dalatilitation		V.	hospital or hospital clinic	
Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Pulmonary Rehabilitation Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Network Evidence of Coverage and the Out-oprovider's office located in a hospital/hospital clinic or provider's office located in a hospital/hospital clinic or prov	Cardiac Renabilitation		res		
the Out-of-Network Evidence of Coverage. Pulmonary Rehabilitation Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood the Out-of-Network Evidence of Services of Fice located in a hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a ho					
Pulmonary Rehabilitation Pulmonary Rehabilitation Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Limited to one pulmonary rehabilitation one pulmonary rehabilitation are pulmonary rehabilitation and Infusion Services) Pulmonary Yes \$50 per visit and \$150 per provider's office located in a hospital clinic or provider's office located in a hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic Benefits are available to the same extent as				hospital/hospital clinic or	
Pulmonary Rehabilitation Limited to one pulmonary rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Limited to one pulmonary rehabilitation and Blood Limited to one pulmonary rehabilitation program per lifetime visit if rendered in the outpatient department of a hospital clinic or provider's office located in a hospital clinic Benefits are available to the same extent as		Coverage.			
rehabilitation program per lifetime combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Other Treatment Services Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Tehabilitation program per lifetime outpatient department of a hospital/hospital clinic or provider's office located in a hospital/hospital clinic or provider's office located in a hospital/hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital or hospital clinic	Dulmonomy Dohobilitation	Limited to one pulmonery	Voc		
combined under the In-Network Evidence of Coverage and the Out- of-Network Evidence of Coverage. Other Treatment Services Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Combined under the In-Network Evidence of Coverage and the Out- hospital clinic or provider's office located in a hospital or hospital clinic or provider's office located in a hospital/hospital clinic or provider's office located in a hospital or hospital or hospital or hospital or hospital clinic Benefits are available to the same extent as	Pullionary Renabilitation	rehabilitation program per lifetime	ies		
Evidence of Coverage and the Out- of-Network Evidence of Coverage. Other Treatment Services Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Evidence of Coverage and the Out- of-Network Evidence of Coverage. Nospital/hospital clinic or provider's office located in a hospital/hospital clinic or provider's office located in a hospital or hospital clinic		combined under the In-Network			
Nother Treatment ServicesOutpatient TherapeuticYes\$50 per visit and \$150 perTreatment Servicesvisit if rendered in the(excluding Cardiacoutpatient department of aRehabilitation, pulmonaryhospital/hospital clinic orrehabilitation and Infusionprovider's office located in aServices)hospital or hospital clinicBlood and BloodBenefits are available to the same extent as		Evidence of Coverage and the Out-		hospital/hospital clinic or	
Other Treatment ServicesOutpatient TherapeuticYes\$50 per visit and \$150 perTreatment Servicesvisit if rendered in the(excluding Cardiacoutpatient department of aRehabilitation, pulmonaryhospital/hospital clinic orrehabilitation and Infusionprovider's office located in aServices)hospital or hospital clinicBlood and BloodBenefits are available to the same extent as		of-Network Evidence of Coverage.			
Outpatient Therapeutic Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Yes \$50 per visit and \$150 per visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital or hospital clinic	Other Treetment Convices			hospital or hospital clinic	
Treatment Services (excluding Cardiac Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood visit if rendered in the outpatient department of a hospital/hospital clinic or provider's office located in a hospital or hospital or hospital clinic Benefits are available to the same extent as			Yes	\$50 per visit and \$150 per	
Rehabilitation, pulmonary rehabilitation and Infusion Services) Blood and Blood Rehabilitation, pulmonary hospital/hospital clinic or provider's office located in a hospital or hospital clinic Benefits are available to the same extent as	Treatment Services			visit if rendered in the	
rehabilitation and Infusion Services) Blood and Blood Benefits are available to the same extent as					
Services) hospital or hospital clinic Blood and Blood Benefits are available to the same extent as					
Blood and Blood Benefits are available to the same extent as					
			Benefits are availab		

BENEFITS				
SERVICE	LIMITATIONS	SUBJECT TO DEDUCTIBLE?	MEMBER PAYS	
Clinical Trial	Benefits are available to the same extent as benefits provided for other services			
Retail Health Clinic		Yes	\$50 per visit	
Telemedicine Services	Benefits are available to the same ex	tent as benefits provi	ided for other services	
Infusion Services				
Physician's Office	Prior authorization is required for Specialty Drugs in the Prescription Guidelines.	Yes	\$40 per session	
Free-Standing Infusion Center	Prior authorization is required for Specialty Drugs in the Prescription Guidelines.	Yes	\$40 per session	
Hospital Outpatient Department	Prior authorization is required for Specialty Drugs in the Prescription Guidelines.	Yes	\$300 per session	
Member's Home	Prior authorization is required for Specialty Drugs in the Prescription Guidelines.	Yes	\$40 per session	
Outpatient Surgical Facility	and Professional Services	<u> </u>	1	
Surgical Care at an Ambulatory Care Facility		Yes	\$150 per visit	
Outpatient Surgical Professional Services Provided at an Ambulatory Care Facility		Yes	\$50 per visit	
Surgical Care at an Outpatient Hospital Facility		Yes	\$250 per visit	
Outpatient Surgical Professional Services Provided at an Outpatient Hospital		Yes	\$50 per visit	
INPATIENT HOSPITAL S	ERVICES			
Inpatient Facility (medical or surgical condition, including maternity and rehabilitation)	Hospitalization solely for rehabilitation limited to ninety (90) days per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	Yes	\$300 per admission	
	Prior authorization is required except for emergency admissions and all maternity admissions.			
Inpatient Professional Services		Yes	\$50 per visit	
Organ and Tissue Transplants	Except for corneal transplants and kidney transplants, prior authorization is required.	Benefits are available to the same extent as benefits provided for other services		
SKILLED NURSING FAC				

BENEFITS			
GEDAN GE		SUBJECT TO	147147777
SERVICE	LIMITATIONS	DEDUCTIBLE?	MEMBER PAYS
Skilled Nursing Facility Services	Limited to 60 days per Benefit Period combined under the In-	Yes	\$50 per admission
Services	Network Evidence of Coverage and		
	the Out-of-Network Evidence of		
	Coverage.		
	Prior authorization is required.		
HOME HEALTH SERVICE Home Health Services		Yes	\$50 man visit
Home Hearm Services	Prior authorization is required.	1 68	\$50 per visit
	Limited to ninety (90) visits per		
	"episode of care" combined under		
	the In-Network Evidence of		
	Coverage and the Out-of-Network		
	Evidence of Coverage. A new		
	episode of care begins if the		
	Member does not receive Home		
	Health Care for the same or a		
	different condition for sixty (60) consecutive days.		
Postpartum Home Visits	Benefits are available to all	No	No Copayment or
1 ostpartam Home Visits	Members.	110	Coinsurance
HOSPICE SERVICES			
Inpatient Care	Prior authorization is required.	Yes	\$50 per admission
	Complete limited to a maximum one		
	Services limited to a maximum one hundred eighty (180) day hospice		
	eligibility period		
	engionity period		
	Limited to sixty (60) days per		
	hospice eligibility period combined		
	under the In-Network Evidence of		
	Coverage and the Out-of-Network		
Orational Comp	Evidence of Coverage.	V.	Φ50 ··············
Outpatient Care	Prior authorization is required.	Yes	\$50 per visit
	Services limited to a maximum one		
	hundred eighty (180) day hospice		
	eligibility period.		
Respite Care	Prior authorization is required.	Yes	\$50 per visit
	Compile the last transfer to the compile to the com		
	Services limited to a maximum one		
	hundred eighty (180) day hospice eligibility period.		
Bereavement	Prior authorization is required.	Yes	\$50 per visit
Services	11101 datio112ddoi 15 foquilod.	103	φο ο per vibit
	Covered only if provided within		
	ninety (90) days following death of		
	the deceased.		
	SUBSTANCE USE DISORDER SEI	RVICES	
Outpatient Services Office Visits		Yes	\$50 per visit
		105	to per visit
Outpatient Hospital Facility		Yes	\$50 per visit
Services			

LIMITATIONS	SUBJECT TO DEDUCTIBLE?	MEMBER PAYS
	Yes	\$50 per visit
	V	Φ50
	Yes	\$50 per visit
	Yes	\$50 per visit
	Yes	\$50 per visit
	Yes	\$50 per visit
Prior authorization is required except for emergency admissions.	Yes	\$300 per admission
, , , , , , , , , , , , , , , , , , ,	Yes	\$50 per visit
AND URGENT CARE		
Limited to unexpected, urgently		under the In-Network Evidence
		under the In-Network Evidence
unexpected, urgently required services.	of Coverage.	
Limited to Emergency Services or unexpected, urgently required services.	Benefits provided under the In-Network Eviden of Coverage.	
Limited to Emergency Services or unexpected, urgently required services.	Benefits provided under the In-Network Evider of Coverage.	
Prior authorization is required for air ambulance services, except for Medically Necessary air ambulance services in an emergency.	Benefits provided under the In-Network Evidence of Coverage.	
SUPPLIES		
	Yes	45% of the Allowed Benefit
Limited to one per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	Yes	45% of the Allowed Benefit
	Yes	45% of the Allowed Benefit
Coverage for Diabetes Supplies will be provided under the Prescription Drug benefit.	Yes	45% of the Allowed Benefit NESS AND DISEASE
	Prior authorization is required except for emergency admissions. SAND URGENT CARE Limited to unexpected, urgently required services. Limited to Emergency Services or unexpected, urgently required services. Limited to Emergency Services or unexpected, urgently required services. Limited to Emergency Services or unexpected, urgently required services. Prior authorization is required for air ambulance services, except for Medically Necessary air ambulance services in an emergency. SUPPLIES Limited to one per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage. Coverage for Diabetes Supplies will be provided under the	Prior authorization is required except for emergency admissions. Yes

BENEFITS			
SERVICE	LIMITATIONS	SUBJECT TO DEDUCTIBLE?	MEMBER PAYS
Associated Costs for the Patient-Centered Medical Home Program (PCMH)		Benefits provided to of Coverage	under the In-Network Evidence
TCCI Program Elements		of Coverage	inder the In-Network Evidence
Services Provided Pursuant to an Active Plan of Care under the BHCC Program, CCC Program, or SUD Program		Benefits provided to of Coverage	under the In-Network Evidence
Health Promotion and Wellness		Benefits provided up of Coverage	inder the In-Network Evidence
Disease Management		Benefits provided u of Coverage	under the In-Network Evidence
PRESCRIPTION DRUGS			

Benefits provided under the In-Network Evidence of Coverage

Pediatric Vision – Benefit limited to Members up to age 19. If Member is under age 19 at the start of the Benefit Period but turns 19 during the Benefit Period, then the Member will receive Covered Vision Services up to the end of the Calendar Year in which the Member turns age 19.

up to the end of the Calendar Year in which the Member turns age 19.				
			MEMBER PAYS	
SERVICE	LIMITATIONS	SUBJECT TO THE DEDUCTIBLE?	NON-CONTRACTING VISION PROVIDER	
Eye Examination	Limited to one per Benefit Period combined under the In- Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	No	Expenses in excess of the Vision Allowed Benefit of \$40 are a non- Covered Vision Service	
as glare resistant t in additional costs	reatment, ultraviolet coating, progre	ents: "Basic" means spec essive lenses, transitional	ctacle lenses with no "add-ons" such lenses and others which may result	
Basic Single vision	Limited to one pair per Benefit Period combined under the In- Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	No	Expenses in excess of the Vision Allowed Benefit of \$40 are a non- Covered Vision Service	
Basic Bifocals	Limited to one pair per Benefit Period combined under the In- Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	No	Expenses in excess of the Vision Allowed Benefit of \$60 are a non- Covered Vision Service	
Basic Trifocals	Limited to one pair per Benefit Period combined under the In- Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	No	Expenses in excess of the Vision Allowed Benefit of \$80 are a non- Covered Vision Service	
Basic Lenticular	Limited to one pair per Benefit Period combined under the In- Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	No	Expenses in excess of the Vision Allowed Benefit of \$100 are a non-Covered Vision Service	
Frames				
Frames	Limited to one frame per Benefit Period combined under the In- Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	No	Expenses in excess of the Vision Allowed Benefit of \$70 are a non- Covered Vision Service	
Low Vision				
Low Vision Eye Examination	Prior authorization is required. It is the Member's responsibility to obtain prior authorization for services obtained from a Non-Contracting Vision Provider. Limited to one comprehensive low vision evaluation every 5 years and 4 follow-up visits in any 5-year period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	No	Expenses in excess of the Vision Allowed Benefit of \$300 are a non-Covered Vision Service	

Pediatric Vision – Benefit limited to Members up to age 19. If Member is under age 19 at the start of the Benefit Period but turns 19 during the Benefit Period, then the Member will receive Covered Vision Services up to the end of the Calendar Year in which the Member turns age 19.

ap to the cha of			MEMDED DAYS
	LIMITATIONS		MEMBER PAYS
SERVICE	LIMITATIONS	SUBJECT TO THE DEDUCTIBLE?	NON-CONTRACTING VISION PROVIDER
Follow-up care	Prior authorization required.	No	Expenses in excess of the Vision
	It is the Member's responsibility to obtain prior authorization for services obtained from a Non- Contracting Vision Provider.		Allowed Benefit of \$100 are a non-Covered Vision Service
	Limited to four visits in any five-year period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.		
High-power Spectacles, Magnifiers and Telescopes	Prior authorization is required. It is the Member's responsibility to obtain prior authorization for services obtained from a Non-Contracting Vision Provider.	No	Expenses in excess of the Vision Allowed Benefit of \$600 are a non-Covered Vision Service
Contact Lenses			
Elective	Includes evaluation, fitting and follow-up fees. Limited to one pair per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	No	Expenses in excess of the Vision Allowed Benefit of \$105 are a non-Covered Vision Service
Medically Necessary	Prior authorization is required. It is the Member's responsibility to obtain prior authorization for services obtained from a Non-Contracting Vision Provider. Limited to one pair per Benefit Period combined under the In-Network Evidence of Coverage and the Out-of-Network Evidence of Coverage.	No	Expenses in excess of the Vision Allowed Benefit of \$225 are a non-Covered Vision Service

Pediatric Dental – Limited to Members up to age 19. If Member is under age 19 at the start of the Benefit Period but turns 19 during the Benefit Period, then the Member will receive Covered Dental Services up to the end of the Calendar Year in which the Member turns age 19.

Pediatric Dental Deductible

The Out-of-Network Deductible of \$50 per Member per Benefit Period applies to all Class II, III, and IV Covered Dental Services.

Pediatric Dental Out-of-Pocket Maximum

Amounts paid by the Member for Covered Pediatric Dental Services will be applied to the Out-of-Network Out-of-Pocket Maximum stated above. Once the Out-of-Network Out-of-Pocket Maximum has been reached, the Member will no longer be required to pay any Deductible or Coinsurance.

		SUBJECT TO	MEMBER PAYS
SERVICE	LIMITATIONS	PEDIATRIC DENTAL DEDUCTIBLE?	NON-PREFERRED DENTIST
Class I		No	20% of the Pediatric Dental Allowed Benefit
Preventive &			
Diagnostic Services			
Class II		Yes	40% of the Pediatric Dental Allowed Benefit
Basic Services			
Class III		Yes	40% of the Pediatric Dental Allowed Benefit
Major Services –			
Surgical			
Class IV		Yes	65% of the Pediatric Dental Allowed Benefit
Major Services –			
Restorative			
Class V	Limited to Medically	No	65% of the Pediatric Dental Allowed Benefit
Orthodontic	Necessary		
Services	Orthodontia		

Group Hospi	talization and Medi	cal Services, Inc.
	[Signature]	
	[Name]	
	[Title]	

doing business as

CareFirst BlueCross BlueShield (CareFirst)

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

OUT-OF-NETWORK INTER-PLAN ARRANGEMENTS DISCLOSURE AMENDMENT

This amendment is effective on the effective date of the Evidence of Coverage to which this amendment is attached.

Out-of-Area Services

Overview

CareFirst has a variety of relationships with other Blue Cross and/or Blue Shield Licensees. Generally, these relationships are called "Inter-Plan Arrangements." These Inter-Plan Arrangements work based on rules and procedures issued by the Blue Cross Blue Shield Association ("Association"). Whenever you access healthcare services outside the geographic area CareFirst serves, the claim for those services may be processed through one of these Inter-Plan Arrangements. The Inter-Plan Arrangements are described below.

When you receive care outside of CareFirst service area, you will receive it from one of two kinds of providers. Most providers ("participating/PPO providers") contract with the local Blue Cross and/or Blue Shield Plan in that geographic area ("Host Blue"). Some providers ("non-participating providers") don't contract with the Host Blue. CareFirst explains below how we pay both kinds of providers.

Inter-Plan Arrangements Eligibility – Claim Types

All claim types are eligible to be processed through Inter-Plan Arrangements, as described above, except for all Dental Care Benefits except when paid as medical claims/benefits, and those Prescription Drug Benefits or Vision Care Benefits that may be administered by a third party contracted by CareFirst to provide the specific service or services.

A. BlueCard® Program

Under the BlueCard® Program, when you receive covered services within the geographic area served by a Host Blue, CareFirst will remain responsible for doing what we agreed to in the contract. However, the Host Blue is responsible for contracting with and generally handling all interactions with its participating/PPO providers.

When you receive covered services outside the CareFirst service area and the claim is processed through the BlueCard Program, the amount you pay for covered services is calculated based on the lower of:

- The billed charges for covered services; or
- The negotiated price that the Host Blue makes available to CareFirst.

Often, this "negotiated price" will be a simple discount that reflects an actual price that the Host Blue pays to your healthcare provider. Sometimes, it is an estimated price that takes into account special arrangements with your healthcare provider or provider group that may include types of settlements, incentive payments and/or other credits or charges. Occasionally, it may be an average price, based on a discount that results in expected average savings for similar types of healthcare providers after taking into account the same types of transactions as with an estimated price.

Estimated pricing and average pricing also take into account adjustments to correct for over- or underestimation of past pricing of claims, as noted above. However, such adjustments will not affect the price CareFirst has used for your claim because they will not be applied after a claim has already been paid.

B. Inter-Plan Programs: Federal/State Taxes/Surcharges/Fees

Federal or state laws or regulations may require a surcharge, tax or other fee that applies to insured accounts. If applicable, CareFirst will include any such surcharge, tax or other fee as part of the claim charge passed on to you.

C. Non-participating Providers Outside CareFirst Service Area

1. Member Liability Calculation

When covered services are provided outside of CareFirst service area by non-participating providers, the amount you pay for such services will normally be based on either the Host Blue's non-participating provider local payment or the pricing arrangements required by applicable state law. In these situations, you may be responsible for the difference between the amount that the non-participating provider bills and the payment CareFirst will make for the covered services as set forth in this paragraph. Federal or state law, as applicable, will govern payments for out-of-network emergency services.

2. Exceptions

In certain situations, CareFirst may use other payment methods, such as billed charges for covered services, the payment we would make if the covered services had been obtained within our service area, or a special negotiated payment to determine the amount CareFirst will pay for services provided by non-participating providers. In these situations, you may be liable for the difference between the amount that the non-participating provider bills and the payment CareFirst will make for the covered services as set forth in this paragraph.

D. Blue Cross Blue Shield Global® Core

If you are outside the United States (hereinafter "BlueCard service area"), you may be able to take advantage of the Blue Cross Blue Shield Global® Core when accessing covered services. The Blue Cross Blue Shield Global® Core is unlike the BlueCard Program available in the BlueCard service area in certain ways. For instance, although the Blue Cross Blue Shield Global® Core assists you with accessing a network of inpatient, outpatient and professional providers, the network is not served by a Host Blue. As such, when you receive care from providers outside the BlueCard service area, you will typically have to pay the providers and submit the claims yourself to obtain reimbursement for these services.

If you need medical assistance services (including locating a doctor or hospital) outside the BlueCard service area, you should call the service center at 1.800.810.BLUE (2583) or call collect at 1.804.673.1177, 24 hours a day, seven days a week. An assistance coordinator, working with a medical professional, can arrange a physician appointment or hospitalization, if necessary.

• Inpatient Services

In most cases, if you contact the service center for assistance, hospitals will not require you to pay for covered inpatient services, except for your deductibles, coinsurance, etc. In such cases, the hospital will submit your claims to the service center to begin claims processing. However, if you paid in full at the time of service, you must submit a claim to receive reimbursement for covered services. You must contact CareFirst to obtain precertification for non-emergency inpatient services.

• Outpatient Services

Physicians, urgent care centers and other outpatient providers located outside the BlueCard service area will typically require you to pay in full at the time of service. You must submit a claim to obtain reimbursement for covered services.

Submitting a Blue Cross Blue Shield Global[®] Core Claim

When you pay for covered services outside the BlueCard service area, you must submit a claim to obtain reimbursement. For institutional and professional claims, you should complete a Blue Cross Blue Shield Global® Core claim form and send the claim form with the provider's itemized bill(s) to the service center (the address is on the form) to initiate claims processing. Following the instructions on the claim form will help ensure timely processing of your claim. The claim form is available from CareFirst, the service center or online at www.bcbsglobalcoare.com. If you need assistance with your claim submission, you should call the service center at 1.800.810.BLUE (2583) or call collect at 1.804.673.1177, 24 hours a day, seven days a week.

This amendment is issued to be attached to the Evidence of Coverage. This amendment does not change the terms and conditions of the Evidence of Coverage, unless specifically stated herein.

Group Hospitalization and Medical Services, Inc.

[Signature.]
[Name.] [Title.]

doing business as

CareFirst BlueCross BlueShield (CareFirst)

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

INTER-PLAN PROGRAM ANCILLARY SERVICES AMENDMENT

This amendment is effective on the effective date or renewal date of the Evidence of Coverage to which this amendment is attached.

As used in this amendment, "Out-of-Area Covered Ancillary Services" mean:

- 1. Independent Clinical Laboratory Tests (performed at non-hospital based labs)
- 2. Medical Devices and Supplies
- 3. Specialty Prescription Drugs (non-routine, biological therapeutics such as injectables, infusion therapies, high-cost therapies, and therapies that require complex care)

Under the BlueCard® Program, Members are able to obtain Covered Ancillary Services outside the geographic area that CareFirst services. This program allows Members to obtain Out-of-Area Covered Ancillary Services from providers that have a contractual agreement (i.e., are "participating providers" or "contracted providers") with the local Blue Cross and/or Blue Shield Licensee in another geographic area, as well as non-participating providers in some instances.

As used in this amendment, the "Local Plan" means the plan that is responsible for processing Out-of-Area Covered Ancillary Services claims under the BlueCard® Program.

Member payment for Out-of-Area Covered Ancillary Services at the participating or non-participating provider payment level is determined by the relationship between the provider and the Local Plan. If the provider of Covered Ancillary Services has a contract with the Local Plan (a participating provider), the Member is responsible for the participating provider member payment as stated in the Inter-Plan Arrangements Disclosure Amendment.

If the provider of Covered Ancillary Services does not have a contract with the Local Plan (a non-participating provider), the Member is responsible for the non-participating provider member payment as stated in the Inter-Plan Arrangements Disclosure Amendment.

For Out-of-Area Covered Ancillary Services, the Local Plan is determined as follows:

Independent Clinical Laboratory Tests - if the referring provider is located in the same service area where the specimen was drawn, the plan of the service area where the specimen was drawn is the Local Plan; if the referring provider is not located in the same service area where the specimen was drawn, the plan of the service area where the referring provider is located is the Local Plan.

Medical Devices and Supplies - the plan of the service area where the equipment was shipped to or purchased at a retail store is the Local Plan.

Specialty Prescription Drugs - the plan of the service area where the ordering physician is located is the Local Plan.

This amendment is subject to all of the terms and conditions of the Evidence of Coverage to which it is attached and does not change any terms or conditions, except as specifically stated herein.

Group Hospitalization and Medical Services, Inc.

doing business as
CareFirst BlueCross BlueShield

840 First Street, NE Washington, DC 20065

(202) 479-8000

An independent licensee of the Blue Cross and Blue Shield Association

2019 AMENDMENT

This amendment is effective on the effective date or renewal date of the Evidence of Coverage to which this amendment is attached.

TABLE OF CONTENTS

SECTION A – DEFINITIONS

SECTION B – SPECIAL ENROLLMENT PERIODS

SECTION C - RESCISSION

SECTION D - SUBROGATION

SECTION E - CLINICAL TRIAL PATIENT COST COVERAGE

SECTION F – MATERNITY SERVICES

SECTION G – EXCLUSIONS AND LIMITATIONS

SECTION A – DEFINITIONS

The definition of "Substance Abuse" and all references to "Substance Abuse" in the Evidence of Coverage are deleted and replaced with the following:

<u>Substance Use Disorder means:</u>

- A. <u>Alcohol Use Disorder</u> means a disease that is characterized by a pattern of pathological use of alcohol with repeated attempts to control its use, and with negative consequences in at least one of the following areas of life: medical, legal, financial, or psycho-social; or
- B. <u>Drug Use Disorder</u> means a disease that is characterized by a pattern of pathological use of a drug with repeated attempts to control the use, and with negative consequences in at least one of the following areas of life: medical, legal, financial, or psycho-social.

SECTION B - SPECIAL ENROLLMENT PERIODS

- 1. Section 2.5C.1.a)(2) in the Evidence of Coverage is deleted. All subsequent provisions are renumbered accordingly.
- 2. Section 2.5C.1.f) in the Evidence of Coverage is deleted and replaced with the following:
 - f) The Qualified Employee or his or her Dependent gains access to new Qualified Health Plans as a result of a permanent move and either:
 - (1) had minimum essential coverage as described in 26 CFR 1.5000A-1(b) for one or more days during the sixty (60) days preceding the date of the permanent move; or,

- (2) was living in a foreign country or in a United States territory for one or more days during the sixty (60) days preceding the date of the permanent move.
- 3. Section 2.5C.1.g) in the Evidence of Coverage is deleted and replaced with the following:
 - g) The Qualified Employee is an Indian, as defined in section 4 of the Indian Health Care Improvement Act, who may enroll in a Qualified Health Plan or change coverage from one Qualified Health Plan to another one time per month or who is or becomes a Dependent of an Indian, as defined by section 4 of the Indian Health Care Improvement Act and is enrolled or is enrolling in a Qualified Health Plan through an Exchange on the same application as the Indian, may change from one Qualified Health Plan to another one time per month, at the same time as the Indian.
- 4. Section 2.5C.1 in the Evidence of Coverage is amended to add the following:
 - i) The Qualified Employee or his or her Dependent:
 - (1) is a victim of domestic abuse or spousal abandonment, as defined by 26 CFR 1.36B-2T, as amended, including a dependent or unmarried victim within a household, is enrolled in minimum essential coverage and seeks to enroll in coverage separate from the perpetrator of the abuse or abandonment; or,
 - (2) is a dependent of a victim of domestic abuse or spousal abandonment, on the same application as the victim, may enroll in coverage at the same time as the victim.
 - j) The Qualified Employee or his or her Dependent:
 - (1) applies for coverage on the Exchange during the annual open enrollment period or due to a qualifying event, is assessed by the SHOP Exchange as potentially eligible for Medicaid or the Children's Health Insurance Program (CHIP), and is determined ineligible for Medicaid or CHIP by the State Medicaid or CHIP agency either after open enrollment has ended or more than sixty (60) days after the qualifying event; or
 - applies for coverage at the State Medicaid or CHIP agency during the annual open enrollment period, and is determined ineligible for Medicaid or CHIP after open enrollment has ended;
 - k) The Qualified Employee, or his or her Dependent, adequately demonstrates to the SHOP Exchange that a material error related to plan benefits, Service Area, or Premium influenced the eligible individual's decision to purchase a QHP through the SHOP Exchange.
- 5. Section 2.5C in the Evidence of Coverage is amended to add the following:
 - 5. A Dependent of a Qualified Employee is not eligible for a Special Enrollment Period if the Group does not extend coverage the offer of coverage to Dependents.
- 6. Section 2.5D.7 in the Evidence of Coverage is deleted and replaced with the following:
 - 7. The Effective Date for a Qualified Employee or Dependent who timely enrolls due to a qualifying event stated in Section 2.5C.1.d) (enrollment or non-enrollment was unintentional, inadvertent, or erroneous and is the result of an error by the SHOP Exchange or the United States Department of Health and Human Services), Section 2.5C.1.e) (a Qualified Health Plan substantially violated a material provision of its contract), Section 2.5C.1.h) (other exceptional circumstances as determined by the SHOP Exchange), Section 2.5C.1.j) (applies for a QHP during annual open enrollment, is assessed

by the SHOP-Exchange as potentially eligible for Medicaid or CHIP and then determined ineligible or applied for Medicaid or CHIP during annual open enrollment and determined to be ineligible) and 2.5C.1.k) (demonstration to the Exchange that a material error related to plan benefits, service area, or premium influenced the individual's decision to purchase a QHP) shall be the appropriate date based on the circumstances of the Special Enrollment Period as determined by the Exchange.

SECTION C – RESCISSION

Section 3.3 in the Evidence of Coverage is amended to add the following:

D. CareFirst demonstrates, to the reasonable satisfaction of the SHOP Exchange, if required by the SHOP Exchange, that the rescission is appropriate

SECTION D – SUBROGATION

Section 5.4, Subrogation, in the Evidence of Coverage is deleted and replaced with the following:

5.4 <u>Subrogation</u>

- A. CareFirst has subrogation and reimbursement rights. Subrogation requires the Member to turn over to CareFirst any rights the Member may have against a third party. A third party is any person, corporation, insurer or other entity that may be liable to a Member for an injury or illness. Subrogation applies to any illness or injury which is:
 - 1. Caused by an act or omission of a third party; or
 - 2. Covered under a member's uninsured or underinsured policy issued to or otherwise covering the Member; or
 - 3. Covered by No Fault Insurance. <u>No Fault Insurance</u> means motor vehicle casualty insurance. This term also refers to motor vehicle insurance issued under any other state or federal legislation of similar purpose.
- B. If the Member receives or is entitled to receive payment from any person, organization or entity in connection with an injury, illness or need for care for which benefits were provided or will be provided under this Evidence of Coverage, the payment will be treated as having been paid to the Member as a recovery for the medical, hospital and other expenses for which CareFirst provided or will provide benefits. CareFirst may recover the amounts paid or will pay in benefits up to the amount received from or on behalf of the third party or applicable first party coverage.
- C. CareFirst's right of recovery is not subject to reduction for attorney's fees and costs under the "common fund" or any other doctrine. If required by law, CareFirst will reduce the amount owed by the Member to CareFirst in accordance with applicable law.
- D. CareFirst will have a lien on all funds the Member recovers up to the total amount of benefits provided. We are entitled under our right of recovery to be reimbursed for our benefit payments even if you are not "made whole" for all of your damages in the recoveries that you receive. CareFirst may give notice of that lien to any party who may have contributed to the Member's loss, or who may be liable for payment as a result of that loss.
- E. CareFirst has the option to be subrogated to the Member's rights to the extent of the benefits provided under this Evidence of Coverage. This includes CareFirst right to bring suit or

file claims against the third party in the Member's name.

F. Members agree to take action, furnish information and assistance, and execute such instruments that CareFirst may require while enforcing CareFirst rights under this Section. The Member agrees to not take any action which prejudices CareFirst's rights and interests under this provision.

SECTION E - CLINICAL TRIAL PATIENT COST COVERAGE

The definition of "Qualified Individual" in Section 1.14, Clinical Trial Patient Cost Coverage, in the Description of Covered Services, is deleted and replaced with the following:

Qualified Individual, as used in this section, means a Member who is eligible to participate in an approved clinical trial according to the trial protocol, with respect to the prevention, early detection, treatment or monitoring of cancer, chronic disease, or other life-threatening illness, and the provider who recommended the Member for the clinical trial has concluded that the Member's participation in such trial is appropriate to prevent, detect early, treat or monitor cancer, chronic disease, or life-threatening illness, or the Member's participation is based on medical and scientific information.

The definition of "Routine Patient Costs" in Section 1.14, Clinical Trial Patient Cost Coverage, in the Description of Covered Services, is deleted and replaced with the following:

Routine Patient Costs means the costs of all Medically Necessary items and health care services consistent with the Covered Services that are typically provided for a Qualified Individual who is not enrolled in a clinical trial that are incurred as a result of the item, device, or service being provided to the Qualified Individual for purposes of the clinical trial. Routine Patient Costs do not include the investigational item, device, or service itself; items or services that are provided solely to satisfy data collection and analysis needs and that are not used in the direct clinical management of the patient; or a service that is clearly inconsistent with widely accepted and established standards of care for a particular diagnosis.

Section B, Covered Services, is deleted and replaced with the following:

B. Covered Services

- 1. Benefits for Routine Patient Costs to a Qualified Individual in a clinical trial will be provided if the Qualified Individual's participation in the clinical trial is the result of prevention, early detection, treatment or monitoring of cancer, chronic disease, or other life-threatening illness
- 2. Coverage for Routine Patient Costs will be provided only if:
 - a) The item device or service is being provided or the studies are being conducted in a Phase I, Phase II, Phase III, or Phase IV clinical trial for cancer; or,
 - b) The item device or service is being provided in a Phase I, Phase II, Phase III, or Phase IV clinical trial for any other life-threatening disease or condition, or chronic disease;
 - c) The item device or service is being provided in a federally funded or approved clinical trial approved by one of the National Institutes of Health, the Centers for Disease Control and Prevention, the Agency for Health Care Research and quality, the Centers for Medicare and Medicaid Services, an NIH Cooperative Group, an NIH Center, the FDA in the form of an investigational new drug or device application, the federal Department of Defense, the federal Department of Veterans Affairs, the federal Department of Energy or a qualified nongovernmental research entity to which the National Cancer Institute has

awarded a support grant,, or an institutional review board of an institution in a state that has a Multiple Project Assurance Contract approved by the Office of Protection from Research Risks of the NIH;

- d) The item device or service is being provided under a drug trial that is exempt from the requirement of an investigational new drug application.
- e) The facility and personnel providing the item device or service are capable of doing so by virtue of their experience, training, and volume of patients treated to maintain expertise;
- 3. Coverage is provided for the Routine Patient Costs incurred for drugs and devices that have been approved for sale by the FDA whether or not the FDA has approved the drug or device for use in treating the Qualified Individual's particular condition, to the extent that the drugs or devices are not paid for by the manufacturer, distributor, or provider of that drug or device.

SECTION F - MATERNITY SERVICES

Description of Covered Services, Section 1.5 C.2.c), Non-Preventive Services, is deleted and replaced with the following:

c) Non-preventive routine professional services rendered to the newborn during a covered hospitalization for delivery. A newborn Dependent child will be automatically covered for the first thirty-one (31) days following the child's birth. The Agreement describes the steps, if any, necessary to enroll a newborn Dependent child.

SECTION G – EXCLUSIONS AND LIMITATIONS

Section 15.1.JJ, Exclusions and Limitations, is deleted and replaced with the following exclusion:

JJ. Services required solely for administrative purposes, for example, employment, insurance, foreign travel, school, camp admissions or participation in sports activities.

This amendment is issued to be attached to the Evidence of Coverage. This amendment does not change the terms and conditions of the Evidence of Coverage, unless specifically stated herein.

Group Hospitalization and Medical Services, Inc.

[Signature]

[Name]
[Title]

doing business as

CareFirst BlueCross BlueShield (CareFirst)

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

PATIENT PROTECTION DISCLOSURE NOTICE

Primary Care Provider Designation

CareFirst generally requires the designation of a primary care provider. You have the right to designate any primary care provider who participates in our network and who is available to accept you or your family members. Until you make this designation, CareFirst designates one for you. For information on how to select a primary care provider, and for a list of the participating primary care providers, contact the CareFirst at the customer service telephone number listed on your identification card.

For children, you may designate a CareFirst pediatrician as the primary care provider.

Obstetrics and Gynecological Care

You do not need prior authorization from CareFirst or from any other person (including a primary care provider) in order to obtain access to obstetrical or gynecological care from a health care professional in our network who specializes in obstetrics or gynecology. The health care professional, however, may be required to comply with certain procedures, including obtaining prior authorization for certain services, following a pre-approved treatment plan, or procedures for making referrals. For a list of CareFirst health care professionals who specialize in obstetrics or gynecology, contact CareFirst at customer service telephone number listed on your identification card.

Group Hospitalization and Medical Services, Inc.

doing business as
CareFirst BlueCross BlueShield (CareFirst)

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association

SUMMARY OF GENERAL PURPOSES, COVERAGE LIMITATIONS AND CONSUMER PROTECTION

General Purposes

Residents of the District of Columbia should know that licensed insurers who sell health insurance, life insurance, and annuities in the District of Columbia are members of the District of Columbia Life and Health Insurance Guaranty Association ("Guaranty Association").

The purpose of the Guaranty Association is to assure that policy or contract holders of certain types of insurance policies and contracts are covered up to the statutory levels of protection of contractual benefits in the unlikely event that a member insurer is unable to meet its financial obligations and found by a court of law to be insolvent. When a member company is found by a court to be insolvent, the Guaranty Association will assess its other member insurers to provide benefits on any outstanding covered claims of persons who reside in the District of Columbia. However, this additional protection provided through the Guaranty Association is subjected to certain statutory limits explained under "Coverage Limitations" section, below. In some cases, the Guaranty Association may facilitate the reassignment of policies or contracts to other licensed insurance companies to keep them in-force, with no change in contractual rights or benefits.

Coverage

The District of Columbia Life and Health Insurance Guaranty Association ("Guaranty Association"), established pursuant to the Life and Health Guaranty Association Act of 1992 ("Act"), effective July 22, 1992 (D.C. Law 9-129; D.C. Official Code §31-5401 et seq.), provides insolvency protection for certain types of insurance policies and contracts. NOTE: Certain policies and contracts may not be covered or fully covered.

The insolvency protections provided by the Guaranty Association are generally conditioned on an individual being a resident of the District and are the insured or owner under a health insurance, life insurance, or annuity contract issued by a member insurer, or they are insured under a under a group policy insurance contract issued by a member insurer. Beneficiaries, payees, or assignees of the District insureds are also covered under the Act, even if they live in another state.

Coverage Limitations

The Act also limits the amount the Guaranty Association is obligated to pay. The benefits for which the Guaranty Association may become liable shall be limited to the lesser of:

- The contractual obligations for which the insurer is liable for which the insurer would have been liable if it were not an impaired or insolvent insurer; or
- With respect to any one life, regardless of the number of policies, contracts, or certificates:
 - ▶ \$300,000 in life insurance death benefits for any one life; including net cash surrender or net cash withdrawal values;
 - ▶ \$300,000 in the present value of annuity benefits, including net cash surrender or net cash withdrawal values;
 - ▶ \$300,000 in the present value of structured settlement annuity benefits, including net cash surrender or net cash withdrawal values;
 - ▶ \$300,000 for long-term insurance care benefits;
 - ► \$300,000 for disability insurance;
 - ▶ \$500,000 for basic hospital, medical, and surgical insurance, or major medical insurance;
 - ▶ \$100,000 for coverage not defined as disability insurance or basic hospital, medical and surgical insurance or major medical insurance or long-term care insurance including any net cash surrender and net cash withdrawal values.

In no event is the Guaranty Association liable for more than \$300,000 with respect to any one life (\$500,000 in the event of basic hospital, medical, and surgical, and major medical claims).

Additionally, the Guaranty Association is not obligated to cover more than \$5,000,000 for multiple non-group policies of life insurance with one owner of regardless of the number of policies owned.

Exclusions Examples

Policy or contract holders are not protected by the Guaranty Association if:

- They are eligible for protection under the laws of another state (this may occur when the insolvent insurer was domiciled in a state whose guaranty association law protects insureds that live outside of that state);
- Their insurer was not authorized to do business in the District of Columbia; or
- Their policy was issued by a charitable organization, a fraternal benefit society, a mandatory state pooling plan, a mutual assessment company, an insurance exchange, a non-profit hospital or medical service organization, a health maintenance organization, or a risk retention group.

The Guaranty Association also does not cover:

- Any policy or portion of a policy which is not guaranteed by the insurer or for which the individual has assumed the risk;
- Any policy of reinsurance (unless an assumption certificate was issued);
- Any plan or program of an employer or association that provides life, health, or annuity benefits to its employees or members and is self-funded;
- Interest rate guarantees which exceed certain statutory limitations;
- Dividends, experience rating credits or fees for services in connection with a policy;
- Credits given in connection with the administration of a policy by a group contract holder; or
- Unallocated annuity contracts.

Consumer Protection

To learn more about the above referenced protections, please visit the Guaranty Association's website at www.dclifega.org. Additional questions may be directed to the District of Columbia Department of Insurance, Securities and Banking (DISB) will respond to questions not specifically addressed in this disclosure document.

Policy or contract holders with additional questions may contact either:

District of Columbia Department of Insurance, Securities and Banking 1050 First Street, N.E., Suite 801 Washington, DC 20002 (202) 727-8000 District of Columbia Life and Health Guaranty Association 1200 G Street, N.W. Washington, DC 20005 (202) 434-8771

Pursuant to the Act (D.C. Official Code § 31-5416), insurers are required to provide notice to policy and contract holders of the existence of the Guaranty Association and statutory coverage protections. Your insurer and agent are prohibited by law from using the existence of the Guaranty Association and the protection it provides to market insurance products. You should not rely on insolvency protection provided under the Act when selecting an insurer or insurance product. If you have obtained this document from an agent in connection with the purchase of a policy or contract, you should be aware that such delivery does not guarantee that the Guaranty Association would cover your policy or contract. Any determination of whether a policy or contract will be covered will be determined solely by the coverage provisions of the Act.

This disclosure is intended to summarize the general purpose of the Act and does not address all the provisions of the Act. Moreover, the disclosure is not intended and should not be relied upon to later any right established in any policy or contract, or under the Act.

Group Hospitalization and Medical Services, Inc. doing business as CareFirst BlueCross BlueShield (CareFirst)

840 First Street, NE Washington, DC 20065 202-479-8000

An independent licensee of the Blue Cross and Blue Shield Association.

ATTACHMENT D **ELIGIBILITY SCHEDULE**

Effective Date	Effective as of the Group Contract Effective Date.		
ELIGIBILITY			
Subscriber	See the In-Network Eligibility Schedule.		
Spouse	See the In-Network Eligibility Schedule.		
Dependent Children	See the In-Network Eligibility Schedule.		
Limiting Age for	See the In-Network Eligibility Schedule.		
Dependent Children (other			
than an incapacitated			
Dependent Child)			
ENROLLMENT PERIODS AND EFFECTIVE DATES			
Annual Open Enrollment and Newly Eligible Qualified Employees			
Annual Open Enrollment	See the In-Network Eligibility Schedule.		
Newly Eligible Qualified	See the In-Network Eligibility Schedule.		
Employee			

Special Envellment Davieds		
Special Enrollment Periods		
Newly Eligible Dependent	The Special Enrollment Period is the thirty-one (31) day period from the	
Child: (Newborn	date of the qualifying event.	
Dependent Child, Newly		
Adopted Dependent Child,	The Effective Date of coverage is the Dependent Child's First Eligibility	
a Minor Dependent Child	Date:	
for whom guardianship		
has been granted by court	A. Newly born Dependent Child: the date of birth.	
or testamentary	The state of the s	
appointment, or	B. Adopted Dependent Child: the date of Adoption, which is the	
MCSO/QMSO)	earlier of the date a judicial decree of adoption is signed; or the	
MCSO/QMSO)	assumption of custody, pending adoption, of a prospective adoptive	
	assumption of custody, pending adoption, of a prospective adoptive	
	child by a prospective adoptive parent.	
	C. Testamentary or court appointed guardianship of a Dependent Child:	
	the date of appointment.	
	D. Dependent child who is the subject of a Medical Child Support	
	Order or Qualified Medical Support Order that creates or recognizes	
	the right of the Dependent Child to receive benefits under a parent's	
	health insurance coverage:	
	neutin insurance coverage.	
	Medical Child Support Order: the date specified in the Medical	
	Child Support Order.	
	Cinia Support Order.	
	Qualified Medical Support Order: the date specified in the	
	Qualified Medical Support Order.	
	If the Qualified Employee is enrolled, the newly eligible Dependent Child	
	will be covered automatically, but only for the first thirty-one (31) days	
	following the child's First Eligibility Date. The Subscriber must enroll such	
	a Dependent Child within thirty-one (31) days of the child's First Eligibility	
	Date.	
Newly Eligible Dependent	The Special Enrollment Period is the thirty-one (31) day period from the	
(including a Spouse or a	date of the qualifying event.	
newly eligible Dependent	date of the qualifying event.	
	The Effective Date of accompanies the first day of the month following the	
Child not described	The Effective Date of coverage is the first day of the month following the	
above)	receipt of enrollment by the SHOP Exchange.	
Eligible Qualified	The Special Enrollment Period is the thirty (30) day period from the date of	
Individual who loses other	the qualifying event.	
Minimum Essential		
Coverage (as provided in	The Effective Date of coverage is the first day of the month following the	
the Evidence of Coverage)	receipt of enrollment by the SHOP Exchange.	
Eligible Qualified	The Special Enrollment Period is the sixty (60) day period from the date of	
Individual who loses	the qualifying event.	
coverage under a		
Medicaid plan or a state	The Effective Date of coverage is the first day of the month following the	
child health plan,(as	receipt of enrollment by the SHOP Exchange.	
provided in the Evidence		
of Coverage)		

Eligible Qualified	The Special Enrollment Period shall be sixty (60) days prior to the end of
Individual or is permitted	coverage under the employer sponsored plan.
to terminate coverage	
under a non-qualifying	The Effective Date of coverage is the first day of the month following the
employer-sponsored health benefit plan (as	receipt of enrollment by the SHOP Exchange.
provided in the Evidence	
of Coverage)	
Eligible Qualified	The Special Enrollment Period is the thirty (30) day period from the date of
Individual who becomes	the qualifying event.
eligible to enroll due to	
any other qualifying event	The Effective Date of the Qualified Individual's coverage will be:
stated in Section 2.5 of the	A E 11 (CHOPE 1 1 (1 C') 1
Evidence of Coverage	A. For enrollment received by the SHOP Exchange between the first and the fifteenth day of the month, the first day of the following month; and
	B. For enrollment received by the SHOP Exchange between the sixteenth and the last day of the month, the first day of the second following month.
	For a Qualified Individual who enrolls during a Special Enrollment Period due to (i) an error by the SHOP Exchange or the United States Department of Health and Human Services; (ii) another Qualified Health Plan substantially violating a material provision of its contract with the Qualified Individual; or (iii) other exceptional circumstances as determined by the SHOP Exchange, the Effective Date of the Qualified Individual's coverage will be established by the SHOP Exchange to be either:
	A. The date of the event that triggered the Special Enrollment Period under these circumstances; or
	B. The Effective Date set forth for all other circumstances stated above.
	TERMINATION OF ENROLLMENT
Termination of Enrollme	
Subscriber terminates	See the In-Network Eligibility Schedule.
enrollment of a Member	
Termination of Enrollme	nt by CareFirst or the SHOP Exchange
Subscriber no longer	See the In-Network Eligibility Schedule.
eligible as a Qualified	
Employee or Qualified	
Individual Dependent Child	Consider In Material Ellis Heller College
Dependent Child Limiting Age (other than	See the In-Network Eligibility Schedule.
a Dependent Child who is	
incapacitated as provided	
in the Evidence of	
Coverage)	
Dependent no longer	See the In-Network Eligibility Schedule.
eligible (on grounds other than Limiting Age)	
Death of Subscriber	
L Death of Nilnscriper	See the In-Network Eligibility Schedule.

Notice of Nondiscrimination and Availability of Language Assistance Services

(UPDATED 7/12/18)

CareFirst BlueCross BlueShield, CareFirst BlueChoice, Inc., CareFirst Diversified Benefits and all of their corporate affiliates (CareFirst) comply with applicable federal civil rights laws and do not discriminate on the basis of race, color, national origin, age, disability or sex. CareFirst does not exclude people or treat them differently because of race, color, national origin, age, disability or sex.

CareFirst:

Provides free aid and services to people with disabilities to communicate effectively with us, such as:

 Qualified sign language interpreters
 Written information in other formats (large print, audio, accessible electronic formats, other formats)

 Provides free language services to people whose primary language is not English, such as:

 Qualified interpreters
 Information written in other languages

If you need these services, please call 855-258-6518.

If you believe CareFirst has failed to provide these services, or discriminated in another way, on the basis of race, color, national origin, age, disability or sex, you can file a grievance with our CareFirst Civil Rights Coordinator by mail, fax or email. If you need help filing a grievance, our CareFirst Civil Rights Coordinator is available to help you.

To file a grievance regarding a violation of federal civil rights, please contact the Civil Rights Coordinator as indicated below. Please do not send payments, claims issues, or other documentation to this office.

Civil Rights Coordinator, Corporate Office of Civil Rights

Mailing Address P.O. Box 8894

Baltimore, Maryland 21224

Email Address civilrightscoordinator@carefirst.com

Telephone Number 410-528-7820 Fax Number 410-505-2011

You can also file a civil rights complaint with the U.S. Department of Health and Human Services, Office for Civil Rights electronically through the Office for Civil Rights Complaint portal, available at https://ocrportal.hhs.gov/ocr/portal/lobby.jsf or by mail or phone at:

U.S. Department of Health and Human Services 200 Independence Avenue, SW Room 509F, HHH Building Washington, D.C. 20201 800-368-1019, 800-537-7697 (TDD)

Complaint forms are available at http://www.hhs.gov/ocr/office/file/index.html.

Foreign Language Assistance

Attention (English): This notice contains information about your insurance coverage. It may contain key dates and you may need to take action by certain deadlines. You have the right to get this information and assistance in your language at no cost. Members should call the phone number on the back of their member identification card. All others may call 855-258-6518 and wait through the dialogue until prompted to push 0. When an agent answers, state the language you need and you will be connected to an interpreter.

አማርኛ (Amharic) ማሳሰቢያ፦ ይህ ማስታወቂያ ስለ መድን ሽፋንዎ መረጃ ይዟል። ከተወሰኑ ቀነ-ገደቦቸ በፊት ሊፌጽሟቸው የሚገቡ ነገሮች ሊኖሩ ስለሚቸሉ እነዚህን ወሳኝ ቀናት ሊይዝ ይቸላል። ይኽን መረጃ የማግኘት እና ያለምንም ከፍያ በቋንቋዎ እንዛ የማግኘት መብት አለዎት። አባል ከሆኑ ከመታወቂያ ካርድዎ በስተጀርባ ላይ ወደተጠቀሰው የስልክ ቁጥር መደወል ይቸላሉ። አባል ካልሆኑ ደግሞ ወደ ስልክ ቁጥር 855-258-6518 ደውለው 0ን እንዲጫኑ እስኪነገርዎ ድረስ ንግግሩን መጠበቅ አለብዎ። አንድ ወኪል መልስ ሲሰጥዎ፣ የሚፌልጉትን ቋንቋ ያሳውቁ፣ ከዚያም ከተርጓሚ ጋር ይገናኛሉ።

Èdè Yorùbá (Yoruba) Ìtétíléko: Àkíyèsí yìí ní ìwífún nípa iṣé adójútòfò rẹ. Ó le ní àwọn déètì pàtó o sì le ní láti gbé ìgbésè ní àwọn ojó gbèdéke kan. O ni ètó láti gba ìwífún yìí àti ìrànlówó ní èdè rẹ lófèé. Àwọn omọ-ẹgbé gbódò pe nómbà fóònù tó wà léyìn káàdì ìdánimò wọn. Àwọn míràn le pe 855-258-6518 kí o sì dúró nípasè ìjíròrò títí a ó fi sọ fún ọ láti tẹ 0. Nígbàtí aṣojú kan bá dáhùn, sọ èdè tí o fé a ó sì so ó pò mó ògbufò kan.

Tiếng Việt (Vietnamese) Chú ý: Thông báo này chứa thông tin về phạm vi bảo hiểm của quý vị. Thông báo có thể chứa những ngày quan trọng và quý vị cần hành động trước một số thời hạn nhất định. Quý vị có quyền nhận được thông tin này và hỗ trợ bằng ngôn ngữ của quý vị hoàn toàn miễn phí. Các thành viên nên gọi số điện thoại ở mặt sau của thẻ nhận dạng. Tất cả những người khác có thể gọi số 855-258-6518 và chờ hết cuộc đối thoại cho đến khi được nhắc nhấn phím 0. Khi một tổng đài viên trả lời, hãy nêu rõ ngôn ngữ quý vị cần và quý vị sẽ được kết nối với một thông dịch viên.

Tagalog (Tagalog) Atensyon: Ang abisong ito ay naglalaman ng impormasyon tungkol sa nasasaklawan ng iyong insurance. Maaari itong maglaman ng mga pinakamahalagang petsa at maaaring kailangan mong gumawa ng aksyon ayon sa ilang deadline. May karapatan ka na makuha ang impormasyong ito at tulong sa iyong sariling wika nang walang gastos. Dapat tawagan ng mga Miyembro ang numero ng telepono na nasa likuran ng kanilang identification card. Ang lahat ng iba ay maaaring tumawag sa 855-258-6518 at maghintay hanggang sa dulo ng diyalogo hanggang sa diktahan na pindutin ang 0. Kapag sumagot ang ahente, sabihin ang wika na kailangan mo at ikokonekta ka sa isang interpreter.

Español (Spanish) Atención: Este aviso contiene información sobre su cobertura de seguro. Es posible que incluya fechas clave y que usted tenga que realizar alguna acción antes de ciertas fechas límite. Usted tiene derecho a obtener esta información y asistencia en su idioma sin ningún costo. Los asegurados deben llamar al número de teléfono que se encuentra al reverso de su tarjeta de identificación. Todos los demás pueden llamar al 855-258-6518 y esperar la grabación hasta que se les indique que deben presionar 0. Cuando un agente de seguros responda, indique el idioma que necesita y se le comunicará con un intérprete.

Русский (Russian) Внимание! Настоящее уведомление содержит информацию о вашем страховом обеспечении. В нем могут указываться важные даты, и от вас может потребоваться выполнить некоторые действия до определенного срока. Вы имеете право бесплатно получить настоящие сведения и сопутствующую помощь на удобном вам языке. Участникам следует обращаться по номеру телефона, указанному на тыльной стороне идентификационной карты. Все прочие абоненты могут звонить по номеру 855-258-6518 и ожидать, пока в голосовом меню не будет предложено нажать цифру «0». При ответе агента укажите желаемый язык общения, и вас свяжут с переводчиком.

हिन्दी (Hindi) ध्यान दें: इस सूचना में आपकी बीमा कवरेज के बारे में जानकारी दी गई है। हो सकता है कि इसमें मुख्य तिथियों का उल्लेख हो और आपके लिए किसी नियत समय-सीमा के भीतर काम करना ज़रूरी हो। आपको यह जानकारी और संबंधित सहायता अपनी भाषा में निःशुल्क पाने का अधिकार है। सदस्यों को अपने पहचान पत्र के पीछे दिए गए फ़ोन नंबर पर कॉल करना चाहिए। अन्य सभी लोग 855-258-6518 पर कॉल कर सकते हैं और जब तक 0 दबाने के लिए न कहा जाए, तब तक संवाद की प्रतीक्षा करें। जब कोई एजेंट उत्तर दे तो उसे अपनी भाषा बताएँ और आपको व्याख्याकार से कनेक्ट कर दिया जाएगा।

Bǎsóò-wùdù (*Bassa*) Tò Đùǔ Cáo! Bỗ nìà ke bá nyo bẽ ké m̀ gbo kpá bó nì fuà-fuá-tiǐn nyee jè dyí. Bỗ nìà ke bédé wé jéé bẽ m̀ ké dẽ wa mó m̀ ké nyuee nyu hwè bế wé bẽa ké zi. O mò nì kpé bế m̀ ké bỗ nìà ke kè gbo-kpá-kpá m̀ mɔ́ee dyé dé nì bídí-wudu mú bế m̀ ké se wídí dò péè. Kpooò nyo bẽ me dá fuun-nɔ́bà nìà dé waà I.D. káàò deín nye. Nyo tòò seín me dá nɔ̂bà nìà ke: 855-258-6518, ké m̀ me fò tee bế wa kée m̀ gbo cẽ bế m̀ ké nɔ̀bà mòà 0 kee dyi pàdàìn hwè. O jǔ ké nyo dò dyi m̀ gỗ jǔǐn, po wudu m̀ mɔ́ poe dyie, ké nyo dò mu bó nììn bế o ké nì wuduò mú zà.

বাংলা (Bengali) লক্ষ্য করুন: এই নোটিশে আপনার বিমা কভারেজ সম্পর্কে তথ্য রয়েছে। এর মধ্যে গুরুত্বপূর্ণ তারিথ থাকতে পারে এবং নির্দিষ্ট তারিথের মধ্যে আপনাকে পদক্ষেপ নিতে হতে পারে। বিনা থরচে নিজের ভাষায় এই তথ্য পাওয়ার এবং সহায়তা পাওয়ার অধিকার আপনার আছে। সদস্যদেরকে তাদের পরিচয়পত্রের পিছনে থাকা নম্বরে কল করতে হবে। অন্যেরা ৪55-258-651৪ নম্বরে কল করে 0 টিপতে না বলা পর্যন্ত অপেক্ষা করতে পারেন। যথন কোনো এজেন্ট উত্তর দেবেন তথন আপনার নিজের ভাষার নাম বলুন এবং আপনাকে দোভাষীর সঙ্গে সংযুক্ত করা হবে।

اردو (Urdu) توجہ :یہ نوٹس آپ کے انشورینس کوریج سے متعلق معلومات پر مشتمل ہے۔ اس میں کلیدی تاریخیں ہو سکتی ہیں اور ممکن ہے کہ آپ کو مخصوص آخری تاریخوں تک کارروائی کرنے کی ضرورت پڑے۔ آپ کے پاس یہ معلومات حاصل کرنے اور بغیر خرچہ کیے اپنی زبان میں مدد حاصل کرنے کا حق ہے۔ ممبران کو اپنے شناختی کارڈ کی پشت پر موجود فون نمبر پر کال کرنی چاہیے۔ سبھی دیگر لوگ 855-258پر کال کر سکتے ہیں اور 0 دبانے کو کہے جانے تک انتظار کریں۔ ایجنٹ کے جواب دینے پر اپنی مطلوبہ زبان بتائیں اور مترجم سے مربوط ہو جائیں گے۔

فارسی (Farsi) توجه: این اعلامیه حاوی اطلاعاتی درباره پوشش بیمه شما است. ممکن است حاوی تاریخ های مهمی باشد و لازم است تا تاریخ مقرر شده خاصی اقدام کنید. مقرر شده خاصی اقدام کنید. شما از این حق برخوردار هستید تا این اطلاعات و راهنمایی را به صورت رایگان به زبان خودتان دریافت کنید. اعضا باید با شماره اعضا باید با شماره درج شده در پشت کارت شناسایی شان تماس بگیرند. سایر افراد می توانند با شماره مقبرند و منتظر بمانند تا از آنها خواسته شود عدد 0 را فشار دهند. بعد از پاسخگویی توسط یکی از اپراتور ها، زبان مورد نیاز را تنظیم کنید تا به مترجم مربوطه وصل شوید.

اللغة العربية (Arabic) تنبيه :يحتوي هذا الإخطار على معلومات بشأن تغطيتك التأمينية، وقد يحتوي على تواريخ مهمة، وقد تحتاج إلى اتخاذ إجراءات بحلول مواعيد نهائية محددة .يحق لك الحصول على هذه المساعدة والمعلومات بلغتك بدون تحمل أي تكلفة .ينبغي على الأعضاء الاتصال على رقم الهاتف المذكور في ظهر بطاقة تعريف الهوية الخاصة بهم يمكن للآخرين الاتصال على الرقم 855-258 والانتظار خلال المحادثة حتى يطلب منهم الضغط على رقم .0 عند إجابة أحد الوكلاء، اذكر اللغة التي تحتاج إلى التواصل بها وسيتم توصيلك بأحد المترجمين الفوريين.

中文繁体(Traditional Chinese) 注意:本聲明包含關於您的保險給付相關資訊。本聲明可能包含重要日期及您在特定期限之前需要採取的行動。您有權利免費獲得這份資訊,以及透過您的母語提供的協助服務。會員請撥打印在身分識別卡背面的電話號碼。其他所有人士可撥打電話 855-258-6518, 並等候直到對話提示按下按鍵 0。當接線生回答時,請說出您需要使用的語言,這樣您就能與口譯人員連線。

Igbo (Igbo) Nrubama: Okwa a nwere ozi gbasara mkpuchi nchekwa onwe gi. O nwere ike inwe ubochi ndi di mkpa, i nwere ike ime ihe tupu ufodu ubochi njedebe. I nwere ikike inweta ozi na enyemaka a n'asusu gi na akwughi ugwo o bula. Ndi otu kwesiri ikpo akara ekwenti di n'azu nke kaadi njirimara ha. Ndi ozo niile nwere ike ikpo 855-258-6518 wee chere ububo ahu ruo mgbe amanyere ipi 0. Mgbe onye nnochite anya zara, kwuo asusu i choro, a ga-ejiko gi na onye okowa okwu.

Deutsch (German) Achtung: Diese Mitteilung enthält Informationen über Ihren Versicherungsschutz. Sie kann wichtige Termine beinhalten, und Sie müssen gegebenenfalls innerhalb bestimmter Fristen reagieren. Sie haben das Recht, diese Informationen und weitere Unterstützung kostenlos in Ihrer Sprache zu erhalten. Als Mitglied verwenden Sie bitte die auf der Rückseite Ihrer Karte angegebene Telefonnummer. Alle anderen Personen rufen bitte die Nummer 855-258-6518 an und warten auf die Aufforderung, die Taste 0 zu drücken. Geben Sie dem Mitarbeiter die gewünschte Sprache an, damit er Sie mit einem Dolmetscher verbinden kann.

Français (French) Attention: cet avis contient des informations sur votre couverture d'assurance. Des dates importantes peuvent y figurer et il se peut que vous deviez entreprendre des démarches avant certaines échéances. Vous avez le droit d'obtenir gratuitement ces informations et de l'aide dans votre langue. Les membres doivent appeler le numéro de téléphone figurant à l'arrière de leur carte d'identification. Tous les autres peuvent appeler le 855-258-6518 et, après avoir écouté le message, appuyer sur le 0 lorsqu'ils seront invités à le faire. Lorsqu'un(e) employé(e) répondra, indiquez la langue que vous souhaitez et vous serez mis(e) en relation avec un interprète.

한국어(Korean) 주의: 이 통지서에는 보험 커버리지에 대한 정보가 포함되어 있습니다. 주요 날짜 및 조치를 취해야 하는 특정 기한이 포함될 수 있습니다. 귀하에게는 사용 언어로 해당 정보와 지원을 받을 권리가 있습니다. 회원이신 경우 ID 카드의 뒷면에 있는 전화번호로 연락해 주십시오. 회원이 아니신 경우 855-258-6518 번으로 전화하여 0을 누르라는 메시지가 들릴 때까지 기다리십시오. 연결된 상담원에게 필요한 언어를 말씀하시면 통역 서비스에 연결해 드립니다.

Diné Bizaad (Navajo) Ge': Díí bee ił hane'ígíí bii' dahóló bee éédahózin béeso ách'ááh naanil ník'ist'i'ígíí bá. Bii' dahólóó doo íiyisíí yoolkáálígíí dóó t'áádoo le'é ádadoolyí(lígíí da yókeedgo t'áá doo bee e'e'aahí ájiil'í(h. Bee ná ahóót'i' díí bee ił hane' dóó niká'ádoowoł t'áá nínizaad bee t'áá jiik'é. Atah danilínígíí béésh bee hane'é bee wółta'ígíí nitł'izgo bee nee hódolzinígíí bikéédéé' bikáá' bich'i' hodoonihjí'. Aadóó náánáła' éí koji' dahódoolnih 855-258-6518 dóó yii diiłts'íijł yałtí'ígíí t'áá níléíjí áádóó éí bikéé'dóó naasbąąs bił adidiilchił. Áká'ánidaalwó'ígíí neidiitáágo, saad bee yániłt'i'ígíí yii diikił dóó ata' halne'é lá níká'ádoolwoł.